Case 25-10979-JKS Doc 50 Filed 06/04/25 Page 1 of 3 Docket #0050 Date Filed: 6/4/2025

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

	Obj. Deadline: June 20, 2025 at 4:00 p.m. (ET) Hearing Date: July 2, 2025 at 11:00 a.m. (ET)
Debtors. ¹	Jointly Administered
VWS Holdco, Inc., et al.,	Case No. 25-10979 (JKS)
In re	Chapter 11

NOTICE OF HEARING REGARDING DEBTORS' MOTION FOR ENTRY OF INTERIM AND FINAL ORDERS PURSUANT TO SECTIONS 105, 361, 362, 363, 364, 503, 506, 507 AND 552 OF THE BANKRUPTCY CODE AND RULES 2002, 4001, 6004 AND 9014 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE (I) AUTHORIZING THE DEBTORS TO (A) USE CASH COLLATERAL, (B) OBTAIN SENIOR SECURED SUPERPRIORITY POSTPETITION FINANCING AND GRANTING LIENS AND SUPERPRIORITY ADMINISTRATIVE CLAIMS, AND (C) PROVIDE ADEQUATE PROTECTION, (II) MODIFYING THE AUTOMATIC STAY, (III) SCHEDULING A FINAL HEARING, AND (IV) GRANTING RELATED RELIEF

PLEASE TAKE NOTICE that on June 1, 2025, the above-captioned debtors and debtors in possession (the "Debtors") filed the Debtors' Motion for Entry of Interim and Final Orders Pursuant to Sections 105, 361, 362, 363, 364, 503, 506, 507 and 552 of the Bankruptcy Code and Rules 2002, 4001, 6004 and 9014 of the Federal Rules of Bankruptcy Procedure (I) Authorizing the Debtors to (A) Use Cash Collateral, (B) Obtain Senior Secured Superpriority Postpetition Financing and Granting Liens and Superpriority Administrative Claims and (C) Provide Adequate Protection, (II) Modifying the Automatic Stay, (III) Scheduling a Final Hearing, and (IV) Granting Related Relief [D.I. 15] (the "Motion"), attached hereto as Exhibit A.

PLEASE TAKE FURTHER NOTICE that, on June 4, 2025, the Bankruptcy Court entered the Interim Order Pursuant to Sections 105, 361, 362, 363, 364, 503, 506, 507 and 552 of the Bankruptcy Code and Rules 2002, 4001, 6004 and 9014 of the Federal Rules of Bankruptcy Procedure (I) Authorizing the Debtors to (A) Use Cash Collateral, (B) Obtain Senior Secured Superpriority Postpetition Financing and Granting Liens and Superpriority Administrative Claims and (C) Provide Adequate Protection, (II) Modifying the Automatic Stay, (III) Scheduling a Final Hearing, and (IV) Granting Related Relief [D.I. 42] (the "Interim Order"), attached hereto as Exhibit B.

The Debtors in these Chapter 11 Cases, along with the last four digits of each Debtor's federal tax identification numbers are as follows: VWS Holdco, Inc. (5412) and Shoosmith Bros., Inc. (6914). The Debtors' mailing address is P.O. Box 2770, Chesterfield, VA 23832.



PLEASE TAKE FURTHER NOTICE that objections, if any, to the final approval of the Motion must (a) be in writing, (b) be filed with the Clerk of the Bankruptcy Court, 824 Market Street, 3rd Floor, Wilmington, Delaware 19801, on or before June 20, 2025 at 4:00 p.m. (ET) (the "Objection Deadline"), and (c) served as to be received on or before the Objection Deadline upon (a) the Debtors, (b) proposed counsel to the Debtors, Pashman Stein Walder Hayden, P.C., 824 N. Market Street, Suite 800, Wilmington, Delaware 19801, Attn: John W. Weiss (jweiss@pashmanstein.com), Leah M. Eisenberg (leisenberg@pashmanstein.com), Richard W. Riley (rriley@pashmanstein.com) and David E. Sklar (dsklar@pashmanstein.com), (b) the Office of the United States Trustee for the District of Delaware, J. Caleb Boggs Building, 844 King Street, Suite 2207, Lockbox 35, Wilmington, Delaware, 19801, Attn: Jane M. Leamy (jane.m.leamy@usdoj.gov), (c) counsel to the DIP Lenders, Cole Schotz P.C., 1325 Avenue of the Americas, New York, NY 10019, Attn: Daniel F. X. Geoghan (DGeoghan@coleschotz.com).

PLEASE TAKE FURTHER NOTICE that only objections made in writing and timely filed and received, in accordance with the procedures above, will be considered by the Bankruptcy Court at such hearing.

PLEASE TAKE FURTHER NOTICE THAT A FINAL HEARING ON THE MOTION WILL BE HELD ON JULY 2, 2025 AT 11:00 A.M. (ET) BEFORE THE HONORABLE J. KATE STICKLES, AT THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE, 824 MARKET STREET, WILMINGTON, DELAWARE 19801.

IF YOU FAIL TO RESPOND IN ACCORDANCE WITH THIS NOTICE, THE COURT MAY GRANT THE RELIEF REQUESTED IN THE MOTION WITHOUT FURTHER NOTICE OR HEARING.

Dated: June 4, 2025 Wilmington, Delaware

PASHMAN STEIN WALDER HAYDEN, P.C.

/s/ Richard W. Riley

John W. Weiss (No. 4160) Richard W. Riley (No. 4052) 824 North Market Street, Suite 800 Wilmington, DE 19801 Telephone: (302) 592-6496

Email: jweiss@pashmanstein.com rriley@pashmanstein.com

-and-

Leah M. Eisenberg (pro hac vice forthcoming)
David E. Sklar (pro hac vice forthcoming)
Court Plaza South, East Wing
21 Main Street, Suite 200
Hackensack, NJ 07601
Telephone: (201) 488-8200
Email: laisenberg@pashmanstein.acm

Email: leisenberg@pashmanstein.com dsklar@pashmanstein.com

Proposed Counsel to the Debtors and Debtors in Possession

Exhibit A

Motion

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re

VWS Holdco, Inc., et al.

Debtors.¹

Chapter 11

Case No. 25-10979 (JKS)

(Joint Administration Requested)

DEBTORS' MOTION FOR ENTRY OF INTERIM
AND FINAL ORDERS PURSUANT TO SECTIONS

105, 361, 362, 363, 364, 503, 506, 507 AND 552 OF THE BANKRUPTCY
CODE AND RULES 2002, 4001, 6004 AND 9014 OF THE FEDERAL
RULES OF BANKRUPTCY PROCEDURE (I) AUTHORIZING THE
DEBTORS TO (A) USE CASH COLLATERAL, (B) OBTAIN SENIOR
SECURED SUPERPRIORITY POSTPETITION FINANCING AND
GRANTING LIENS AND SUPERPRIORITY ADMINISTRATIVE
CLAIMS, AND (C) PROVIDE ADEQUATE PROTECTION,
(II) MODIFYING THE AUTOMATIC STAY, (III) SCHEDULING
A FINAL HEARING, AND (IV) GRANTING RELATED RELIEF

VWS Holdco, Inc. ("<u>VWS Holdco</u>") and Shoosmith Bros., Inc. ("<u>Shoosmith</u>"), the debtors and debtors-in-possession in the above-captioned cases (together, the "<u>Debtors</u>" or the "<u>Company</u>") hereby move (this "<u>Motion</u>") the Court for the entry of an order, substantially in the form attached as <u>Exhibit A</u> hereto (the "<u>Interim Order</u>"),² and a final order (the "<u>Final Order</u>," and together with the Interim Order, the "<u>DIP Orders</u>"),³ pursuant to sections 105, 361, 362, 363, 364, 503, 506, 507, and 552 of title 11 of the United States Code (as amended, the "Bankruptcy

The Debtors in these Chapter 11 Cases, along with the last four digits of each Debtor's federal tax identification numbers are as follows: VWS Holdco, Inc. (5412) and Shoosmith Bros., Inc. (6914). The Debtors' mailing address is P.O. Box 2770, Chesterfield, VA 23832.

² Capitalized terms not otherwise defined herein shall have the meanings ascribed to them in the DIP Credit Agreement (as defined below) or the proposed Interim Order, as applicable.

The proposed Final Order shall be substantially the same as the Interim Order except that (i) those provisions in the Interim Order that are subject to the entry of the Final Order shall be included in the Final Order without such qualification, and (ii) where appropriate, references to the Interim Order shall be changed to references to the Final Order.

Code"), rules 2002, 4001, 6004, and 9014 of the Federal Rules of Bankruptcy Procedure (as amended, the "Bankruptcy Rules") and rules 2002-1, 4001-1, 4001-2, 9006-1, and 9013-1 of the Local Rules of Bankruptcy Practice and Procedure for the District of Delaware (the "Local Rules") (i) authorizing the Debtors to (a) use cash collateral, (b) obtain senior secured superpriority postpetition financing and granting liens and superpriority administrative expenses claims, and (c) provide adequate protection, (ii) modifying the automatic stay, (iii) scheduling interim and final hearings, and (iv) granting related relief. In support of this Motion, the Debtors rely upon and incorporate by reference the Declaration of Steven Agran in Support of Chapter 11 Petitions and First Day Pleadings [D.I. 12] (the "First Day Declaration"), and respectfully submit the Declaration of Steven Agran in Support of Debtor's Motion for Entry of Interim and Final Orders (I) Authorizing the Debtors to (A) Use Cash Collateral, (B) Obtain Senior Secured Superpriority Postpetition Financing and: Grant Liens and Superpriority Administrative Claims, and (C) Provide Adequate Protection, (II) Scheduling a Final Hearing, and (III) Granting Related Relief filed concurrently herewith (the "DIP Declaration," and together with the First Day Declaration, the "Declarations"). In further support of the Motion, the Debtors, by and through their undersigned proposed counsel, respectfully state as follows:

PRELIMINARY STATEMENT

1. To address immediate liquidity issues, the Debtors have negotiated with their prepetition secured creditor Volunteer Enterprises, LLC ("Volunteer") and the DIP Lenders (as defined below) the terms of a senior secured superpriority debtor-in-possession term loan credit facility of up to an aggregate principal amount of \$5,400,000 (with \$4,900,000 in new money) (the "DIP Facility") to be provided as follows: (i) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.1 of the DIP Credit Agreement, the DIP Lenders shall make available during the period from the date hereof through and including the date of entry

of the Final Order by this Court, in one or more additional draws the aggregate amount of \$1,800,000.00, to include a roll-up based on the interim funding of \$500,000.00; (ii) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.2 of the DIP Credit Agreement, the DIP Lenders shall make available in one or more additional draws the aggregate amount of \$1,550,000.00; and (iii) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.3 of the DIP Credit Agreement, the DIP Lenders shall make available in one or more additional draws the aggregate amount of \$1,550,000.00.

- 2. Specifically, the DIP Facility will provide new financing to fund these Chapter 11 Cases, preserve the value of the Debtors' assets, and maximize value for all parties as the Debtors work expeditiously toward a sale of their primary asset, the Shoosmith Landfill (the "Landfill") to a new owner and operator. As described in further detail in the DIP Declaration, it is critical that a new party be put in place to replace the Debtors (as owner and operator of the Landfill) to address and remove the effluent leachate being generated in the Landfill so as to avoid any significant environmental impact on the surrounding area.
- 3. By this Motion, the Debtors seek authorization to access the DIP Facility and use Cash Collateral (as such term is defined in section 363(a) of title 11 of the United States Code), each of which is critical to (i) the continued, uninterrupted operation of the Debtors' business in chapter 11 while they pursue a court-approved sale process of the Landfill; (ii) the payment of necessary expenses associated with administering the Chapter 11 Cases; and (iii) the Debtors' ability to continue to meet their obligations to their insurers, taxing authorities, critical vendors, and banks, and otherwise manage their businesses as close to the ordinary course as possible.
- 4. Access to the DIP Facility is necessary to avoid the immediate and irreparable harm that would otherwise result if the Debtors were denied incremental liquidity. Without the DIP

Facility, the Debtors will have insufficient cash on hand to pay their ordinary course obligations as they come due during the pendency of these Chapter 11 Cases, as well as to fund their contemplated sale process. The DIP Facility and use of Cash Collateral are thus critical to ensure the smooth entry by the Debtors into chapter 11 and their ability to operate their business prudently during the pendency of these Chapter 11 Cases while also running the contemplated sale process.

- 5. As will be discussed further below, the terms of the DIP Facility, including the Interim Roll-Up Amount of \$500,000 in Bridge Notes Obligations, are market and reasonable under the circumstances.
- 6. The DIP Facility was evaluated by the Debtors, their advisors and recently retained Chief Restructuring Officer (the "CRO") and was thoroughly negotiated at arm's length between the Debtors and the DIP Lenders.⁴ The Debtors' advisors conducted a reach-out to over 20 potential middle market lenders experienced in the distressed space, as well as larger, institutional banks for completeness. Despite such efforts, the Debtors have been unable to obtain any alternative source of financing due to, among other things, the timing and the amount of the financing needed and the Debtors' environmental issues.
- 7. For these reasons, the relief requested by this Motion is a necessary step to preserve the Debtors' operations so that the Debtors can focus their efforts on (i) addressing the immediate need to remove the effluent leachate being generated in the Landfill so as to avoid any significant environmental impact on the surrounding area; and (ii) the sale process and obtain the highest and best purchase price for the Landfill, thereby providing the maximum possible value for the Debtors' estates and stakeholders. As the Debtors will be prepared to demonstrate at the Interim

_

⁴ Although the DIP Lenders are insiders of the Debtors due to their ownership of equity interests in Volunteer (which owns 28.19% of the voting units of VWS Acquisitions), the DIP Lenders took no part in the Debtors' consideration or negotiation of, or voting on, approval of the DIP Facility. The CRO and the Debtors' legal and financial advisors negotiated the DIP Facility on behalf of the Debtors.

Hearing and the Final Hearing, the relief requested herein represents a sound exercise of business judgment, meets the heightened scrutiny required of a transaction between a debtor and an insider, and should therefore be approved.

RELIEF REQUESTED

- 8. By this Motion, pursuant to sections 105, 361, 362, 363, 364, 503, 506, 507 and 552 of the Bankruptcy Code, Bankruptcy Rules 2002, 4001, 6003, 6004, and 9014, and Local Rules 4001-2 and 9013-1 of the United States Bankruptcy Court for District of Delaware (the "Court"), the Debtors seek entry of the Interim Order and a Final Order providing for the following relief:
 - authorizing Debtor VWS Holdco, as borrower, (the "Borrower") and Debtor a. Shoosmith. as guarantor, (the "Guarantor") to obtain a senior secured superpriority DIP Facility in an aggregate amount of \$5,400,000 (with \$4,900,000 in new money) pursuant to the terms of the DIP Orders, the terms and conditions set forth in the credit agreement attached to the Interim Order as Exhibit 1 (as may be amended, restated, supplemented or otherwise modified from time to time, in each case with the consent of the DIP Lenders (as defined below), the "DIP Credit Agreement,"⁵ collectively with the DIP Orders, the DIP Budget (as defined below), and all other agreements, documents, and instruments delivered or executed in connection therewith to the extent that the DIP Agent (as defined below) and the DIP Lenders determine there is a need for such other agreements, documents, and instruments, in each case as amended, restated, supplemented, or otherwise modified from time to time, the "DIP Loan Documents") provided by Volunteer Enterprises, LLC, as administrative agent (together with any successor administrative agents and collateral agents, the "DIP Agent"), and the parties listed on Annex B to the DIP Credit Agreement, as lenders (in such capacity, together with any successors and assigns permitted under the DIP Loan Documents, the "DIP Lenders" and, together with the DIP Agent, the "DIP Secured Parties"), which DIP Facility shall be available, subject to the terms and conditions set forth in the Interim Order and the other DIP Loan Documents, as follows: (i) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.1 of the DIP Credit Agreement, the DIP Lenders shall make available during the period (the "Interim Period") from the date hereof through and including the date of entry of the Final Order by this Court, in one or more additional draws the aggregate amount of \$1,800,000.00 (the "First Interim DIP Loan"), to include a roll-up based on the

Capitalized terms used but not defined herein have the same meaning ascribed to such terms in the DIP Credit Agreement or the Motion, as appropriate.

interim funding of \$500,000.00 (the "Interim Roll-Up Amount"); (ii) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.2 of the DIP Credit Agreement, the DIP Lenders shall make available in one or more additional draws the aggregate amount of \$1,550,000.00 (the "Second Interim DIP Loan"); and (iii) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.3 of the DIP Credit Agreement, the DIP Lenders shall make available in one or more additional draws the aggregate amount of \$1,550,000.00 (the "Third Interim DIP Loan", and with the First Interim DIP Loan and the Second Interim DIP Loan, collectively, the "DIP Loans")

- b. the Court's authorization for the Debtors to execute, deliver, and perform, as applicable, under the DIP Credit Agreement and all other DIP Loan Documents, and to perform such other and further acts as may be necessary or appropriate in connection therewith;
- c. the Court's authorization for the Debtors to use proceeds of the DIP Loans immediately upon approval of the Interim Order solely in accordance with the budget and cash flow forecast prepared by the Debtors and annexed to the Interim Order as Exhibit 2 (as updated from time to time in accordance with the terms of the DIP Credit Agreement and the other DIP Loan Documents, subject to the prior approval of the DIP Lenders in their sole discretion, the "Initial DIP Budget"), subject to the Permitted Variances;
- d. the Court's authorization to grant to the DIP Agent, for the benefit of the DIP Secured Parties, in respect of the DIP Obligations (as defined below), subject to the Carve-Out (as defined below) a superpriority administrative claim pursuant to section 364(c)(1) of the Bankruptcy Code and first priority priming liens on and security interests in substantially all assets and property of the Debtors (now owned or hereafter acquired) pursuant to sections 364(c)(2), (c)(3), and (d)(1) of the Bankruptcy Code, in each case as and to the extent, set forth more fully below and in the DIP Loan Documents (including the DIP Credit Agreement);
- e. the Court's authorization for the Debtors to use "cash collateral" as such term is defined in section 363 of the Bankruptcy Code in which the Prepetition Secured Parties (as defined below) or the DIP Secured Parties has an interest (the "Cash Collateral") solely in accordance with the DIP Budget (subject to Permitted Variances) and the DIP Credit Agreement;
- f. the Court's authorization to grant, as of the Petition Date, adequate protection for the benefit of the Prepetition Secured Parties, as set forth more fully below, including the Adequate Protection Superpriority Claim, Adequate Protection Liens, and Adequate Protection Payments (each as defined below), to the extent of and as compensation for any Diminution in Value (as defined below), the payment of fees of the Prepetition Secured Parties, if any, and the Debtors' satisfaction of the Milestones (as defined below) in accordance with the DIP Credit Agreement;

- g. the modification or waiver by the Court of the automatic stay imposed by section 362 of the Bankruptcy Code and any other applicable stay (including Bankruptcy Rule 6004) to the extent necessary to implement and effectuate the terms and provisions of the DIP Facility and the Interim Order and the other DIP Loan Documents, and to provide for the immediate effectiveness of the Interim Order;
- h. the scheduling by the Court of a final hearing (the "<u>Final Hearing</u>") to consider entry of an order (the "<u>Final Order</u>") granting the relief requested in the Motion on a final basis and approving the form of notice with respect to the Final Hearing and the transactions contemplated by the Motion; and
- i. entry of a Final Order.
- 9. For the reasons set forth herein, the Debtors submit that the relief requested is in the best interests of the Debtors, their estates, creditors, and other parties in interest, and therefore should be granted.

JURISDICTION, VENUE, AND PREDICATES FOR RELIEF

- 10. This Court has jurisdiction to consider this Motion under 28 U.S.C. §§ 157 and 1334 and the *Amended Standing Order of Reference*, dated February 29, 2012 (Sleet, C.J.). This is a core proceeding under 28 U.S.C. § 157(b). Venue of these Chapter 11 Cases and this Motion is proper in this District under 28 U.S.C. §§ 1408 and 1409.
- 11. The predicates for the relief requested by this Motion are sections 105, 361, 362, 363, 364, 506, 507, and 552 of the Bankruptcy Code, Rules 2002, 4001, 6004, and 9014 of the Bankruptcy Rules, and Local Rules 4001-2 and 9013-1.
- 12. Pursuant to Local Rule 9013-1(f), the Debtors consent to the entry of a final judgment or order with respect to this Motion if it is determined that this Court lacks Article III jurisdiction to enter such final order or judgment absent consent of the parties.

BACKGROUND

A. Overview

- 13. On June 1, 2025 (the "Petition Date"), the Debtors each commenced with this Court a voluntary case under chapter 11 of the Bankruptcy Code (collectively, the "Chapter 11 Cases"). The Debtors have filed a separate procedural motion requesting that the Chapter 11 Cases be jointly administered. The Debtors continue to operate their businesses and manage their properties as debtors in possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No creditors' committee has been appointed by the Office of the United States Trustee for the District of Delaware (the "U.S. Trustee"), nor has a trustee or examiner been appointed in these Chapter 11 Cases.
- 14. Additional factual background and information regarding the Debtors, including their business operations, their corporate and capital structure, their restructuring activities, and the events leading to the commencement of these Chapter 11 Cases, is set forth in detail in the First Day Declaration.

B. The Debtors' Prepetition Capital Structure

- i. Equity Interests
- 15. Debtor Shoosmith is a Virginia corporation that is 100% owned by Debtor VWS Holdco, Inc.
- 16. VWS Holdco is a Delaware corporation that is 100% owned by non-Debtor VWS Acquisitions, LLC ("VWS Acquisitions"), a Delaware limited liability company.
- 17. Volunteer (the DIP Lenders), a Texas limited liability company, owns 41.36% of the common membership units and 28.19% of the voting units of VWS Acquisitions.
- 18. The membership units of Volunteer are owned 50/50 between Fred Nichols ("Mr. Nichols") and Larry McGee ("Mr. McGee").

ii. Prepetition Indebtedness

- 19. Effective as of April 30, 2020, VWS, as borrower (in such capacity, the "SPA Borrower"), entered into that certain Amended and Restated Securities Purchase Agreement (as amended, modified, or supplemented from time to time, the "Prepetition Term Loan Agreement") with VWS Acquisitions and the investors party thereto from time to time (in such capacity, collectively, the "Investors").
- 20. Pursuant to the Prepetition Term Loan Agreement, the SPA Borrower issued Senior Notes and Junior Notes (each as defined in the Prepetition Term Loan Agreement, and collectively, the "Notes" and the obligations under the DIP Credit Agreement and under the related Transaction Documents (as defined in the Prepetition Term Loan Agreement), the "Note Obligations") in the original principal amounts of \$35,777,223.52 and \$64,942,214.21, respectively. The Notes are secured by the collateral (the "Note Collateral") described in Section 2.1 of that certain Security Agreement dated June 27, 2008 (the "Notes Security Agreement")⁶ between the SPA Borrower and the Notes Collateral Agent (as defined in the Notes Security Agreement).
- 21. On April 30, 2020, the SPA Borrower entered into that certain Amended and Restated Subordination and Intercreditor Agreement with Comerica Bank (in such capacity, the "Senior Agent") and the Investors (as may be amended, modified, or supplemented from time to time, the "Senior Intercreditor Agreement") with respect to the Note Obligations and the Security Documents (each as defined in the Prepetition Term Loan Agreement).
- 22. Pursuant to that certain Non-Recourse Securities Assignment Agreement, effective as of January 27, 2023, Volunteer purchased and received all right, title, and interest in the Notes, the related Note Obligations, and all rights and obligations under the applicable transaction

⁶ The Notes Security Agreement was executed in connection with the original Prepetition Term Loan Agreement.

documents related thereto (the "Note Purchase"). From and after the closing of the Note Purchase, Volunteer became the Notes Collateral Agent and an Investor under the Prepetition Term Loan Agreement.

- 23. As of the Petition Date, approximately \$24,677,963.78 in Senior Note Obligations and approximately \$158,372,630.41 in Junior Note Obligations remain outstanding, plus other interest, fees and all other obligations under the Prepetition Term Loan Agreement, each of the Prepetition Loans and the other documents related thereto (collectively, the "Prepetition Note Obligations"), which obligations are secured by liens (the "Senior Note Liens" and "Junior Note Liens," respectively) on the Notes Collateral (collectively, the "Prepetition Note Liens").
- 24. On May 27, 2025, the SPA Borrower, VWS Acquisitions, and the Investors entered into a third amendment to the Prepetition Term Loan Agreement, which created a new class of notes (*i.e.*, the Bridge Notes) to provide funding for the Debtors to prepare for and commence these Chapter 11 Cases, on the condition that the SPA Borrower and VWS Acquisitions file necessary and appropriate pleadings with the Court seeking that the Court approve the "roll up" of the Bridge Notes Obligations in connection with the DIP Facility. As of the Petition Date, approximately \$500,000 in obligations of the SPA Borrower under the Bridge Notes, the Security Documents, and, to the extent that they relate to or arise out of the Bridge Notes, the other Transaction Documents (the "Bridge Note Obligations") remain outstanding, which obligations are secured by senior liens on the Prepetition Notes Collateral (the "Bridge Note Liens"). The Bridge Note Liens rank *pari passu* with the Senior Notes Liens.

C. The Debtors' Immediate Need for Postpetition Financing and Access to Cash Collateral.

25. The Debtors have limited liquidity to continue operating in the ordinary course and to cover ongoing administrative expenses. To maintain administrative solvency, the Debtors

require immediate debtor-in-possession financing. Without access to the DIP Facility and authorization to use cash that is currently subject to the Prepetition Liens and thus constitutes Cash Collateral, the Debtors would be unable to sustain operations, pay employees or meet critical vendor obligations.

26. The CRO, the Debtors and their advisors considered the Debtors' current liquidity position and developed a 13-week cash flow budget, copy of which is attached to the proposed Interim Order as Exhibit 2 (the "Initial Budget"). The Initial Budget is premised on reasonable assumptions, including the Debtors' working capital requirements, cash disbursements, and required payments and includes all expenditures for which the Debtors seek authority to pay pursuant to the First Day Motions, if approved by the Court. The Debtors believe the Initial Budget will be adequate to pay all administrative expenses due or accruing during the period covered by the Initial Budget. As cash on hand is insufficient to fund the costs associated with the Debtors' operations and Chapter 11 Cases, the Debtors determined that the postpetition financing provided by the DIP Facility is immediately necessary to cover substantial shortfalls in working capital for the Debtors' business and satisfy costs associated with the Chapter 11 Cases. As such, the Debtors believe the DIP Facility is fundamental to the preservation and maintenance of the Debtors' going concern value.

D. The Debtors' Prepetition Efforts to Obtain DIP Financing

27. The DIP Lenders' willingness to provide the DIP Facility on the terms and subject to the conditions set forth therein is predicated on the Debtors implementing a sale process to sell the Landfill. Thus, the Debtors, in consultation with their advisors and the CRO, engaged in negotiations with the DIP Lenders to establish the Initial Budget that would secure the DIP Lenders' commitment to finance these Chapter 11 Cases. After good faith, arm's length

negotiations, the Debtors and the DIP Lenders reached an agreement on the terms of the proposed DIP Facility and the Debtors believe this financing will ensure adequate liquidity on reasonable and market terms to cover accruing administrative expenses throughout these Chapter 11 Cases, supporting the Debtors' pursuit of a meaningful sale process. This financing will allow the Debtors immediate access to up to \$1,800,000.00 upon approval of the Interim Order and the remainder upon entry of the Final Order, to continue paying its operating expenses and proceed with a value-maximizing sale process that is in the best interests of all of the Debtors' stakeholders.

- As described in the First Day Declaration, due to a combination of (a) the massive increase in costs to remove effluent leachate from the Landfill and (b) royalty payments being significantly reduced based on inappropriately claimed credits against the royalties by Swift Creek Renewables ("SCR"), the Debtors determined that there was an immediate need for additional liquidity to fund their operations in the near term pending a bankruptcy filing and approval of a debtor-in-possession financing facility, and to fund professionals retained for the preparation of the bankruptcy within approximately ten days prior to the Petition Date. As a result, the Debtors did not have sufficient time or funds prior to the Petition Date to explore alternatives to the Bridge Notes and DIP Facility. As of the date hereof, the DIP Facility represents the most favorable—indeed, the *only*—financing option to provide a runway for the Debtors to conduct a value-maximizing, court-approved sale process for the Landfill.
- 29. As set forth in the DIP Declaration, as a condition to provide the DIP Facility, the Debtors were required to provide both the Interim Roll-Up Amount and the stipulations as provided in the Interim Order. In addition, the terms of the DIP Facility are fair and reasonable under the circumstances, and reflective of the market for financings of this type. The DIP Lenders required the Interim Roll-Up Amount in the context of their commitment to provide the DIP

Facility on the terms described herein. The Debtors, on the one hand, and the DIP Lenders, on the other, engaged in arms' length negotiations and ultimately agreed to the Interim Roll-Up Amount as consideration for, among other things, the DIP Lenders making available additional new money to fund the Chapter 11 Cases. Indeed, without the Interim Roll-Up Amount, the DIP Lenders would not have agreed to provide debtor-in-possession financing.

E. The DIP Facility is Necessary to Preserve the Value of the Debtors' Estates.

30. It is essential to the Debtors' administration that they have sufficient funds to operate in a manner sufficient to maintain the Debtors' going-concern value as the Debtors undertake a sale process. Absent access to the DIP Facility, the Debtors will have extremely limited liquidity and would be unlikely to cover their working capital needs and the costs of these Chapter 11 Cases. Thus, the DIP Facility is necessary to preserve the value of the Debtors' estates for the benefit of the Debtors' stakeholders and avoid irreparable harm.

CONCISE SUMMARY OF TERMS OF THE DIP FACILITY

A. Concise Statement Regarding the DIP Facility

31. Under the disclosure requirements of Bankruptcy Rule 4001(b), (c), and (d) and Local Rule 4001-2(a)(i) and (ii), the following table concisely summarizes the significant terms of the DIP Facility and the Interim Order:

Summary of Material Terms	Reference
WS Holdco, Inc.	DIP Credit Agreement, Introduction

Bankruptcy Rules	Summary of Material Terms	Reference
Guarantors Bankruptcy Rule 4001(c)(1)(B)	Shoosmith Bros., Inc.	DIP Credit Agreement, Introduction
DIP Agent Bankruptcy Rule 4001(c)(1)(B)	Volunteer Enterprises, LLC	DIP Credit Agreement, Introduction
DIP Lenders Bankruptcy Rule 4001(c)(1)(B)	The Persons listed on <u>Annex B</u> to the DIP Credit Agreement and any other Person that becomes party to the DIP Credit Agreement pursuant to an Assignment and Assumption, other than any such Person that ceases to be a party to the DIP Credit Agreement pursuant to an Assignment and Assumption.	DIP Credit Agreement, Introduction
Amount and Facility Bankruptcy Rule 4001(c)(1)(B) Local Rule 4001-2(a)(i)(E) Local Rule 4001-2(a)(ii)	A senior secured superpriority debtor-in-possession financing facility in an aggregate amount of \$5,400,000 (with \$4,900,000 in new money), which DIP Facility shall be available, subject to the terms and conditions set forth in the Interim Order and the other DIP Loan Documents, as follows (in each case, subject to the terms and conditions set forth in the DIP Loan Documents: (i) the DIP Lenders shall make available during the Interim Period, in one or more additional draws, the First Interim DIP Loan in the aggregate amount of \$1,800,000.00, to include a \$500,000.00 Interim Roll-Up Amount; (ii) the DIP Lenders shall make available in one or more additional draws the Second Interim DIP Loan in the aggregate amount of \$1,550,000.00; and (iii) the DIP Lenders shall make available in one or more additional draws the Third Interim DIP Loan in the aggregate amount of \$1,550,000.00.	DIP Credit Agreement § 2.1
Maturity Bankruptcy Rule 4001(c)(1)(B)	"Maturity Date" means the date that is the earliest of (i) the date that is one-hundred ten (110) days after the Petition Date, (ii) the effective date or date of the substantial consummation (as defined in Section 1101(2) of the Bankruptcy Code) of a Plan of Reorganization or similar dispositive restructuring plan	DIP Credit Agreement, Annex A

Bankruptcy Rules	Summary of Material Terms	Reference
Local Rule 4001-2(a)(ii)	that has been confirmed by an order of the Bankruptcy Court; (iii) the date the Bankruptcy Court orders the conversion of the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code; (iv) the date the Bankruptcy Court orders the dismissal of the Chapter 11 Cases; (v) the date on which the Borrower consummates a sale of all or substantially all of the assets of the Borrower or its Subsidiaries pursuant to Section 363 of the Bankruptcy Code or otherwise; and (vi) the occurrence of any Event of Default or the date of acceleration of the DIP Term Loans or early termination of the DIP Term Loan Commitments under the DIP Credit Agreement or the DIP Orders, including as a result of the occurrence of an Event of Default.	Interim Order ¶ 21(a)
Use of Proceeds Bankruptcy Rule 4001(c)(1)(B)	Unless otherwise approved by the DIP Agent in writing (at the direction of the Required Lenders), the proceeds of the DIP Term Loans shall be used by the Borrower strictly in accordance with the Budget and subject to the covenant set forth in Section 6.18(B) of the DIP Credit Agreement (subject to the Permitted Variances).	DIP Credit Agreement § 6.14
Security and Priority Bankruptcy Rule 4001(c)(1)(B)(i) Local Rule 4001-2(a)(i)(d) and (G) Local Rule 4001-2(a)(ii)	"DIP Collateral" shall mean all assets, interests, rights, and property of any nature whatsoever of the Debtors, including, without limitation, all property in which the Debtors and their estates have an interest (whether tangible, intangible, real, personal or mixed), whether now owned or hereafter acquired and wherever located, before or after the Petition Date, including, without limitation, all accounts, proceeds of leases, inventory, equipment, equity interests or capital stock in subsidiaries, investment property, instruments, chattel paper, contracts, patents, copyrights, trademarks and other general intangibles, commercial litigation claims, cash, any investment of such cash, inventory, accounts receivable, including intercompany accounts (and all rights associated therewith), other rights to payment whether arising before or after the Petition Date, any deposit accounts, and "cash collateral accounts", and, in each case all amounts on deposit therein from time to time, the proceeds of all claims or causes of action, and all rents, products, offspring, profits, proceeds, and substitutions thereof (including, without limitation, all Prepetition Collateral and, subject to the entry of the	Interim Order ¶ 2(e), (f)

Bankruptcy Rules	Summary of Material Terms	Reference
	Final Order, all claims or causes of action of the Borrower arising under sections 502(d), 542, 544, 545, 547, 548, 549, 550, and 553 of the Bankruptcy Code and any other avoidance or similar action under the Bankruptcy Code or similar state law avoidance actions under chapter 5 of the Bankruptcy Code (the "Avoidance Actions") provided that the lien on Avoidance Actions shall be limited to the proceeds and property recovered in connection therewith).	
	 All DIP Obligations shall be secured by the following: pursuant to section 364(c)(2) of the Bankruptcy Code, valid, binding, continuing, enforceable, fully perfected first-priority liens on and security interests in all DIP Collateral that is not subject to any liens or encumbrances immediately prior to the Petition Date, subject only to the Carve-Out; and pursuant to section 363(c)(3) and 364(d)(1) of the Bankruptcy Code, valid, binding, continuing, enforceable, fully perfected first priority, priming 	
	liens on and security interests in all other DIP Collateral, which liens and security interests shall be subject only to (a) any valid, enforceable, perfected, and non-avoidable lien or security interest in favor of any person other than the Prepetition Secured Parties that was in existence immediately prior to the Petition Date or that is perfected as permitted by Section 546(b) of the Bankruptcy Code, in each case, with respect to any DIP Collateral comprised of Prepetition Collateral, solely to the extent such lien is senior to the Prepetition Liens (a "Permitted Encumbrance") and (b) the Carve-Out, and senior to all other liens and encumbrances in respect of the DIP Collateral.	

Bankruptcy Rules	Summary of Material Terms	Reference
Interest Rate Bankruptcy Rule 4001(c)(1)(B) Local Rule 4001-2(a)(ii)	Each Loan (including the PIK Amounts) shall bear interest at a rate <i>per annum</i> equal to the Interest Rate. "Interest Rate" means a rate <i>per annum</i> equal to twelve percent (12.0%).	DIP Credit Agreement § 3.2(A); Annex A
Default Interest Bankruptcy Rule 4001(c)(1)(B)	Subject to Section 8.3 of the DIP Credit Agreement (Rights and Remedies Cumulative; Non-Waiver), immediately upon the occurrence and during the continuance of an Event of Default, the Borrower shall pay interest on the aggregate outstanding principal amount of the DIP Term Loans and Interim Roll-Up Amount at the Default Rate, such interest being due and payable on demand. "Default Rate" means an interest rate equal to the Interest Rate plus 2.0%.	DIP Credit Agreement § 3.2(B); Annex A
Fees	Not applicable.	
Bankruptcy Rule 4001(c)(1)(B)		
Local Rule 4001-2(a)(ii)		
Optional Prepayments Bankruptcy Rule 4001(c)(1)(B)	Generally permitted without premium or penalty, provided that each optional prepayment of the DIP Facility must be in an aggregate principal amount of at least \$500,000 or any whole multiple of \$100,000 in excess thereof.	DIP Credit Agreement § 2.5
Mandatory Prepayments Bankruptcy Rule 4001(c)(1)(B)	Required with respect to Net Cash Proceeds in respect of any voluntary or involuntary Asset Disposition, Extraordinary Receipts, or Debt Issuances.	DIP Credit Agreement § 2.6

Bankruptcy Rules	Summary of Material Terms	Reference
Events of Default Bankruptcy Rule 4001(c)(1)(B) Local Rule 4001-2(a)(ii)	Customary events of default (e.g., failure to make payments when due, defaults under other agreements, breach of covenants, judgments exceeding an aggregate amount of \$2,000,000) plus, including but not limited to, (i) any party contesting the validity of the DIP Term Loan Documents or any such documents cease to be effective, (ii) an order lifting the stay is entered permitting foreclosure of assets exceeding \$100,000 in value or causing some other material adverse effect, (iii) inability to conduct business affairs, and (iv) an order appointing a Chapter trustee or converting the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code.	DIP Credit Agreement § 8.1
Conditions Precedent to Borrowing Bankruptcy Rule 4001(c)(1)(B) Local Rule 4001-2(a)(ii)	Customary conditions precedent (e.g., executed DIP Credit Agreement, executed of related documents, payment of fees and expenses, delivery of certain certificates, bring-down of representations and warranties, no defaults) plus, including but not limited to: (i) delivery of the Initial Approved Budget no later than two (2) day prior to the Petition Date, (ii) the Petition Date having occurred, (iii) no material adverse effect since the Petition Date, and (iv) entry of the Interim Order within five (5) business days of the Petition Date and being in full force and effect.	DIP Credit Agreement §§ 4.1 & 4.2
Indemnification Bankruptcy Rule 4001(c)(1)(B)(ix)	Under the DIP Credit Agreement, the Borrower agrees to defend, indemnify, pay, and hold harmless the DIP Agent, the Prepetition Secured Parties, and the DIP Secured Parties, including the Affiliates, partners, directors, officers, employees, agents, trustees, administrators, managers, advisors, and representatives of each party and its affiliates.	DIP Credit Agreement § 10.2(B)

Bankruptcy Rules	Summary of Material Terms	Reference
506(c) and 552(b) Waiver; Marshaling Bankruptcy Rule 4001(c)(1)(B)(x) Local Rule 4001-2(a)(i)(C)	Subject to entry of the Final Order, except to the extent of the Carve Out, the DIP Liens and Adequate Protection Liens shall not be subject to section 510 of the Bankruptcy Code, to the "equities of the case" exception of section 552 of the Bankruptcy Code, or to section 506(c) of the Bankruptcy Code, or sections 549, 550, or 551 of the Bankruptcy Code.	Interim Order ¶ L
DIP Budget and Related Covenants Bankruptcy Rule 4001(c)(1)(B)	The DIP Budget attached to the Interim Order as Exhibit 2 constitutes an "Initial DIP Budget" under the DIP Credit Agreement. The Borrower will subsequently provide the DIP Agent with an updated budget before noon every Friday of each week commencing with the second week following the Petition Date, prior to the end of the DIP Budget Period (defined below) for the remaining weeks left under the Initial DIP Budget. The "DIP Budget Period" shall refer to period commencing on the date that the Interim Order is entered and ending on the Outside Effective Date (as defined in the DIP Credit Agreement). Each updated budget shall be subject to the approval of the DIP Lenders in their sole discretion (each such approved budget, a "Supplemental Approved DIP Budget" and together with the Initial DIP Budget, the "DIP Budget"); provided that until such time as the DIP Agent approves any updated budget, the Borrower shall be subject to and be governed by the terms of the Initial DIP Budget or such Supplemental Approved DIP Budget, as applicable, then in effect in accordance with the Interim Order. The DIP Budget may only be amended, supplemented, modified, restated, replaced, or extended in accordance with the Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement). Any such amendment, supplement, modification, restatement, replacement, or extension in accordance with the DIP Credit Agreement may be effected without further order of the Court, so long as approved by the Required Lenders (as such term is defined in the DIP Credit Agreement) of DIP Lenders.	Interim Order ¶ 5

Bankruptcy Rules	Summary of Material Terms	Reference
	Pursuant to the terms of the DIP Loan Documents (including the DIP Credit Agreement), the Borrower shall provide to the DIP Agent a budget variance report (the "Budget Variance Report") before noon on Friday that is the second week following the Petition Date, and measured weekly thereafter, and the DIP Budget Variance Report shall set forth in reasonable detail actual receipts and disbursements of Borrower, together with a statement certifying compliance with the DIP Budget Covenant (as defined below) set forth below. The Borrower shall not permit or suffer to exist, a negative variance of 15% or more of total the total spending set forth in the Budget for such period (as reflected in the DIP Budget) in the aggregate from the amounts set forth in the DIP Budget (each, a "Permitted Variance"), tested on a cumulative basis (the "DIP Budget Covenant"). The Debtors shall operate in accordance with the DIP Budget and all disbursements of the Debtors shall be consistent with the provisions of the DIP Budget, subject in each case to any Permitted Variance provided in the Interim Order and in the DIP Credit Agreement.	
Carve Out Bankruptcy Rule 4001(c)(1)(B) Local Rule 4001-2(a)(i)(F)	The sum of: (i) all fees required to be paid to the Clerk of the Bankruptcy Court ("Court Fees") and all statutory fees required to be paid to the U.S. Trustee under section 1930(a) of title 28 of the United States Code plus interest at the statutory rate ("US Trustee Fees") (without regard to the notice set forth in paragraph 13(iii) of the Interim Order), (ii) up to the amounts set forth in the respective DIP Budget line items and to the extent allowed by the Court pursuant to a fee application or fee statement on notice or other procedure permitted by any Court order allowing interim compensation or the payment of fees of ordinary course professionals, whether by interim order, final order, procedural order, or otherwise, all reasonable and documented unpaid fees and expenses, all unpaid fees and expenses (the "Allowed Professional Fees") incurred by the Debtors' directors or persons or firms retained by the Debtors pursuant to section 327, 328, or 363 of the Bankruptcy Code (collectively, the "Debtor Professionals") and the	Interim Order ¶ 13

Bankruptcy Rules	Summary of Material Terms	Reference
	Committee, if any, appointed in the Bankruptcy Case (the "Committee Professionals," and together with the Debtor Professionals, the "Professionals") at any time on or before one (1) business day following delivery by the DIP Agent of a Carve-Out Trigger Notice, whether allowed by the Bankruptcy Court prior to or after delivery of a Carve-Out Trigger Notice, and (iii) Allowed Professional Fees of Professionals in an aggregate amount not to exceed \$100,000 incurred after delivery by the DIP Agent of the Carve Out Trigger Notice (the amount set forth in clause (iii) being the "Post Carve-Out Trigger Notice Cap"); provided, that the Post Carve-Out Trigger Notice Cap shall be reduced, dollar-for-dollar, by the amount of any fees and expenses incurred and accruing by the Debtors and paid to the applicable Professionals following delivery of a Carve-Out Trigger Notice. For purposes of the Interim Order, "Carve-Out Trigger Notice" shall mean a written notice delivered by email (or other electronic means) by the DIP Agent to counsel to the Debtors, the U.S. Trustee, and counsel to the Committee (if any), which notice may be delivered following the occurrence and during the continuation of an Event of Default, stating that the Post Carve-Out Trigger Notice Cap has been invoked.	
Entities with Interests in Cash Collateral Bankruptcy Rule 4001(b)(l)(B)(i)	The DIP Secured Parties and the Prepetition Secured Parties.	Interim Order ¶ (i)
Effect of Stipulations/ Challenge Bankruptcy Rule 4001(c)(1)(B)(iii) Local Rule 4001-	The Debtors' Stipulations shall be binding on the Debtors and their estates upon entry of the Interim Order and shall be binding on each party in interest (other than the Debtors), except to the extent such party in interest with standing: • first, timely commences a Challenge (A) objecting to or challenging any of the Stipulations or (B) otherwise asserting or prosecuting any action	Interim Order ¶ 11

Bankruptcy Rules	Summary of Material Terms	Reference
2(a)(i)(B)	against the Prepetition Secured Parties or any of their respective affiliates, agents, attorneys, advisors, professionals, officers, directors, or employees in connection with or related to the matters covered by the Stipulations (each such proceeding or appropriate pleading commencing a proceeding or other contested matter, a "Challenge"), no later than seventy-five (75) calendar after the entry of the Interim Order for any party in interest with requisite standing (such period, the "Challenge Period"), as such date may be extended to any such party in interest by the Prepetition Agent and each applicable Prepetition Secured Party that is the subject of a Challenge or by any such later date as has been ordered by the Court for cause upon a motion filed and served within the Challenge Period (before giving effect to such extension); and • second, the Court enters judgment in favor of the plaintiff or movant in any such timely and properly commenced Challenge proceeding, and any such judgment has become final and is not subject to any further review or appeal.	
Adequate Protection Bankruptcy Rules 4001(b)(l)(B)(iv) & 4001(c)(1)(B)(ii)	The Prepetition Secured Parties are entitled to receive adequate protection to the extent of any aggregate Diminution of Value of their interests in the Prepetition Collateral. As adequate protection, subject in all respects to the Carve-Out, the Prepetition Secured Parties will receive: • replacement liens and security interests in all DIP Collateral, provided that the replacement liens granted to the Prepetition Secured Parties will be immediately junior to the DIP Liens; the Carve-Out, and any Permitted Encumbrances; • allowed adequate protection superpriority	Interim Order ¶ 4
	administrative expense claims with priority over all administrative expense claims and other unsecured claims against the Debtors and their estates, junior	

Bankruptcy Rules	Summary of Material Terms	Reference
	 only to the DIP Superpriority Claim and the Carve-Out; and adequate protection payments from the Debtors consisting of (1) payment in cash, as and when due under the terms of the Prepetition Term Loan Agreement, of all interest due under the Prepetition Term Loan Agreement, calculated at the default rates under the Prepetition Term Loan Agreement, and (2)(x) upon entry of the Interim Order, immediate cash payment of all accrued and unpaid fees and disbursements owing to the Prepetition Secured Parties under the Prepetition Term Loan Agreement, whether incurred prior to or after the Petition Date and (y) thereafter, when due, all fees and other amounts (including all reasonable and documented legal and advisory fees, out-of-pocket costs and expenses of legal, financial, and other advisory professionals of the Prepetition Secured Parties), provided that payment of any fees and expenses of the Prepetition Secured Parties' professionals incurred after entry of the Interim Order shall be subject to the notice and objection provisions of the Interim Order. 	
Milestones Bankruptcy Rule 4001(c)(1)(B)(v) & (vi)	 The Loan Parties shall comply with and meet the following milestones as the same may be extended, waived, or otherwise modified with the prior written consent of the Required Lenders in their sole discretion (which may be by email) without further order by the Bankruptcy Court (collectively, the "Chapter 11 Milestones" and each a "Chapter 11 Milestone"): The Bankruptcy Court shall have entered the Interim Order by the date that is no later than five (5) Business Days after the Petition Date; The Debtors shall have filed the motion seeking entry of the Bidding Procedures Order and Sale Order (the "Bidding Procedures Motion") by no later than ten (10) days after the Petition Date; 	DIP Credit Agreement § 6.16

Bankruptcy Rules	Summary of Material Terms	Reference
	• The Bankruptcy Court shall have entered the Final Order by the date that is no later than thirty-three (33) days after the Petition Date;	
	• The Bankruptcy Court shall have entered the order approving the Bidding Procedures Motion by no later than thirty-three (33) days after the filing of the Bidding Procedures Motion;	
	• The Debtors shall have (a) entered into and delivered a fully executed and binding Acceptable Stalking Horse Purchase Agreement and (b) satisfied any other applicable Chapter 11 Milestones by no later than the date of entry of the Final Order;	
	• The Bankruptcy Court shall have entered the order approving the Sale Transaction by no later than eighty (80) days after the Petition Date;	
	• The Debtors shall have closed the Sale Transaction by no later than eighty-five (85) days after the Petition Date; and	
	• The Plan Effective Date shall have occurred by no later than one hundred ten (110) days after the Petition Date.	
Automatic Stay Waiver/ Modification Bankruptcy Rule 4001(c)(1)(B)(iii)	The automatic stay imposed under section 362(a) of the Bankruptcy Code will be modified pursuant to the Interim Order and the DIP Loan Documents (including the DIP Credit Agreement) as necessary to (i) permit the Debtors to grant the Adequate Protection Liens and the DIP Liens and to incur all DIP Obligations and all liabilities and obligations to the Prepetition Secured Parties under the Interim Order and under the DIP Credit Agreement, as the case may be, and (ii) authorize the DIP Agent and Prepetition Agent to retain and apply payments and otherwise enforce their respective rights and remedies of the Interim Order, subject to the provisions of paragraph 21(b) of the Interim Order.	Interim Order ¶ 21(d)

Bankruptcy Rules	Summary of Material Terms	Reference
Liens on Avoidance Actions	DIP liens granted on proceeds and property recovered in connection with Avoidance Actions under the Bankruptcy Code or similar state law, subject to entry of Final Order.	Interim Order ¶ 2(e)
Bankruptcy Rule 4001(c)(1)(B)(xi)		
Local Rule 4001-2(a)(i)(D)		

PROVISIONS TO BE HIGHLIGHTED PURSUANT TO LOCAL RULE 4001-2

- Cross-Collateralization Provisions (Local Rule 4001-2(a)(i)(A)): Neither the Interim Order nor the Final Order provides for cross collateralization, other than replacement liens as adequate protection.
- Findings of Fact (Local Rule 4001-2(a)(i)(B)): Pursuant to paragraph G of the Interim Order, the Debtors have provided certain agreements and stipulations concerning the Prepetition Obligations. These stipulations and findings of fact were required by the DIP Lenders in exchange for providing the DIP Facility and are reasonable under the circumstances discussed in this Motion. Moreover, the stipulations are subject to challenge by a Committee (if any) or any other party in interest (subject to the limitations set forth in paragraph 11 of the Interim Order) and will not be binding upon a Committee (if any) or any other party in interest until after the Challenge Period. The Challenge Period is seventy-five (75) calendar days after the entry of the Interim Order for any party in interest with requisite standing, as such date may be extended to any such party in interest by the Prepetition Agent and each applicable Prepetition Secured Party that is the subject of a Challenge or by any such later date as has been ordered by the Court for cause upon a motion filed and served within the Challenge Period (before giving effect to such extension). See Interim Order¶ 11.
- 506(c) Waiver (Local Rule 4001-2(a)(i)(C)): Subject to the entry of a Final Order, no costs or expenses of administration of the Chapter 11 Cases shall be charged against or recovered from or against any or all of the DIP Secured Parties, the Prepetition Secured Parties or the Prepetition Collateral, the DIP Collateral, and the Cash Collateral, in each case pursuant to section 506(c) of the Bankruptcy Code or otherwise, without the prior written consent of the DIP Agent and the Prepetition Agent (with respect to Prepetition Collateral). No such consent shall be implied from any other action, inaction, or acquiescence of any or all of the DIP Secured Parties and the Prepetition Secured Parties. This waiver was required by the DIP Lenders in exchange for providing the DIP Facility and is reasonable under the

circumstances discussed in this Motion. As such relief is only being requested in connection with the Final Order, parties will have notice of (and an opportunity to object to) the Debtors' proposed waiver. See Interim Order¶ 25.

- Liens on Avoidance Actions (Local Rule 4001-2(a)(i)(D)): Pursuant to paragraph 2(e) of the Interim Order, the Debtors are seeking approval of DIP Liens and Adequate Protection Liens on the proceeds of Avoidance Actions (but not the Avoidance Actions themselves), upon entry of the Final Order. Liens on the Avoidance Action proceeds were required by the DIP Lenders in exchange for providing the DIP Facility. Furthermore, since such relief is not being sought until the Final Order, parties will have notice of (and an opportunity to object to) such proposed liens. See Interim Order ¶ 2(e).
- Provisions Authorizing "Roll Up" or Repayment of Prepetition Debt (Local Rule 4001-2(a)(i)(E)):

Roll Up Loans. Pursuant to paragraph 2(c) of the Interim Order, upon entry of the Interim Order and the Final Order, and subject to paragraph 11 below, the Prepetition Obligations, on a 1 to 1 cashless basis, in the aggregate amounts equal to the Interim Roll-Up Amount, shall be converted into DIP Obligations, without any further action by the Debtors or any other party. Upon any of the foregoing Interim Roll-Up Amount, and subject to paragraph 11 below, the Interim Roll-Up Amount shall cease to be indebtedness under the Prepetition Term Loan Agreement and shall be deemed DIP Obligations under the DIP Facility in all respects, including for purposes of having the benefit of section 364(e) of the Bankruptcy Code. See Interim Order ¶ 2(c).

As discussed in Section VI *infra* and the DIP Declaration, the Interim Roll-Up Amount are justified under the circumstances.

- Disparate Treatment of Debtor and Creditors' Committee Professionals in Carve Out (Local Rule 4001-2(a)(i)(F)): The Interim Order (and ultimately Final Order will) contain no provision for disparate treatment for professionals retained by the Debtors and professionals retained by a Committee, if appointed, with respect to the Carve-Out. See Interim Order ¶ 13.
- Nonconsensual Priming of Prepetition Liens (Local Rule 4001-2(a)(i)(G)): The Interim Order does not (and the Final Order will not) provide for the nonconsensual priming of the Prepetition Secured Parties. The Prepetition Secured Parties have either expressly consented, or are deemed pursuant to the Prepetition Loan Documents to have consented, to the DIP Facility and the DIP Liens. See Interim Order ¶ K.
- 552(b) Waiver (Local Rule 4001-2(a)(i)(H)): Pursuant to paragraphs L and 25 of the Interim Order, the Debtors are seeking approval for the DIP Lenders and Prepetition Secured Parties to each be entitled to all of the rights and benefits of section 552(b) of the Bankruptcy Code, and for the "equities of the case" exception

under section 552(b) of the Bankruptcy Code to not apply. Furthermore, since it is not requested until the Final Order, parties will have notice of (and an opportunity to object to) the Debtors' proposed waiver. See Interim Order ¶¶ L and 25(j).

BASIS FOR RELIEF

- I. The Debtors Have Exercised Their Business Judgment in Entering Into the DIP Facility and Satisfies the Heightened Scrutiny Required for an Insider.
- the Bankruptcy Code is generally governed by the business judgment standard. *See In re Ames Dep't Stores, Inc.*, 115 B.R. 34, 40 (Bankr. S.D.N.Y. 1990) ("Cases consistently reflect that the court's discretion under section 364 is to be utilized on grounds that permit reasonable business judgment to be exercised so long as the financing agreement does not contain terms that leverage the bankruptcy process and powers or their purpose is not so much to benefit the estate as it is to benefit parties in interest."); *see also In re Farmland Indus., Inc.*, 294 B.R. 855, 881 (Bankr. W.D. Mo. 2003) (noting that approval of postpetition financing requires, inter alia, an exercise of "sound and reasonable business judgment"); *In re Trans World Airlines, Inc.*, 163 B.R. 964, 974 (Bankr. D. Del. 1994) (noting that the interim loan, receivable facility and asset-based facility were approved because they "reflect[ed] sound and prudent business judgment . . . [were] reasonable under the circumstances and in the best interest of [the debtor] and their creditors.").
- 33. However, when a transaction is between a debtor and its insider, "courts apply a 'heightened scrutiny' test in assessing the bona fides of a transaction among a debtor and an insider of the debtor." *In re LATAM Airlines Grp. S.A.*, 620 B.R. 722, 769 (Bankr. S.D.N.Y. 2020). "A claim arising from the dealings between a debtor and an insider is to be rigorously scrutinized by the courts." *Schubert v. Lucent Techs. Inc. (In re Winstar Commc'ns, Inc.)*, 554 F.3d 382, 412 (3d Cir. 2009) (quoting *Fabricators Inc. v. Tech. Fabricators, Inc. (In re Fabricators, Inc.)*, 926 F.2d 1458, 1465 (5th Cir. 1991)).

- 34. The Debtors' decision to enter into the DIP Facility is an exercise of their sound business judgment for a number of reasons, even taking into account the heightened scrutiny test for entering into transactions with insiders. First, without access to the DIP Facility, the Debtors will not be able to (i) address the immediate need to remove the effluent leachate being generated in the Landfill or (ii) pay expenses necessary to sustain their operations as they undergo the contemplated sale process, which would result in undue harm to the Debtors' estates and stakeholders. As set forth in the DIP Declaration, the Debtors' use of their current cash reserves and reliance on business operations are insufficient to meet the Debtors' working capital needs to operate their businesses in the ordinary course while simultaneously administering these Chapter 11 Cases. There is therefore an immediate need for postpetition financing as the Debtors' current finances are insufficient to allow the Debtors to continue to operate as a going concern, meet their obligations in the Chapter 11 Cases and facilitate a sale process. Indeed, failure to obtain approval of the DIP Facility potentially would lead to a wind-down of the Debtors' business operations, and create potential environmental hazards which, in turn, would adversely affect their creditors, stakeholders and their estates.
- 35. Second, the Debtors believe the terms of the DIP Facility are market, reasonable and are the best (and only) available that will provide the Debtors with sufficient liquidity to effectively operate the Debtors' businesses through the Chapter 11 Cases and make necessary administrative payments. This Court has previously approved similar debtor-in-possession financing agreements where the debtor was not able to obtain postpetition financing under other conditions. See, e.g.; In re Viridos, Inc., Case No. 25-10697 (CTG) (Bankr. D. Del. May 8, 2025) [D.I. 94]; In re BLH Topco LLC, Case No. 25-10576 (CTG) (Bankr. D. Del. Apr. 24, 2025) [D.I. 129]; In re Brightmark Plastics Renewal LLC, Case No. 25-10472 (LSS) (Bankr. D. Del. Apr. 22,

2025) [D.I. 152]; In re PetroQuest Energy, Inc., No. 24-12609 (CTG) (Bankr. D. Del. Dec. 10, 2024) [D.I. 145]; In re Smallhold, Inc., Case No. 24-10267 (CTG) (Bankr. D. Del. Apr. 17, 2024) [D.I. 129].

36. Third, the DIP Facility was the product of arms'-length discussions and negotiations. The Debtors, with the assistance of their advisors, analyzed their cash needs to determine the liquidity levels necessary to maintain the Debtors' going concern value and undertake a sale process. In considering whether to pursue postpetition financing, the Debtors and their advisors considered several factors in the exercise of their business judgment, including (i) the need for financing based on the Debtors' operations, (ii) the Debtors' inability to administer these Chapter 11 Cases in an orderly and stabilized manner without additional financing and (iii) the Debtors' inability to preserve value of their operations and estates without additional financing. After receiving an initial proposal from the DIP Lenders, the Debtors' management team determined that the proposal was an attractive offer and, most importantly, provided the much-needed liquidity to allow the Debtors to continue to fund their operations and administrative obligations during the course of the Chapter 11 Cases. The Debtors' management team, as well as the Debtors' advisors, continued to engage with the DIP Lenders to obtain more favorable terms than its initial proposal, resulting in a materially improved proposal that will allow for a smooth operation in the Chapter 11 Cases and a path towards successful Chapter 11 Cases. During this time the Debtors' advisors also conducted a reach-out to over 20 potential middle market lenders experienced in the distressed space, as well as larger, institutional banks for completeness. Despite such efforts, as set forth above, the Debtors were unable to obtain any alternative source of financing due to, among other things, the timing and the amount of the financing needed and the Debtors' environmental issues.

37. Accordingly, pursuant to sections 364(c) and (d) of the Bankruptcy Code, the Debtors respectfully submit that they should be granted authority to obtain financing from the DIP Lenders on the terms set forth in the DIP Facility.

II. The Debtors Should Be Authorized to Obtain Postpetition Financing on a Senior Secured Superpriority Basis.

- 38. In connection with the DIP Facility, the Debtors will provide the DIP Lenders with priming security interests, liens, and superpriority claims pursuant to section 364(c) and (d)(1) of the Bankruptcy Code. The Debtors meet the requirements of section 364(c) of the Bankruptcy Code, which authorizes a debtor to incur secured or superpriority debt under certain circumstances. Specifically, section 364(c) of the Bankruptcy Code provides that:
 - (c) If the trustee is unable to obtain unsecured credit allowable under section 503(b)(1) of this title as an administrative expense, the court, after notice and a hearing, may authorize the obtaining of credit or the incurring of debt
 - (1) with priority over any or all administrative expenses of the kind specified in section 503(b) or 507(b) of this title;
 - (2) secured by a lien on property of the estate that is not otherwise subject to a lien; or
 - (3) secured by a junior lien on property of the estate that is subject to a lien[.]

11 U.S.C. § 364(c).

39. In evaluating proposed postpetition financing under section 364(c) of the Bankruptcy Code, courts perform a qualitative analysis and consider whether (a) the debtor is unable to obtain unsecured credit under section 364(b), *i.e.*, by allowing a lender only an administrative claim pursuant to section 503(b)(1)(A), (b) the credit transactions are necessary to preserve assets of the estate, and (c) the terms of the credit agreement are fair, reasonable, and adequate. See, e.g., In re L.A. Dodgers, 457 B.R. at 312; In re Aqua Assoc., 123 B.R. 192,

- 195–96 (Bankr. E.D. Pa. 1991); Norris Square Civic Ass'n v. Saint Mary Hosp., 86 B.R. 393, 401 (Bankr. E.D. Pa. 1988); In re Crouse Grp., Inc., 71 B.R. 544, 549 (Bankr. E.D. Pa. 1987).
- 40. Furthermore, in the event that a debtor is unable to obtain unsecured credit allowable as an administrative expense under section 503(b)(1) of the Bankruptcy Code, section 364(c) provides that a court "may authorize the obtaining of credit or the incurring of debt (1) with priority over any or all administrative expenses of the kind specified in section 503(b) or 507(b) of the Bankruptcy Code; (2) secured by a lien on property of the estate that is not otherwise subject to a lien; or (3) secured by a junior lien on property of the estate that is subject to a lien." 11 U.S.C. § 364(c).
- 41. As described in the DIP Declaration, the terms of the DIP Facility are the product of extensive and good faith negotiations between the Debtors, their advisors, the CRO, and the DIP Lenders, which are the best and only reasonable financing option currently available under the circumstances. The DIP Facility is the Debtors' only viable source of funding, whether on an out-of-court or in-court basis, and no other, better alternative, taken as a whole, is currently available to the Debtors under the circumstances.
- 42. The Debtors also meet the requirements for relief under section 364(d) of the Bankruptcy Code, which permits a debtor (with Court authorization) to obtain postpetition financing secured by liens that are senior to prepetition liens. Specifically, section 364(d) of the Bankruptcy Code provides as follows:
 - (1) The court, after notice and a hearing, may authorize the obtaining of credit or the incurring of debt secured by a senior or equal lien on property of the estate that is subject to a lien only if:
 - (A) the trustee is unable to obtain such credit otherwise; and
 - (B) there is adequate protection of the interest of the holder of the lien on the property of the estate on which such senior or equal lien is proposed to be granted,

(2) In any hearing under this subsection, the trustee has the burden of proof on the issue of adequate protection.

11 U.S.C. § 364(d). Here, the Debtors may incur "priming" liens under the DIP Facility because the Prepetition Secured Parties have consented. Therefore, the Debtors submit that the requirements of section 364 of the Bankruptcy Code are satisfied.

III. The Prepetition Secured Parties Are Adequately Protected.

- 43. Under section 361 of the Bankruptcy Code, the Debtors may provide adequate protection by making a cash payment or periodic cash payments "to such entity, to the extent that ...any grant of a lien under section 364 of this title results in a decrease in the value of such entity's interest in such property." 11 U.S.C. § 361(1). Adequate protection may also be provided by providing to such entity an additional or replacement lien to the extent that such stay, use, sale, lease, or grant results in a decrease in the value of such entity's interest in such property. *Id.* § 361(2).
- 44. The Debtors request that the Court approve certain protections of the Prepetition Secured Parties' interests in the Prepetition Collateral from any aggregate diminution in value resulting from (i) the use, sale, or lease by the Debtors of the applicable Prepetition Collateral (including any Cash Collateral), (ii) the imposition or enforcement of the automatic stay of section 362(a) of the Bankruptcy Code, or (iii) any other reason for which adequate protection may be granted under the Bankruptcy Code. The Debtors request that the Court approve the following forms of adequate protection:
 - Pursuant to sections 361, 363(e), and 364(d) of the Bankruptcy Code, effective as of the Petition Date and perfected without the need for execution by the Borrower or the Guarantors or the recordation or other filing by the DIP Lenders of security agreements, control agreements, pledge agreements, financing statements, or other similar documents or the possession or control by the of any DIP Collateral, solely to the extent of any Diminution in Value, shall be granted valid, binding, continuing, enforceable, fully perfected replacement and additional liens on and security interests in (collectively, the "Adequate Protection Liens") all DIP

Collateral. The Adequate Protection Liens shall be junior only to the DIP Liens, the Carve-Out, and any Permitted Encumbrance, and senior to all other security interests in, liens on, or claims against the DIP Collateral. The Adequate Protection Liens shall be non-avoidable and shall not be subject to sections 510, 549, 550, or 551 of the Bankruptcy Code. The Adequate Protection Liens shall be enforceable against and binding upon the Borrower, the Guarantors, their estates and any successors thereto, including, without limitation, any trustee or other estate representative appointed in any Successor Case.

- To the extent of the aggregate Diminution in Value, the Prepetition Secured Parties shall be granted allowed superpriority administrative expense claims against the Borrower or the Guarantors, and their estates, as provided in section 507(b) of the Bankruptcy Code, immediately junior to the DIP Obligations and subject only to the Carve-Out, and with priority in payment over any and all other claims and administrative expense claims against the Borrower or the Guarantors, now existing or hereafter arising, of any kind or nature whatsoever, including, without limitation, administrative expenses of the kinds specified in or ordered pursuant to sections 105, 326, 327, 328, 330, 331, 361, 362, 363, 364, 365, 503(a), 503(b), 507(a), 507(b), 546(c), 546(d), 552(b), 726, 1113, 1114, and any other provision of the Bankruptcy Code, and shall at all times be senior to the rights of the Borrower and the Guarantors, and any successor trustee or creditor in the Chapter 11 Cases or any Successor Case (the "Adequate Protection Superpriority Claims").
- As further adequate protection, from and after entry of the Interim Order, the Prepetition Agent (on behalf of the Prepetition Secured Parties) shall receive from the Debtors, (1) payment in cash, as and when due under the terms of the Prepetition Term Loan Agreement, of all interest due under the Prepetition Term Loan Agreement, calculated at the default rates under the Prepetition Term Loan Agreement, and (2) (x) upon entry of the Interim Order, immediate cash payment of all accrued and unpaid fees and disbursements (including all reasonable and documented fees, out-of-pocket costs and expenses of legal, financial, and other advisory professionals of the Prepetition Secured Parties (including, without limitation, Cole Schotz P.C.) owing to the Prepetition Secured Parties under the Prepetition Term Loan Agreement, whether incurred prior to or after the Petition Date and (y) thereafter, when due, all fees and other amounts (including all reasonable and documented legal and advisory fees, out-of-pocket costs and expenses of legal, financial, and other advisory professionals of the Prepetition Secured Parties (including, without limitation, Cole Schotz P.C.) owing to the Prepetition Secured Parties by the Debtors under the Prepetition Term Loan Agreement, provided that payment of any fees and expenses of the Prepetition Secured Parties' professionals incurred after entry of the Interim Order shall be subject to the notice and objection provisions of paragraph 27(b) (the "Adequate Protection Payments"). Payment of any amounts set forth in this clause (c)(2) shall not be subject to disgorgement or recharacterization.

45. In addition to the above, the DIP Facility itself provides adequate protection against any diminution in value by maximizing the going-concern value of the Debtors' enterprise and the Prepetition Secured Parties' collateral value, each of which would be in severe peril if the DIP Facility is not approved. Courts have held enhancement of collateral is a critical component of adequate protection and have considered "whether the value of the debtor's property will increase as a result of the" use of the collateral. In re 495 Cent. Park Ave. Corp., 136 B.R. 626, 631 (Bankr. S.D.N.Y. 1992) (finding that improvements to collateral financed by postpetition financing proceeds would improve collateral value in excess of loans and, therefore, provided adequate protection); see In re Hubbard Power & Light, 202 B.R. 680, 685 (Bankr. E.D.N.Y. 1996) (approving postpetition financing to be used, in part, to fund cleanup costs of encumbered property anticipated to improve the value of the collateral, thereby serving the goal of adequate protection); VWI Props., LLC v. Mt. Olive Hosp., LLC (In re Mt. Olive Hospitality), LLC, No. 13-3395 (RBK), 2014 U.S. Dist. LEXIS 42886, at *14–16 (D.N.J. Mar. 31, 2014) (affirming the bankruptcy court's holding that renovation of a hotel provided adequate protection because the value would be enhanced by the use of cash collateral). The Debtors submit that the proposed Adequate Protection Obligations are appropriate to protect the Prepetition Secured Parties from any diminution in value of their interests in the Prepetition Collateral.

IV. The Debtors Should Be Authorized to Use Cash Collateral

46. Section 363 of the Bankruptcy Code generally governs the use of estate property and permits a debtor in possession to use cash collateral with the consent of the secured party. 11 U.S.C. § 363(c). Specifically, section 363(c)(2) of the Bankruptcy Code provides that the Debtors may not use, sell or lease cash collateral unless "(a) each entity that has an interest in such cash collateral consents; or (b) the court, after notice and hearing, authorizes such use, sale, or lease in accordance with the provisions of this section." 11 U.S.C. § 363(c)(2). Therefore, the

Debtors may not use the Cash Collateral without the consent of the Prepetition Secured Parties or authority granted by the Court. Further, section 363(e) of the Bankruptcy Code provides that on request of an entity that has an interest in property to be used by a debtor, the Court shall prohibit or condition such use as necessary to provide adequate protection of such interest. *Id.* § 363(e).

- 47. Here, the Debtors require use of Cash Collateral to, among other things, pay: (a) postpetition expenses, including vendors and service providers necessary to maintain their business operations consistent with prepetition practices; (b) certain prepetition obligations as further described in the Debtors' motions fled contemporaneously herewith; and (c) disbursements as described in the Budget. Because the Prepetition Secured Parties have consented to the use of Cash Collateral and will (subject to this Court's approval) receive the Adequate Protection Obligations, the Debtors submit that the requirements of sections 363(c)(2) and 363(e) have been satisfied, and that they should be authorized to use Cash Collateral pursuant to the Budget, the DIP Orders, and other DIP Loan Documents.
- 48. Accordingly, the Debtors respectfully request that the Court authorize the Debtors to use the Cash Collateral in accordance with the terms set forth in the Interim Order.

V. The DIP Secured Parties Should Be Afforded Good Faith Protection Under Section 364(e) of the Bankruptcy Code.

49. Section 364(e) of the Bankruptcy Code protects a good faith lender's right to collect on loans extended to a debtor, and its right in any lien securing those loans, even if the authority of the debtor to obtain such loans or grant such lien is later reversed or modified on appeal.

11 U.S.C. § 364(e). Section 364(e) of the Bankruptcy Code provides that:

The reversal or modification on appeal of an authorization under this section [364 of the Bankruptcy Code] to obtain credit or incur debt, or of a grant under this section of a priority or a lien, does not affect the validity of any debt so incurred, or any priority or lien so granted, to an entity that extended such credit in good faith, whether or not such entity knew of the pendency of the appeal, unless such authorization and the incurring of such debt, or the granting of such priority or lien,

were stayed pending appeal.;

Id.

50. The DIP Facility is the result of (a) the Debtors' reasonable and informed determination that the DIP Lenders offered the most favorable terms on which to obtain vital postpetition financing and (b) arms' length negotiations between the Debtors and the DIP Lenders. The Debtors submit that the terms and conditions of the proposed DIP Loan Documents are reasonable and appropriate under the circumstances, and the proceeds of the DIP Facility will be used only for purposes that are permissible under the Bankruptcy Code. Further, the terms and conditions of the DIP Loan Documents were negotiated in good faith and at arms' length with all parties represented by experienced counsel. Accordingly, the Court should find that the obligations arising under the DIP Facility and other financial accommodations made to the Debtors have been extended by the DIP Agent and the DIP Lenders in "good faith" within the meaning of section 364(e) of the Bankruptcy Code, and therefore, the DIP Secured Parties are entitled to all of the protections afforded thereby

VI. The Roll Ups of the Prepetition Obligations Is Appropriate

- 51. The funding provided under the Bridge Notes was done on the eve and in contemplation of the commencement of these Chapter 11 Cases to allow the Debtors to maintain their business while completing the negotiations of the DIP Facility. The Debtors submit that the proposed roll-ups of the Prepetition Obligations owed to the DIP Lenders is justified under the circumstances.
- 52. <u>First</u>, the DIP Lenders specifically negotiated for the Interim Roll-Up Amount in the context of their commitment to provide the DIP Facility on the terms described herein. The Debtors, on the one hand, and the DIP Lenders, on the other, engaged in arms' length negotiations and ultimately agreed to the Interim Roll-Up Amount as consideration for, among other things, the

DIP Lenders making available \$4,900,000 in new money to fund the Chapter 11 Cases. Indeed, without the Interim Roll-Up Amount, the DIP Lenders would not have agreed to provide the DIP Facility. Second, as discussed in the DIP Declaration, the Debtors have an immediate need to obtain the DIP Facility and use Cash Collateral and are unable to obtain financing on more favorable terms from sources other than the DIP Lenders. As such, the refinancing of the Bridge Notes Obligations as part of the DIP Facility is critical to the Debtors' obtaining access to the liquidity necessary to preserve and maximize the value of their estates. The Interim Roll-Up Amount benefit the Debtors' estates by allowing the Debtors to run a robust bidding and sale process. As such, this relief is warranted.

Finally, roll-ups and repayments of prepetition debt, such as the roll-up of the Bridge Notes Obligations, are a common feature of debtor-in-possession financings and have been approved in a variety of cases, including pursuant to interim financing orders. *See, e.g.*; *In re Kognitiv US, LLC*, No. 25-10648 (BLS) (Bankr. D. Del. Apr. 29, 2025) [D.I. 57] (authorizing a \$2 million DIP facility comprised of \$400,000 of new money and a roll-up of a \$1.6 million emergency bridge loan); *In re PetroQuest Energy, Inc.*, No. 24-12609 (CTG) (Bankr. D. Del. Dec. 10, 2024) [D.I. 145] (authorizing a \$14.2 million DIP facility comprised of \$1.7 million of new money and a roll-up of \$12.5 million, including a full roll-up of a \$2 million emergency bridge loan); *In re Never Slip Holdings, Inc.*, No. 24-10663 (LSS) (Bankr. D. Del. Apr. 26, 2024) [D.I. 133] (authorizing \$120.8 million DIP facility comprised of \$30.8 million of new money and a \$90 million roll-up); *In re Sientra, Inc.*, No. 24-10245 (JTD) (Bankr. D. Del. Mar. 11, 2024) [D.I. 168] (authorizing \$90 million DIP facility comprised of \$22.5 million of new money and a \$67.5 million roll-up); *In re Virgin Orbit Holdings, Inc.*, No. 23-10405 (KBO) (Bankr. D. Del. May 1, 2023) [D.I. 202] (authorizing a \$74 million DIP facility comprised of \$15.2 million of new money and a

roll-up of \$31.6 million, including a full roll-up of a \$10.9 million emergency bridge loan); *In re Vesta Holdings*, *LLC*, No. 22-11019 (LSS) (Bankr. D. Del. Dec. 2, 2022) [D.I. 141] (authorizing \$37.87 million DIP facility comprised of \$12.62 of new money and a \$25.25 million roll-up); *In re Phoenix Servs. Topco LLC*, Case No. 22-10906 (MFW) (Bankr. D. Del. Sept. 29, 2022) [D.I. 237] (authorizing \$100 million DIP facility comprised of \$25 million of new money and a \$75 million roll-up); *In re Nine Point Energy Holdings, Inc.*, No. 21-10570 (MFW) (Bankr. D. Del. Apr. 12, 2021) [D.I. 240] (authorizing \$13 million of new money and a roll-up of \$39 million pursuant to interim order).

VII. Modification of the Automatic Stay Is Warranted

54. The relief requested contemplates a modification of the automatic stay to permit the DIP Secured Parties to enforce their rights under the DIP Loan Documents and the proposed Interim Order. Stay modifications of this kind are typical and standard features of debtor in possession financing arrangements, and, in the Debtors' business judgment, are appropriate under the circumstances in the Chapter 11 Cases. *See, e.g., In re Express, Inc.*, No. 24-10831 (KBO) (Bankr. D. Del. April 24, 2024) [D.I. 85]; *In re Sientra, Inc.*, No. 24-10245 (JTD) (Bankr. D. Del. Feb. 14, 2024) [D.I. 61] (modifying automatic stay as necessary to effectuate the terms of the order and following occurrence of an event of default); *In re PGX Holdings, Inc.*, No. 23-10718 (CTG) (Bankr. D. Del. Jun. 6, 2023) [D.I. 70] (same); *In re SiO2 Medical Prods., Inc.*, No. 23-10366 (JTD) (Bankr. D. Del. Mar. 30, 2023) [D.I. 76] (same); *In re Carestream Health, Inc.*, No. 22-10778 (JKS) (Bankr. D. Del. Aug. 24, 2022) [D.I. 65] (same); *In re Town Sports Int'l, LLC*, No. 20-12168 (CTG) (Bankr. D. Del. Oct. 2, 2020) [D.I. 159] (same).

Because of the voluminous nature of the orders cited herein, such orders have not been attached to this Motion. Copies of these orders are available upon request of the Debtors' proposed counsel.

VIII. The Scope of the Carve-Out Is Appropriate

- 55. The liens, security interests, and administrative expense claims under the DIP Facility and with respect to the Adequate Protection Obligations are all subject to the Challenge Period and the Carve-Out. The Carve-Out protects against administrative insolvency during the course of the Chapter 11 Cases by ensuring that, notwithstanding the grant of superpriority claims and liens under the DIP Facility and Adequate Protection Obligations, assets remain for the payment of the fees of the U.S. Trustee and professional fees of the Debtors and any official committee. *See In re Ames Dep't Stores, Inc.*, 115 B.R. at 38 ("Absent such protection, the collective rights and expectations of all parties-in-interest are sorely prejudiced"); *In re Molycorp, Inc.*, 562 B.R. 67, 76 (Bankr. D. Del. 2017) ("To deal with some of the abovementioned risks[, including administrative insolvency and conversion of the case,] professionals usually negotiate a carve-out to provide for the payment of their allowed fees.").
- 56. Courts in this district and others routinely approve carve-out provisions similar to those agreed to by the Debtors and the DIP Lenders. See, e.g., In re Yellow Corp., No. 23-11069 (CTG) (Bankr. D. Del. Sept. 15, 2023) [D.I. 571]; In re Lannett Co., Inc., No. 23-10559 (JKS) (Bankr. D. Del. June 7, 2023) [D.I. 230]; In re Carestream Health, Inc., No. 22-10778 (JKS) (Bankr. D. Del. Sep. 27, 2022) [D.I. 179]; In re Armstrong Flooring, Inc., No. 22-10426 (MFW) (Bankr. D. Del. Jun. 7, 2022) [D.I. 311]; In re MD Helicopters, Inc., No. 22-10263 (KBO) (Bankr. D. Del. Apr. 26, 2022) [D.I. 205]; In re BHCosmetics Holdings, LLC, No. 22-10050 (CSS) (Bankr. D. Del. Mar. 9, 2022) [D.I. 217]; In re Alto Maipo Del. LLC, No. 21-11507 (KBO) (Bankr. D. Del. Dec. 17, 2021) [D.I. 166]; In re PWM Prop. Mgmt. LLC, No. 21-11445 (MFW) (Bankr. D. Del. Dec. 13, 2021) [D.I. 241]; In re Riverbed Tech., Inc., No. 21-11503 (CTG) (Bankr. D. Del. Nov. 18, 2021) [D.I. 89].9

IX. Failure to Obtain Immediate Interim Access to the DIP Facility and Cash Collateral Would Cause Immediate and Irreparable Harm.

- 57. Bankruptcy Rule 4001(b) and (c) provide that a final hearing on a motion to use cash collateral pursuant to section 363 of the Bankruptcy Code or to obtain credit pursuant to section 364 of the Bankruptcy Code may not be commenced earlier than fourteen (14) days after the service of such motion. Fed. R. Bankr. P. 4001. Upon request, however, pursuant to Bankruptcy Rule 4001(b) and (c), the Court may conduct a preliminary, expedited hearing on the motion and authorize the use of cash collateral and the obtaining of credit to the extent necessary to avoid immediate and irreparable harm to the debtor's estate. Local Rule 4001-2(b) provides that with respect to interim orders approving postpetition financing, "[i]nterim relief granted under the DIP Credit Agreement shall be only what is necessary to avoid immediate and irreparable harm to the estate pending a final hearing."
- 58. Here, the Debtors seek authorization to borrow under the DIP Facility and to use the Cash Collateral immediately. The Debtors will suffer immediate and irreparable harm if the Interim Order approving the DIP Facility and use of Cash Collateral on the terms and conditions thereof (and outlined here) is not entered sooner than 14 days after service of this Motion, and if the Debtors are not permitted interim access to the DIP Facility. The DIP Lenders have indicated that such terms, including those outlined above, are integral to its decision to provide the DIP Facility. The Debtors require access to the DIP Facility and Cash Collateral to, among other things, fund the administration of these Chapter 11 Cases and the operation of their business in accordance with the DIP Loan Documents and the DIP Orders, and to allow them to consummate the resolution of the Chapter 11 Cases contemplated by the milestones in the DIP Credit Agreement, for the benefit of their estates.

59. Additionally, "[i]n the absence of extraordinary circumstances, the Court shall not approve interim financing orders that include any of the provisions previously identified in Local Rule 4001-2(a)(i)(P) through 4001-2(a)(i)(X)." Del. Bankr. L.R. 4001-2(b). None of the extraordinary relief specified in Local Rule 4001-2(a) is being sought on an interim basis other than interim approval of the proposed roll up of the Interim Roll-Up Amount owed to the DIP Lenders. For the reasons set forth above, the Debtors submit that the DIP Facility is necessary to avoid immediate and irreparable harm to the Debtors' estates, and the terms thereof are appropriate under the circumstances.

X. Scheduling a Final Hearing

60. The Debtors are requesting that, following the Interim Hearing, the Court schedule the Final Hearing to be held more than fourteen (14) days after service of this Motion as required by Bankruptcy Rule 4001 and Local Rule 9006-1. In compliance with Local Rule 9006-1(c)(ii), the Debtors are requesting that the deadline to object to approval of the DIP Facility on a final basis be set at seven (7) days before the Final Hearing.

RESERVATION OF RIGHTS

61. The Debtors reserve all rights. Without limiting the generality of the foregoing, except as set forth specifically in the DIP Orders, nothing contained herein is or should be construed as: (a) an admission as to the validity, extent, perfection, priority, allowability, enforceability, or character of any claim or any security interest which purportedly secures such claim or other asserted right or obligation, or a waiver or other limitation on the Debtors' ability to contest the same on any ground permitted by bankruptcy or applicable nonbankruptcy law; (b) a waiver of the Debtors' or any appropriate party in interest's rights to dispute the amount of, basis for, or validity of any claim against the Debtors; (c) a promise to pay any claim; (d) a waiver of any claim or causes of action which may exist against any creditor or interest holder; (e) an

assumption or rejection of any executory contract or unexpired lease pursuant to section 365 of the Bankruptcy Code, and nothing herein otherwise affects the Debtors' rights under section 365 of the Bankruptcy Code to assume or reject any executory contract or unexpired lease with any party subject to this Motion; (f) granting third-party beneficiary status or bestowing any additional rights on any third party; or (g) being otherwise enforceable by any third party.

62. To implement the foregoing successfully, and given the nature of the relief requested herein, the Debtors respectfully request a finding that (x) the notice requirements under Bankruptcy Rule 6004(a) are met and (y) the 14-day stay under Bankruptcy Rule 6004(h) is waived. As described above, immediate access to the DIP Facility and Cash Collateral is essential to prevent potentially irreparable harm to the Debtors' business, value, and ability to implement a restructuring.

NOTICE

63. Notice of this Motion will be provided to: (i) the Office of the United States Trustee for the District of Delaware, J. Caleb Boggs Building, 844 King Street, Suite 2207, Lockbox 35, Wilmington, Delaware, 19801, Jane M. Leamy (jane.m.leamy@usdoj.gov); (ii) counsel to the DIP Lenders, Cole Schotz P.C., 1325 Avenue of the Americas, New York, NY 10019, Attn: Daniel F. X. Geoghan (DGeoghan@coleschotz.com) (iii) the Virginia Department of Environmental Quality, Piedmont Regional Office, 4949-A Cox Road, Glen Allen, Virginia 23060, Attn: Russell Deppe (russell.deppe@deq.virginia.gov); (iv) the Internal Revenue Service; (v) the Securities and Exchange Commission; (vi) the Delaware Secretary of State; (vii) the Delaware State Treasury; (viii) the Debtors' consolidated twenty (20) largest unsecured creditors; and (ix) all parties requesting notice pursuant to Bankruptcy Rule 2002. Notice of this Motion and any order entered hereon will be served in accordance with Local Rule 9013-1(m). In light of the nature of the relief requested herein, the Debtors submit that no other or further notice is necessary.

NO PRIOR REQUEST

64. No previous request for the relief sought herein has been made by the Debtors to this or any other Court.

[Remainder of page left blank intentionally]

CONCLUSION

WHEREFORE, for the reasons set forth herein, the Debtors respectfully request that the Court grant the relief requested in this Motion and the DIP Orders, and such other and further relief as is just and proper.

Dated: June 2, 2025

Wilmington, Delaware

PASHMAN STEIN WALDER HAYDEN, P.C.

/s/ Richard W. Riley

John W. Weiss (No. 4160) Richard W. Riley (No. 4052) 824 North Market Street, Suite 800 Wilmington, DE 19801 Telephone: (302) 592-6496

Email: jweiss@pashmanstein.com rriley@pashmanstein.com

-and-

Leah M. Eisenberg (*pro hac vice* forthcoming) David E. Sklar (*pro hac vice* forthcoming) Court Plaza South, East Wing 21 Main Street, Suite 200 Hackensack, NJ 07601 Telephone: (201) 488-8200

Email: leisenberg@pashmanstein.com dsklar@pashmanstein.com

Proposed Counsel to the Debtors and Debtors in Possession

Exhibit A

Proposed Interim Order

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Chapter 11
) Case No. 25-10979 (JKS)
VWS HOLDCO, INC., et al.,)
Debtors. ¹) Jointly Administration Requested)
	Re: Docket No

INTERIM ORDER PURSUANT TO

SECTIONS 105, 361, 362, 363, 364, 503, 506, 507 AND 552 OF THE BANKRUPTCY CODE AND RULES 2002, 4001, 6004 AND 9014 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE (I) AUTHORIZING THE DEBTORS TO (A) USE CASH COLLATERAL, (B) OBTAIN SENIOR SECURED SUPERPRIORITY POSTPETITION FINANCING AND GRANTING LIENS AND SUPERPRIORITY ADMINISTRATIVE CLAIMS, AND (C) PROVIDE ADEQUATE PROTECTION, (II) MODIFYING THE AUTOMATIC STAY, (III) SCHEDULING A FINAL HEARING, AND (IV) GRANTING RELATED RELIEF

Upon consideration of the motion (the "Motion") of the above-captioned debtors and debtors in possession (the "Debtors") in the above-captioned chapter 11 cases (the "Chapter 11 Cases"), for the entry of an order pursuant to sections 105, 361, 362, 363, 364, 503, 506, 507 and 552 of title 11 of the United States Code (as amended, the "Bankruptcy Code"), rules 2002, 4001, 6004, and 9014 of the Federal Rules of Bankruptcy Procedure (as amended, the "Bankruptcy Rules") and rules 2002-1, 4001-1, 4001-2, 9006-1, and 9013-1 of the Local Rules of Bankruptcy Practice and Procedure for the District of Delaware (the "Local Rules") (i) authorizing the Debtors to (a) use cash collateral, (b) obtain senior secured superpriority postpetition financing and granting liens and superpriority administrative expenses claims and (c) provide adequate

The Debtors in these Chapter 11 Cases, along with the last four digits of each Debtor's federal tax identification numbers are as follows: VWS Holdco, Inc. (5412) and Shoosmith Bros., Inc. (6914). The Debtors' mailing address is P.O. Box 2770, Chesterfield, VA 23832.

protection; (ii) modifying the automatic stay; (iii) scheduling interim and final hearings; and (iv) granting related relief, the Debtors sought, among other things, the following relief:

(i) authorizing Debtor VWS Holdco, Inc, as borrower, (the "Borrower") and Debtor Shoosmith Bros., Inc. as guarantor, (the "Guarantor") to obtain a senior secured superpriority debtor-in-possession financing facility in an aggregate amount of \$5,400,000.00 (the "DIP Facility"), comprised of \$4,900,000.00 of new money, pursuant to the terms herein, the terms and conditions set forth in the credit agreement attached hereto as Exhibit 1 (as may be amended, restated, supplemented or otherwise modified from time to time, in each case with the consent of the DIP Lenders (as defined below), the "DIP Credit Agreement," collectively with this order, this "Interim Order", the Final Order (as defined below), the DIP Budget (as defined below), and all other agreements, documents, and instruments delivered or executed in connection therewith to the extent that the DIP Agent (as defined below) and the DIP Lenders determine there is a need for such other agreements, documents, and instruments, in each case as amended, restated, supplemented, or otherwise modified from time to time, the "DIP Loan Documents") provided by Volunteer Enterprises, LLC, as administrative agent (together with any successor administrative agents and collateral agents, the "DIP Agent"), and the parties listed on Annex B to the DIP Credit Agreement, as lenders (in such capacity, together with any successors and assigns permitted under the DIP Loan Documents, the "DIP Lenders" and, together with the DIP Agent, the "DIP Secured Parties"), which DIP Facility shall be available, subject to the terms and conditions set forth in this Interim Order and the other DIP Loan Documents, as follows: (i) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.1, the DIP Lenders shall make available during the period (the "Interim Period") from the date hereof through and including the

² Capitalized terms used but not defined herein have the same meaning ascribed to such terms in the DIP Credit Agreement or the Motion, as appropriate.

date of entry of the Final Order by this Court, in one or more additional draws the aggregate amount of \$1,800,000.00 (the "First Interim DIP Loan"), to include a roll-up based on the interim funding of \$500,000.00 (the "Interim Roll-Up Amount"); (ii) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.2, the DIP Lenders shall make available in one or more additional draws the aggregate amount of \$1,550,000.00 (the "Second Interim DIP Loan"); and (iii) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.3, the DIP Lenders shall make available in one or more additional draws the aggregate amount of \$1,550,000.00 (the "Third Interim DIP Loan", and with the First Interim DIP Loan and the Second Interim DIP Loan, collectively, the "DIP Loans;

- (ii) the Court's authorization for the Debtors to execute, deliver, and perform, as applicable, under the DIP Credit Agreement and all other DIP Loan Documents, and to perform such other and further acts as may be necessary or appropriate in connection therewith;
- (iii) the Court's authorization for the Debtors to use proceeds of the DIP Loans immediately upon entry of the Interim Order solely in accordance with the budget and cash flow forecast prepared by the Debtors and annexed hereto as **Exhibit 2** (as updated from time to time in accordance with the terms of the DIP Credit Agreement and the other DIP Loan Documents, subject to the prior approval of the DIP Lenders in their sole discretion, the "Initial DIP Budget"), subject to the Permitted Variances;
- (iv) the Court's authorization to grant to the DIP Agent, for the benefit of the DIP Secured Parties, in respect of the DIP Obligations (as defined below), subject to the Carve-Out (as defined below) a superpriority administrative claim pursuant to section 364(c)(1) of the Bankruptcy Code and first priority priming liens on and security interests in substantially all assets and property of the Debtors (now owned or hereafter acquired) pursuant to sections 364(c)(2),

(c)(3), and (d)(1) of the Bankruptcy Code, in each case as and to the extent, set forth more fully below and in the DIP Loan Documents (including the DIP Credit Agreement);

- (v) the Court's authorization for the Debtors to use "cash collateral" as such term is defined in section 363 of the Bankruptcy Code in which the Prepetition Secured Parties (as defined below) or the DIP Secured Parties has an interest (the "Cash Collateral") solely in accordance with the DIP Budget (subject to Permitted Variances) and the DIP Credit Agreement;
- (vi) the Court's authorization to grant, as of the Petition Date, adequate protection for the benefit of the Prepetition Secured Parties, as set forth more fully below, including the Adequate Protection Superpriority Claim, Adequate Protection Liens, and Adequate Protection Payments (each as defined below), to the extent of and as compensation for any Diminution in Value (as defined below), the payment of fees of the Prepetition Secured Parties, if any, and the Debtors' satisfaction of the Milestones (as defined below) in accordance with the DIP Credit Agreement;
- (vii) the modification or waiver by the Court of the automatic stay imposed by section 362 of the Bankruptcy Code and any other applicable stay (including Bankruptcy Rule 6004) to the extent necessary to implement and effectuate the terms and provisions of the DIP Facility and this Interim Order and the other DIP Loan Documents, and to provide for the immediate effectiveness of this Interim Order;
- (viii) the scheduling by the Court of a final hearing (the "<u>Final Hearing</u>") to consider entry of an order (the "<u>Final Order</u>") granting the relief requested in the Motion on a final basis and approving the form of notice with respect to the Final Hearing and the transactions contemplated by the Motion; and
 - (ix) entry of the Final Order.

Cassee 255-10099799-JK\$\$ Doorc 505-11 FFileed 00660092255 FPaggre 551 off 12938

The Court having considered the Motion, the terms of the DIP Facility and the DIP Loan Documents (including the DIP Credit Agreement), the Declaration of Steven Agran in Support of Debtors' Motion for Entry of Interim and Final Orders (I) Authorizing the Debtors to (A) Use Cash Collateral, (B) Obtain Senior Secured Superpriority Postpetition Financing and Granting Liens and Superpriority Administrative Claims, and (C) Provide Adequate Protection, (II) Scheduling a Final Hearing, and (III) Granting Related Relief, and the Declaration of Steven Agran, as Chief Restructuring Officer, in Support of Chapter 11 Petitions and First Day Pleadings (collectively, the "Declarations") and the evidence submitted at the hearing held before this Court on June, 2025, to consider entry of this Interim Order (the "Interim Hearing"); and due and sufficient notice of the Motion and the Interim Hearing having been provided by the Debtors as set forth in paragraph E of this Interim Order; and the Interim Hearing having been held and concluded; and it appearing that approval of the interim relief requested in the Motion is necessary to avoid immediate and irreparable harm to the Debtors pending the Final Hearing and is otherwise fair and reasonable and in the best interests of the Debtors, their creditors, and their estates and stakeholders, and essential for the continued operation of the Debtors' businesses and preservation of the value of the Debtors' assets; and it appearing that the Debtors' entry into the DIP Documents is a sound and prudent exercise of the Debtors' business judgment; and all objections, if any, to the entry of this Interim Order having been withdrawn, resolved, or overruled by the Court; and after due deliberation and consideration, and for good and sufficient cause appearing therefor:

BASED UPON THE RECORD ESTABLISHED AT THE INTERIM HEARING, IT IS FOUND AND DETERMINED THAT:³

- A. <u>Petition Date</u>. On June 1, 2025 (the "<u>Petition Date</u>"), the Debtors filed voluntary petitions under Chapter 11 of the Bankruptcy Code with the United States Bankruptcy Court for the District of Delaware (the "<u>Court</u>").
- B. <u>Debtors in Possession</u>. The Debtors are authorized to continue in the management and operation of their businesses and property as debtors in possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee or examiner has been appointed in the Chapter 11 Cases.
- C. <u>Jurisdiction and Venue</u>. The Court has jurisdiction over this proceeding, pursuant to 28 U.S.C. §§ 157 and 1334 and the *Amended Standing Order of Reference* dated February 29, 2012, from the United States District Court for the District of Delaware. Consideration of the Motion constitutes a core proceeding under 28 U.S.C. § 157(b)(2), and the Court may enter a final order consistent with Article III of the United States Constitution. Venue for the Chapter 11 Cases and the proceeding on the Motion is proper in this district pursuant to 28 U.S.C. §§ 1408 and 1409.
- D. <u>Committee Formation</u>. As of the date hereof, no official committee of unsecured creditors has been appointed in the Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code (a "<u>Committee</u>").
- E. <u>Notice</u>. The Interim Hearing was held pursuant to Bankruptcy Rule 4001(b)(2) and (c)(2). Proper, timely, adequate, and sufficient notice of the Interim Hearing and the relief

6

³ The findings and conclusions set forth herein constitute the Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent that any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent that any of the following conclusions of law constitute findings of fact, they are adopted as such.

requested in the Motion has been provided to (a) the Office of the United States Trustee for the District of Delaware (the "<u>U.S. Trustee</u>"), (b) counsel to the Prepetition Secured Parties and DIP Secured Parties, (c) all other parties asserting a lien on or a security interest in the assets of the Debtors to the extent reasonably known to the Debtors, (d) the Office of the United States Attorney General for the State of Delaware, (e) the Internal Revenue Service, (f) those creditors holding the 20 largest unsecured claims against the Debtors' estates (excluding insiders), (g) the Securities and Exchange Commission, and (h) all parties who have filed a notice of appearance and request for service of papers pursuant to Bankruptcy Rule 2002 (the "<u>Notice Parties</u>"). Under the circumstances, such notice of the Interim Hearing and the relief requested in the Motion complies with section 102(1) of the Bankruptcy Code, Bankruptcy Rules 2002 and 4001(b), (c), and (d), and the Local Rules, and no other or further notice need be provided for entry of this Interim Order.

- F. <u>Prepetition Indebtedness</u>. The Borrower is a party to that certain Amended and Restated Securities Purchase Agreement (as amended, the "<u>Prepetition Term Loan Agreement</u>"), originally dated as of April 30, 2020, by and among the Borrower, as borrower, Guarantor, Volunteer Enterprises, LLC (as successor in interest to PNC Mezzanine Partners III, L.P., the "<u>Prepetition Agent</u>") and the lender parties identified on Annex B thereto (the "<u>Prepetition</u> Lenders" or the "<u>Prepetition Secured Parties</u>").
- G. <u>Stipulations as to Prepetition Obligations</u>. After consultation with their attorneys and financial advisors, as a condition to provide the DIP Facility as set forth herein and subject to the rights of any party in interest as and to the extent set forth in paragraph 11 hereof, the Debtors, on their behalf, acknowledge, stipulate, and agree with the following: (collectively, the "Stipulations"):

- (i) **Prepetition Obligations**. Pursuant to that certain Prepetition Term Loan Agreement, the Debtors issued the Senior Notes, Junior Notes and Bridge Notes (collectively, the "<u>Prepetition Loans</u>") to the Prepetition Secured Parties. As of the Petition Date, the aggregate principal and accrued interest owed to the Prepetition Secured Parties was not less than \$183,050,584.19, plus other interest, fees and all other obligations under the Prepetition Term Loan Agreement, each of the Prepetition Loans and the other documents related thereto (collectively, the "Prepetition Obligations").
- (ii) Enforceability, etc. of the Prepetition Obligations. The Debtors acknowledge that the Prepetition Term Loan Agreement and the Prepetition Obligations are legal, valid, binding and enforceable obligations against the Debtors. As a condition to providing the DIP Facility, the DIP Lenders have required and the Debtors have agreed not to contest or challenge the Prepetition Obligations under the Bankruptcy Code, under applicable non-bankruptcy law or otherwise.
- that the liens granted by the Debtors under the Prepetition Term Loan Agreement to the Prepetition Agent (the "Prepetition Liens") for the benefit of the Prepetition Secured Parties as security for the Prepetition Obligations encumber all Collateral (as defined in the Prepetition Term Loan Agreement), which includes substantially all of the Debtors' assets and property and all proceeds thereof (collectively, the "Prepetition Collateral") were granted to or for the benefit of the Prepetition Secured Parties for fair consideration and reasonably equivalent value. The Debtors acknowledge that as of the Petition Date, (a) the Debtors granted to the Prepetition Secured Parties the Prepetition Liens, which Prepetition Liens are subject only to any permitted Liens set forth in section 2 of Schedule 3.14 of the Prepetition Term Loan Agreement (the "Prepetition Permitted Liens"); (b) the

Prepetition Obligations constitute legal, valid and binding obligations of the Debtors enforceable in accordance with the terms of the Prepetition Term Loan Agreement; (c) the Debtors have no claims, objections, causes of action, and/or choses in action, including avoidance claims under Chapter 5 of the Bankruptcy Code or applicable state law equivalents or actions for recovery or disgorgement, against any of the Prepetition Secured Parties or any of their respective affiliates, agents, attorneys, advisors, professionals, officers, directors, and employees arising out of, based upon, or related to the Prepetition Term Loan Agreement; (d) the Debtors have waived, discharged, and released any right to challenge any of the Prepetition Obligations, the priority of the Debtors' obligations thereunder, and the validity, extent and priority of the liens securing the Prepetition Obligations; and (e) the Prepetition Obligations constitute allowed, secured claims within the meaning of sections 502 and 506 of the Bankruptcy Code.

(iv) Indemnity. The Prepetition Secured Parties and the DIP Secured Parties and their Related Parties (as defined below) have acted in good faith and without negligence, misconduct, or violation of public policy or law, in respect of all actions taken by them in connection with or related in any way to negotiating, implementing, documenting, or obtaining requisite approvals of the DIP Facility and the use of Cash Collateral, including in respect of the granting of the DIP Liens, the DIP Superpriority Claim, the Adequate Protection Liens, the Adequate Protection Superpriority Claim (each as defined below), and any of the other rights, privileges, remedies, and protections granted hereunder or under the DIP Loan Documents (including the DIP Credit Agreement), any challenges or objections to the DIP Facility or the use of Cash Collateral, and all documents related to and all transactions contemplated by the foregoing. Accordingly, without limitation to any other right to indemnification (including any and all rights of the Prepetition Secured Parties to indemnification under the Prepetition Term Loan Agreement),

the Prepetition Secured Parties and the DIP Secured Parties and each of their respective affiliates, officers, directors, fiduciaries, employees, agents, advisors, attorneys, and representatives (collectively, the "Related Parties") shall be and hereby are indemnified and held harmless by the Debtors in respect of any claim or liability incurred in respect thereof or in any way related thereto; provided, however, that the Debtors shall not indemnify the Prepetition Secured Parties against (i) any successful Challenge under the Challenge Period (each as defined herein) or (ii) any losses, claims, damages, liabilities, or related expenses to the extent determined by a final, unappealable judgment of a court of competent jurisdiction to have been incurred solely by reason of the gross negligence, willful misconduct, or fraud of such Related Parties. No exception or defense in contract, law, or equity exists as to any obligation set forth, as the case may be, in this paragraph G (iv), in the Prepetition Term Loan Agreement, or in the DIP Credit Agreement, to indemnify and/or hold harmless any Prepetition Secured Party, DIP Secured Party, and any Related Party, as the case may be, and any such defenses are hereby waived.

- (v) **No Control**. None of the DIP Secured Parties or Prepetition Secured Parties are control persons or insiders of the Debtors or any of their affiliates by virtue of any of the actions taken with respect to, in connection with, related to, or arising from the DIP Facility, the DIP Loan Documents (including the DIP Credit Agreement), and/or the Prepetition Term Loan Agreement.
- (vi) **No Claims, Causes of Action**. As of the date hereof, other than as previously disclosed to the Lenders, there exist no other claims or causes of action against any of the Prepetition Secured Parties, the DIP Secured Parties, or their Related Parties with respect to, in connection with, related to, or arising from the Prepetition Term Loan Agreement and/or the DIP Loan Documents (including the DIP Credit Agreement) that may be asserted by the Debtors.

- (vii) **Cash Collateral**. Any and all of the Debtors' cash, whether existing as of the Petition Date or thereafter, wherever located (including, without limitation, all cash or cash equivalents in the control of or on deposit or maintained by the Debtors in any account or accounts, any amounts generated by the sale or other disposition of Prepetition Collateral, and all income, proceeds, products, rents or profits of any Prepetition Collateral), constitutes Cash Collateral of the Prepetition Secured Parties.
- (viii) **Default**. The Debtors are in default under the Prepetition Term Loan Agreement and an event of default has occurred thereunder.
- (ix) Release. Effective as of the date of entry of this Interim Order, but subject to entry of the Final Order, the Debtors hereby forever and irrevocably release, discharge, and acquit all former, current and future (a) DIP Secured Parties, (b) Prepetition Secured Parties, (c) the Related Parties, (d) Affiliates of the DIP Secured Parties and Prepetition Secured Parties, and (e) officers, employees, directors, agents, representatives, owners, members, partners, financial and other advisors and consultants, legal advisors, shareholders, managers, consultants, accountants, attorneys, investment committee members, sub advisors and predecessors and successors in interest of each of the DIP Secured Parties and Prepetition Secured Parties and each of their respective Affiliates, in each case acting in their respective capacities as such (collectively, the "Releasees") of and from any and all claims, demands, liabilities, responsibilities, disputes, remedies, causes of action, indebtedness and obligations, rights, assertions, allegations, actions, suits, controversies, proceedings, losses, damages, injuries, attorneys' fees, costs, expenses, or judgments of every type, whether known, unknown, asserted, unasserted, suspected, unsuspected, accrued, unaccrued, fixed, contingent, pending or threatened including, without limitation, all legal and equitable theories of recovery, arising under common law, statute or regulation or by contract,

of every nature and description, arising out of, in connection with, or relating to the Prepetition Loans, the DIP Facility, the DIP Loan Documents (including the DIP Credit Agreement), the Prepetition Term Loan Agreement, and/or the transactions contemplated hereunder or thereunder including, without limitation, (x) any so-called "lender liability" or equitable subordination claims or defenses, (y) any and all claims and causes of action arising under the Bankruptcy Code, and (z) any and all claims and causes of action with respect to the validity of the liens or claims of any of the Prepetition Secured Parties or the DIP Secured Parties (the "Releases"); provided that nothing in this paragraph is intended to limit or release (i) any commitments and obligations of any of the DIP Secured Parties under the DIP Credit Agreement or (ii) any claims, causes of action, costs or demands of whatever kind or nature, whether known or unknown, liquidated or unliquidated, fixed or contingent, asserted or unasserted, foreseen or unforeseen, or matured or unmatured, resulting solely from the gross negligence or willful misconduct or fraud of the Releasees, as determined by a court of competent jurisdiction in a final and unappealable judgment or order. The Debtors further waive and release any defense, right of counterclaim, right of setoff or deduction to the payment of the Prepetition Obligations and the DIP Obligations which the Debtors now have or may claim to have against the Releasees arising out of, connected with, or relating to any and all acts, omissions, or events occurring prior to the entry of this Interim Order by the Court.

Sale and Credit Bidding. The DIP Secured Parties and the Prepetition Secured Parties shall have the right, as provided for in section 363(k) of the Bankruptcy Code, to credit bid (independently or together) up to the full amount of the applicable outstanding Prepetition Obligations (including any Adequate Protection Superpriority Claims) and the DIP Obligations, in each case including, without limitation, any accrued interest and fees, in a sale of any DIP

Collateral (as defined below) or Prepetition Collateral, as applicable, and whether such sale is effectuated through sections 363 or 1129 of the Bankruptcy Code, or otherwise.

H. Immediate Need for Postpetition Financing and Use of Cash Collateral. The Debtors have requested immediate entry of this Interim Order pursuant to Bankruptcy Rules 4001(b)(2) and (c)(2). Good cause has been shown for entry of this Interim Order as the Debtors do not have sufficient liquidity to operate and maintain their business in the ordinary course of business without the financing requested in the Motion. The Debtors' ability to maintain business relationships with their vendors and suppliers, to pay their employees and to otherwise fund their operations, payroll and other efforts and activities is essential to the Debtors' continued viability as the Debtors seek to maximize and preserve the value of their estates pending the Final Hearing. The ability of the Debtors to finance their operations through the incurrence of the First Interim DIP Loan and use of Cash Collateral is vital to the preservation and maintenance of the going concern value of the Debtors' estates, to permit the orderly continuation of the operation and maintenance of their businesses, minimize the disruption of their operations, and implement and pursue a comprehensive marketing and sale process that maximizes the value of the Debtors' assets for the benefit of their creditors, and to otherwise avoid immediate and irreparable harm to the Debtors, their estates, and their creditors.

I. No Credit Available on More Favorable Terms. The Debtors have been unable to obtain credit on more favorable terms and conditions than those provided in this Interim Order, including for (a) adequate unsecured credit allowable under section 503(b)(1) of the Bankruptcy Code as an administrative expense, (b) credit for money borrowed with priority over any or all administrative expenses of the kind specified in sections 503(b) or 507(b) of the Bankruptcy Code, (c) credit for money borrowed secured by a lien on property of the estates that is not otherwise

subject to a lien, or (d) credit for money borrowed secured by a junior lien on property of the estates which is subject to a lien. The Debtors are unable to obtain credit for borrowed money without granting the DIP Liens and the DIP Superpriority Claim to (or for the benefit of) the DIP Secured Parties and without granting the adequate protection as set forth herein (in each case, subject to the Carve-Out (as defined below)). The Debtors have therefore concluded, in the exercise of their sound business judgment, that the DIP Loans represent the best financing available to the Debtors at this time and are in the best interests of their estates and all stakeholders.

- J. <u>Use of Cash Collateral and Proceeds of the DIP Facility, DIP Collateral, and Prepetition Collateral.</u> Subject to the Challenge Period, the Debtors represent and stipulate that all of the Debtors' cash, cash equivalents, negotiable instruments, investment property, and securities constitute Cash Collateral of the Prepetition Agent on behalf of the Prepetition Secured Parties and the DIP Agent on behalf of the DIP Secured Parties. All Cash Collateral, all proceeds of the Prepetition Collateral and the DIP Collateral, including proceeds realized from any sale or disposition thereof, or from payment thereon, and all proceeds of the DIP Facility (net of any amounts used to pay fees, costs, and expenses payable under the DIP Facility pursuant to this Interim Order or the Final Order) shall be used or applied in accordance with the terms and conditions of this Interim Order, the DIP Budget (subject to the Permitted Variances), and the DIP Loan Documents (including the DIP Credit Agreement) and for no other purpose.
- K. <u>Adequate Protection for the Prepetition Secured Parties</u>. The Prepetition Secured Parties have negotiated in good faith regarding the Debtors' use of the Prepetition Collateral (including the Cash Collateral) to fund the administration of the Debtors' estates and the continued operation of their businesses, in accordance with the terms hereof and the DIP Budget (subject to the Permitted Variances). The Prepetition Secured Parties have agreed to permit

the Debtors to use the Prepetition Collateral, including the Cash Collateral, in accordance with the terms hereof and the DIP Budget (subject to the Permitted Variances) during the Interim Period subject to the terms and conditions set forth herein, including the protections afforded parties acting in "good faith" under section 364(e) of the Bankruptcy Code. The Prepetition Secured Parties are entitled to the adequate protection as and to the extent set forth herein pursuant to sections 361, 362, 363, and 364 of the Bankruptcy Code. Based on the Motion and on the record presented to the Court at the Interim Hearing, the terms of the proposed adequate protection arrangements and of the use of the Prepetition Collateral (including the Cash Collateral) are fair and reasonable, reflect the Debtors' prudent exercise of business judgment and constitute reasonably equivalent value and fair consideration for the Prepetition Agent's consent thereto; provided that nothing in this Interim Order shall (i) be construed as a consent by any Prepetition Secured Party (a) that it would be adequately protected in the event debtor-in-possession financing is provided by a third party (i.e., other than the DIP Lenders) or (b) to the terms of any other such financing, including the consent to any lien encumbering the Prepetition Collateral (whether senior or junior) or to the use of Cash Collateral (except under the terms hereof), or (ii) prejudice, limit, or otherwise impair the rights of the Prepetition Agent (for the benefit of the Prepetition Secured Parties) to seek new, different, or additional adequate protection under any circumstances. The Prepetition Secured Parties' consent to the Debtors' use of Cash Collateral and the granting of DIP Liens on the Prepetition Collateral is expressly conditioned upon entry of this Interim Order and does not and shall not constitute consent other than pursuant to this Interim Order and on the terms set forth herein.

L. <u>Sections 506(b) and 552; Marshaling</u>. In light of and in exchange for (i) the DIP Secured Parties' willingness to provide the DIP Facility to the extent set forth herein, (ii) the DIP

Secured Parties' agreement that their liens and superpriority claims shall be subject to the Carve-Out, as set forth herein, (ii) the Prepetition Secured Parties' agreement that their Prepetition Liens and claims, including any adequate protection liens and claims, shall be subject to the Carve-Out, as set forth herein, (iii) the consensual use of Cash Collateral consistent with the DIP Budget and the terms of this Interim Order, and (iv) the DIP Secured Parties' and the Prepetition Secured Parties' agreement to the payment (in a manner consistent with the DIP Budget (subject to the Permitted Variances) and subject to the terms and conditions of this Interim Order) of certain expenses of administration of these Chapter 11 Cases, the DIP Secured Parties, and subject to entry of the Final Order, the Prepetition Secured Parties are each entitled to a waiver of any "equities of the case" exception under section 552(b) of the Bankruptcy Code and a waiver of the provisions of section 506(c) of the Bankruptcy Code and of the equitable doctrine of marshaling and other similar doctrines upon entry of the Final Order.

M. Extension of Financing. The DIP Secured Parties have indicated a willingness to provide financing to the Debtors in accordance with the DIP Loan Documents (including the DIP Credit Agreement and the DIP Budget) and subject to (i) the entry of this Interim Order and the Final Order, (ii) approval of the terms and provisions of this Interim Order and the DIP Loan Documents (including the DIP Credit Agreement), including the Milestones (as defined in the DIP Credit Agreement) and the waivers set forth herein, and (iii) findings by this Court that such financing is essential to the Debtors' estates, that the DIP Secured Parties are good faith financiers, and that the reversal or modification on appeal of the authorization hereunder for the Debtors to incur the debt under the DIP Facility, or the grant hereunder of the priority of the DIP Liens and the Adequate Protection Liens, does not affect the validity of such debt, or any priority of any such lien so granted as and to the extent provided in section 364(e) of the Bankruptcy Code.

N. <u>Limitation of Liability</u>. The Debtors stipulate and, subject to paragraph 11 and solely with respect to the DIP Loans, this Court finds that in making decisions to advance loans to the Debtors, in administering any loans, in permitting the Debtors to use Cash Collateral, in accepting the DIP Budget, or in taking any other actions permitted by this Interim Order or the DIP Loan Documents (including the DIP Credit Agreement), none of the DIP Secured Parties or Prepetition Secured Parties shall be deemed to be in control of the operations of the Debtors or to be acting as a "responsible person" or "owner or operator" with respect to the operation or management of the Debtors.

O. Business Judgment and Good Faith Pursuant to Section 364(e).

- (i) Based on the Declarations and the record at the Interim Hearing, the extensions of credit under the DIP Facility are fair, reasonable, and the best available under the circumstances, reflect the Debtors' exercise of prudent business judgment, and are supported by reasonably equivalent value and fair consideration;
- (ii) all obligations incurred, payments made, and transfers or grants of security set forth in this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement) by the Debtors are granted to or for the benefit of the DIP Secured Parties for fair consideration and reasonably equivalent value and are granted contemporaneously with the making of the loans and commitments and other financial accommodations secured thereby;
- (iii) the DIP Facility was negotiated in good faith and at arm's length among the Debtors and the DIP Secured Parties; and
- (iv) the use of the proceeds to be extended under the DIP Facility will be so extended in good faith and for valid business purposes and uses, as a consequence of which the

DIP Secured Parties are entitled to the protection and benefits of section 364(e) of the Bankruptcy Code.

- P. Relief Essential; Best Interest. The relief requested in the Motion (and provided in this Interim Order) is necessary, essential, and appropriate for the continued operation of the Debtors' businesses and the management and preservation of the Debtors' assets and property, and satisfies the requirements of Bankruptcy Rule 6003. It is in the best interest of the Debtors' estates and consistent with the Debtors' exercise of their fiduciary duties that the Debtors are allowed to enter into the DIP Facility, incur the DIP Obligations, grant the liens and claims contemplated herein and under the DIP Loan Documents (including the DIP Credit Agreement) to the DIP Secured Parties and the Prepetition Secured Parties, and use Prepetition Collateral, including Cash Collateral, as it will provide the Debtors with the immediate and necessary liquidity to (i) minimize disruption to the Debtors' businesses and on-going operations, (ii) preserve and maximize the value of the Debtors' estates for the benefit of all the Debtors' creditors and stakeholders, (iii) avoid immediate and irreparable harm to the Debtors' estates and (iv) to otherwise maximize the value of the Debtors' estates.
- Q. <u>Findings Regarding Corporate Authority</u>. The Debtors have all requisite corporate power and authority to enter into, ratify, and perform all of their obligations under the DIP Loan Documents (including the DIP Credit Agreement) to which they are a party.
- R. <u>Immediate Entry</u>. Sufficient cause exists for immediate entry of this Interim Order pursuant to Bankruptcy Rules 4001(b)(2), (c)(2) and (d).

NOW, THEREFORE, on the Motion of the Debtors and the record before this Court with respect to the Motion, including the record made during the Interim Hearing, and with

the consent of the Debtors, the Prepetition Secured Parties, and the DIP Secured Parties, and good and sufficient cause appearing therefor,

IT IS HEREBY ORDERED, ADJUDGED AND DECREED that:

1. <u>Motion Granted</u>. The Motion is granted on an interim basis in accordance with the terms and conditions set forth in this Interim Order. Any objections to the Motion with respect to entry of this Interim Order, to the extent not withdrawn, waived or otherwise resolved, and all reservation of rights included therein, are hereby denied and overruled on the merits.

2. **DIP Facility**.

DIP Obligations, Etc. The Debtors are expressly and immediately authorized and (a) empowered to enter into the DIP Facility and to incur and to perform the DIP Obligations in accordance with and subject to this Interim Order (and, upon its entry, a Final Order) and the other DIP Loan Documents, to execute and deliver all DIP Loan Documents and all other related instruments, certificates, agreements, and documents, and to take all actions which may be reasonably required or otherwise necessary for the performance by the Debtors under the DIP Facility, including the creation and perfection of the DIP Liens described and provided for herein. The Debtors are hereby authorized and directed to pay all principal, interest, fees, expenses, indemnities, and other amounts described herein, in the DIP Credit Agreement, and in the other DIP Loan Documents as such shall accrue and become due hereunder or thereunder, including, without limitation, the reasonable and documented fees and expenses of the attorneys and financial and other advisors and consultants of the DIP Agent and the DIP Lenders, as and to the extent provided for herein, in the DIP Credit Agreement, and in the other DIP Loan Documents (collectively, all loans, advances, extensions of credit, financial accommodations, fees (which shall be fully earned and nonrefundable upon entry of this Interim Order), expenses, and other liabilities and obligations

(including indemnities and similar obligations) in respect of DIP Loans, the DIP Facility, the DIP Credit Agreement, and the other DIP Loan Documents, including all "Obligations" under and as defined in the DIP Credit Agreement, the "DIP Obligations"). The DIP Credit Agreement, the other DIP Loan Documents, and all DIP Obligations shall represent, constitute, and evidence, as the case may be, valid and binding obligations of the Debtors, enforceable against the Debtors, their estates, and any successor thereto in accordance with their terms. All obligations incurred, payments made, and transfers or grants of security set forth in this Interim Order and in the other DIP Loan Documents by any DIP Loan Party are granted to or for the benefit of the DIP Secured Parties for fair consideration and reasonably equivalent value and are granted contemporaneously with the making of the loans and commitments and other financial accommodations secured thereby. No obligation, payment, transfer, or grant of security under the DIP Loan Documents (including the DIP Credit Agreement) as approved under this Interim Order shall be voided, voidable, or recoverable under the Bankruptcy Code or under any applicable non-bankruptcy law, or subject to any defense, reduction, setoff, recoupment, or counterclaim; provided, however, that the adequate protection provided in paragraph 4 of this Interim Order shall be subject to paragraph 11 below. The term of the DIP Facility shall commence on the date of entry of this Interim Order and end on the Maturity Date (as defined below), subject to the terms and conditions set forth herein and in the DIP Loan Documents (including the DIP Credit Agreement).

(b) Authorization to Borrow. In order to continue to operate its business, subject to the terms and conditions of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement), the Debtors are hereby authorized to borrow under the DIP Facility and incur DIP Obligations during the Interim Period.

- (c) Roll-Up. Upon entry of this Interim Order and the Final Order, and subject to paragraph 11 below, the Prepetition Obligations, on a 1 to 1 cashless basis, in the aggregate amounts equal to the Interim Roll-Up Amount, shall be converted into DIP Obligations, without any further action by the Debtors or any other party. Upon any of the foregoing Interim Roll-Up Amount, and subject to paragraph 11 below, the Interim Roll-Up Amount shall cease to be indebtedness under the Prepetition Term Loan Agreement and shall be deemed DIP Obligations under the DIP Facility in all respects, including for purposes of having the benefit of section 364(e) of the Bankruptcy Code.
- (d) Conditions Precedent. The DIP Lenders shall have no obligation to make any DIP Loans or any other financial accommodation hereunder or under the other DIP Loan Documents (including the DIP Credit Agreement) (and the Debtors shall not make any request therefor) unless all conditions precedent to making DIP Loans under the DIP Credit Agreement have been satisfied or waived in accordance with the terms of the DIP Credit Agreement.
- (e) **DIP Collateral**. As used herein, "<u>DIP Collateral</u>" shall mean all assets, interests, rights, and property of any nature whatsoever of the Debtors, including, without limitation, all property in which the Debtors and their estates have an interest (whether tangible, intangible, real, personal or mixed), whether now owned or hereafter acquired and wherever located, before or after the Petition Date, including, without limitation, all accounts, proceeds of leases, inventory, equipment, equity interests or capital stock in subsidiaries, investment property, instruments, chattel paper, contracts, patents, copyrights, trademarks and other general intangibles, commercial litigation claims, cash, any investment of such cash, inventory, accounts receivable, including intercompany accounts (and all rights associated therewith), other rights to payment whether arising before or after the Petition Date, any deposit accounts, and "cash collateral accounts", and,

in each case all amounts on deposit therein from time to time, the proceeds of all claims or causes of action, and all rents, products, offspring, profits, proceeds, and substitutions thereof (including, without limitation, all Prepetition Collateral and, subject to the entry of the Final Order, all claims or causes of action of the Borrower arising under sections 502(d), 542, 544, 545, 547, 548, 549, 550, and 553 of the Bankruptcy Code and any other avoidance or similar action under the Bankruptcy Code or similar state law avoidance actions under chapter 5 of the Bankruptcy Code (the "Avoidance Actions") provided that the lien on Avoidance Actions shall be limited to the proceeds and property recovered in connection therewith).

- (f) **DIP Liens**. Effective immediately upon the entry of this Interim Order and subject to the Carve-Out, as set forth more fully in this Interim Order, the DIP Agent for the benefit of the DIP Secured Parties is hereby granted the following security interests and liens, which shall immediately be valid, binding, perfected, continuing, enforceable, and non-avoidable without the need for execution by the Borrower or Guarantors or the recordation or other filing by the DIP Secured Parties of security agreements, control agreements, pledge agreements, financing statements, or other similar documents or the possession or control by the DIP Secured Parties of any DIP Collateral (all liens and security interests granted to the DIP Agent for the benefit of the DIP Secured Parties pursuant to this Interim Order, any Final Order, and the DIP Credit Agreement, the "DIP Liens"):
 - (i) pursuant to section 364(c)(2) of the Bankruptcy Code, valid, binding, continuing, enforceable, fully perfected first-priority liens on and security interests in all DIP Collateral that is not subject to any liens or encumbrances immediately prior to the Petition Date, subject only to the Carve-Out; and
 - (ii) pursuant to section 363(c)(3) and 364(d)(1) of the Bankruptcy Code, valid, binding, continuing, enforceable, fully perfected first priority, priming liens on and security interests in all other DIP Collateral, which liens and security interests shall be subject only to (a) any valid, enforceable, perfected, and non-avoidable lien or security interest in favor of any person other than the

Prepetition Secured Parties that was in existence immediately prior to the Petition Date or that is perfected as permitted by Section 546(b) of the Bankruptcy Code, in each case, with respect to any DIP Collateral comprised of Prepetition Collateral, solely to the extent such lien is senior to the Prepetition Liens (a "Permitted Encumbrance")⁴ and (b) the Carve-Out, and senior to all other liens and encumbrances in respect of the DIP Collateral.

of this Interim Order, subject to the Carve-Out, the DIP Liens shall secure all of the DIP Obligations, to the extent and subject to the priorities set forth herein, and such liens (and any superpriority claims and other senior liens, including adequate protection liens, granted hereunder) shall at all times have a higher priority and shall remain senior to the rights of the Debtors, any chapter 7 or chapter 11 trustee, and any secured, administrative priority, unsecured or other claims of any party in these Chapter 11 Cases under the Bankruptcy Code (except as (and solely to the extent) expressly provided herein), and the DIP Liens, the DIP Superpriority Claim, the Adequate Protection Claims, and the Adequate Protection Liens granted herein shall not be made or become subject, junior, or subordinated to any "priming" or other liens, nor made *pari passu* with any other lien, security interest, or claim heretofore or hereafter granted under Bankruptcy Code section 364 or otherwise, in these Cases (except as (and solely to the extent) expressly provided herein) or any Successor Case (as defined below), including to (i) any lien or security interest that is avoided and preserved for the benefit of the Debtors and their estates under section 551 of the Bankruptcy Code,

Nothing herein shall constitute a finding or ruling by this Court that any such Permitted Encumbrance is valid, senior, enforceable, prior, perfected, or non-avoidable. Moreover, nothing shall prejudice the rights of any party in interest, including but not limited to the Debtors, the DIP Secured Parties, the Prepetition Secured Parties, or Committee (if any), to challenge the validity, priority, enforceability, seniority, avoidability, perfection, or extent of any alleged Permitted Encumbrance and/or security interest. Notwithstanding anything to the contrary in this Interim Order or the DIP Loan Documents, nothing in this Interim Order or the DIP Loan Documents limits or adversely affects the rights, claims or defenses of any party asserting any Prepetition Permitted Liens or Excepted Liens (as defined in the DIP Credit Agreement) and the rights, claims or defenses of any such party are expressly reserved and preserved including but not limited to any rights of recoupment and/or setoff, subject to the Debtors' right to oppose or contest any of the foregoing.

- (ii) any lien or security interest existing on or arising on or after the Petition Date, including, without limitation, any lien or security interests granted in favor of any federal, state, municipal or other domestic or foreign governmental unit (including any regulatory body), commission, board or court for any liability of the Debtors, (iii) any intercompany or affiliate lien or claim; or (iv) any other lien, claim, or security interest under sections 361, 363, or 364 of the Bankruptcy Code or otherwise, in each case other than as (and solely to the extent) expressly set forth herein. The DIP Liens and the Adequate Protection Liens shall be valid and enforceable against any trustee appointed in the Chapter 11 Cases, upon the conversion of the Chapter 11 Cases to cases under chapter 7 of the Bankruptcy Code or in any other proceedings related to any of the foregoing (each such case or proceeding, a "Successor Case"), and/or upon the dismissal of the Chapter11 Cases. The DIP Liens and, subject to the entry of the Final Order, the Adequate Protection Liens shall not be subject to section 510 of the Bankruptcy Code, to the "equities of the case" exception of section 552 of the Bankruptcy Code, or to section 506(c) of the Bankruptcy Code, or sections 549, 550, or 551 of the Bankruptcy Code.
- (h) Superpriority Administrative Claim Status. The DIP Obligations shall, subject to the Carve-Out, pursuant to section 364(c)(1) of the Bankruptcy Code, at all times constitute an allowed superpriority claim (the "DIP Superpriority Claim") of the DIP Agent for the benefit of the DIP Secured Parties against the Debtors, and be payable from all assets and properties of the Debtors, with priority over any and all other administrative expenses, adequate protection claims, diminution in value claims, and all other claims asserted against the Debtors now existing or hereafter arising of any kind whatsoever, including, without limitation, all administrative expenses of the kind specified in sections 503(b) and 507(b) of the Bankruptcy Code, and over any and all other administrative expenses or other claims arising under any other provisions of the Bankruptcy

Code, including sections 105, 326, 327, 328, 330, 331, 363, 364, 365, 503, 506(b), 506(c), 507(a), 507(b), 507(d), 546, 726, 1113, or 1114 of the Bankruptcy Code, whether or not such expenses or claims may become secured by a judgment lien or other nonconsensual lien, levy, or attachment, subject only to the Carve-Out, which DIP Superpriority Claim shall for purposes of section 1129(a)(9)(A) of the Bankruptcy Code be considered an administrative expense allowed under section 503(b) of the Bankruptcy Code and shall be payable from all pre- and postpetition assets and property, whether existing on the Petition Date or thereafter acquired, of the Debtors and all proceeds thereof; provided, however, that the DIP Superpriority Claim shall not be payable from the proceeds of Avoidance Actions or the proceeds thereof before entry of the Final Order. The DIP Superpriority Claim shall be subject only to the Carve-Out. Other than as expressly provided herein, including in paragraph 13 hereof with respect to the Carve-Out, no costs or expenses of administration, including, without limitation, professional fees allowed and payable under sections 328, 330, and 331 of the Bankruptcy Code, or otherwise, that have been or may be incurred in these Cases or in any Successor Case and no priority claims are or will be senior to, prior to, or pari passu with the DIP Liens, the DIP Superpriority Claim, or any of the DIP Obligations or with any other claims of the DIP Secured Parties arising hereunder or under the DIP Loan Documents (including the DIP Credit Agreement), or otherwise in connection with the DIP Facility.

(i) **Proceeds of DIP Loans**. All proceeds of the DIP Loans shall be funded and held in the account specified under the terms of the DIP Loan Documents (including the DIP Credit Agreement) and as otherwise agreed by the DIP Secured Parties, and such account (and any funds therein) shall be subject to a first-priority senior security interest and lien in favor of the DIP Agent, and shall be subject to a deposit control agreement in favor of the DIP Agent.

- (j) **Professional Fees and Expenses**. The Debtors shall pay in cash and in full on the dates provided for in the Budget, all reasonable and documented out of pocket fees, costs and expenses of the DIP Agent and the DIP Lenders (including, without limitation, fees and disbursements of counsel, including Cole Schotz P.C., financial and accounting advisors, and any other advisors) in connection with the Chapter 11 Cases, including, without limitation (i) the negotiation, preparation, execution, and entry, as applicable, of this DIP Credit Agreement, (ii) any litigation, contest, dispute, suit, proceeding or action (whether instituted by DIP Agent, the Borrower, the Guarantors, or any other person) in any way relating to this DIP Credit Agreement, and (iii) the enforcement of any rights and remedies under the DIP Credit Agreement, including, without limitation, any accrued and unpaid fees, costs, and expenses of the DIP Agent and the DIP Lenders (including, without limitation, any fees and disbursements of counsel and any other advisors to the DIP Agent and to the DIP Lenders) prior to the Petition Date. Payment of any amounts set forth in this clause (j) shall not be subject to disgorgement.
- 3. Authorization and Approval to Use Cash Collateral and Proceeds of DIP Facility. Subject to the terms and conditions of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement) the Debtors are authorized during the Interim Period (and not beyond) to (a) use the Cash Collateral and (b) request and use proceeds of the DIP Loans, in each case as set forth in the DIP Budget (subject to the Permitted Variances) and in accordance with this Interim Order the other DIP Loan Documents (including the DIP Credit Agreement). Notwithstanding anything herein to the contrary, subject only to the Debtors' rights under paragraph 21(b) hereof and the Carve-Out, the Debtors' right (a) to request proceeds of the DIP Loans shall terminate on the earlier of (i) the date the DIP Term Loans are fully funded or reduced to zero and (ii) the Maturity Date and (b) to use proceeds of the DIP Loans or Cash Collateral shall

terminate on the Maturity Date. Nothing in this Interim Order shall authorize the disposition of any assets of the Debtors or their estates or proceeds resulting therefrom outside the ordinary course of business except as expressly permitted herein and in the other DIP Loan Documents (including the DIP Credit Agreement) (subject to any required Court approval).

- 4. Adequate Protection for Prepetition Secured Parties. The Prepetition Secured Parties shall receive adequate protection of their interests in the Prepetition Collateral, including the Cash Collateral, in an amount equal to the aggregate diminution in the value of the Prepetition Secured Parties' interests in the Prepetition Collateral (including Cash Collateral) from and after the Petition Date, if any, for any reason provided for under the Bankruptcy Code, including, without limitation, as a result of the use of Prepetition Collateral, including Cash Collateral, the priming of their interests in Prepetition Collateral by the DIP Facility and the Carve-Out, and the imposition or enforcement of the automatic stay (collectively, "Diminution in Value"). As adequate protection, the Prepetition Secured Parties will be granted, subject to paragraph 11 below, the following:
- (a) Adequate Protection Liens. Pursuant to sections 361, 363(e), and 364(d) of the Bankruptcy Code, effective as of the Petition Date and perfected without the need for execution by the Borrower or the Guarantors or the recordation or other filing by the DIP Lenders of security agreements, control agreements, pledge agreements, financing statements, or other similar documents or the possession or control by the of any DIP Collateral, solely to the extent of any Diminution in Value, shall be granted valid, binding, continuing, enforceable, fully perfected replacement and additional liens on and security interests in (collectively, the "Adequate Protection Liens") all DIP Collateral. The Adequate Protection Liens shall be junior only to the DIP Liens, the Carve-Out, and any Permitted Encumbrance, and senior to all other security

interests in, liens on, or claims against the DIP Collateral. The Adequate Protection Liens shall be non-avoidable and shall not be subject to sections 510, 549, 550, or 551 of the Bankruptcy Code. The Adequate Protection Liens shall be enforceable against and binding upon the Borrower, the Guarantors, their estates and any successors thereto, including, without limitation, any trustee or other estate representative appointed in any Successor Case.

- (b) Adequate Protection Superpriority Claim. To the extent of the aggregate Diminution in Value, the Prepetition Secured Parties shall be granted allowed superpriority administrative expense claims against the Borrower or the Guarantors, and their estates, as provided in section 507(b) of the Bankruptcy Code, immediately junior to the DIP Obligations and subject only to the Carve-Out, and with priority in payment over any and all other claims and administrative expense claims against the Borrower or the Guarantors, now existing or hereafter arising, of any kind or nature whatsoever, including, without limitation, administrative expenses of the kinds specified in or ordered pursuant to sections 105, 326, 327, 328, 330, 331, 361, 362, 363, 364, 365, 503(a), 503(b), 507(a), 507(b), 546(c), 546(d), 552(b), 726, 1113, 1114, and any other provision of the Bankruptcy Code, and shall at all times be senior to the rights of the Borrower and the Guarantors, and any successor trustee or creditor in the Chapter 11 Cases or any Successor Case (the "Adequate Protection Superpriority Claims").
- (c) Adequate Protection Payments, Etc. As further adequate protection, from and after entry of this Interim Order, the Prepetition Agent (on behalf of the Prepetition Secured Parties) shall receive from the Debtors, (1) payment in cash, as and when due under the terms of the Prepetition Term Loan Agreement, of all interest due under the Prepetition Term Loan Agreement, calculated at the default rates under the Prepetition Term Loan Agreement, and (2) (x) upon entry of this Interim Order, immediate cash payment of all accrued and unpaid fees

and disbursements (including all reasonable and documented fees, out-of-pocket costs and expenses of legal, financial, and other advisory professionals of the Prepetition Secured Parties (including, without limitation, Cole Schotz P.C.) owing to the Prepetition Secured Parties under the Prepetition Term Loan Agreement, whether incurred prior to or after the Petition Date and (y) thereafter, when due, all fees and other amounts (including all reasonable and documented legal and advisory fees, out-of-pocket costs and expenses of legal, financial, and other advisory professionals of the Prepetition Secured Parties (including, without limitation, Cole Schotz P.C.) owing to the Prepetition Secured Parties by the Debtors under the Prepetition Term Loan Agreement, provided that payment of any fees and expenses of the Prepetition Secured Parties' professionals incurred after entry of this Interim Order shall be subject to the notice and objection provisions of paragraph 27(b) (the "Adequate Protection Payments"). Payment of any amounts set forth in this clause (c)(2) shall not be subject to disgorgement or recharacterization.

5. **DIP Budget**.

under the DIP Credit Agreement. The Borrower will subsequently provide the DIP Agent with an updated budget before noon every Friday of each week commencing with the second week following the Petition Date, prior to the end of the DIP Budget Period (defined below) for the remaining weeks left under the Initial DIP Budget. The "DIP Budget Period" shall refer to period commencing on the date that this Interim Order is entered and ending on the Outside Effective Date (as defined in the DIP Credit Agreement). Each updated budget shall be subject to the approval of the DIP Lenders in their sole discretion (each such approved budget, a "Supplemental Approved DIP Budget" and together with the Initial DIP Budget, the "DIP Budget"); provided that until such time as the DIP Agent approves any updated budget, the Borrower shall be subject to

and be governed by the terms of the Initial DIP Budget or such Supplemental Approved DIP Budget, as applicable, then in effect in accordance with this Interim Order. The DIP Budget may only be amended, supplemented, modified, restated, replaced, or extended in accordance with this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement). Any such amendment, supplement, modification, restatement, replacement, or extension in accordance with the DIP Credit Agreement may be effected without further order of the Court, so long as approved by the Required Lenders (as such term is defined in the DIP Credit Agreement) of DIP Lenders. The DIP Budget is an integral part of this Interim Order and has been relied upon by the DIP Secured Parties to provide the DIP Facility and consent to this Interim Order.

- Agreement), the Borrower shall provide to the DIP Agent a budget variance report (the "Budget Variance Report") before noon on Friday that is the second week following the Petition Date, and measured weekly thereafter, and the DIP Budget Variance Report shall set forth in reasonable detail actual receipts and disbursements of Borrower, together with a statement certifying compliance with the DIP Budget Covenant (as defined below) set forth below. The Borrower shall not permit or suffer to exist, a negative variance of 15% or more of total the total spending set forth in the Budget for such period (as reflected in the DIP Budget) in the aggregate from the amounts set forth in the DIP Budget (each, a "Permitted Variance"), tested on a cumulative basis (the "DIP Budget Covenant"). The Debtors shall operate in accordance with the DIP Budget and all disbursements of the Debtors shall be consistent with the provisions of the DIP Budget, subject in each case to any Permitted Variance provided herein and in the DIP Credit Agreement.
- (c) The DIP Secured Parties shall have no obligation to permit the use of proceeds of DIP Loans or Cash Collateral, and the Borrower shall have no authority to use proceeds of DIP

Loans or Cash Collateral, other than in accordance with the DIP Budget, subject to the DIP Budget Covenant and as set forth in the DIP Order.

- 6. Monitoring of Collateral. The Prepetition Secured Parties, the DIP Secured Parties, and their consultants and advisors shall be given reasonable access to the Debtors' books, records, assets, and properties for purposes of monitoring the Debtors' businesses and the value of the DIP Collateral, including view access to the Debtors' accounts, at the Debtors' cost and expense, as provided in the DIP Credit Agreement.
- Financial Reporting, Etc. Without limitation of the requirements of the DIP Credit 7. Agreement, the Debtors shall provide to the DIP Secured Parties and to the Prepetition Secured Parties (and, in each case, their consultants, advisors, and professionals) (a) all financial information required under the DIP Loan Documents (including the DIP Credit Agreement) and (b) access upon reasonable notice and during regular business hours to the Debtors' books and records, assets, and properties for purposes of monitoring its businesses and the value of the DIP Collateral. The Debtors shall also provide such reports and information required to be provided in the DIP Credit Agreement and reasonably cooperate, discuss with, and provide to the DIP Secured Parties and the Prepetition Secured Parties (and, in each case, their professionals) all such information as may be reasonably requested. In addition, the Debtors hereby authorize their accountants, attorneys, financial advisors, bankruptcy professionals, and consultants to cooperate, consult with, and provide to the DIP Secured Parties and the Prepetition Secured Parties (and, in each case, their consultants, advisors and professionals) all such information as may be reasonably requested with respect to the business, results of operations and financial condition of the Debtors consistent with the requirements set forth in the DIP Loan Documents (including the DIP Credit Agreement).

- 8. <u>Milestones</u>. It is a condition to the DIP Facility and the use of Cash Collateral that the Debtors shall comply with the milestones set forth in the DIP Credit Agreement (the "<u>Milestones</u>"). The failure to comply with any Milestone shall constitute an Event of Default (as defined in the DIP Credit Agreement) in accordance with the DIP Credit Agreement.
- 9. <u>Subordination of Intercompany, Affiliate Liens</u>. All intercompany or affiliate liens of the Debtors, if any (other than any liens granted to the DIP Secured Parties), will be contractually subordinated to the DIP Liens and the Adequate Protection Liens granted pursuant to this Interim Order and under the DIP Credit Agreement, on terms satisfactory to the DIP Lenders.
- 10. **DIP Lien and Adequate Protection Replacement Lien Perfection**. Automatically upon entry of this Interim Order, the DIP Liens and the Adequate Protection Liens, shall be deemed to be valid, binding, perfected, enforceable, non-avoidable, and effective by operation of law, and not subject to challenge as of the Petition Date, without the need of any further action of any kind; provided, however, that the liens associated with the Interim Roll-Up Amount and the Adequate Protection Liens shall both be subject to the Challenge Period. This Interim Order shall be sufficient and conclusive evidence of the validity, perfection, and priority of the DIP Liens and the Adequate Protection Liens without the necessity of filing or recording any financing statement, deed of trust, mortgage, or other instrument or document which may otherwise be required under the law of any jurisdiction or the taking of any other action to validate or perfect the DIP Liens and the Adequate Protection Liens or to entitle the DIP Liens and the Adequate Protection Liens to the priorities granted herein. Notwithstanding the foregoing, the DIP Agent may, at the direction of the Required Lenders, file such financing statements, deeds of trust, mortgages, security agreements, notices of liens, and other similar documents and is hereby granted relief from the automatic stay of section 362 of the Bankruptcy Code in order to do so, and all such financing

statements, deeds of trust, mortgages, security agreements, notices, and other agreements or documents shall be deemed to have been filed or recorded at the time and on the date of the commencement of the Chapter 11 Cases. The Debtors shall execute and deliver to the DIP Agent all such financing statements, mortgages, security agreements, notices and other documents as the DIP Agent may reasonably request to evidence, confirm, validate, or perfect or to insure the contemplated priority of the DIP Liens and the Adequate Protection Liens, and the Debtors shall take all such further actions that may be required under any applicable law or that the DIP Secured Parties or Prepetition Secured Parties, as applicable, may reasonably request in order to grant, preserve, protect or perfect the DIP Liens and Adequate Protection Liens granted pursuant to this Interim Order. The DIP Agent, at the direction of the Required Lenders, may file a photocopy of this Interim Order as a financing statement with any recording officer designated to file financing statements or with any registry of deeds or similar office in any jurisdiction in which the Debtors have real or personal property and, in such event, the subject filing or recording officer shall be authorized to file or record such copy of this Interim Order. To the extent that a Prepetition Secured Party is the secured party under any account control agreements, listed as loss payee under any of the Debtors' insurance policies or is the secured party under any loan document, financing statement, deed of trust, mortgage, or other instrument or document which may otherwise be required under the law of any jurisdiction to create, validate, attach, perfect, or prioritize liens (any such instrument or document, a "Security Document"), then (i) such Prepetition Secured Party shall be deemed to maintain such possession or notation or exercise such control as a gratuitous bailee and/or gratuitous agent for perfection for the benefit of the DIP Agent and the DIP Lenders, and such Prepetition Secured Party shall comply with the instructions of the DIP Agent with respect to the exercise of such control and (ii) the DIP Agent is also hereby deemed to be the

secured party under such account control agreements, loss payee under the Debtors' insurance policies and the secured party under each such Security Document, and shall have all rights and powers attendant to that position (including, without limitation, rights of enforcement) and shall act in that capacity and distribute any proceeds recovered or received in accordance with the terms of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement).

11. Reservation of Certain Third-Party Rights and Bar of Challenges and Claims.

- (a) The Debtors' stipulations, admissions, agreements, Releases, and waivers contained in this Interim Order, including the Stipulations, are and shall be irrevocably binding upon the Debtors and any and all of the Debtors' successors in interest and assigns (subject to the Challenge Period, including, without limitation, any chapter 7 or chapter 11 trustee, responsible person, examiner with expanded powers, or other estate representative) in all circumstances and for all purposes.
- (b) The Stipulations and all other admissions, agreements, releases, and waivers set forth in this Interim Order also are and shall be binding upon all other persons and entities (including any Committee) and each of their respective successors in interest and assigns in all circumstances and for all purposes, unless, and solely to the extent that (i) such parties in interest (including any Committee) in each case with standing and requisite authority to do so (subject in all respects to any agreement or applicable law which may limit or affect such entity's right or ability to do so) have timely filed the proper pleadings, and timely commenced the appropriate proceedings under the Bankruptcy Code and Bankruptcy Rules, including, without limitation, as required pursuant to Part VII of the Bankruptcy Rules (in each case subject to the limitations set forth below in this paragraph 11), (x) objecting to or challenging any of the Stipulations or (y) otherwise asserting or prosecuting any action against the Prepetition Secured Parties or any of

their respective affiliates, agents, attorneys, advisors, professionals, officers, directors, or employees in connection with or related to the matters covered by the Stipulations (each such proceeding or appropriate pleading commencing a proceeding or other contested matter, a "Challenge"), no later than seventy-five (75) calendar days after the date of entry of this Interim Order for any party in interest with requisite standing (such period, the "Challenge Period" and the date that is the next calendar day after the termination of the Challenge Period shall be referred to as the "Challenge Period Termination Date"), as such date may be extended to any such party in interest by the Prepetition Agent and each applicable Prepetition Secured Party that is the subject of a Challenge or by any such later date as has been ordered by the Court for cause upon a motion filed and served within the Challenge Period (before giving effect to such extension) and (ii) the Court enters judgment in favor of the plaintiff or movant in any such timely and properly commenced Challenge proceeding, and any such judgment has become final and is not subject to any further review or appeal.

- (c) Any Challenge not asserted by the timely and proper filing of a pleading by a party in interest with the requisite standing and authority as contemplated herein prior to the Challenge Period Termination Date (or any extension thereto) shall be deemed forever waived, released, and barred with respect to such party in interest. To the extent a party in interest with requisite standing and authority timely and properly commences a Challenge prior to the Challenge Period Termination Date, all claims, causes of action and other matters not specifically set forth in such Challenge shall be deemed forever waived, released, and barred with respect to such party in interest.
- (d) To the extent any Stipulations are (i) not subject to a Challenge timely and properly commenced prior to the Challenge Period Termination Date or (ii) subject to a Challenge timely

and properly commenced prior to the Challenge Period Termination Date, to the extent any such Challenge does not result in a final and non-appealable judgment or order of the Court that is inconsistent with such Stipulations, then, in each case, without further notice, motion or application to, or order of, or hearing before this Court and without the need or requirement to file any proof of claim: (1) any and all such Challenges by any Committee or any other party in interest shall be deemed to be forever waived, released, and barred (including in any Successor Case), and (2) the Prepetition Obligations shall be deemed to be an allowed secured claim within the meaning of sections 502 and 506 of the Bankruptcy Code for all purposes in connection with these Chapter 11 Cases, and all of the Stipulations and all other waivers, releases, and affirmations set forth in this Interim Order (or any not properly and timely challenged) shall be in full force and effect and shall be binding, conclusive, and final on any person, entity, or party in interest, including any Committee, (in each case, and their successors and assigns) in these Chapter 11 Cases and in any Successor Case for all purposes, without any further order of the Court, and shall not be subject to challenge or objection by the Committee or any other party in interest, including, without limitation, any other statutory committee, any trustee, responsible individual, examiner with expanded powers or other representative of the Debtors' estates.

(e) If any Challenge is timely and properly filed during the Challenge Period, the Stipulations and all other waivers, releases, and affirmations contained in this Interim Order shall nonetheless remain binding and preclusive, as provided in this paragraph 11, on all other parties in interest (including any Committee) and on any other person or entity, except for the plaintiff or movant timely and successfully asserting such Challenge, as set forth in a final, non-appealable order of a court of competent jurisdiction. Notwithstanding anything to the contrary herein, the right to commence any Challenge under this Interim Order is preserved only as against any

particular Prepetition Obligation or against any Prepetition Secured Party to the extent such Challenge is commenced timely and properly prior to the Challenge Period Termination Date and, in respect of such Prepetition Obligation or Prepetition Secured Party, is otherwise waived as set forth in this paragraph 11.

- (f) All remedies or defenses of any party with respect to any Challenge are hereby preserved. Nothing in this Interim Order vests or confers on any person (as defined in the Bankruptcy Code), including any Committee (if any) appointed in these Chapter 11 Cases, standing or authority to pursue any cause of action belonging to the Debtors or their estates, including, without limitation, any Challenges with respect to the Prepetition Obligations, and an order of the Court (or any other court of competent jurisdiction) conferring such standing on a Committee or other party in interest shall be a prerequisite for the prosecution of a Challenge by the Committee or such other party in interest. If a Committee or other party in interest files a motion for standing to pursue a Challenge that includes a draft of the complaint or motion such party in interest seeks to pursue, then the Challenge Period Termination Date shall be stayed as to such movant(s) until the Court rules on the motion for standing.
- Challenge Period expires, then such trustee shall be deemed to be a party in interest other than the Debtors and shall not be bound by the waivers, releases, affirmations, and Stipulations of the Debtors in this Interim Order until the Challenge Period expires. Notwithstanding anything to the contrary herein, if a Committee or other party in interest timely and properly files a Challenge within the Challenge Period that is pending as of the time a chapter 7 or chapter 11 trustee is appointed or elected in these cases, such trustee shall have the right (but not the obligation) to intervene in such Challenge and, upon such intervention, shall be deemed to have timely and

properly filed such Challenge within the Challenge Period to the same extent as the party originally instituting the Challenge.

12. Limitations on Use of DIP Facility and Cash Collateral. Notwithstanding anything herein to the contrary, no portion of the proceeds of the DIP Facility, the DIP Collateral, the Prepetition Collateral, any Cash Collateral, or the Carve-Out and no disbursements set forth in the DIP Budget may be used, in any way, directly or indirectly (a) for any investigation, adversary action, suit, arbitration, proceeding, application, motion, objection, defense, other contested matter, or other litigation of any type, including any Challenge, adverse to the interests of any or all of DIP Lenders, the DIP Agent, the Prepetition Secured Parties, or their respective rights and remedies under the DIP Facility, this Interim Order, the Final Order, the DIP Credit Agreement, or the Prepetition Term Loan Agreement and other documents and instruments in respect of the Prepetition Obligations, including without limitation (i) any action arising under the Bankruptcy Code, (ii) any so-called "lender liability" claims and causes of action, (iii) any action with respect to the validity and extent of the DIP Obligations or Prepetition Obligations or the validity or extent of the DIP Liens or Prepetition Liens, (iv) any action seeking to invalidate, set aside, avoid, reduce, set off, offset, recharacterize, subordinate (whether equitable, contractual, or otherwise), recoup against, disallow, impair, raise any defenses, cross-claims, or counterclaims, or raise any other challenges under the Bankruptcy Code or any other applicable domestic or foreign law or regulation against or with respect to the DIP Liens or the Prepetition Liens, in whole or in part, or (v) appeal or otherwise challenge this Interim Order or the Final Order; *provided* that no more than \$25,000 in the aggregate of the proceeds of the DIP Facility, the DIP Collateral, the Prepetition Collateral, and the Cash Collateral (including any proceeds of the DIP Facility, the DIP Collateral, the Prepetition Collateral, or the Cash Collateral used to fund the Carve-Out) may be used by a

Committee (if any) appointed in these Chapter 11 Cases to investigate (but not prosecute or Challenge, or commence, or initiate the prosecution of, any Challenge, including the preparation of any complaint or motion on account of, or objection to) the Stipulations before termination of the Challenge Period, (b) for any other action, which with the giving of notice or passing of time, would result in an Event of Default under the DIP Facility, (c) for any purpose that is prohibited under the Bankruptcy Code, DIP Loan Documents (including the DIP Credit Agreement), this Interim Order, or the Final Order or that is not in accordance with the DIP Budget, (d) to make any payment, advance, intercompany advance or transfer, or any other remittance or transfer whatsoever (including any intercompany loans and investments) that is not in accordance with the express terms of the DIP Budget, (e) to make any payment in settlement of any claim, action, or proceeding without Bankruptcy Court approval and the prior written consent of the DIP Lenders, which consent shall not be unreasonably withheld or delayed, (f) to prevent, hinder, impede, or delay any of the DIP Secured Parties or Prepetition Secured Parties' enforcement or realization upon or exercise of rights in respect of any of the DIP Collateral or Prepetition Collateral in accordance with the DIP Credit Agreement, or (g) to seek to amend or modify any of the rights or interests granted to any of the DIP Secured Parties or Prepetition Secured Parties under this Interim Order, the DIP Credit Agreement, or the Prepetition Term Loan Agreement, in a manner adverse to any DIP Secured Party or Prepetition Secured Party, including seeking to use Cash Collateral on a contested basis, without the prior written consent of the DIP Agent.

13. Carve-Out.

(a) As used in this Interim Order, the "<u>Carve-Out</u>" means the sum of (i) all fees required to be paid to the Clerk of the Bankruptcy Court ("<u>Court Fees</u>") and all statutory fees required to be paid to the U.S. Trustee under section 1930(a) of title 28 of the United States Code plus interest

at the statutory rate ("US Trustee Fees") (without regard to the notice set forth in (iii) below), (ii) up to the amounts set forth in the respective DIP Budget line items and to the extent allowed by the Court pursuant to a fee application or fee statement on notice or other procedure permitted by any Court order allowing interim compensation or the payment of fees of ordinary course professionals, whether by interim order, final order, procedural order, or otherwise, all reasonable and documented unpaid fees and expenses, all unpaid fees and expenses (the "Allowed Professional Fees") incurred by persons or firms retained by the Debtors pursuant to section 327, 328, or 363 of the Bankruptcy Code (collectively, the "Debtor Professionals") and the Committee, if any, appointed in the Bankruptcy Case (the "Committee Professionals," and together with the Debtor Professionals, the "Professionals") at any time on or before one (1) business day following delivery by the DIP Agent of a Carve-Out Trigger Notice (as defined below), whether allowed by the Bankruptcy Court prior to or after delivery of a Carve-Out Trigger Notice, and (iii) Allowed Professional Fees of Professionals in an aggregate amount not to exceed \$100,000 incurred after delivery by the DIP Agent of the Carve-Out Trigger Notice (the amount set forth in this clause (iii) being the "Post Carve-Out Trigger Notice Cap"); provided that, under no circumstances shall any success, completion, transaction or similar fees be payable from the Carve-Out; provided, further, that the Post Carve-Out Trigger Notice Cap shall be reduced, dollar-for-dollar, by the amount of any fees and expenses incurred and accruing by the Debtors and paid to the applicable Professionals following delivery of a Carve-Out Trigger Notice. For purposes of this Interim Order, "Carve-Out Trigger Notice" shall mean a written notice delivered by email (or other electronic means) by the DIP Agent to counsel to the Debtors, the U.S. Trustee, and counsel to the Committee (if any), which notice may be delivered following the occurrence and during the continuation of an Event of Default, stating that the Post Carve-Out Trigger Notice Cap has been invoked.

- (b) The Debtors shall be authorized and permitted to establish an escrow account, to be held by a qualified institutional escrow agent to be agreed upon by the Debtors and the DIP Secured Parties (the "Escrow Holder"), for the benefit of the Professionals and to fund by wire transfer such escrow on a weekly basis, in each case as provided herein and in the DIP Loan Documents (including the DIP Credit Agreement) (such account, to the extent of such deposited amounts, the "Professional Fee Reserve Account"), in an amount equal to, but not to exceed, the professional fees and costs set forth in the DIP Budget line item for Debtor Professionals for such week. The Professional Fee Reserve Account shall be funded on a weekly basis in accordance with the DIP Budget and ending on the date upon which a Carve-Out Trigger Notice is delivered, plus any amounts provided for in the DIP Budget, if any, since the most recent funding of the Professional Fee Reserve Account through the delivery of the Carve-Out Trigger Notice, in accordance with the DIP Budget. After the delivery of a Carve-Out Trigger Notice, notwithstanding anything herein to the contrary, the Debtors shall be authorized to deposit any cash on hand into the Professional Fee Reserve Account up to the amount of any remaining unpaid professional fees for each of the Professionals set forth in the DIP Budget for the period until delivery of the Carve-Out Trigger Notice plus an amount equal to the Post Carve-Out Trigger Notice Cap.
- (c) From the funds held in the Professional Fee Reserve Account, the Escrow Holder shall release to the Debtor Professionals solely such amounts as are payable from time to time pursuant to the DIP Budget and an applicable order of the Bankruptcy Court, including an order approving interim compensation procedures in the Chapter 11 Cases and any order granting interim or final fee applications for Professionals (each, a "Fee Order"). For the avoidance of

doubt, (a) in making payments from the Professional Fee Reserve Account, Escrow Holder shall be entitled to rely upon written certifications of each Debtor Professional as to the amount such Debtor Professional is authorized pursuant to the DIP Budget and an applicable order of the Bankruptcy Court to be paid at such time from the Professional Fee Reserve Account; and (b) in no circumstances shall the Escrow Holder be obligated to pay any Debtor Professional other than from the funds held, from time to time, in the Professional Fee Reserve Account in accordance with the DIP Budget and an applicable order of the Bankruptcy Court. Funds held in the Professional Fee Reserve Account shall be applied to Allowed Professional Fees that have been incurred following the Petition Date in accordance with procedures to be established in the Chapter 11 Cases. Payments and reimbursements made to a Professional prior to delivery of the Carve-Out Trigger Notice shall reduce the amounts available to such Professional under section (ii) of the Carve-Out, and neither the Professional Fee Reserve Account nor payments therefrom shall in any way increase the Carve-Out. All Allowed Professional Fees payable to Debtor Professionals shall be paid first from funds in the Professional Fee Reserve Account, if any, and, upon exhaustion thereof, from the Carve-Out (less any amounts previously funded into the Professional Fee Reserve Account). For the avoidance of doubt the DIP Liens, Prepetition Liens, and Adequate Protection Liens shall attach to, and shall have a residual right to and lien on, any excess funds in the Professional Fee Reserve Account after satisfaction of the Carve-Out in respect of Allowed Professional Fees, which excess funds shall then be used to satisfy amounts due under the DIP Facility, the Prepetition Obligations, and the Adequate Protection Payments, as applicable, in accordance with their rights and priorities as of the Petition Date and under this Interim Order, in each case, until paid in full in cash.

- (d) None of the DIP Secured Parties or the Prepetition Secured Parties shall be responsible for the payment or reimbursement of any fees or disbursements of any Professional incurred in connection with the Chapter 11 Cases or any Successor Case under any chapter of the Bankruptcy Code. Nothing in this Interim Order or otherwise shall be construed to obligate the DIP Secured Parties or the Prepetition Secured Parties, in any way, to pay compensation to, or to reimburse expenses of, any Professional or to guarantee that the Debtors have sufficient funds to pay such compensation or reimbursement. The DIP Secured Parties and the Prepetition Secured Parties reserve the right to review and object to any fee statement, interim application, or monthly application issued or filed by the Professionals.
- 14. Payment of Compensation. Nothing herein shall be construed as a consent to the allowance of any professional fees or expenses of any of the Debtors or the Committee (if any) or shall limit or otherwise affect the right of the DIP Secured Parties or the Prepetition Secured Parties or any other party in interest to object to the allowance and payment of any such fees and expenses. No professional fees shall be paid absent a Court order allowing such payment, pursuant to a fee application or fee statement on notice, or other procedure permitted by any Court order allowing interim compensation or the payment of fees of ordinary course professionals. So long as no Event of Default exists that has not been waived in writing, the Debtors shall be permitted to pay compensation and reimbursement of expenses allowed by the Court and payable under sections 330 and 331 of the Bankruptcy Code or compensation procedures approved by the Court and in form and substance reasonably acceptable to the Debtors and the DIP Secured Parties, as the same may be due and payable, and the same shall not reduce the Post Carve-Out Trigger Notice Cap.
- 15. <u>Section 506(c) Claims</u>. As a further condition of the DIP Facility, any obligation of the DIP Secured Parties to make DIP Loans, and the Debtors' authorization to use the Cash Collateral

and the consent of the DIP Secured Parties and the Prepetition Secured Parties to the payment of the Carve-Out to the extent provided herein, the Debtors (and any successors thereto or any representatives thereof, including any trustees appointed in the Chapter 11 Cases or any Successor Case) shall be deemed to have waived any rights, benefits, or causes of action under section 506(c) of the Bankruptcy Code as they may relate to or be asserted against the DIP Secured Parties, the DIP Liens, and the DIP Collateral and, subject to the entry of the Final Order, the Prepetition Secured Parties, the Adequate Protection Liens, the Prepetition Liens, and the Prepetition Collateral, and except to the extent of the Carve-Out, nothing contained in this Interim Order, the Final Order, or the DIP Credit Agreement shall be deemed a consent by the Prepetition Secured Parties or the DIP Secured Parties to any charge, lien, assessment, or claim against or in respect of the DIP Collateral or the Prepetition Collateral under sections 105 or 506(c) of the Bankruptcy Code or otherwise, and no such consent shall be implied from any other action, inaction, or acquiescence by any by any such parties.

16. Collateral Rights; Limitations in Respect of Subsequent Court Orders. Without limiting any other provisions of this Interim Order, unless the DIP Agent and the Prepetition Agent have provided their prior written consent, no credit shall be obtained, or indebtedness incurred that is secured by a security, mortgage, or collateral interest or other lien on all or any portion of the DIP Collateral or entitled to priority administrative status which is superior to or *pari passu* with those granted pursuant to this Interim Order to or for the benefit of the DIP Secured Parties or the Prepetition Secured Parties. Further, the proceeds of the DIP Facility and the Cash Collateral shall not be used for any purpose other than as permitted under the DIP Credit Agreement and consistent with this Interim Order and the DIP Budget (subject to Permitted Variances).

- Proceeds of Subsequent Financing. Without limiting the provisions and protections of paragraph 16 hereof, if at any time prior to the Maturity Date, the Debtors, the Debtors' estates, any trustee, any examiner with enlarged powers or any responsible officer subsequently appointed shall obtain credit or incur debt in violation of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement), then all of the cash proceeds derived from such credit or debt and all Cash Collateral in the amount necessary to satisfy all DIP Obligations then outstanding shall immediately be turned over to the DIP Agent for the indefeasible repayment and satisfaction in full in cash of all DIP Obligations then outstanding. For the avoidance of doubt, if the Debtors, any trustee, any examiner with expanded powers, or any responsible officer subsequently appointed in these Chapter 11 Cases or any Successor Case shall obtain credit or incur debt pursuant to section 364(d) of the Bankruptcy Code at any time prior to the Maturity Date, the Prepetition Secured Parties' rights to object to the Debtors' use of Cash Collateral and assert a lack of adequate protection shall be fully preserved.
- 18. <u>Cash Management</u>. From and after the date of the entry of this Interim Order, the Debtors shall maintain a cash management system that is the same as or substantially similar to their prepetition cash management system or as otherwise agreed by the DIP Lenders and authorized by DIP Lenders (in their sole discretion) and the Court. The Debtors shall maintain no accounts except those identified in any order of this Court approving the maintenance of the Debtors' cash management system (the "<u>Cash Management Order</u>") or that may be subsequently opened pursuant to the Cash Management Order. It shall constitute an Event of Default (as defined in the DIP Credit Agreement) if the Debtors' cash management system shall not at all times be maintained in accordance with the terms of the DIP Credit Agreement, the Cash Management Order, and this Interim DIP Order in all material respects, and otherwise in a manner which shall

be reasonably satisfactory to the DIP Secured Parties in accordance with the terms of the DIP Loan Documents (including the DIP Credit Agreement). The Debtors and the financial institutions where the Debtors maintain deposit accounts (as identified in the Cash Management Order) are authorized to remit, without offset or deduction, funds in such deposit accounts upon receipt of any direction to that effect from the DIP Agent (acting at the direction of the DIP Lenders). The DIP Agent shall be deemed to have "control" over all cash management accounts for all purposes of perfection under the Uniform Commercial Code pursuant to this Interim Order and, if required or requested by the DIP Agent, pursuant to control agreements reasonably acceptable to the DIP Agent.

- 19. <u>Disposition of DIP Collateral</u>. It shall constitute an Event of Default (as defined in the DIP Credit Agreement) if the Debtors sell (including, without limitation, any sale and leaseback transaction), transfer (including any assignment of rights), lease, encumber or otherwise dispose of any portion of the DIP Collateral, except as expressly permitted under the DIP Credit Agreement.
- 20. <u>Survival of Certain Provisions</u>. In the event of the entry of any order converting any of these Chapter 11 Cases into a Successor Case, the DIP Liens, the DIP Superpriority Claim, the Adequate Protection Liens, the Adequate Protection Superpriority Claim, and the Carve-Out shall continue in this proceeding and in any Successor Case, and such DIP Liens, DIP Superpriority Claim, Adequate Protection Liens, Adequate Protection Superpriority Claim, and Carve-Out shall maintain their respective priorities as provided by this Interim Order.

21. Events of Default; Maturity Date; Rights and Remedies Upon Event of Default.

(a) Any automatic stay otherwise applicable to the DIP Secured Parties and the Prepetition Agent, whether arising under sections 105 or 362 of the Bankruptcy Code or otherwise,

is hereby modified so that, (i) upon the occurrence and during the continuance of an Event of Default, the DIP Secured Parties, in their sole and absolute discretion, may immediately (w) deliver a notice of an Event of Default; (x) terminate any pending DIP Loans and use of Cash Collateral; (y) sweep all cash in any controlled accounts, and (z) terminate the DIP Facility and all commitments thereunder and (ii) upon and after the occurrence of the Maturity Date, the DIP Agent and the Prepetition Agent shall, without further notice to, hearing of, or order from this Court, to the extent necessary to permit the DIP Secured Parties to take any or all of the following actions, at the same time or different times, unless the Court orders otherwise, and subject only to subparagraph (b) of this paragraph 21, as applicable, be immediately entitled to exercise all of their rights and remedies in respect of the DIP Collateral and the Prepetition Collateral, in accordance with this Interim Order, the DIP Documents, and/or the Prepetition Term Loan Agreement, as applicable. The term "Maturity Date" shall mean the date that is the earliest of (i) the date that is one-hundred ten (110) days after the Petition Date, (ii) the effective date or date of the substantial consummation (as defined in Section 1101(2) of the Bankruptcy Code) of a Plan of Reorganization or similar dispositive restructuring plan that has been confirmed by an order of the Bankruptcy Court; (iii) the date the Bankruptcy Court orders the conversion of the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code; (iv) the date the Bankruptcy Court orders the dismissal of the Chapter 11 Cases; (v) the date on which the Borrower consummates a sale of all or substantially all of the assets of the Borrower or its Subsidiaries pursuant to Section 363 of the Bankruptcy Code or otherwise; (vi) with the consent of the DIP Lenders (solely to the extent the DIP Obligations have not been satisfied in full), the confirmation of a chapter 11 plan pursuant to section 1129 of the Bankruptcy Code; and (vii) the occurrence of any Event of Default or the date of acceleration of the DIP Term Loans or early termination of the DIP Term Loan Commitments

hereunder or the DIP Financing Orders, including as a result of the occurrence of an Event of Default.

Notwithstanding the foregoing subparagraph (a) of this paragraph 21, immediately (b) following the giving of notice by the DIP Agent to counsel to the Debtors, the U.S. Trustee, and counsel to the Committee, if any, of the occurrence of an Event of Default under the DIP Loan Documents (the "Termination Notice") (i) all Commitments of the DIP Lenders to provide any DIP Loans shall immediately be suspended; (ii) the Debtors shall have no right to request or use any proceeds of any DIP Loans or DIP Collateral, or to use Cash Collateral, other than towards the satisfaction of the DIP Obligations and the Carve-Out, as provided in this Interim Order and the DIP Loan Documents (including the DIP Credit Agreement); provided that, during the Default Notice Period (as defined below), the Debtors shall be permitted solely to continue to use drawn proceeds of the DIP Facility and any Cash Collateral subject to the consent of the DIP Lenders for any critical business-related expenses necessary to operate the Debtors' businesses and preserve the DIP Collateral as determined by the Debtors in their reasonable discretion and in good faith; provided, further that the only basis on which the Debtors shall have the right to contest a Termination Notice shall be with respect to the validity of the Event of Default giving rise to such Termination Notice (i.e., whether or not such Event of Default has occurred or not), (iii) the Debtors shall deliver and cause the delivery of the proceeds of the DIP Loans and the DIP Collateral to the DIP Agent as provided herein and in the DIP Loan Documents (including the DIP Credit Agreement) subject to the funding of the Carve-Out, and (iv) the DIP Agent shall be permitted to apply such proceeds in accordance with the terms of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement). The Debtors and the Committee (if any) shall be entitled to an emergency hearing before this Court (subject to the Court's availability)

within five (5) calendar days after the Termination Notice is sent by the DIP Agent to the Debtors, the U.S. Trustee, and counsel to the Committee (if any) (such 5-calendar-day period, the "Default Notice Period"). If the Debtors, the Committee (if any), or any other party in interest does not contest the occurrence of the Event of Default within the Default Notice Period or if there is a timely contest of the occurrence of an Event of Default and the Court after notice and a hearing declines to stay the enforcement thereof, the Maturity Date shall be deemed to have occurred for all purposes and the automatic stay shall be modified to permit the DIP Agent and the DIP Lenders to exercise all remedies under this Interim Order, the Final Order, the other DIP Loan Documents (including the DIP Credit Agreement), and applicable law, including, without limitation, to (i) setoff any and all amounts in accounts maintained by the Debtors with the DIP Agent or the DIP Lenders against the DIP Obligations, (ii) to otherwise enforce any and all rights against the DIP Collateral in the possession of any of the applicable DIP Lenders, including, without limitation, disposition of the DIP Collateral solely for application towards the DIP Obligations; and (iii) take any other actions or exercise any other rights or remedies permitted under this Interim Order, the Final Order, the other DIP Loan Documents (including the DIP Credit Agreement), or applicable law to effect the repayment of the DIP Obligations.

- (c) Upon the occurrence of the Maturity Date (but subject, only in the case of the occurrence of the Maturity Date resulting from an Event of Default, to the provisions of paragraph 21(b)), the DIP Secured Parties and the Prepetition Secured Parties are authorized to exercise all remedies and proceed under or pursuant to the applicable DIP Loan Documents (including the DIP Credit Agreement), the Prepetition Term Loan Agreement, this Interim Order, and applicable law.
- (d) The automatic stay imposed under section 362(a) of the Bankruptcy Code is hereby modified pursuant to this Interim Order and the DIP Loan Documents (including the DIP Credit

Agreement) as necessary to (i) permit the Debtors to grant the Adequate Protection Liens and the DIP Liens and to incur all DIP Obligations and all liabilities and obligations to the Prepetition Secured Parties hereunder and under the DIP Credit Agreement, as the case may be, and (ii) authorize the DIP Agent and Prepetition Agent to retain and apply payments and otherwise enforce their respective rights and remedies hereunder, subject to the provisions of paragraph 21(b) hereof.

- (e) The Debtors shall reasonably cooperate with the DIP Secured Parties and/or the Prepetition Secured Parties in their efforts to enforce their liens and security interests in the DIP Collateral or the Prepetition Collateral, as applicable, in accordance with this Interim Order and (other than the right to contest whether a DIP Termination Event has occurred and is continuing) the Debtors shall not take or direct any entity to take any action designed or intended to hinder or restrict in any respect such party from enforcing its rights or remedies in the DIP Collateral or the Prepetition Collateral, as applicable.
- (f) Nothing included herein shall prejudice, impair, or otherwise affect the Prepetition Agent's or the DIP Agent's rights to seek (on behalf of the Prepetition Secured Parties and the DIP Secured Parties, respectively) any other or supplemental relief in respect of the Debtors (including, as the case may be, other or additional adequate protection).

22. Applications of Proceeds of Collateral, Payments and Collections.

(a) As a condition to the DIP Loans and the authorization to use Cash Collateral, the Debtors have agreed that proceeds of any DIP Collateral and Prepetition Collateral, any amounts held on account of the DIP Collateral or Prepetition Collateral, and all payments and collections received by the Debtors with respect to all proceeds of DIP Collateral and Prepetition Collateral shall be used and applied in accordance with the DIP Loan Documents (including the DIP Credit

Agreement) (including repayment and reduction of the DIP Obligations), the DIP Budget (subject to Permitted Variances) and this Interim Order.

- (b) Subject to the Debtors' rights under paragraph 21(b) hereof and the funding of the Carve-Out, if applicable, upon and after the occurrence of the Maturity Date, all proceeds of DIP Collateral and Prepetition Collateral, whenever received, shall be paid and applied as follows: (i) *first*, to permanently and indefeasibly repay and reduce the DIP Obligations then due and owing in accordance with the DIP Loan Documents (including the DIP Credit Agreement), until paid and satisfied in full in cash; (ii) *second*, subject to paragraph 11 hereof, to permanently and indefeasibly repay and reduce the Prepetition Obligations (including any adequate protection obligations) then due and owing in accordance with the Prepetition Term Loan Agreement, until paid and satisfied in full in cash; and (iii) *third*, to the Debtors' estates. For avoidance of doubt, nothing in this Interim Order shall be construed to limit the voluntary and mandatory repayment provisions set forth in the DIP Loan Documents (including the DIP Credit Agreement).
- 23. Proofs of Claim, etc. None of the DIP Secured Parties or the Prepetition Secured Parties shall be required to file proofs of claim in the Chapter 11 Cases or any Successor Case for any claim described herein. Notwithstanding any order entered by the Court in relation to the establishment of a bar date in the Chapter 11 Cases or any Successor Case to the contrary, the DIP Agent, on behalf of itself and the DIP Secured Parties, and the Prepetition Agent, on behalf of itself and the Prepetition Secured Parties, respectively, are hereby authorized and entitled (but not required), in each of their sole and absolute discretion to file (and amend or supplement, as each sees fit) a proof of claim or aggregate proofs of claim in the Chapter 11 Cases or any Successor Case for any claim described herein; any proof of claim filed by the DIP Agent or the Prepetition Agent shall be deemed to be in addition to and not in lieu of any other proof of claim that may be

filed by any of the respective DIP Secured Parties or Prepetition Secured Parties. Any order entered by the Court in relation to the establishment of a bar date for any claim (including without limitation administrative claims) in the Chapter 11 Cases or any Successor Case shall not apply to the DIP Agent, the other DIP Secured Parties, the Prepetition Agent or the other Prepetition Secured Parties.

Payments Free and Clear. Any and all payments or proceeds remitted to the DIP Secured Parties or, except as otherwise provided herein, subject to paragraph 11 hereof, the Prepetition Secured Parties, pursuant to the provisions of this Interim Order, the DIP Loan Documents (including the DIP Credit Agreement) or any subsequent order of the Court shall be irrevocable, received free and clear of any claim, charge, assessment or other liability.

25. Other Rights and Obligations.

(a) Good Faith Under Section 364(e) of the Bankruptcy Code; No Modification or Stay of this Interim Order. The DIP Secured Parties and the Prepetition Secured Parties have acted in good faith in connection with the DIP Facility, the DIP Loans, the DIP Loan Documents (including the DIP Credit Agreement), the Prepetition Term Loan Agreement, and this Interim Order and are entitled to rely upon the protections granted herein and by section 364(e) of the Bankruptcy Code. Based on the findings set forth in this Interim Order and in accordance with section 364(e) of the Bankruptcy Code, which is applicable to the DIP Facility as approved by this Interim Order, and notwithstanding any modification, amendment, reversal, vacatur, or stay of any or all of this Interim Order by a subsequent order of the Court or any other court, the DIP Secured Parties and the Prepetition Secured Parties are entitled to the protections provided in section 364(e) of the Bankruptcy Code, and no such appeal, modification, amendment, reversal, vacatur, or stay shall affect the validity and enforceability of any DIP Obligation, DIP Lien, Adequate Protection

Superpriority Claim, or Adequate Protection Lien, or any other advances made hereunder or the claims, liens, security interest, or priority authorized or created hereby or pursuant to the DIP Loan Documents (including the DIP Credit Agreement). Notwithstanding any such modification, amendment, reversal, vacatur, or stay, any claim granted to the DIP Secured Parties hereunder arising prior to the effective date of such modification, amendment, reversal vacatur, or stay of any DIP Liens or of the DIP Superpriority Claim granted to or for the benefit of the DIP Secured Parties shall be governed in all respects by the original provisions of this Interim Order, and the DIP Secured Parties shall be entitled to all of the rights, remedies, privileges, and benefits, including the DIP Liens and the DIP Superpriority Claim granted herein, with respect to any such claim.

(b) Expenses. To the fullest extent provided in the Prepetition Term Loan Agreement and this Interim Order, the Debtors shall pay all expenses incurred by the DIP Secured Parties and the Prepetition Secured Parties (including, without limitation, the reasonable and documented fees and disbursements of their counsel, including Cole Schotz P.C., any other advisors and any other local counsel that they shall retain and any internal or third-party appraisers, consultants, financial, restructuring or other advisors and auditors advising any such counsel) without duplication, in connection with (i) the preparation, execution, delivery, funding, and administration of the DIP Credit Agreement and the other DIP Loan Documents, including, without limitation, all due diligence fees and expenses incurred or sustained in connection therewith, (ii) the administration of the Prepetition Term Loan Agreement, (iii) the Chapter 11 Cases or any Successor Case, or (iv) enforcement of any rights or remedies under the DIP Loan Documents (including the DIP Credit Agreement) or the Prepetition Term Loan Agreement, or applicable law, in each case whether or not the transactions contemplated hereby are fully consummated. Notwithstanding any

Casse 225-10097/9-JK\$ Door 505-11 FFiled 006 004 225 Page 1500 off 1288

other provisions of this Interim Order, the Prepetition Secured Parties and the DIP Secured Parties, and their advisors and professionals shall submit an invoice in summary form, which shall not be required to include time entry detail and may be redacted or modified to the extent necessary to delete any information subject to the attorney-client privilege, any information constituting attorney work product, or any other confidential information, and the provision of such invoices shall not constitute any waiver of the attorney-client privilege or of any benefits of the attorney work-product doctrine for fees and expenses incurred after entry of this Interim Order to the Debtors, the U.S. Trustee and counsel for the Committee (if any). Thereafter, within ten (10) calendar days of presentment of such statements, if no written objections to the reasonableness of the fees and expenses charged in any such invoice (or portion thereof) is made, the Debtors shall pay in cash all such fees and expenses of the Prepetition Secured Parties, the DIP Secured Parties, and their advisors and professionals. Any objection to the payment of such fees or expenses shall be made only on the basis of "reasonableness," and shall specify in writing the amount of the contested fees and expenses and the detailed basis for such objection. To the extent an objection only contests a portion of an invoice, the undisputed portion thereof shall be promptly paid. If any such objection to payment of an invoice (or any portion thereof) is not otherwise resolved between the objecting party and the issuer of the invoice, either party may submit such dispute to the Court for a determination as to the reasonableness of the relevant disputed fees and expenses set forth in the invoice. This Court shall resolve any dispute as to the reasonableness of any fees and expenses. For the avoidance of doubt and without limiting any of the foregoing or any other provision of this Interim Order, the fees specified in the "Costs and Expenses" section of the DIP Credit Agreement are, upon entry of this Interim Order and irrespective of any subsequent order approving or denying the DIP Facility or any other financing pursuant to section 364 of the Bankruptcy Code, entitled

to all of the protections of section 364(e) of the Bankruptcy Code that may apply to such fees, and are deemed fully earned, indefeasibly paid, non-refundable, irrevocable and non-avoidable as of the date of this Interim Order.

Credit Bid. As provided in section 363(k) of the Bankruptcy Code, the DIP (c) Secured Parties and the Prepetition Secured Parties shall each have the right to credit bid (either directly or through one or more acquisition vehicles), up to the full amount of the DIP Obligations, the Prepetition Obligations, and the Adequate Protection Superpriority Claims, in connection with any sale of all or any portion of the Prepetition Collateral or the DIP Collateral, including (without limitation) any sale occurring pursuant to section 363 of the Bankruptcy Code or included as part of any chapter 11 plan subject to confirmation under section 1129(b)(2)(A)(ii)-(iii) of the Bankruptcy Code, by a chapter 7 trustee under section 725 of the Bankruptcy Code, or otherwise, without the need for further court order authorizing the same. The Debtors shall not object to any DIP Secured Party or any Prepetition Secured Party credit bidding up to the full amount of the applicable outstanding DIP Obligations or Prepetition Obligations (including any Adequate Protection Superpriority Claims), in each case including any accrued interest, fees, and expenses, in any sale of any DIP Collateral or Prepetition Collateral, as applicable, whether such sale is effectuated through sections 363 or 1129 of the Bankruptcy Code, by a chapter 7 trustee under section 725 of the Bankruptcy Code, or otherwise. If the DIP Agent, DIP Secured Parties, and the Prepetition Secured Parties (either directly or through one or more acquisition vehicles) make a credit bid in connection with any auction or other sale process relating to the sale or other disposition of any Prepetition Collateral or DIP Collateral, then for purposes of such auction or sale process or any applicable order of this Court, each of the DIP Agent, DIP Secured Parties, and Prepetition Secured Parties shall be deemed to be a "qualified bidder" and its bid shall be a "qualified bid" regardless of whether any qualified bidder or qualified bid requirements are satisfied.

- Lenders Not Responsible Persons. In (i) making the decision to make the DIP Loans and consent to the use of Cash Collateral, (ii) administering the DIP Facility and the DIP Loans, (iii) extending other financial accommodations to the Debtors under the DIP Documents, and (iv) making the decision to collect the indebtedness and obligations of the Debtors, neither the DIP Agent, the DIP Secured Parties nor the Prepetition Secured Parties shall be considered to (x) owe any fiduciary obligation to the Debtors or any other party with respect to their exercise of any consent or other rights afforded them under the DIP Documents or this Interim Order or (y) be exercising control over any operations of the Debtors or acting in any way as a responsible person, or as an owner or operator under any applicable law, including without limitation, any environmental law (including but not limited to the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. 9601, et seq., and the Resource Conservation and Recovery Act, 42 U.S.C. 6901, et seq., as either may be amended from time to time, or any similar federal or state statute, but in both cases solely by reason of having made the DIP Loans).
- (e) **Binding Effect**. The provisions of this Interim Order shall be binding upon and inure to the benefit of the DIP Secured Parties and the Prepetition Secured Parties, the Debtors, and their successors and assigns (including any trustee or other fiduciary hereinafter appointed as a legal representative of the Debtors or with respect to the property of the estates of the Debtors) whether in the Chapter 11 Cases, in any Successor Case, or upon dismissal of any such Chapter 11 or Chapter 7 case.

(f) No Waiver.

- The failure of the DIP Secured Parties or the Prepetition Secured Parties to (i) seek relief or otherwise exercise their rights and remedies under this Interim Order, the other DIP Loan Documents (including the DIP Credit Agreement), or the Prepetition Term Loan Agreement or otherwise, as applicable, shall not constitute a waiver of any of the DIP Secured Parties' or Prepetition Secured Parties' rights hereunder, thereunder, or otherwise. Notwithstanding anything herein, the entry of this Interim Order is without prejudice to and does not constitute a waiver of, expressly or implicitly, or otherwise impair any of the rights, claims, privileges, objections, defenses, or remedies of the DIP Secured Parties or the Prepetition Secured Parties under the Bankruptcy Code or under non-bankruptcy law against any other person or entity in any court, including without limitation, the rights of the DIP Agent and the Prepetition Agent (1) to request conversion of the Chapter 11 Cases to a case under Chapter 7, dismissal of the Chapter 11 Cases, or the appointment of a trustee in the Chapter 11 Cases, (2) to propose, subject to the provisions of section 1121 of the Bankruptcy Code, a Plan, or (3) to exercise any of the rights, claims or privileges (whether legal, equitable, or otherwise) on behalf of the DIP Secured Parties or the Prepetition Secured Parties.
- (ii) The failure or delay on the part of any DIP Secured Parties or Prepetition Secured Parties to seek relief or otherwise exercise their rights and remedies under this Interim Order, the other DIP Loan Documents (including the DIP Credit Agreement), the Prepetition Term Loan Agreement, or applicable law, as the case may be, shall not constitute a waiver of any of their respective rights hereunder, thereunder, or otherwise. No delay on the part of any party in the exercise of any right or remedy under this Interim Order shall preclude any other or further exercise of any such right or remedy or the exercise of any other right or remedy. None of the

rights or remedies of any party under this Interim Order shall be deemed to have been amended, modified, suspended, or waived unless such amendment, modification, suspension, or waiver is express, in writing and signed by the party against whom such amendment, modification, suspension, or waiver is sought. No consents required hereunder by any DIP Secured Parties or Prepetition Secured Parties shall be implied by any inaction or acquiescence by any of the DIP Secured Parties or Prepetition Secured Parties, respectively.

- (g) **No Third-Party Rights**. Except as explicitly provided for herein, this Interim Order does not create any rights for the benefit of any third party, creditor, equity holder or any direct, indirect, third party, or incidental beneficiary.
- (h) Limitation on Surcharge. Without limiting the terms of the Carve-Out and subject to the entry of the Final Order, no costs or expenses of administration which have been or may be incurred in the Chapter 11 Cases or any Successor Case at any time shall be surcharged against, and no person may seek to surcharge any costs or expenses of administration against the DIP Secured Parties, the Carve-Out (other than parties entitled to assert a right to be paid under the Carve-Out), the DIP Collateral or the Prepetition Collateral, pursuant to sections 105 or 506(c) of the Bankruptcy Code or otherwise, without the prior written consent of the DIP Agent and the Prepetition Agent (and the beneficiaries of the Carve-Out in the case of a surcharge against the Carve-Out). No action, inaction or acquiescence by the DIP Secured Parties or the Prepetition Secured Parties shall be deemed to be or shall be considered evidence of any alleged consent to a surcharge against the DIP Secured Parties, the DIP Collateral, the Prepetition Secured Parties or the Prepetition Collateral.
- (i) **No Marshaling**. Neither the DIP Secured Parties nor, subject to the entry of the Final Order, the Prepetition Secured Parties shall be subject to the equitable doctrine of

"marshaling" or any other similar doctrine with respect to any of the DIP Collateral or the Prepetition Collateral, as applicable.

- (j) Section 552(b). The Prepetition Secured Parties and the DIP Secured Parties shall each be entitled to all of the rights and benefits of section 552(b) of the Bankruptcy Code; and, subject to the entry of the Final Order, to the extent provided therein, the "equities of the case" exception under section 552(b) of the Bankruptcy Code shall not apply to the DIP Secured Parties or the Prepetition Secured Parties with respect to proceeds, product, offspring, or profits of any of the Prepetition Collateral or the DIP Collateral.
- (k) Amendment. The Debtors and the DIP Agent may amend, modify, supplement, or waive any provision of the DIP Credit Agreement (a "DIP Loan Amendment") without further approval of the Court; provided that (i) such DIP Loan Amendment is not material (for purposes hereof, a "Material DIP Loan Amendment" shall mean any modification or amendment that (1) increases the interest rate (other than as a result of the imposition of the default rate) or fees charged in connection with the DIP Facility, (2) increases the commitments of the DIP Lenders to make DIP Loans under the DIP Credit Agreement, or (3) changes the Maturity Date) and is undertaken in good faith by the DIP Lenders and DIP Agent and (ii) the Debtors provide written notice of the DIP Loan Amendment on the docket. Any Material DIP Loan Amendment to the DIP Credit Agreement must be set forth in writing, signed by, or on behalf of the Debtors and the DIP Agent (after having obtained the approval of the requisite DIP Secured Parties as provided in the DIP Credit Agreement) and approved by the Court on five (5) days' notice to parties in interest in order to be effective, provided that the DIP Secured Parties and the Debtors reserve the right to seek approval of any Material DIP Loan Amendment on an expedited basis.

- (1) **Priority of Terms**. To the extent of any conflict between or among (i) the express terms or provisions of any of the DIP Loan Documents (including the DIP Credit Agreement), the Motion, or any other agreements, on the one hand, and (b) the terms and provisions of this Interim Order, on the other hand, unless such term or provision herein is phrased in terms of "defined in" or "as set forth in" the DIP Loan Documents (including the DIP Credit Agreement), the terms and provisions of this Interim Order prevail.
- (m) Survival of Interim Order. The provisions of this Interim Order and any actions taken pursuant hereto shall survive entry of any order which may be entered (i) confirming any plan in the Chapter 11 Cases, (ii) converting of the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code, (iii) to the extent authorized by applicable law, dismissing the Chapter 11 Cases, (iv) withdrawing of the reference of the Chapter 11 Cases from this Court, or (v) providing for abstention from handling or retaining of jurisdiction of the Chapter 11 Cases in this Court. The terms and provisions of this Interim Order and the Prepetition Term Loan Agreement, including the DIP Liens and DIP Superpriority Claim granted pursuant to this Interim Order, the DIP Credit Agreement, and the Prepetition Term Loan Agreement and any priorities and protections granted to or for the benefit of the DIP Secured Parties and Prepetition Secured Parties (including the Adequate Protection Liens and the Adequate Protection Superpriority Claim) hereunder and thereunder shall continue in full force and effect to the fullest extent provided by section 364(e) of the Bankruptcy Code.
- (n) **Enforceability**. This Interim Order shall constitute findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052 and shall take effect and be fully enforceable *nunc pro tunc* to the Petition Date immediately upon execution hereof.

- (o) **No Waivers or Modification of Interim Order**. The Debtors irrevocably waive any right to seek any modification or extension of this Interim Order without the prior written consent of the DIP Agent and the Prepetition Agent, and no such consent shall be implied by any other action, inaction, or acquiescence of the DIP Agent or the Prepetition Agent. The Debtors may not seek to modify or to alter relative lien priority of the DIP Liens, the Prepetition Liens and the Adequate Protection Liens set forth in this Interim Order.
- (p) Waiver of any Applicable Stay. Any applicable stay (including, without limitation, under Bankruptcy Rule 6004(h)) is hereby waived and shall not apply to this Interim Order.
- 26. **Reservations to Final Order**. Notwithstanding language in this Interim Order that provides that certain relief is subject to or conditioned upon entry of a Final Order, such provisions are without prejudice to rights of parties in interest to object to the relief on a final basis and the Court's authority to determine the final relief.

27. Final Hearing.

- (a) The Final Hearing to consider entry of the Final Order and final approval of the DIP Facility is scheduled for June _____, 2025, at ______.m. (prevailing Eastern Time) at the United States Bankruptcy Court for the District of Delaware.
- (b) On or before two (2) business days after entry of this Interim Order, the Debtors shall cause to be served, by United States mail, first-class postage prepaid, notice of the entry of this Interim Order and of the Final Hearing (the "Final Hearing Notice"), together with copies of this Interim Order and the Motion, on any known party affected by the terms of this Interim Order, and to any other party that has filed a request for notices with this Court prior thereto. The Final Hearing Notice shall state that any party in interest objecting to the entry of the proposed Final

Casse 225-1009799-JK\$ Door 505-11 FFiled 0060002225 Prage 1608 off 1288

Order shall file written objections with the Clerk of the Court no later than _________, at 4:00 p.m. (prevailing Eastern Time), which objections shall be served on (i) proposed counsel for the Debtors, Pashman Stein Walder Hayden, P.C., 824 North Market Street, Suite 800, Wilmington, Delaware 19801 (Attn.: John Weiss (jweiss @pashmanstein.com), Leah M. Eisenberg (leisenberg@pashmanstein.com), Richard W. Riley (rriley@pashmanstein.com) and David E. Sklar (dsklar@pashmanstein.com); (ii) counsel to the DIP Lenders, Cole Schotz P.C., 500 Delaware Avenue, Suite 1410, Wilmington, Delaware 19801 (Attn: Patrick J. Reilley (PReilley@coleschotz.com) and Melissa M. Hartlipp (MHartlipp@coleschotz.com)) and 1325 Avenue of the Americas, 19th Floor, New York, NY 10019 (Attn: Daniel F.X. Geoghan (DGeoghan@coleschotz.com)); (iii) the U.S. Trustee, 844 King Street, Suite 2207, Wilmington, DE 19801 (Attn: Jane M. Leamy) (jane.m.leamy@usdoj.gov); and (iv) counsel to any official committee appointed in the Chapter 11 Cases.

28. **Retention of Jurisdiction**. The Court has and will retain jurisdiction to enforce this Interim Order according to its terms.

EXHIBIT 1

DIP CREDIT AGREEMENT

SENIOR SECURED SUPERPRIORITY DEBTOR-IN-POSSESSION TERM LOAN AGREEMENT

dated as of June 2, 2025,

among

VWS HOLDCO, INC.,

a Debtor and Debtor-in-Possession under Chapter 11 of the Bankruptcy Code as Borrower,

SHOOSMITH BROS., INC.,

a Debtor and Debtor-in-Possession under Chapter 11 of the Bankruptcy Code as Guarantor

the Lenders party hereto from time to time

and

VOLUNTEER ENTERPRISES, LLC,

as Administrative Agent

Cassae 225-110097/99-JKSS DDoorc 5105-11 FFi lead 0066/004/225 PPagge 16161.off 12938

TABLE OF CONTENTS

		Page
SECTION 1.	INTERPRETATION:	2
SECTION 2.	DIP FACILITY:	2
2.1	DIP Term Loans	2
2.2	Roll-Up Loans	3
2.3	Procedure for Advances	3
2.4	Repayment	5
2.5	Optional Prepayments	5
2.6	Mandatory Prepayments	6
2.7	No Reborrowings	7
SECTION 3.	GENERAL LOAN PROVISIONS:	7
3.1	Repayment of Principal and Outstanding Obligations	7
3.2	Interest	8
3.3	ReservedError! Bookmark not	defined.
3.4	Manner of Payment	8
3.5	Evidence of Indebtedness	9
3.6	Sharing of Payments by Lenders	9
3.7	Increased Costs	10
3.8	Taxes	11
3.9	Mitigation Obligations; Replacement of Lenders	14
3.10	Defaulting Lenders	15
3.11	No Discharge; Survival.	16
SECTION 4.	CONDITIONS OF CLOSING AND BORROWING:	16
4.1	Conditions to Closing and Funding of First DIP Term Loan Advance	16
4.2	Conditions to Funding the Second DIP Term Loan Advance	19
SECTION 5.	REPRESENTATIONS:	20
5.1	Organization; Powers Error! Bookmark not	defined.
5.2	Authority; Enforceability	20
5.3	Approvals; No Conflicts	21
5.4	Financial Condition; No Material Adverse Effect Error! Bookmark not	defined.
5.5	LitigationError! Bookmark not	defined.

TABLE OF CONTENTS

(continued)

Page

5.6	Environmental Matters	Error! Bookmark not defined.
5.7	Compliance with the Laws; No Defaults	Error! Bookmark not defined.
5.8	Investment Company Act	Error! Bookmark not defined.
5.9	Taxes	Error! Bookmark not defined.
5.10	ERISA	Error! Bookmark not defined.
5.11	Disclosure; No Material Misstatements	Error! Bookmark not defined.
5.12	Insurance	Error! Bookmark not defined.
5.13	Restriction on Liens	Error! Bookmark not defined.
5.14	Subsidiaries	Error! Bookmark not defined.
5.15	Location of Business and Offices	Error! Bookmark not defined.
5.16	Properties; Titles	Error! Bookmark not defined.
5.17	Maintenance of Properties	Error! Bookmark not defined.
5.18	Gas Imbalances, Prepayments	Error! Bookmark not defined.
5.19	Marketing of Production	Error! Bookmark not defined.
5.20	Swap Agreements	Error! Bookmark not defined.
5.21	Use of Proceeds	23
5.22	Anti-Corruption Laws and Sanctions	Error! Bookmark not defined.
5.23	Budget	23
SECTION 6	6. AFFIRMATIVE COVENANTS:	23
6.1	Financial Statements; Other Information	23
6.2	Notices of Material Events	25
6.3	Existence; Conduct of Business	Error! Bookmark not defined.
6.4	Payment of Obligations	Error! Bookmark not defined.
6.5	Performance of Obligations	Error! Bookmark not defined.
6.6	Operation and Maintenance of Properties	Error! Bookmark not defined.
6.7	Insurance	Error! Bookmark not defined.
6.8	Books and Records; Inspection Rights	Error! Bookmark not defined.
6.9	Compliance with Laws	Error! Bookmark not defined.
6.10	Environmental Matters	Error! Bookmark not defined.
6.11	Further Assurances	25

TABLE OF CONTENTS

(continued)

Page

6.12	Reserved	Error! Bookmark not defined.
6.13	Title Information	Error! Bookmark not defined.
6.14	Additional Collateral; Additional Guarantors	26
6.15	ERISA Compliance	Error! Bookmark not defined.
6.16	Swap Agreements	Error! Bookmark not defined.
6.17	Marketing Activities	Error! Bookmark not defined.
6.18	Use of Proceeds and Budget.	26
6.19	Collateral Matters	27
6.20	Chapter 11 Milestones.	28
6.21	Compliance with Anti-Corruption Laws and Sa	nctions29
6.22	Deposit Accounts	29
SECTION 7.	NEGATIVE COVENANTS:	29
7.1	[Reserved]	29
7.2	Debt	Error! Bookmark not defined.
7.3	Liens	Error! Bookmark not defined.
7.4	Restricted Payments	Error! Bookmark not defined.
7.5	Investments	Error! Bookmark not defined.
7.6	Nature of Business; International Operations	Error! Bookmark not defined.
7.7	Limitation on Leases	Error! Bookmark not defined.
7.8	Use of Proceeds.	Error! Bookmark not defined.
7.9	ERISA Compliance	Error! Bookmark not defined.
7.10	Sale or Discount of Receivables	Error! Bookmark not defined.
7.11	Mergers	Error! Bookmark not defined.
7.12	Asset Dispositions	Error! Bookmark not defined.
7.13	New Deposit Accounts	Error! Bookmark not defined.
7.14	Environmental Matters	Error! Bookmark not defined.
7.15	Transactions with Affiliates	Error! Bookmark not defined.
7.16	Subsidiaries	Error! Bookmark not defined.
7.17	Negative Pledge Agreements; Dividend Restrict defined.	tions Error! Bookmark not

TABLE OF CONTENTS

(continued)

n			
ν	a	Œ	c
	a	~	•

7.1	8 Gas Imbalances, Take-or-Pay or Other Prepayments Erro defined.	r! Bookmark not
7.1	9 Swap Agreements Error! Bookn	nark not defined.
7.2	20 Holding CompanyError! Bookn	nark not defined.
7.2	21 [Reserved]Error! Bookn	nark not defined.
7.2	Prepayment of Prepetition Debt	29
7.2	23 Compliance with the Budget	30
7.2	24 Chapter 11 Modifications	30
7.2	25 Additional Bankruptcy Matters	30
SECTION	8. DEFAULT AND REMEDIES:	32
8.1	Events of Default	32
8.2	Remedies	36
8.3	Rights and Remedies Cumulative; Non-Waiver	36
8.4	Crediting of Payments and Proceeds	36
8.5	Credit Bidding	37
8.6	Injunctive Relief	38
SECTION	9. AGENCY:	38
9.1	Appointment and Authority	38
9.2	Rights as a Lender	38
9.3	Exculpatory Provisions	38
9.4	Reliance by Administrative Agent	41
9.5	Delegation of Duties	42
9.6	Resignation of Administrative Agent	42
9.7	Non-Reliance on Administrative Agent and Other Lenders	43
9.8	Collateral and Guaranty Matters	43
9.9	Erroneous Payment	44
SECTION	10. MISCELLANEOUS:	48
10.	.1 Governing Law	48
10.	.2 Expenses; Indemnity; Damage Waiver	48
10	3 Amendments Waivers and Consents	50

Casse 225-10097/9-JKSS Doorc 5105-11 FFileed 0066/00/2/25 Pragge 17/05 off 12338

TABLE OF CONTENTS

(continued)

		Page
10.4	Successors and Assigns	51
10.5	Treatment of Certain Information; Confidentiality	55
10.6	Submission to Jurisdiction; Venue; Waiver of Jury Trial	56
10.7	Notices	57
10.8	Reinstatement of Obligations	58
10.9	Severability	58
10.10	Integration	58
10.11	No Implied Waivers	58
10.12	USA PATRIOT Act	59
10.13	Counterparts; Effectiveness	59

Casse 225-1009799-JKSS Doorc 5105-11 FFiled 0006 00 92 225 Pragge 1716 off 12338

ANNEXES

Annex A - Rules of Construction

Annex B - DIP Term Loan Commitments and Commitment Percentages

Annex C - Addresses for Notice Annex D - Closing Deliverables

EXHIBITS

Exhibit A - Form of Notice of Borrowing
Exhibit B - Form of Notice of Prepayment

Exhibit C - Form of Responsible Officer's Compliance Certificate

Exhibit D - Form of Assignment and Assumption
Exhibit E - Proposed Interim DIP Financing Order

SCHEDULES

Schedule 5.4 - Ownership and Control
Schedule 6.1 - Financial Reporting
Schedule 7.1 - Negative Covenants

SENIOR SECURED SUPERPRIORITY DEBTOR-IN-POSSESSION TERM LOAN AGREEMENT

This Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement, dated as of June 2, 2025 (this "<u>Agreement</u>"), is among VWS Holdco, Inc. a Delaware corporation (the "<u>Borrower</u>"), Shoosmith Bros., Inc., a Virginia Corporation ("<u>Shoosmith</u>" or the "<u>Guarantor</u>"), the Lenders (as defined below) party hereto from time to time, and Volunteer Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the "<u>Administrative Agent</u>" and, together with the Borrowers, the Guarantor, and the Lenders, the "<u>Parties</u>" and each, a "<u>Party</u>").

PRELIMINARY STATEMENT:

The Borrower, the Lenders party hereto, the Administrative Agent and VWS Acquisitions, L.L.C. (the "<u>Parent</u>") are parties to that certain Amended and Restated Securities Purchase Agreement, dated as of April 30, 2020 (as amended from time to time, the "<u>Prepetition Loan Agreement</u>").

The Borrower and the Guarantor (each a "<u>Debtor</u>" and together, the "<u>Debtors</u>") each commenced a voluntary case (the "<u>Chapter 11 Cases</u>" or the "<u>Bankruptcy Case</u>") on June 1, 2025 (the "<u>Petition Date</u>") under Chapter 11 of the Bankruptcy Code (as hereinafter defined) in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>"), and the Debtors continue to operate their businesses and manage them as debtors-in-possession pursuant to Sections 1107 and 1108 of the Bankruptcy Code.

The Borrower has asked the Lenders to provide it with a senior secured superpriority debtor-in-possession term loan credit facility (the "<u>DIP Facility</u>"), consisting of (a) \$4,900,000 of new money delayed draw term loans that will be made available to the Borrower pursuant to the terms and subject to the conditions set forth in this Agreement and the DIP Financing Orders (the "<u>DIP Term Loans</u>"); and (b) Prepetition Loans in the amount of \$500,000 will be deemed "rolled up" as Loans hereunder on a dollar-for-dollar basis pursuant to the terms and subject to the conditions set forth in this Agreement and the DIP Financing Orders (the "<u>Roll-Up Loans</u>").

The Guarantor has agreed to guarantee the obligations of the Borrower hereunder;

To provide for the repayment of the DIP Facility, the Borrower and the Guarantor have agreed to secure, and have secured, their respective Obligations by granting the Administrative Agent, for its benefit and the benefit of the other Secured Parties, subject to the Carve-Out, certain rights and protections pursuant to the terms hereof, security interests and liens pursuant to Sections 364(c)(2), 364(c)(3) and 364(d) of the Bankruptcy Code and superpriority administrative expense claims pursuant to Section 364(c)(1) of the Bankruptcy Code, in each case having the relative priorities as set forth in the DIP Financing Orders, and other rights and protections as more fully described herein and in the DIP Financing Orders.

The Borrower believes that the loans and other financial accommodations provided to the Borrower under this Agreement will preserve the value of the Borrower's business and assets during the Chapter 11 Case.

Casse 225-1009799-JK\$ Door 505-11 FFiled 0060002225 Prage 1708 off 1288

The Borrower has requested, and subject to the terms and conditions set forth in this Agreement, the Lenders have agreed to extend, the DIP Facility to the Borrower.

AGREEMENT:

In consideration of the foregoing and the mutual agreements contained in this Agreement, the receipt and sufficiency of which are acknowledged, the Parties hereby agree as follows:

SECTION 1. INTERPRETATION:

This Agreement is to be interpreted in accordance with the rules of construction set forth on Annex A. Capitalized terms used in this Agreement and not otherwise defined have the meanings set forth for such terms on Annex A. All annexes, schedules and exhibits to this Agreement are deemed to be a part of this Agreement.

SECTION 2. DIP FACILITY:

2.1 DIP Term Loans.

- (A) Subject to the terms and conditions of this Agreement and in the DIP Financing Orders, and in reliance upon the representations set forth in this Agreement and the DIP Financing Orders, the Lenders shall make the DIP Term Loans available to be drawn by the Borrower from time to time in an aggregate original principal amount equal to the Lenders' total DIP Term Loan Commitment (with each Lender severally making DIP Term Loans in an amount equal to such Lender's DIP Term Loan Commitment). Each Lender's DIP Term Loan Commitment shall be permanently reduced on a dollar for dollar basis by the principal amount of the DIP Term Loan made by such Lender.
- (B) Subject to the terms and conditions of this Agreement and the Interim DIP Financing Order, including the satisfaction of the conditions set forth in Section 4.1, on or after the Closing Date the Lenders shall make DIP Term Loans to the Borrower in one or more additional draws (each, a "First DIP Term Loan Advance") in an aggregate amount up to the First DIP Term Loan Commitment.
- (C) Subject to the terms and conditions of this Agreement and the applicable DIP Financing Order, including this Section 2.1(C) and the satisfaction of the conditions set forth in Section 4.2, the Lenders shall make the DIP Term Loans to the Borrower (after making First DIP Term Loan Advances in the amount of the First DIP Term Loan Commitment), in one or more additional draws (each, a "Second DIP Term Loan Advance") in an aggregate amount up to the Second DIP Term Loan Commitment.
- (D) Subject to the terms and conditions of this Agreement and the applicable DIP Financing Order, including this Section 2.1(D) and the satisfaction of the conditions set forth in Section 4.3, the Lenders shall make the DIP Term Loans to the Borrower (after making First DIP Term Loan Advances in the amount of the First DIP Term Loan Commitment and Second DIP Term Loan Advances in the amount of the

- Second DIP Term Loan Commitment), in one or more additional draws (each, a "<u>Third DIP Term Loan Advance</u>") in an aggregate amount up to the Third DIP Term Loan Commitment.
- (E) The obligations of the Lenders hereunder to make the First DIP Term Loan Advances, Second DIP Term Loan Advances and Third DIP Term Loan Advances are several and not joint. The failure of any Lender to make any DIP Term Loan or, as applicable, to make any such payment on any date required hereunder shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so make its applicable DIP Term Loan (including the First DIP Term Loan Advances, Second DIP Term Loan Advances and Third DIP Term Loan Advances).
- 2.2 Roll-Up Loans.
- (A) Subject to the terms and conditions of the Interim DIP Financing Order, simultaneously with the First DIP Term Loan Commitment being available to Borrower, the Prepetition Loans, including accrued and unpaid interest thereon, shall be deemed Loans advanced under this Agreement, shall constitute Roll-Up Loans and, subject to the terms and conditions set forth herein and without any further action by any party to this Agreement, shall be administered hereunder. Roll-Up Loans under this Section 2.2(A) may be prepaid or repaid but may not be reborrowed.
- (B) [Reserved]
- 2.3 Procedure for Advances.
- (A) First DIP Term Loan Advances.
 - (i) The Borrower shall give the Administrative Agent and the Lenders an irrevocable Notice of Borrowing prior to 11:00 a.m. at least two Business Days prior to any proposed date of borrowing (or such shorter period as may be acceptable to the Lenders in their sole discretion), specifying (i) the amount of funds to be disbursed, (ii) wire instructions of the Borrower's account(s) to which funds are to be disbursed, (iii) the requested date of the borrowing, which shall be a Business Day, and (iv) the Type of DIP Term Loans to be borrowed. A Notice of Borrowing received after 11:00 a.m. is deemed received on the next Business Day. Not later than 1:00 p.m. on the date the Borrower requests a First DIP Term Loan Advance pursuant to a Notice of Borrowing provided by Borrower in accordance with this paragraph (A), and after the acceptance by the Lenders of the latest approved Budget as set forth in Section 6.14(B) and the Borrower having met all conditions as set forth in Section 4.1, each Lender will make available to the Administrative Agent, for the account of the Borrower, at the office of the Administrative Agent in funds immediately available to the Administrative Agent, such Lender's Commitment Percentage of the First

DIP Term Loan Advance requested by the Borrower to be made on such date. The Borrower hereby irrevocably authorizes the Administrative Agent to disburse the proceeds of any First DIP Term Loan Advance received by the Administrative Agent in immediately available funds by wire transfer to such Persons as may be designated by the Borrower in writing. The Administrative Agent shall only be required to fund after the Bankruptcy Court approves the Interim DIP Financing Order in the Chapter 11 Cases.

(B) Second DIP Term Loan Advance.

(i) The Borrower shall give the Administrative Agent and the Lenders an irrevocable Notice of Borrowing prior to 11:00 a.m. at least two Business Days prior to the date of any Second DIP Term Loan Advance, specifying (i) the amount of such advance, (ii) wire instructions of the Borrower's account(s) to which funds are to be disbursed, (ii) the requested date of the borrowing, which shall be a Business Day, (iv) the Type of DIP Term Loans to be borrowed, and (v) the Final DIP Financing Order approving such draw has been (or will be) entered and all of the conditions to such draw set forth in Section 4.2 have been (or will be) satisfied on or prior to the date of such draw. A Notice of Borrowing received after 11:00 a.m. is deemed received on the next Business Day. Not later than 1:00 p.m. on the date the Borrower requests a Second DIP Term Loan Advance pursuant to a Notice of Borrowing provided pursuant to a Notice of Borrowing provided by Borrower in accordance with this paragraph (B), and after the acceptance by the Lenders of the latest approved Budget as set forth in Section 6.18(B) and the Borrower having met all conditions as set forth in Sections 4.1 and 4.2, each Lender will make available to the Administrative Agent, for the account of the Borrower, at the office of the Administrative Agent in funds immediately available to the Administrative Agent, such Lender's Commitment Percentage of the Second DIP Term Loan Advance to be made The Borrower hereby irrevocably authorizes the on such date. Administrative Agent to disburse the proceeds of the Second DIP Term Loan Advance received by the Administrative Agent in immediately available funds by wire transfer to such Persons as may be designated by the Borrower in writing.

(C) Third DIP Term Loan Advance.

(i) The Borrower shall give the Administrative Agent and the Lenders an irrevocable Notice of Borrowing prior to 11:00 a.m. at least two Business Days prior to the date of any Third DIP Term Loan Advance, specifying (i) the amount of such advance, (ii) wire instructions of the Borrower's account(s) to which funds are to be disbursed, (ii) the requested date of the borrowing, which shall be a Business Day, (iv) the Type of DIP Term Loans to be borrowed, and (v) the Final DIP Financing Order approving such draw has been (or will be) entered and all of the conditions to such draw set forth

in Section 4.3 have been (or will be) satisfied on or prior to the date of such draw. A Notice of Borrowing received after 11:00 a.m. is deemed received on the next Business Day. Not later than 1:00 p.m. on the date the Borrower requests a Third DIP Term Loan Advance pursuant to a Notice of Borrowing provided by Borrower in accordance with this paragraph (C), and after the acceptance by the Lenders of the latest approved Budget as set forth in Section 6.18(B) and the Borrower having met all conditions as set forth in Sections 4.1, 4.2 and 4.3 each Lender will make available to the Administrative Agent, for the account of the Borrower, at the office of the Administrative Agent in funds immediately available to the Administrative Agent, such Lender's Commitment Percentage of the Third DIP Term Loan Advance to be made on such date. The Borrower hereby irrevocably authorizes the Administrative Agent to disburse the proceeds of the Third DIP Term Loan Advance received by the Administrative Agent in immediately available funds by wire transfer to such Persons as may be designated by the Borrower in writing.

- (D) The Administrative Agent is not obligated to disburse the portion of the proceeds of any DIP Term Loans requested pursuant to this <u>Section 2</u> to the extent that any Lender has not made available to the Administrative Agent its Commitment Percentage of such DIP Term Loans. If a Lender does not make available to the Administrative Agent its Commitment Percentage of any requested DIP Term Loan, then the Administrative Agent shall return the funds it received from the other Lenders with respect to such requested DIP Term Loan.
- 2.4 <u>Repayment</u>. The Borrower shall repay the aggregate outstanding principal amount of the DIP Term Loans and Roll-Up Loans in full, together with accrued interest thereon, on the Maturity Date.
 - 2.5 Optional Prepayments.
 - (A) <u>Prepayments</u>.
 - The Borrower may from time to time, without premium or penalty, prepay (i) the DIP Facility, in whole or in part, upon delivery to the Administrative Agent and the Lenders of a Notice of Prepayment not later than 11:00 a.m. at least three (3) Business Days before its intention to prepay specifying the date, which must be a Business Day, and amount of prepayment. Each optional prepayment of the DIP Facility hereunder must be in an aggregate principal amount of at least \$500,000 or any whole multiple of \$100,000 in excess thereof. A Notice of Prepayment received after 11:00 a.m. is deemed received on the next Business Day. The Administrative Agent shall promptly notify the Lenders of each Notice of Prepayment; provided, however, that the delivery by the Borrower of such Notice of Prepayment to each Lender directly shall satisfy this requirement. Any prepayment of the DIP Facility shall be accompanied by all accrued interest thereon. Notwithstanding the foregoing, any Notice of Prepayment delivered in

- connection with any refinancing of all of the DIP Facility with the proceeds of such refinancing may be, if expressly so stated to be, contingent upon the consummation of such refinancing or incurrence and may be revoked by the Borrower in the event such refinancing or incurrence is not consummated.
- shall be applied first to all costs, expenses, indemnities and other amounts due and payable hereunder, then proportionately (based on the relation of such amounts to the total amount of the relevant payment under this Section 2.5 to the payment or prepayment (as applicable) of the following amounts of the Obligations: default interest, if any, accrued interest and principal. All payments hereunder will be applied on a pro rata basis to the thenoutstanding DIP Facility.

(B) <u>Commitment Termination</u>.

- (i) The DIP Term Loan Commitments of each Lender to fund the DIP Term Loans pursuant to Section 2.1(A), (B) and (C) shall expire upon the earlier of the date (the "DIP Term Loan Commitment Termination Date") that (a) the DIP Term Loans are fully funded or reduced to zero and (b) the Maturity Date.
- (ii) The Borrower may at any time terminate or reduce (including to zero) any undrawn amount of the DIP Term Loan Commitments upon prior notice to the Administrative Agent and the Lenders.
- (iii) The Borrower shall notify the Administrative Agent and the Lenders of any election to reduce (including to zero) the DIP Term Loan Commitments under paragraph (ii) of this Section at least three (3) Business Days prior to the effective date of such reduction, specifying such election and the effective date thereof.

2.6 <u>Mandatory Prepayments</u>.

- (A) <u>Asset Dispositions and Involuntary Asset Dispositions.</u>
 - (i) Subject to (x) the priority of Liens and application of funds set forth in the DIP Financing Orders, and (y) the use of such Net Cash Proceeds as set forth in the most-recent Budget, the Borrower shall promptly (and, in any event, within three (3) Business Days) prepay the DIP Facility in an aggregate amount equal to 100% of the Net Cash Proceeds of any Asset Disposition or Involuntary Asset Disposition received by the Borrower or any other Debtor. Any prepayment pursuant to this Section 2.6(A)(i) shall be applied as set forth in Section 2.6(D) below.
- (B) Extraordinary Receipts.

(i) Subject to (x) the priority of Liens and application of funds set forth in the DIP Financing Orders, and (y) the use of such Net Cash Proceeds as set forth in the most-recent Budget, the Borrower shall promptly (and, in any event, within three (3) Business Days) upon the receipt by the Borrower or any other Debtor of the Net Cash Proceeds of any Extraordinary Receipt, prepay the DIP Facility in an aggregate amount equal to 100% of such Net Cash Proceeds. Any prepayment pursuant to this this Section 2.6(B)(i) shall be applied as set forth in Section 2.6(D) below.

(C) <u>Debt Issuances</u>.

- (i) Subject to the priority of Liens and application of funds set forth in the DIP Financing Orders, the Borrower shall promptly (and, in any event, within three (3) Business Days) upon the receipt by the Borrower or any other Debtor of the Net Cash Proceeds of any Debt Issuance, prepay the DIP Facility in an aggregate amount equal to 100% of such Net Cash Proceeds. Any prepayment pursuant to this Section 2.6(C)(i) shall be applied as set forth in Section 2.6(D) below.
- (D) Application of Mandatory Prepayments. All payments under this Section 2.6 shall be applied first to all costs, expenses, indemnities and other amounts due and payable hereunder, then proportionately (based on the relation of such amounts to the total amount of the relevant payment under this Section 2.6 to the payment or prepayment (as applicable) of the following amounts of the Obligations: default interest, if any, accrued interest and principal. All payments hereunder will be applied on a pro rata basis to the then-outstanding DIP Facility.
- (E) <u>Notice of Prepayment</u>. Upon the occurrence of any event triggering the prepayment requirement under this <u>Section 2.6</u>, the Borrower shall promptly deliver a Notice of Prepayment to the Administrative Agent and the Lenders.
- (F) <u>Accrued Interest</u>. Any prepayment of the DIP Facility pursuant to this <u>Section 2.6</u> shall be accompanied by all accrued interest thereon.
- 2.7 <u>No Reborrowings</u>. The Borrower may not reborrow any amounts repaid or prepaid under the DIP Term Loans or Roll-Up Loans.

SECTION 3. GENERAL LOAN PROVISIONS:

- 3.1 Repayment of Principal and Outstanding Obligations.
- (A) The Borrower hereby unconditionally promises to pay the Administrative Agent, for the benefit of the Lenders, all amounts owed hereunder with respect to the DIP Term Loans and the Roll-Up Loans (including, for the avoidance of doubt, any PIK Amounts) on the Maturity Date. Each repayment of a DIP Term Loan and/or Roll-Up Loan shall be applied ratably to the Loans. Repayments of Loans shall be accompanied by accrued interest on the amounts repaid. Any reference in this

Agreement or the Loan Documents to the Loans or the outstanding principal balance of the Loans shall include the PIK Amounts.

3.2 Interest.

- (A) <u>Interest Rate</u>. Each Loan (including the PIK Amounts) shall bear interest at a rate *per annum* equal to the Interest Rate.
- (B) <u>Default Rate</u>. Subject to <u>Section 8.3</u>, immediately upon the occurrence and during the continuance of an Event of Default, the Borrower shall pay interest on the aggregate outstanding principal amount of the DIP Term Loans and Roll-Up Loans at the Default Rate, such interest being due and payable on demand.
- (C) <u>Interest Payment and Computation</u>. The Borrower shall pay interest (at the Interest Rate or Default Rate, as applicable) on the DIP Facility in arrears on each Interest Payment Date. The aggregate outstanding principal amount of the DIP Facility shall be automatically increased on each such Interest Payment Date by the amount of such interest paid in kind (the "<u>PIK Amounts</u>"). For the avoidance of doubt interest shall accrue on the principal amount of the DIP Facility increased as a result of payments of interest in kind. All computations of interest are made on the basis of a 360-day year and actual days elapsed.
- (D) Maximum Rate. In no contingency or event whatsoever will the aggregate of all amounts deemed interest under this Agreement charged or collected pursuant to the terms of this Agreement exceed the highest rate permissible under any Applicable Law that a court of competent jurisdiction, in a final determination, deems applicable hereto. In the event that such a court determines that the Lenders have charged or received interest hereunder in excess of the highest applicable rate, the rate in effect hereunder automatically is reduced to the maximum rate permitted by Applicable Law and the Lenders shall apply such excess to the principal balance of the Obligations. It is the intent hereof that the Borrower not pay or contract to pay, and that neither the Administrative Agent nor any Lender receive or contract to receive, directly or indirectly in any manner whatsoever, interest in excess of that which may be paid by the Borrower under Applicable Law.

3.3 Reserved.

3.4 Manner of Payment. The Borrower shall make each payment on account of the principal of or interest on the DIP Facility or of any fee, commission or other amounts payable to the Lenders under this Agreement not later than 1:00 p.m. on the date specified for payment under this Agreement to the Administrative Agent at the Administrative Agent's Office for the account of the Lenders entitled to such payment in Dollars, in immediately available funds and without any set off, counterclaim or deduction whatsoever. Any payment received after 1:00 p.m. is deemed to have been made on the next succeeding Business Day for all purposes. Upon receipt by the Administrative Agent of each such payment, the Administrative Agent shall distribute to each Lender its Commitment Percentage (or other applicable share as provided herein) of such payment and shall provide notice of the amount of such credit to each Lender. The Borrower shall

pay each payment to the Administrative Agent of Administrative Agent's fees, expenses or other amounts payable to the Administrative Agent for the account of the Administrative Agent and shall pay any amount payable to any Lender under Sections 3.7, 3.8 or 10.2 to the Administrative Agent for the account of the applicable Lender. If any payment under this Agreement is specified to be made upon a day that is not a Business Day, it shall be made on the next succeeding day that is a Business Day and such extension of time is included in computing any interest if payable along with such payment. Notwithstanding the foregoing, each payment by the Borrower to a Defaulting Lender is applied in accordance with Section 3.10(A)(ii).

- 3.5 Evidence of Indebtedness. The DIP Term Loans made by each Lender will be evidenced by one or more accounts or records maintained by such Lender in the ordinary course of business and by the Administrative Agent in accordance with this Agreement. The accounts or records maintained by the Administrative Agent and each Lender are conclusive absent manifest error of the amount of the DIP Term Loans made by the Lenders to the Borrower and the interest and payments thereon. Any failure to so record or any error in doing so do not, however, limit or otherwise affect the obligation of the Borrower hereunder to pay any amount owing with respect to the Obligations. In the event of any conflict between the accounts and records maintained by any Lender and the accounts and records of the Administrative Agent in respect of such matters, the accounts and records of the Administrative Agent control in the absence of manifest error.
- 3.6 Sharing of Payments by Lenders. Subject to the DIP Financing Orders, if any Lender, by exercising any right of setoff or counterclaim or otherwise, obtains payment in respect of any principal of or interest on any of its DIP Term Loans or other obligations hereunder resulting in such Lender receiving payment of a proportion of the aggregate amount of its DIP Term Loans and accrued interest thereon or other such obligations greater than its pro rata share thereof as provided herein, then the Lender receiving such greater proportion shall (a) notify the Administrative Agent of such fact and (b) purchase (for cash at face value) participations in the DIP Term Loans and such other obligations of the other Lenders, or make such other adjustments as are equitable, so that the benefit of all such payments are shared by the Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective DIP Term Loans and other amounts owing them; provided:
 - (A) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations are rescinded and the purchase price restored to the extent of such recovery, without interest; and
 - (B) the provisions of this Section 3.6 may not be construed to apply to (i) any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement (including the application of funds arising from the existence of a Defaulting Lender) or (ii) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its DIP Term Loans to any assignee or participant.

Each Loan Party consents to the foregoing and agrees, to the extent it may effectively do so under Applicable Law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against each Loan Party rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of each Loan Party in the amount of such participation.

3.7 Increased Costs.

(A) In the event that any Lender shall have determined (which determination shall, absent manifest error, be final and conclusive and binding upon all parties hereto) any Change in Law shall subject any Recipient to any Taxes (other than (A) Indemnified Taxes, (B) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes and (C) Connection Income Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto;

then, and in any such event, such Lender shall promptly give written notice to the Borrower and to the Administrative Agent of such determination (which notice the Administrative Agent shall promptly transmit to each of the other Lenders). The Borrower agrees to pay to such Lender, upon such Lender's written request therefor, such additional amounts (in the form of an increased rate of, or a different method of calculating, interest or otherwise as such Lender shall determine after consultation with the Borrower) as shall be required to compensate such Lender for such increased costs or reductions in amounts received or receivable hereunder (a written notice as to the additional amounts owed to such Lender, showing in reasonable detail the basis for the calculation thereof, submitted to the Borrower by such Lender shall, absent manifest error, be final and conclusive and binding on all the parties hereto).

(B) If any Lender determines that after the date of this Agreement the introduction of or any change in any Applicable Law or governmental rule, regulation, order, guideline, directive or request (whether or not having the force of law) concerning capital adequacy, or any change in interpretation or administration thereof by the NAIC or any Governmental Authority, central bank or comparable agency, will have the effect of increasing the amount of capital required or expected to be maintained by such Lender or any corporation controlling such Lender based on the existence of such Lender's DIP Term Loan Commitments hereunder or its obligations hereunder, then the Borrower agrees to pay to such Lender, upon its written demand therefor, such additional amounts as shall be required to compensate such Lender or such other corporation for the increased cost to such Lender or such other corporation or the reduction in the rate of return to such Lender or such other corporation as a result of such increase of capital. In determining such additional amounts, each Lender will act reasonably and in good faith and will use averaging and attribution methods which are reasonable; provided that such Lender's determination of compensation owing under this Section 3.7(B) shall, absent manifest error, be final and conclusive and binding on all the parties hereto. Each Lender, upon determining that any additional amounts, will be payable pursuant to this Section 3.7(B), will give prompt written notice thereof to the Borrower, which notice shall show in reasonable detail the basis for calculation of such additional amounts, although the failure to give any such notice shall not release or diminish a Borrower's obligations to pay additional amounts pursuant to this <u>Section 3.7(B)</u> upon the subsequent receipt of such notice.

- (C) Notwithstanding anything in this Agreement to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines, requirements and directives thereunder, issued in connection therewith or in implementation thereof and (y) all requests, rules, guidelines or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States regulatory authorities, in each case pursuant to Basel III, shall be deemed to be a change after the date of this Agreement in a requirement of law or government rule, regulation or order, regardless of the date enacted, adopted, issued or implemented (including for purposes of this Section 3.6).
- 3.8 Taxes.
- (A) <u>Defined Terms</u>. For purposes of this <u>Section 3.8</u>, the term "applicable law" includes FATCA.
- (B) Payments Free of Taxes. Any and all payments by or on account of any obligation of any Loan Party under any Loan Document shall be made without deduction or withholding for any Taxes, except as required by applicable law. If any applicable law (as determined in the good faith discretion of an applicable Withholding Agent) requires the deduction or withholding of any Tax from any such payment by a Withholding Agent, then the applicable Withholding Agent is entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with applicable law and, if such Tax is an Indemnified Tax, then the sum payable by the applicable Loan Party is increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section 3.8) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made.
- (C) <u>Payment of Other Taxes by the Borrower</u>. The Loan Parties shall timely pay to the relevant Governmental Authority in accordance with applicable law, or at the option of the Administrative Agent timely reimburse it for the payment of, any Other Taxes.
- (D) Indemnification by the Borrower. The Loan Parties shall jointly and severally indemnify each Recipient, within 10 days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by such Recipient or required to be withheld or deducted from a payment to such Recipient and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority, and shall indemnify the Administrative Agent, as Withholding Agent, in connection with any tax withholdings. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, is conclusive absent manifest error.

- (E) Indemnification by the Lenders. Each Lender shall severally indemnify the Administrative Agent, within 10 days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that any Loan Party has not already indemnified the Administrative Agent for such Indemnified Taxes and without limiting the obligation of the Loan Parties to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of Section 10.4(D) relating to the maintenance of a Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by the Administrative Agent in connection with any Loan Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent is conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under any Loan Document or otherwise payable by the Administrative Agent to the Lender from any other source against any amount due to the Administrative Agent under this Section 3.8(E).
- (F) Evidence of Payments. As soon as practicable after any payment of Taxes by any Loan Party to a Governmental Authority pursuant to this Section 3.8, such Loan Party shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Required Lenders.
- (G) Status of Lenders. (i) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to the Borrower and the Administrative Agent, at the time or times reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation reasonably requested by the Borrower or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by the Borrower or the Administrative Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Section 3.8(G)(ii) is not required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.
- (ii) Any Lender that is a U.S. Person shall deliver to the Borrower and the Administrative Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the

Administrative Agent), executed copies of IRS Form W-9 certifying that such Lender is exempt from U.S. federal backup withholding tax. Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify the Borrower and the Administrative Agent in writing of its legal inability to do so.

Volunteer Enterprises, LLC, both in its individual capacity and in its capacity as the Administrative Agent, has no liability to the Borrower, the Lenders or any other Person in connection with any tax withholding amounts paid or withheld from any payment pursuant to Applicable Law or arising from the Borrower's or a Lender's failure, as applicable, to timely provide an accurate, correct and complete IRS Form W-9, an appropriate IRS Form W-8 or such other documentation contemplated under this Agreement.

- (H) Treatment of Certain Refunds. If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this Section 3.8 (including by the payment of additional amounts pursuant to this Section 3.8), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this Section 3.8 with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnified party and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this Section 3.8(H) (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) in the event that such indemnified party is required to repay such refund to such Governmental Authority. Notwithstanding anything to the contrary in this Section 3.8(H), in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this Section 3.8(H) the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. This Section 3.8(H) may not be construed to require any indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party or any other Person.
- (I) <u>Survival</u>. Each Party's obligations under this <u>Section 3.8</u> survives the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the DIP Term Loan Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.
- (J) <u>Tax Reporting</u>. The Lenders and any transferees or assignees after the Closing Date will be required to provide to Administrative Agent or its agents all information, documentation or certifications reasonably requested by Administrative Agent to

- permit Administrative Agent to comply with its tax reporting obligations under Applicable Laws, including any applicable cost basis reporting obligations.
- (K) Not less than four (4) Business Days (or such shorter period as may be agreed to by Administrative Agent) prior to any payment, distribution or transfer of funds by Administrative Agent to any Person under the Loan Documents, the payee shall provide to Administrative Agent such documentation and information as may be requested by Administrative Agent (unless such Person has previously provided the documentation or information, and so long as such documentation or information remain accurate and true). Administrative Agent shall have no duty, obligation or liability to make any payment to any Person unless it has timely received such documentation and information with respect to such Person, which documentation and information shall be reasonably satisfactory to Administrative Agent.
- 3.9 <u>Mitigation Obligations; Replacement of Lenders.</u>
- (A) <u>Designation of a Different Lending Office</u>. If any Lender requests compensation under <u>Section 3.7</u>, or requires the Borrower to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.8</u>, then such Lender shall (at the request of the Borrower) use reasonable efforts to designate a different lending office for funding or booking its DIP Term Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to <u>Section 3.7</u> or <u>3.8</u>, as the case may be, in the future and (ii) would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender. The Borrower shall pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment.
- (B) Replacement of Lenders. If any Lender requests compensation under Section 3.7, or if the Borrower is required to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 3.8 and, in each case, such Lender has declined or is unable to designate a different lending office in accordance with Section 3.9(A), or if any Lender is a Defaulting Lender or a Non-Consenting Lender, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 10.4), all of its interests, rights (other than its existing rights to payments pursuant to Section 3.7 or 3.8) and obligations under this Agreement and the related Loan Documents to an Eligible Assignee that assumes such obligations (which assignee may be another Lender, if a Lender accepts such assignment); provided:
- (i) [reserved];

- (ii) such Lender has received payment of an amount equal to the outstanding principal of its DIP Term Loans or Roll-Up Loans, accrued interest thereon and all other amounts payable to it hereunder and under the other Loan Documents from the assignee (to the extent of such outstanding principal and accrued interest) or the Borrower (in the case of all other amounts);
- (iii) in the case of any such assignment resulting from a claim for compensation under <u>Section 3.7</u> or payments required to be made pursuant to <u>Section 3.8</u>, such assignment will result in a reduction in such compensation or payments thereafter;
- (iv) such assignment does not conflict with Applicable Law; and
- (v) in the case of any assignment resulting from a Lender becoming a Non-Consenting Lender, the applicable assignee has consented to the applicable amendment, waiver or consent.

A Lender is not required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

3.10 <u>Defaulting Lenders</u>.

- (A) <u>Defaulting Lender Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as such Lender is no longer a Defaulting Lender, to the extent permitted by Applicable Law:
- (i) Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement is restricted as set forth in the definition of Required Lenders.
- Any payment of principal, interest, fees or other amounts received by the (ii) Administrative Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Section 8 or otherwise) shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; second, as the Borrower may request (so long as no Default or Event of Default exists), to the funding of any DIP Term Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement; third, if so determined by the Administrative Agent and the Borrower, to be held in a deposit account and released pro rata in order to satisfy such Defaulting Lender's potential future funding obligations with respect to its DIP Term Loans under this Agreement; fourth, to the payment of any amounts owing to the Lenders as a result of any judgment of a court of competent jurisdiction obtained by any Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; fifth, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction

obtained by the Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and sixth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (a) such payment is a payment of the principal amount of any DIP Term Loans in respect of which such Defaulting Lender has not fully funded its appropriate share and (b) such DIP Term Loans were made at a time when the conditions set forth in Section 4.1 or 4.2 were satisfied or waived, such payment is applied solely to pay the DIP Term Loans of all Non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any DIP Term Loans of such Defaulting Lender until such time as all DIP Term Loans are held by the Lenders pro rata in accordance with the DIP Term Loan Commitments. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender pursuant to this Section 3.9(A)(ii) are deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.

(B) <u>Defaulting Lender Cure</u>. If the Borrower and the Required Lenders agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent shall so notify the Parties, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein, that Lender shall, to the extent applicable, purchase at par that portion of outstanding DIP Term Loans of the other Lenders or take such other actions as may be necessary to cause the DIP Term Loans to be held pro rata by the Lenders in accordance with the DIP Term Loan Commitments, whereupon such Lender will cease to be a Defaulting Lender. Except to the extent otherwise expressly agreed by the affected Parties, no change hereunder from Defaulting Lender to Lender constitutes a waiver or release of any claim of any Party arising from that Lender's having been a Defaulting Lender.

3.11 No Discharge; Survival.

Unless in accordance with an Acceptable Plan, the Borrower agrees that prior to payment in full in cash of the Obligations and termination of the DIP Term Loan Commitments in accordance herewith, (a) its obligations under the Loan Documents shall not be discharged by the entry of an order confirming a Plan of Reorganization and (b) the superpriority claims granted to the Administrative Agent and the Lenders pursuant to the DIP Financing Orders and the Liens granted to the Administrative Agent and the Lenders pursuant to the DIP Financing Orders shall not be affected in any manner by the entry of an order confirming a Plan of Reorganization, except to the extent set forth in an Acceptable Plan confirmed pursuant to a final, non-appealable order from the Bankruptcy Court that is consummated.

SECTION 4. CONDITIONS OF CLOSING AND BORROWING:

4.1 <u>Conditions to Closing and Funding of any First DIP Term Loan Advance</u>. The obligation of the Lenders to close this Agreement and to make any First DIP Term Loan Advance is subject to the satisfaction of each of the following conditions:

- (A) Executed Loan Documents. Each agreement, instrument or other document set forth on Annex D has been duly authorized, executed and delivered to the Administrative Agent on behalf of the Lenders and to the Lenders by the parties thereto and is in full force and effect.
- (B) <u>Interim DIP Financing Order</u>. Subject to the availability of the Bankruptcy Court shall have entered the Interim DIP Financing Order, in a form acceptable to the Lenders (in their sole and exclusive discretion) and the Administrative Agent (in its reasonable discretion), no later than five (5) Business Days after the Petition Date, which Interim DIP Financing Order shall be in full force and effect and shall not have been reversed, vacated or stayed, and shall not have been amended, supplemented or otherwise modified without the prior written consent of the Lenders; and the Loan Parties shall be in compliance with the Interim DIP Financing Order.
- (C) <u>Governing Documents</u>. The Lenders shall have received a certificate executed by a Responsible Officer of the Borrower certifying:
 - (i) the names and signatures of the officers of each Loan Party authorized to execute the Loan Documents;
 - (ii) the resolutions duly adopted by the Board of Directors (or equivalent governing body) of each Loan Party approving the Loan Documents;
 - (iii) the correctness and completeness of the copy of the Operating Agreements (or equivalent governing document) of each Loan Party attached thereto; and
 - (iv) the correctness and completeness of the copy of the certificate of incorporation (or equivalent governing document) of each Loan Party attached thereto.
- (D) <u>Lender Approval</u>. The Lenders shall have received approval of the transaction contemplated hereby.
- (E) <u>Initial Approved Budget.</u> No later than two (2) days after the Petition Date, the Lenders shall have received, and approved, a cash forecast for the period from the Petition Date through the following thirteen (13) weeks setting forth projected cash flows and disbursements (the "<u>Initial Approved Budget</u>").
- (F) <u>First Day Motions</u>.
 - (i) The Borrower shall have provided the Lenders with a copy of the "first day" motions, including the cash management motion, and proposed orders to be filed with the Bankruptcy Court in connection with the commencement of the Chapter 11 Cases, in form and substance satisfactory to the Required Lenders.

- (ii) The Bankruptcy Court shall have entered orders approving all "first day" motions, either on an interim or final basis as the case may be, other than the Interim DIP Financing Order or as may be waived by the Lenders.
- (G) Restructuring Support Agreement or Plan of Reorganization. Other than as set forth herein, the Borrower shall not have executed, entered into or otherwise committed to any plan or restructuring support agreement or any other agreement or understanding concerning the terms of a Chapter 11 plan or other exit strategy without the consent of the Lenders.
- (H) <u>No Trustee, Examiner, or Receiver</u>. No order shall have been entered appointing a trustee, examiner or receiver (or local law equivalent) with respect to the Borrower or its Subsidiaries' business, properties or assets, and no motion shall be pending seeking any such relief.
- (I) Administrative Agent and Lender Expenses. The Borrower shall have paid to the Administrative Agent and the Lenders, the fees, costs and expenses then earned, due and payable under the Loan Documents (including, without limitation, the fees, costs and expenses of counsel and the other advisors to the Administrative Agent and the Lenders incurred in connection with the Chapter 11 Cases and the negotiation, preparation, execution and delivery of the Loan Documents).
- (J) Payment at Closing. The Borrower has paid or made arrangements to pay contemporaneously with closing (i) all reasonable fees, charges and disbursements of counsel to the Administrative Agent and the Lenders (directly to such counsel if requested by the Administrative Agent) to the extent accrued and unpaid and submitted to the Borrower prior to or on the Closing Date, plus such additional amounts of such fees, charges and disbursements as constitute its reasonable estimate of such fees, charges and disbursements incurred or to be incurred by it through the closing proceedings (provided that such estimate does not thereafter preclude a final settling of accounts between the Borrower and the Administrative Agent and the Lenders) and (ii) to any other Person such amount as may be due thereto in connection with the transactions contemplated hereby, including all taxes, fees and other charges in connection with the execution, delivery, recording, filing and registration of any of the Loan Documents.
- (K) The Administrative Agent and the Lenders shall have received a Notice of Borrowing from the Borrower in accordance with <u>Section 2.3</u>.
- (L) The representations contained in this Agreement and the other Loan Documents are true and correct in all material respects, on and as of the Closing Date with the same effect as if made on and as of such date (except for any such representation that by its terms is made only as of an earlier date, which representation remains true and correct in all material respects as of such earlier date).
- (M) No Default or Event of Default exists on the Closing Date with respect to such DIP Term Loans or after giving effect to the DIP Term Loans to be made on such date.

- (N) <u>Material Adverse Effect</u>. Since the Petition Date, other than those customarily resulting from the commencement of the Chapter 11 Cases:
 - (i) no event, development or circumstance has occurred or exists that has resulted in, or could reasonably be expected to result in, a Material Adverse Effect; and
 - (ii) there has not been any fact, event, change, effect, development, circumstance or occurrence that, individually or together with any other fact, event, change, effect, development, circumstance or occurrence, has had or would reasonably be expected to have a material and adverse effect on the ability of the Loan Parties taken as a whole to perform their respective obligations under, or to consummate the transactions contemplated by, the Loan Documents.

Notwithstanding anything to the contrary contained herein, funding of any First DIP Term Loan Advance shall be subject to entry of the Interim DIP Financing Order, which shall not have been vacated, reversed or stayed, appealed (and for which the appeal period has expired or has been waived), or modified or amended without the prior written consent of the Lenders in their sole discretion.

- 4.2 <u>Conditions to Funding any Second DIP Term Loan Advance</u>. The obligation of the Lenders to make any Second DIP Term Loan Advance is subject to the satisfaction of each of the following conditions, in addition to all of the conditions set forth in <u>Section 4.1</u>:
 - (A) <u>Final DIP Financing Order</u>. The Final DIP Financing Order shall have been entered by the Bankruptcy Court, in a form acceptable to the Lenders, in their sole and exclusive discretion, shall be in full force and effect, shall not have been vacated or reversed, shall not be subject to a stay and shall not have been modified or amended other than as acceptable to the Required Lenders in their sole and exclusive discretion; and the Loan Parties shall be in compliance with the Final DIP Financing Order.
 - (B) <u>Second Day Orders</u>. All "second day orders" approving on a final basis any first day orders intended to be entered on or prior to the date of entry of the Final DIP Financing Order shall have been entered by the Bankruptcy Court, shall be in full force and effect, shall not have been vacated or reversed, shall not be subject to a stay, and shall be in form and substance satisfactory to the Required Lenders.
 - (C) The Debtors shall have (a) entered into and delivered a fully executed and binding Acceptable Stalking Horse Purchase Agreement and (b) satisfied any other applicable Chapter 11 Milestones;
 - (D) The Bankruptcy Court shall have approved the Bidding Procedures Motion in a form satisfactory to the Lenders and scheduled the auction.
 - (E) <u>Compliance with Orders</u>. The Loan Parties shall be in compliance in all material respects with each First Day Order and Second Day Order then in effect.

- (F) <u>Approved Budget</u>. The Lenders shall have received, and approved, the Budget in form and substance acceptable to the Lenders in their sole discretion, at least two (2) Business Days before funding the Second DIP Term Loan Advance.
- (G) The representations contained in this Agreement and the other Loan Documents are true and correct in all material respects, on and as of the Closing Date with the same effect as if made on and as of such date (except for any such representation that by its terms is made only as of an earlier date, which representation remains true and correct in all material respects as of such earlier date).
- (H) <u>Material Adverse Effect</u>. No event shall have occurred, or circumstance exists, that has or could reasonably be expected to have a Material Adverse Effect.
- (I) <u>Legal Restriction</u>. Such advance or financial accommodation shall not be prohibited by any law or regulation or any order of any court or Governmental Authority or agency.
- (J) <u>No Repudiation</u>. No Loan Party shall have repudiated or made any anticipatory breach or repudiation of any of its obligations under any Loan Document.
- 4.3 <u>Conditions to Funding any Third Term Loan Advance</u>. The obligation of the Lenders to make any Third DIP Term Loan Advance shall become effective from and after July ___, 2025, and shall be subject to the satisfaction of each of the conditions set forth in <u>Sections 4.1</u> and 4.2.

SECTION 5. REPRESENTATIONS:

To induce the Administrative Agent and Lenders to enter into this Agreement and to induce the Lenders to make the DIP Term Loans, the Loan Parties represent to the Administrative Agent and the Lenders both before and after giving effect to the transactions contemplated hereunder, which representations and warranties are deemed made on the Closing Date, that:

- Organization; Powers. Subject to any restrictions arising on account of the Loan Parties' status as "debtors" under the Bankruptcy Code, the Loan Parties (a) are duly organized, incorporated or formed, validly existing and (where applicable) in good standing under the Laws of the jurisdiction of its incorporation or organization, (b) have all requisite power and authority and all requisite governmental licenses, authorizations, consents and approvals to (i) own or lease its assets and carry on their businesses and (ii) execute, deliver and perform their obligations under the Loan Documents to which they are a party, and (c) are duly qualified and are licensed and (where applicable) in good standing under the Laws of each jurisdiction where their ownership, lease or operation of properties or the conduct of their businesses require such qualification or license; except in each case referred to in clause (b)(i) or (c), to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect.
- 5.2 <u>Authority; Enforceability</u>. Subject to the entry of the DIP Financing Orders, the Transactions are within the Borrower's and Guarantor's corporate powers and have been duly authorized by all necessary corporate and, if required, stockholder action (including any action required to be taken by any class of directors of the Borrower or any other Person, whether

interested or disinterested, in order to ensure the due authorization of the Transactions). Subject to the entry of the DIP Financing Orders, each Loan Document to which the Borrower and each Guarantor is a party has been duly executed and delivered by the Borrower and such Guarantor and constitutes a legal, valid and binding obligation of the Borrower and such Guarantor, as applicable, enforceable in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally and subject to general principles of equity, regardless of whether considered in a proceeding in equity or at law.

- Approvals; No Conflicts. Other than as otherwise previously disclosed to the Lenders, subject to the entry of the DIP Financing Orders Transactions (a) do not require any consent or approval of, registration or filing with, or any other action by, any Governmental Authority or any other third Person (including shareholders or any class of directors, whether interested or disinterested, of the Borrower or any other Person), nor is any such consent, approval, registration, filing or other action necessary for the validity or enforceability of any Loan Document or the consummation of the transactions contemplated thereby, except such as have been obtained or made and are in full force and effect other than the recording and filing of the Security Documents as required by this Agreement and those third party approvals or consents which, if not made or obtained, would not cause an Event of Default hereunder, could not reasonably be expected to have a Material Adverse Effect or do not have an adverse effect on the enforceability of the Loan Documents, (b) will not violate any Applicable Law or regulation or the charter, by-laws or other organizational documents of the Loan Parties or any order of any Governmental Authority, (c) will not violate or result in a default under any indenture, agreement or other instrument binding upon it or any of its Subsidiaries or its Properties, or give rise to a right thereunder to require any payment to be made by it or such Subsidiary, (d) breach or result in a default (or an event which, with the giving of notice or the passage of time, or both, would constitute a default) under, require any consent under, result in the imposition of any Lien under or give to others any rights of termination, acceleration, suspension, revocation, cancellation or amendment of any Material Agreement and (e) will not result in the creation or imposition of any Lien on any Property of it or any of its Subsidiaries (other than the Liens created by the Loan Documents).
- Ownership and Control. Schedule 5.4 sets forth a correct and complete list of (a) 5.4 the authorized capitalization of Parent and each Loan Party, (b) the number of Equity Interests of Parent and each Loan Party issued and outstanding, and (c) the names of the record and beneficial owners of the Equity Interests of Parent and each Loan Party and the number and percentage of Parent's and each Loan Party's Equity Interests owned by such Persons, in each case as of the Closing Date. Each Loan Party's and Parent's issued and outstanding Equity Interests are duly authorized, validly issued, fully paid and non-assessable as of the Closing Date. Such securities have been or, at the Closing Date, will be issued in compliance with all applicable state and federal securities laws. There are no outstanding (w) options, warrants or other rights issued by Parent or any Loan Party for the acquisition of its Equity Interests, (x) securities or other obligations of Parent or any Loan Party which are convertible into its Equity Interests, (y) agreements of Parent or any Loan Party to issue or sell its Equity Interests, or (z) options, sale agreements, equityholder agreements, pledges, proxies, voting trusts, powers of attorney, restrictions on transfer or other agreements or instruments which are binding on Parent's or any Loan Party's equityholders and which relate to the ownership, voting or transfer of Parent's or such Loan Party's Equity Interests.

- 5.5 <u>Subsidiaries</u>. Parent has no Subsidiaries other than the Borrower, and the Borrower has no Subsidiaries other than Shoosmith. As of the Closing Date, the Borrower is the sole direct and wholly-owned Subsidiary of Parent, Shoosmith is the sole direct and wholly-owned Subsidiary of the Borrower and Shoosmith has no Subsidiaries. No Loan Party is party to a joint venture arrangement or owns any Equity Interest in any Person who is not a Loan Party.
 - 5.6 <u>Financial Condition.</u>
 - (a) [reserved]
 - (b) [reserved]
- (c) Except as and to the extent otherwise disclosed in the most recent Financial Statements or the most recent balance sheet (collectively, the "Current Financial Statements"), the Loan Parties have no material liabilities of any kind, whether direct or indirect, fixed or contingent or otherwise, other than (i) executory obligations under Material Agreements which are not required to be set forth in the Current Financial Statements in accordance with GAAP and (ii) liabilities incurred in the ordinary course of business since the date of the Current Financial Statements.
- (d) The Loan Parties have previously delivered to the Lenders projected consolidated and consolidating balance sheets and statements of income, retained earnings and cash flows of Parent and its consolidated Subsidiaries for Fiscal Year 2025. Such projections were prepared taking into account the past operations and results of the Loan Parties and are based upon the estimates and assumptions stated therein, all of which the Loan Parties believe to be reasonable and fair in light of current conditions and current facts actually known to the Loan Parties. Such projections reflect, as of the Closing Date, the Loan Parties' good faith and reasonable estimates of the future financial performance of the Loan Parties and of the other information projected therein for the periods set forth therein.
- 5.7 <u>Litigation</u>. Other than the Chapter 11 Cases or as otherwise previously disclosed to the Lenders, there is neither pending nor, to the actual knowledge of the Loan Parties' knowledge, threatened against any Loan Party or its assets any action, proceeding or investigation by or before any Governmental Authority or arbitrator.
- 5.8 <u>Compliance with Requirements of Law, Constituent Documents and Material Agreements</u>. Other than as otherwise previously disclosed to the Lenders, each Loan Party is in compliance with (a) all Requirements of Law applicable to such Loan Party or to its business or assets, (b) its constituent documents, and (c) all Material Agreements, except in each case for such non-compliance as would not, individually or in the aggregate, have a Material Adverse Effect.
- 5.9 Governmental Permits, Intellectual Property. Other than as otherwise previously disclosed to the Lenders, each Loan Party has obtained and maintained in full force and effect, and is in compliance with, (a) all permits, licenses, franchises, authorizations, consents and approvals from governmental entities necessary for the ownership or operation of its assets or for the conduct of its business as presently conducted and as presently planned to be conducted, and (b) all licenses and other agreements necessary for it to lawfully use any intellectual property used in the conduct of its business as presently conducted and as presently planned to be conducted, except in each

case where the failure to obtain or maintain the same or such non-compliance would not, individually or in the aggregate, have a Material Adverse Effect.

5.10 <u>Title to Assets</u>. Other than as otherwise previously disclosed to the Lenders, each Loan Party has (a) good and marketable fee simple title to all real property purported to be owned by it, (b) good and marketable title to all other assets purported to be owned by it, and (c) good leasehold title to all assets purported to be leased by it, in each case free and clear of all Liens other than Excepted Liens.

5.11 Certain Requirements of Law.

- (a) The Loan Parties are not subject to regulation under the Investment Company Act of 1940, the Federal Power Act, the Interstate Commerce Act, the Commodity Exchange Act or any Requirement of Law limiting its ability to incur indebtedness for borrowed money or consummate the transactions contemplated by the Loan Documents.
- (b) No Loan Party (i) is a Person whose property or interest in property is blocked or subject to blocking pursuant to Section 1 of Executive Order 13224 of September 23, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)), (ii) engages in any dealings or transactions prohibited by Section 2 of such executive order, or is otherwise associated with any such Person in any manner violative of Section 2, or (iii) is a Person on the list of Specially Designated Nationals and Blocked Persons or subject to the limitations or prohibitions under any other U.S. Department of Treasury's Office of Foreign Assets Control regulation or executive order.
 - (c) The Loan Parties are in compliance, in all material respects, with the Patriot Act.
- 5.12 <u>Use of Proceeds</u>. The Loan Parties have used the proceeds of the DIP Term Loans solely as permitted in <u>Section 6.14</u>. The Loan Parties are not engaged principally, or as one of its or their important activities, in the business of extending credit for the purpose, whether immediate, incidental or ultimate, of buying or carrying margin stock (within the meaning of Regulation T, U or X of the Board).
- 5.13 <u>Budget</u>. Each Budget has been prepared in good faith on the basis of the assumptions stated therein, which assumptions were believed by the Borrower to be reasonable on the date such Budget was delivered.

SECTION 6. AFFIRMATIVE COVENANTS:

Until all of the Obligations (other than contingent indemnification obligations not then due) have been paid and satisfied in full in cash and the DIP Term Loan Commitments terminated or expired, each Loan Party shall, and shall cause each of its Subsidiaries to:

- 6.1 <u>Financial Statements; Other Information</u>. Comply at all times with the covenants regarding information and reporting requirements set forth on <u>Schedule 6.1</u>.
- 6.2 <u>Visitation and Access to Records</u>. The Loan Parties will permit such Persons as the Administrative Agent may designate to visit and inspect any of their respective properties, to examine

and to make copies or make extracts from their respective books and records and to discuss their respective affairs with their respective officers, employees and independent accountants at such times (during normal business hours and upon reasonable prior notice (which prior notice shall not be required following the occurrence and during the continuance of an Event of Default)) and as often as the Administrative Agent may reasonably request. The Loan Parties will bear the reasonable costs and expenses of such visitations and inspections, provided that so long as no Event of Default or Default exists, the Loan Parties shall not be required to reimburse the Administrative Agent for such visitations or inspections more frequently than twice each Fiscal Year.

- 6.3 <u>Financial Accounting Practices</u>. The Loan Parties will make and keep books, records and accounts which, in reasonable detail, accurately and fairly reflect their respective financial positions and operations and which otherwise comply with good accounting practices and GAAP.
- 6.4 Preservation of Existence and Franchises. The Loan Parties will (a) continue to engage in their respective business and operations substantially as conducted immediately prior to the Closing Date, (b) maintain their respective existences, rights and franchises in full force and effect in their respective jurisdictions of organization, (c) qualify and remain qualified as foreign business entities in good standing in each jurisdiction in which the failure to obtain or maintain such qualification would, individually or in the aggregate, have a Material Adverse Effect, and (d) obtain and maintain in full force and effect, and comply with, all permits, licenses, franchises, authorizations, consents and approvals from governmental entities necessary for the ownership or operation of their assets or the conduct of their businesses, except where the failure to obtain or maintain the same or such non-compliance would not, individually or in the aggregate, have a Material Adverse Effect.
- assets and businesses and against such liabilities, casualties and contingencies and of such types, in such amounts and with such financially sound and reputable third party insurers as is customary in the case of Persons engaged in the same or a similar business or having similar properties similarly situated. The Loan Parties will cause to be delivered to the Administrative Agent, upon their written request, certificates for all of the foregoing policies. The activities and operations of the Loan Parties will be conducted at all times in a manner so as to conform in all material respects to such insurance policies, and no Loan Party will take or fail to take any action which would cause any such insurance policy to lapse.
- 6.6 <u>Maintenance of Assets</u>. The Loan Parties will (a) maintain in good repair (ordinary wear and tear excepted), working order and condition the assets now or in the future owned, leased or otherwise possessed by them, (b) make all material repairs, renewals, replacements, improvements and additions to such assets, and (c) not use or permit the use of any of such assets for any unlawful purpose or permit any nuisance to exist thereon.
- 6.7 <u>Payment of Liabilities</u>. The Loan Parties will pay or discharge the following, except those which are being disputed in good faith and by appropriate proceedings, diligently prosecuted to final resolution and as to which appropriate reserves are being maintained to the extent required by GAAP: (a) on or prior to the date on which penalties attach, all taxes imposed upon them or any of their respective assets, (b) on or prior to the date when due, all lawful claims of materialmen,

mechanics, carriers, warehousemen, landlords and other like Persons which, if unpaid, might result in the creation of a Lien upon any of their respective properties or income (other than an Excepted Lien), and (c) on or prior to the date when due, all other lawful claims the nonpayment of which would, individually or in the aggregate, have a Material Adverse Effect.

- 6.8 <u>Compliance With Requirements of Law, Constituent Documents and Material Agreements</u>. The Loan Parties will comply with all applicable Requirements of Law, ¹ their respective constituent documents and all Material Agreements, except for such non-compliance as would not, individually or in the aggregate, have a Material Adverse Effect.
- 6.9 <u>Federal Reserve Board Regulations</u>. The Loan Parties will neither take nor permit to be taken on their behalf any action which might cause any transaction, obligation or right created by any of the Loan Documents to violate any regulation of the Federal Reserve Board applicable to the Administrative Agent.
- 6.10 <u>Anti-Corruption</u>. The Loan Parties shall (i) not permit any Covered Entity to become a Sanctioned Person, (ii) not permit any Covered Entity, either in its own right or through any third party, to (A) have any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Corruption Law; (B) do business in or with, or derive any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Corruption Law; or (C) engage in any dealings or transactions prohibited by any Anti-Corruption Law, and (iii) cause each Covered Entity to comply with all Anti-Corruption Laws.

6.11 Further Assurances

- (A) It shall, and shall cause each of its Subsidiaries to, at its own expense, promptly execute and deliver to the Administrative Agent all such other documents, agreements and instruments reasonably requested by the Required Lenders to comply with, cure any defects or accomplish the conditions precedent, covenants and agreements of it or any of its Subsidiaries, as the case may be, in the Loan Documents or to further evidence and more fully describe the collateral intended as security for the Obligations, or to correct any omissions in this Agreement or the other Loan Documents, or to state more fully the obligations secured by the Loan Documents, or to perfect, protect or preserve any Liens created pursuant to this Agreement or any of the Loan Documents or the priority thereof, or to make any recordings, file any notices or obtain any consents, all as may be reasonably necessary or appropriate, in the sole discretion of the Required Lenders, in connection therewith.
- (B) Each Loan Party authorizes the Administrative Agent and the Lenders to file one or more financing or continuation statements, and amendments thereto, relative to all or any part of the Collateral without the signature of such Loan Party where permitted by law. A carbon, photographic or other reproduction of the Security

¹ NTD – to discuss.

- Documents or any financing statement covering the Collateral or any part thereof shall be sufficient as a financing statement where permitted by law.
- (C) Notwithstanding any provision in any of the Loan Documents to the contrary, in no event will any Excluded Deposit Account be encumbered by any Security Document; <u>provided</u>, the Borrower shall not, and shall not permit any of its Subsidiaries to, permit to exist any Lien on any Excluded Deposit Account.

6.13 Additional Collateral; Additional Guarantors

(A) Borrower shall promptly cause each of its Subsidiaries to guarantee the Obligations pursuant to the Guaranty Agreement. Borrower shall cause such Subsidiary to promptly (but with respect to any Subsidiary formed or acquired after the date hereof, no later than ten days after the date of such formation or acquisition), (i) execute and deliver the Guaranty Agreement or a supplement to the Guaranty Agreement as required by the Required Lenders, (ii) pledge all of the Equity Interests of such Subsidiary (including delivery of original certificates evidencing the Equity Interests of such Subsidiary, together with an appropriate undated stock or equity powers for each certificate duly executed in blank by the registered owner thereof) and (iii) execute and deliver such other additional closing documents, certificates and legal opinions as are reasonably requested by the Required Lenders in connection with this Section 6.13.

6.14 <u>Use of Proceeds and Budget</u>.

- (A) Unless otherwise approved by the Administrative Agent in writing (at the direction of the Required Lenders), the proceeds of the DIP Term Loans shall be used by the Borrower strictly in accordance with the Budget and subject to the covenant set forth below in Section 6.18(B) (including variances).
- (B) The Borrower shall also deliver to the Administrative Agent and Lenders the Budget, delivered before noon on Friday of each week commencing with the second week following the Petition Date, showing actual receipts and disbursements through and including the immediately preceding week. The Budget attached to the Interim DIP Financing Order constitutes the Initial Approved Budget. Commencing with the second week following the Petition Date, the Borrower will subsequently provide the Lenders with an updated budget before noon every Friday prior to the end of the Budget Period for the remaining weeks left under the Initial Approved Budget.
- (C) Every week, before noon beginning on the Friday that is one week following the Petition Date, the Borrower shall deliver to the Administrative Agent and the Lenders supplements to the Budget showing projected receipts and disbursements for the subsequent 13-week period. The timing and delivery of the Budget may be modified by written agreement of the Borrower and the Administrative Agent (acting at the direction of a majority in interest of Lenders). Each updated budget shall be subject to the approval of the Lenders in their sole discretion (each such

approved budget, a "<u>Supplemental Approved DIP Budget</u>" and together with the Initial Approved Budget, the "<u>Budget</u>"); <u>provided</u> that until such time as the Lenders approve any updated budget, the Borrower shall be subject to and be governed by the terms of the Initial Approved Budget or such Supplemental Approved DIP Budget, as applicable, then in effect.

- (D) Beginning on the Friday that is the first week following the Petition Date, and measured weekly thereafter, the Borrower shall not permit the actual total disbursements (excluding any disbursements that are or would be reimbursable by a liquidator) to exceed, on a cumulative basis, 115% of the total disbursements set forth in the Budget for such period (each, a "Permitted Variance").
- (E) The Borrower shall hold weekly calls with the Lenders following the delivery of the updated Budget on a day and time to be mutually agreed upon by the Borrower and the Lenders.

6.15 Collateral Matters.

Subject to the Carve-Out and the DIP Financing Orders, all Obligations of the Loan Parties shall be:

- (A) Entitled to superpriority claim status under section 364(c)(1) of the Bankruptcy Code with priority over all administrative expense claims and unsecured claims existing as of the Petition Date or arising thereafter under the Bankruptcy Code, including, without limitation, the prepetition claims and adequate protection claims of the Prepetition Lenders (the "DIP Superpriority Claims"). The DIP Superpriority Claims may be repaid from any cash of the Debtors, including without limitation, Cash Collateral (within the meaning of the DIP Financing Orders) and, following entry of the Final DIP Financing Order, the proceeds of Avoidance Actions and Avoidance Action Proceeds;
- (B) secured, pursuant to section 364(c)(2) of the Bankruptcy Code, by valid, enforceable, first priority, security interests in and liens on all of the Debtors' rights in property of the Debtors' estates as of the Petition Date that, as of the Petition Date, were unencumbered (and do not become perfected subsequent to the Petition Date as permitted by section 546(b) of the Bankruptcy Code) (including, following entry of the Final DIP Financing Order, Avoidance Action Proceeds);
- (C) secured, pursuant to section 364(c)(3) of the Bankruptcy Code, by valid, enforceable, security interests in and liens on all of the Debtors' rights in property of the Debtors' estates as of the Petition Date that, as of the Petition Date, were subject to valid liens and unavoidable liens in existence immediately prior to the Petition Date, if any, that are perfected subsequent to the Petition Date as permitted by section 546(b) of the Bankruptcy Code (the "Permitted Prior Liens"), which security interests and liens shall be junior and subordinate only to such Permitted Prior Liens and the Carve-Out; and

(D) secured, pursuant to section 364(d)(1) of the Bankruptcy Code, by valid, enforceable, priming first priority security interests in and liens upon all of the Debtors' rights in property of the Debtors' estates as of the Petition Date, and all of the Debtors' rights in property acquired post-petition (and proceeds thereof), whether now existing or hereafter acquired or arising, that secure the obligations in respect of the Prepetition Term Loan Agreement.

The Collateral shall also include any and all hereafter acquired or arising property, rents, issues, products, offspring, proceeds and profits generated by any item of Collateral and, following entry of the Final DIP Financing Order, liens on proceeds of any Avoidance Actions.

The Liens shall not be subject or subordinate to (i) any lien or security interest that is avoided and preserved for the benefit of any Debtor and their estates under section 551 of the Bankruptcy Code, (ii) any liens arising after the Petition Date including, without limitation, any liens or security interests granted in favor of any federal, state, municipal or other governmental unit, commission, board or court for any liability of any Debtor, or (iii) any intercompany or affiliate liens of any Debtor.

The Collateral will be free and clear of other liens, claims and encumbrances, except valid, enforceable liens, rights of recoupment enforceable in bankruptcy, and rights of setoff permissible under section 553 of the Bankruptcy Code, in each case except as otherwise agreed by the applicable creditor or lienholder, as applicable, in existence as of the Petition Date and permitted pursuant to Prepetition Credit Agreements, if any, and any other Permitted Prior Liens.

The Borrower acknowledges that the Liens shall automatically attach to the Collateral and became valid immediately upon entry of the Interim DIP Financing Order without the requirement of any further action by the Lender; provided that if the Lender determines to file any financing statements, notice of liens or similar instruments, the Debtors will cooperate and assist in any such filings.

6.16 Chapter 11 Milestones.

The Loan Parties shall comply with and meet the following milestones as the same may be extended, waived, or otherwise modified with the prior written consent of the Required Lenders in their sole discretion (which may be by email) without further order by the Bankruptcy Court (collectively, the "Chapter 11 Milestones" and each a "Chapter 11 Milestone"):

- (A) The Bankruptcy Court shall have entered the Interim DIP Financing Order by the date that is no later than five (5) Business Days after the Petition Date;
- (B) The Debtors shall have filed the motion seeking entry of the Bidding Procedures Order and Sale Order (the "Bidding Procedures Motion") by no later than ten (10) days after the Petition Date;
- (C) The Bankruptcy Court shall have entered the Final DIP Financing Order by the date that is no later than thirty-three (33) days after the Petition;

- (D) The Bankruptcy Court shall have entered the order approving the Bidding Procedures Motion by no later than thirty-three (33) days after the filing of the Bidding Procedures Motion;
- (E) The Debtors shall have (a) entered into and delivered a fully executed and binding Acceptable Stalking Horse Purchase Agreement and (b) satisfied any other applicable Chapter 11 Milestones by no later than the date of entry of the Final DIP Financing Order;
- (F) The Bankruptcy Court shall have entered the order approving the Sale Transaction by no later than eighty (80) days after the Petition Date;
- (G) The Debtors shall have closed the Sale Transaction by no later than eighty-five (85) days after the Petition Date; and
- (H) The Plan Effective Date shall have occurred by no later than one hundred ten (110) days after the Petition Date.

The Loan Parties shall also provide the Administrative Agent and the Lenders with drafts of all first day pleadings, second day pleadings, sale pleadings and plan pleadings in the Bankruptcy Case two (2) days prior to the anticipated filing date, subject to waiver by the Lenders in their sole discretion.

- 6.17 <u>Compliance with Anti-Corruption Laws and Sanctions</u>. The Borrower shall maintain in effect and enforce policies and procedures designed to ensure compliance by the Borrower and its Subsidiaries and their respective directors, officers, employees and agents with Anti-Corruption Laws and applicable Sanctions.
- 6.18 <u>Deposit Accounts</u>. Each Loan Party shall provide the Administrative Agent with an Account Control Agreement in form and substance reasonably satisfactory to the Required Lenders duly executed by such Loan Party and the financial institute maintaining each Deposit Account of such Loan Party other than (a) Excluded Deposit Accounts and (b) Deposit Accounts used for the payment of corporate credit cards so long as the aggregate amount on deposit in such Deposit Accounts does not exceed \$100,000 at any time.

SECTION 7. NEGATIVE COVENANTS:

Until all of the Obligations (other than contingent, indemnification obligations not then due) have been paid and satisfied in full in cash and the DIP Term Loan Commitments terminated or expired, the Loan Parties shall not, and shall not permit any of their respective Subsidiaries to:

- 7.1 <u>Negative Covenants</u>. Fail to be in compliance at any time with the negative covenants set forth in <u>Schedule 7.1</u>.
 - 7.2 <u>Prepayment of Prepetition Debt.</u>
 - (A) It shall not make (or give any notice with respect thereto) any voluntary or optional payment or prepayment or redemption or acquisition for value of (including without

limitation, by way of depositing money or securities with the Administrative Agent with respect thereto before due for the purpose of paying when due), refund, refinance or exchange of any Debt of the Loan Parties (other than (i) the Obligations arising under the Loan Documents and (ii) payments in respect of Debt expressly contemplated under the Budget (subject to Permitted Variances), in each case, to the extent permitted by the DIP Financing Orders, the First and Second Day Orders and adequate protection payments as set forth in the DIP Financing Orders).

7.3 Compliance with the Budget.

- (A) It shall not, except as approved in advance in writing by the Required Lenders (or the Administrative Agent at the direction of the Required Lenders), directly or indirectly:
 - (i) use any cash or the proceeds of any DIP Term Loans in a manner or for a purpose other than those consistent with this Agreement, the DIP Financing Orders and the approved Budget (other than a Permitted Variance);
 - (ii) permit a disbursement causing any variance from the approved Budget other than a Permitted Variance; or
 - (iii) transfer cash or Cash Equivalents to a Subsidiary or Affiliate other than payments set forth in and consistent with the approved Budget (subject to any Permitted Variance).

It is understood and agreed, for the avoidance of doubt, that Permitted Variances under this Agreement shall be limited to variances in excess of aggregate disbursements of items contemplated by the approved Budget and not for the purposes of additional categories or types of expenditures or disbursements not expressly contemplated in the approved Budget.

7.4 Chapter 11 Modifications.

(A) Borrower shall not, without the prior written consent of the Required Lenders (or the Administrative Agent at the direction of the Required Lenders): (a) make or permit to be made any change, amendment or modification, to the DIP Financing Orders or (b) incur, create, assume or suffer to exist or permit any claim or Lien against the Borrower ranking *pari passu* with or senior to the claims and Liens of the Administrative Agent and the other Secured Parties hereunder, except with respect to the Carve-Out or as expressly permitted under the Interim DIP Financing Order approved by the Bankruptcy Court as of the Effective Date or under the Final DIP Financing Order, as applicable.

7.5 Additional Bankruptcy Matters.

(A) Borrower shall not do any of the following other than as expressly permitted by the DIP Financing Orders or as may be agreed to in writing by the Required Lenders (or the Administrative Agent at the direction of the Required Lenders):

- (i) enter into any agreement to return any of its inventory to any of its creditors for application against any Prepetition Debt, prepetition trade payables or other prepetition claims under Section 546(c) of the Bankruptcy Code or agree that any creditor may take any set-off or recoupment against any of its Prepetition Debt, prepetition trade payables or other prepetition claims based upon any such return pursuant to Section 553(b)(1) of the Bankruptcy Code or otherwise if, after giving effect to any such agreement, set-off or recoupment, the aggregate amount applied to Prepetition Debt, prepetition trade payables and other prepetition claims subject to all such agreements, set-offs and recoupments since the Petition Date would exceed \$100,000;
- (ii) seek, consent to, or permit to exist, without timely filing an objection, without the prior written consent of the Required Lenders (or the Administrative Agent at the direction of the Required Lenders), any order granting authority to take any action that is prohibited by the terms of this Agreement, the DIP Financing Orders or the other Loan Documents or refrain from taking any action that is expressly required to be taken by the terms of this Agreement, the DIP Financing Orders or any of the other Loan Documents;
- (iii) assert, join, investigate, support or prosecute any claim or cause of action against (i) any of the Secured Parties, unless such claim or cause of action is solely to enforce the Loan Documents against any of the Secured Parties or (ii) any Prepetition Secured Party;
- (iv) subject to the terms of the DIP Financing Orders and subject to Section 8, object to, contest, prevent or interfere with in any manner the exercise of rights and remedies by the Administrative Agent, the Lenders or other Secured Parties with respect to the Collateral following the occurrence of an Event of Default, including, without limitation, a motion or petition by any Secured Party to lift an applicable stay of proceedings to do the foregoing (provided, that, the Loan Parties may contest or dispute whether an Event of Default has occurred in accordance with the terms of the DIP Financing Orders);
- (v) hold any proceeds of the DIP Term Loans in any account other than a Controlled Account, pending withdrawal and application thereof in accordance with this Agreement;
- (vi) other than in connection with an Asset Disposition permitted by <u>Section 4(g) of Schedule 7.1</u> or an Acceptable Plan, without the prior written consent of the Required Lenders, move to assume or reject any material lease, license, or other Material Contract of the Borrower pursuant to Section 365 of the Bankruptcy Code; or
- (vii) except as expressly provided or permitted hereunder (including, without limitation, to the extent pursuant to any First and Second Day Orders) and

the then approved Budget (including Permitted Variances thereto), make any payment or distribution to any Subsidiary, Affiliate or insider of the Borrower.

SECTION 8. DEFAULT AND REMEDIES:

- 8.1 <u>Events of Default</u>. Each of the following constitutes an Event of Default:
- (A) the Borrower fails to pay (i) when and as required to be paid herein, any amount of principal of any Loan, or (ii) within three (3) Business Days after the same becomes due, any interest on any Loan, or any repayment premium or fee due hereunder, or any other amount payable hereunder or under any other Loan Document;
- (B) any representation or warranty made or deemed made by or on behalf of the Parent, the Borrower or any Subsidiary in or in connection with any Loan Document or any amendment or modification of any Loan Document or waiver under such Loan Document, or in any report, certificate, financial statement or other document furnished pursuant to or in connection with any Loan Document or any amendment or modification thereof or waiver thereunder, proves to have been materially incorrect when made or deemed made;
- (C) the Borrower or any Subsidiary fails to observe or perform any covenant, condition or agreement contained in <u>Section 6</u> or in <u>Section 7</u>, including, but not limited to those requirements concerning the budget contained in <u>Section 6.14</u>;
- (D) the Borrower or any Subsidiary fails to observe or perform any covenant, condition or agreement contained in this Agreement (other than those specified in Section 8.1(A), 8.1(B), or 8.1(C)) or any other Loan Document, and such failure continues unremedied for a period of 30 days after the earlier to occur of (i) notice thereof from the Administrative Agent (at the written direction of any Lender) or any Lender to the Borrower and (ii) a Responsible Officer of the Borrower or such Subsidiary otherwise becoming aware of such default;
- (E) any event or condition occurs that results in any Debt having a principal amount greater than One Hundred Thousand Dollars (\$100,000) ("Material Debt") becoming due prior to its scheduled maturity or that enables or permits (with or without the giving of notice, the lapse of time or both) the holder or holders of any Material Debt or any trustee or agent on its or their behalf to cause any Material Debt to become due, or to require the Redemption thereof or any offer to Redeem to be made in respect thereof, prior to its scheduled maturity or require the Borrower or any Subsidiary to make an offer in respect thereof;
- (F) the appointment of or application for a receiver, trustee, custodian, examiner, sequestrator, conservator or similar official for the Borrower or any Subsidiary or for a substantial part of its or their assets, and, in any such case, such proceeding or petition continues undismissed for 30 days or an order or decree approving or ordering any of the foregoing is entered.

- (G) the Borrower or any Subsidiary who is not party to the Bankruptcy Case (i) voluntarily commences any proceeding or files any petition seeking liquidation, reorganization or other relief under any Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, aside from the contemplated Bankruptcy Case, (ii) consents to the institution of, or fails to contest in a timely and appropriate manner, any proceedings described in Section 8.1(F), (iii) applies for or consents to the appointment of a receiver, trustee, custodian, examiner, sequestrator, conservator, or similar official for the Borrower, or any Subsidiary or for a substantial part of its or their assets, (iv) files an answer admitting the material allegations of a petition filed against it or them in any such proceeding, (v) makes a general assignment for the benefit of creditors, or (vi) takes any action for the purpose of effecting any of the forgoing;
- (H) (i) one or more judgments for the payment of money in an aggregate amount in excess of \$2,000,000 (to the extent not covered by independent third party insurance provided by insurers of the highest claims paying rating or financial strength as to which the insurer does not dispute coverage and is not subject to an insolvency proceeding) or (ii) any one or more non-monetary judgments that have, or could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, are rendered against the Borrower, any Subsidiary or any combination thereof and the same remain undischarged for a period of 30 consecutive days during which execution is not effectively stayed, or any action is legally taken by a judgment creditor to attach or levy upon any assets of the Borrower or any Subsidiary to enforce any such judgment;
- (I) any Loan Document, for any reason, except to the extent permitted by the terms thereof, ceases to be in full force and effect and valid, binding and enforceable in accordance with their terms against the Borrower or a Guarantor party thereto or is repudiated by any of them, or cease to create a Lien of the priority required thereby on any of the collateral purported to be covered thereby, except to the extent permitted by the terms of this Agreement, or the Borrower or any Subsidiary or any of their Affiliates so state in writing;
- (J) there occurs any circumstance or circumstances that could reasonably be expected, either individually or in the aggregate, to have a Material Adverse Effect;
- (K) any subordination provision in any document or instrument governing Debt that is purported to be subordinated to the Obligations or any subordination provision in any subordination agreement that relates to any Debt that is to be subordinated to the Obligations, or any subordination provision in any guaranty by the Borrower or any of its Subsidiaries of any such Debt, shall cease to be in full force and effect, or any Person (including the holder of any such Debt) shall contest in any manner the validity, binding nature or enforceability of any such provision;
- (L) any court order enjoins, restrains, or prevents the Borrower or any of its Subsidiaries from conducting any material part of its business;

- (M) the Borrower shall file a motion in the Chapter 11 Case without the prior express written consent of the Required Lenders (or the Administrative Agent at the direction of the Required Lenders), to obtain additional financing from a party other than the Lenders under Section 364(d) of the Bankruptcy Code or to use Cash Collateral of a Lender under Section 363(c) of the Bankruptcy Code;
- (N) the Bankruptcy Court shall enter an order (i) approving payment of any prepetition claim other than (x) as provided for in the First and Second Day Orders, (y) as otherwise contemplated hereunder and by the approved Budget or (z) as otherwise consented to by the Required Lenders (or the Administrative Agent at the direction of the Required Lenders) in writing in their reasonable discretion, (ii) granting relief from the automatic stay applicable under Section 362 of the Bankruptcy Code to any holder of any security interest to permit foreclosure on any assets having a book value in excess of \$100,000 in the aggregate, or (iii) except, as expressly provided in the DIP Financing Orders, approving any settlement or other stipulation not approved by the Required Lenders (or the Administrative Agent at the direction of the Required Lenders) with any prepetition secured creditor of the Borrower or its Subsidiary providing for payments as adequate protection or otherwise to such secured creditor;
- (O) an order with respect to the Chapter 11 Cases shall be entered by the Bankruptcy Court (i) appointing a trustee under Section 1104, (ii) appointing an examiner with enlarged powers relating to the operation of the business (powers beyond those set forth in Section 1106(a)(3) and (4) of the Bankruptcy Code) under Section 1106(b) of the Bankruptcy Code, or (iii) converting the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code;
- (P) The entry of a final non-appealable order in the Chapter 11 Cases charging any of the DIP Collateral under section 506(c) of the Bankruptcy Code against the Lenders or Administrative Agent;
- (Q) an order shall be entered by the Bankruptcy Court dismissing the Chapter 11 Cases which does not contain a provision for termination of all DIP Term Loan Commitments, and indefeasible payment in full in cash of all Obligations upon entry thereof;
- (R) an order with respect to the Chapter 11 Cases shall be entered by the Bankruptcy Court without the express prior written consent of the Required Lenders (and with respect to any provisions that affect the rights or duties of the Administrative Agent, the Administrative Agent, not to be unreasonably withheld, conditioned or delayed), (i) to revoke, vacate, reverse, stay, modify, supplement or amend the DIP Financing Orders (other than the modifications to the Interim DIP Financing Order by the Final DIP Financing Order) or (ii) to permit, unless otherwise contemplated by the DIP Financing Orders, any administrative expense or any claim (now existing or hereafter arising, of any kind or nature whatsoever) to have administrative priority equal or superior to the administrative priority of the Obligations (other than the Carve Out);

(S) No Contestation; Surcharge.

- (i) an application for any of the orders described in Section 8.1(O) above shall be made by the Borrower or any such application shall be made by a Person other than the Borrower and such application is not contested by the Borrower in good faith or the relief requested is not withdrawn, dismissed or denied within ten (10) days after the filing or (ii) the Borrower shall file a motion seeking (or the Borrower shall not object to any other Person seeking), or any other Person shall obtain an order authorizing, any of the Collateral or collateral securing the Prepetition Debt to be surcharged pursuant to Section 506(c) of the Bankruptcy Code;
- (T) the entry of an order by the Bankruptcy Court terminating or modifying the exclusive right of the Borrower to file a Plan of Reorganization pursuant to Section 1121 of the Bankruptcy Code, without the prior written consent of the Required Lenders;
- (U) the Borrower or any of its Subsidiaries shall seek to, or shall support (in any such case by way of any motion or other pleading filed with the Bankruptcy Court or any other writing to another party-in-interest executed by or on behalf of the Borrower or any of its Subsidiaries) any other Person's motion to, avoid or disallow in whole or in part the Lenders' claims or Liens in respect of the Obligations or contest any provision of any Loan Document or any provision of any Loan Document shall cease to be effective provided, however, that it shall not constitute an Event of Default if the Borrower or any of its Subsidiaries provide information with respect to the Loan Documents or documents relating to the Lenders' Prepetition Debt to a party in interest, or are compelled to provide information by an order of the Bankruptcy Court so long as the Lenders have been served with the request for such an order or, if no such service has been made prior to the time the Lenders are required to provide information, so long as the Borrower or any of its Subsidiaries provide prior written notice to the Lenders of any intention or requirement to do so;
- (V) the Borrower, Guarantor or any of their Subsidiaries or Affiliates shall fail to comply with the Interim DIP Financing Order or the Final DIP Financing Order;
- (W) any order by the Bankruptcy Court is entered granting any administrative or superpriority claim that is *pari passu* with or senior to those of the Secured Parties or any Lien that is senior to the Liens securing the Obligations, other than in accordance with the DIP Financing Order (except with respect to the Carve-Out);
- (X) without the Required Lenders' consent, the filing by the Borrower of any motion or other request with the Bankruptcy Court seeking authority to consummate a sale of material assets or equity or all or substantially all of the Borrower's or any of its Subsidiaries' assets or equity, or the consummation of any such sale;

- (Y) the Borrower or any of its Subsidiaries, or any person claiming by or through the Borrower or any of its Subsidiaries, obtain court authorization to commence, or commence, join in, assist or otherwise participate as an adverse party in any suit or other proceeding against any of the Administrative Agent, the Lenders, or the Prepetition Parties, in each case relating to this Agreement or Prepetition Debt, as applicable;
- (Z) the filing by the Borrower of a Plan of Reorganization other than an Acceptable Plan; or
- (AA) a Change in Control occurs without the Required Lenders consent.
- 8.2 <u>Remedies</u>. Notwithstanding anything in Section 362 of the Bankruptcy Code, but subject to the applicable DIP Financing Order, if any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Lenders, take any or all of the following actions:
 - (A) declare the DIP Term Loan Commitment of each Lender to be terminated, whereupon such DIP Term Loan Commitments shall be terminated;
 - (B) declare the unpaid principal amount of all outstanding DIP Term Loans and, if after the Interim DIP Financing Order is entered, the Roll-Up Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower; and
 - (C) exercise on behalf of any Lender all of its other rights and remedies under this Agreement, the other Loan Documents and Applicable Law, in order to satisfy all of the Obligations.
- Rights and Remedies Cumulative; Non-Waiver. The enumeration of the rights and remedies of the Administrative Agent and the Lenders set forth in this Agreement is not intended to be exhaustive and the exercise by the Administrative Agent and the Lenders of any right or remedy does not preclude the exercise of any other rights or remedies, all of which are cumulative, and are in addition to any other right or remedy given hereunder or under the other Loan Documents or that may now or hereafter exist at law or in equity or by suit or otherwise. No delay or failure to take action on the part of the Administrative Agent or any Lender in exercising any right, power or privilege operates as a waiver thereof, nor does any single or partial exercise of any such right, power or privilege preclude any other or further exercise thereof or the exercise of any other right, power or privilege or be construed to be a waiver of any Event of Default. No course of dealing between the Loan Parties, the Administrative Agent and the Lenders or their respective agents or employees is effective to change, modify or discharge any provision of this Agreement or any of the other Loan Documents or to constitute a waiver of any Event of Default.
- 8.4 <u>Crediting of Payments and Proceeds</u>. In the event that the Obligations have been accelerated pursuant to <u>Section 8.2</u> or the Administrative Agent or any Lender has exercised any remedy set forth in this Agreement or any other Loan Document, all payments received on account

of the Obligations and all net proceeds from the enforcement of the Obligations shall be applied by the Administrative Agent as follows:

- (A) <u>first</u>, to fund the Carve-Out (to the extent not fully funded by Borrower at such time);
- (B) <u>second</u>, to payment of that portion of the Obligations constituting fees, losses, indemnities, expenses and other amounts, including attorney fees, payable to the Administrative Agent under the Loan Documents in its capacity as such;
- (C) third, to payment of that portion of the Obligations constituting fees, indemnities and other amounts (other than principal and interest) payable to the Lenders under the Loan Documents, including attorney fees, ratably among the Lenders in proportion to the respective amounts described in this clause second payable to them;
- (D) <u>fourth</u>, to payment of that portion of the Obligations constituting accrued and unpaid interest on the DIP Facility ratably among the Lenders in proportion to the respective amounts described in this clause third payable to them;
- (E) <u>fifth</u>, to payment of that portion of the Obligations constituting unpaid principal of the DIP Facility, ratably among the Lenders in proportion to the respective amounts described in this clause <u>fourth</u> payable to them; and
- (F) <u>last</u>, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to the Borrower or as otherwise required by Applicable Law.

8.5 <u>Credit Bidding.</u>

(A) The Administrative Agent (at the written direction of the Required Lenders), on behalf of itself and the Lenders, has the right to credit bid and purchase for the benefit of the Administrative Agent and the Lenders all or any portion of Collateral at any sale thereof conducted by the Administrative Agent under the provisions of the Uniform Commercial Code, including pursuant to Sections 9-610 or 9-620 of the Uniform Commercial Code, at any sale thereof conducted under the provisions of the United States Bankruptcy Code, including Section 363 thereof, or a sale under a plan of reorganization, or at any other sale or foreclosure conducted by the Administrative Agent (whether by judicial action or otherwise) in accordance with Applicable Law. Such credit bid or purchase may be completed through one or more acquisition vehicles formed by the Administrative Agent to make such credit bid or purchase and, in connection therewith, the Administrative Agent is authorized, on behalf of itself and the Lenders, to adopt documents providing for the governance of the acquisition vehicle or vehicles, and assign the applicable Obligations to any such acquisition vehicle in exchange for Equity Interests or debt issued by the applicable acquisition vehicle (which shall be deemed to be held for the ratable account of the applicable Lenders on the basis of the Obligations so assigned by each Lender).

- (B) Except as otherwise provided in any Loan Document or with the written consent of the Required Lenders, no Lender shall take any enforcement action, accelerate obligations under any of the Loan Documents, or exercise any right that it might otherwise have under Applicable Law to credit bid at foreclosure sales, UCC sales or other similar dispositions of Collateral.
- 8.6 <u>Injunctive Relief.</u> The Borrower recognizes that, in the event the Borrower fails to perform, observe or discharge any of its obligations or liabilities under this Agreement, any remedy of law may prove to be inadequate relief to the Lenders. Therefore, the Borrower agrees that the Lenders, at the Lenders' option, are entitled to temporary and permanent injunctive relief in any such case without the necessity of proving actual damages.

SECTION 9. AGENCY:

- 9.1 Appointment and Authority. Each of the Lenders irrevocably appoints Volunteer Enterprises, LLC to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents to which the Administrative Agent is a party and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof. The provisions of this Section 9 are solely for the benefit of the Administrative Agent and the Lenders and neither the Borrower nor any other Loan Party have rights as a third-party beneficiary of any of such provisions. It is understood and agreed that the use of the term "agent" herein or in any other Loan Documents (or any other similar term) with reference to the Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any Applicable Law. Instead such term is used as a matter of market custom, and is intended to create or reflect only an administrative relationship between contracting parties.
- 9.2 Rights as a Lender. The Person serving as the Administrative Agent hereunder has the same rights and powers in its capacity as a Lender as any other Lender, if applicable, and may exercise the same as though it were not the Administrative Agent, and the term "Lender" or "Lenders" includes, unless otherwise expressly indicated or unless the context otherwise requires, the Person serving as the Administrative Agent hereunder in its individual capacity, if such Person is a Lender. Such Person and its Affiliates may accept deposits from, lend money to, own securities of, act as the financial advisor or in any other advisory capacity for, and generally engage in any kind of business with, the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.
- 9.3 <u>Exculpatory Provisions</u>. (A) The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents to which it is a party, and its duties hereunder are administrative in nature. Without limiting the generality of the foregoing, the Administrative Agent:
 - (i) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;

- discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as is expressly provided for herein or in the other Loan Documents); provided that the Administrative Agent is not required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or Applicable Law, including for the avoidance of doubt any action that may be in violation of the automatic stay under any Debtor Relief Law or that may affect a forfeiture, modification or termination of property of a Defaulting Lender in violation of any Debtor Relief Law; and
- (iii) shall not, except as expressly set forth herein and in the other Loan Documents to which it is a party, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.
- (B) The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request or direction of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent believes in good faith shall be necessary, under the circumstances) or (ii) in the absence of its own gross negligence or willful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment. The Administrative Agent is not deemed to have knowledge of any Default or Event of Default unless and until notice describing such Default or Event of Default is given to the Administrative Agent in writing by the Borrower or a Lender.
- (C) The Administrative Agent is not responsible for and has no duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith and the Administrative Agent is not required to calculate, recalculate, verify, review or certify any calculation or any other mathematical or numerical information, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default or Event of Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document, or the perfection or priority of any Lien created or purported to be created by the Security Documents or (v) the satisfaction of any condition set forth in Section 4 or elsewhere herein other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.
- (D) The Lenders direct the Administrative Agent to execute and deliver the Loan Documents to which it is a party on the Closing Date. Whether or not so expressly stated therein, in entering into, or taking (or forbearing from) any action under or pursuant to, the Loan Documents, the Administrative Agent has all of the rights, immunities, indemnities and other

protections granted to it under this Agreement (in addition to those that may be granted to it under the terms of such other agreement or agreements).

- (E) The Person serving as the Administrative Agent may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Loan Parties or any Affiliate thereof as if such Person was not the Administrative Agent and without any duty to account therefor to the Lenders.
- (F) The powers conferred on the Administrative Agent under the Security Documents are solely to protect the Administrative Agent's interest in the Collateral, do not impose any duty upon it to exercise any such powers and are subject to the provisions of this Agreement. The Administrative Agent is accountable only for amounts that it actually receives as a result of the exercise of such powers and the Administrative Agent and any of its officers, directors, employees or agents are not responsible for any act or failure to act, except for gross negligence or willful misconduct (as determined by a final non-appealable judgment of a court of competent jurisdiction). Unless otherwise directed by the Required Lenders, the Administrative Agent has no responsibility for taking any necessary steps to protect, preserve or exercise rights against any Person with respect to any of the Collateral and is relieved of all responsibility for the Collateral upon surrendering it to the Borrower or any other designated Person. In no event does the Administrative Agent have any duty, responsibility, obligation or liability with respect to monitoring the Collateral or the condition thereof, and the Administrative Agent is not required to take any action to protect against any diminution in value of the Collateral.
- (G) Notwithstanding any provision of this Agreement or the other Loan Documents to the contrary, the Administrative Agent is not required to (i) make or give any determination (including whether a matter is satisfactory to the Administrative Agent or whether to deem a matter necessary, desirable, proper or advisable), agreement, consent, approval, request, notice, consultation, designation, appointment, election, judgment or direction, (ii) file, record or prepare any Uniform Commercial Code financing or continuation statements or similar documents or instruments in any jurisdiction for purposes of creating, perfecting or maintaining any Lien or security interest or otherwise perfect any security interest or maintain any perfection or priority, (iii) make any inspection or (iv) release or sell Collateral or otherwise exercise any rights or remedies of a secured party (including voting rights), in each case, without the written direction of the Required Lenders.
- (H) Notwithstanding any provision of this Agreement or the other Loan Documents to the contrary, before taking or omitting any action to be taken or omitted by the Administrative Agent under the terms of this Agreement and the other Loan Documents, the Administrative Agent may seek the written direction of the Required Lenders (which written direction may be in the form of an e-mail), and the Administrative Agent is entitled to rely (and is fully protected in so relying) upon such direction. The Administrative Agent is not liable with respect to any action taken or omitted to be taken by it in accordance with such direction. If the Administrative Agent requests such direction with respect to any action, the Administrative Agent is entitled to refrain from such action unless and until the Administrative Agent has received such direction, and the Administrative Agent does not incur liability to any Person by reason of so refraining. If the Administrative Agent so requests, it must first be indemnified to its satisfaction by the directing Lenders against any and all liability and expense which may be incurred by the Administrative

Agent by reason of taking or continuing to take, or omitting, any action directed by any of the Lenders. Any provision of this Agreement or the other Loan Documents authorizing the Administrative Agent to take any action does not obligate the Administrative Agent to take such action.

- (I) The Administrative Agent is never required to use, risk or advance its own funds or otherwise incur financial liability in the performance of any of its duties or the exercise of any of its rights and powers hereunder or under the other Loan Documents (including no obligation to grant any credit extension or to make any advance hereunder).
- (J) Delivery of reports, documents and other information to the Administrative Agent is for informational purposes only and the Administrative Agent's receipt of the foregoing does not constitute constructive knowledge of any event or circumstance or any information contained therein or determinable from information contained therein.
- (K) The parties hereto acknowledge that in accordance with the Customer Identification Program (CIP) requirements under the USA PATRIOT Act and its implementing regulations, the Administrative Agent, in order to help fight the funding of terrorism and money laundering, is required to obtain, verify, and record information that identifies each person or legal entity that establishes a relationship or opens an account with the Administrative Agent. The parties hereto shall provide the Administrative Agent with such information as it may request including each party's name, physical address, tax identification number and other information that will help the Administrative Agent identify and verify each party's identity such as organizational documents, certificate of good standing, license to do business, or other pertinent identifying information.
- (L) In no event is the Administrative Agent liable on any theory of liability for any special, indirect, consequential or punitive damages (including lost profits) even if the Administrative Agent has been advised of the possibility of such damages and regardless of the form of action. The Administrative Agent is not responsible for delays or failures in performance resulting from acts beyond its control. Such acts include, but are not limited to, acts of God, strikes, lockouts, riots, acts of war, epidemics, governmental regulations superimposed after the fact, fire, communication line failures, computer viruses, power failures, earthquakes, terrorist attacks or other disasters.
- (M) The Administrative Agent shall have no responsibility for interest or income on any funds held by it under the Loan Documents and any funds so held shall be held un-invested pending distribution thereof.
- (N) In no event shall the Administrative Agent be liable on account of any loss, including without limitation, any loss of principal or interest, or for any breakage fees or penalties in connection with the purchase or liquidation of any investment made in accordance with a written instruction of the Borrower or the Required Lenders.
- 9.4 <u>Reliance by Administrative Agent</u>. The Administrative Agent is entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed,

sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a DIP Term Loan that by its terms must be fulfilled to the satisfaction of a Lender, the Administrative Agent may presume that such condition is satisfactory to such Lender unless the Administrative Agent has received written notice to the contrary from such Lender prior to the making of such DIP Term Loans. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and is not liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

- 9.5 <u>Delegation of Duties</u>. The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent, which sub-agents may be Affiliates of the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this <u>Section 9</u> apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent. The Administrative Agent is not responsible for the negligence or misconduct of any sub-agents except to the extent that a court of competent jurisdiction determines in a final and nonappealable judgment that the Administrative Agent acted with gross negligence or willful misconduct in the selection of such sub-agents.
- Resignation of Administrative Agent. (A) The Administrative Agent may at any time give notice of its resignation to the Lenders and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders have the right, in consultation with the Borrower, to appoint a successor, which must be a bank or trust company, but in no event may be a Defaulting Lender. If no such successor shall have been so appointed by the Required Lenders and has accepted such appointment within 30 days after the retiring Administrative Agent gives notice of its resignation (or such earlier day as is agreed by the Required Lenders) (the "Resignation Effective Date"), then in any event such resignation becomes effective in accordance with such notice on the Resignation Effective Date and (i) the retiring Administrative Agent is discharged from its duties and obligations hereunder and under the other Loan Documents, except that until a successor Administrative Agent is appointed, any Collateral held by the Administrative Agent on behalf of the Secured Parties under any of the Loan Documents shall continue to be held by the resigning Administrative Agent as bailee until such time as a successor Administrative Agent is appointed, and (ii) except for any fees, expenses or indemnity payments owed to the retiring Administrative Agent, all payments, communications and determinations provided to be made by, to or through such Administrative Agent shall instead be made by or to each Lender directly, until such time as the Required Lenders appoint a successor Administrative Agent as provided in this Section 9.6(A). The Administrative Agent has the right, at the cost and expense of the Borrower, to petition a court of competent jurisdiction regarding the delivery of any Collateral it holds as bailee.
- (B) Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor succeeds to and becomes vested with all of the rights, powers, privileges and duties of the retiring Administrative Agent (other than any rights to fees, expenses and

indemnity payments owed to the retiring Administrative Agent) and the retiring Administrative Agent is discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not discharged earlier as provided in Section 9.6(A)). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Section 9 and Section 10.2 continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

- (C) Any Person into which the Administrative Agent may be merged or converted or with which it may be consolidated, or any Person resulting from any merger, conversion or consolidation to which the Administrative Agent is a party, or any Person succeeding to the corporate trust services business of the Administrative Agent is the successor of the Administrative Agent hereunder without the execution or filing of any paper with any Party or any further act on the part of any of the Parties.
- 9.7 Non-Reliance on Administrative Agent and Other Lenders. Each Lender acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it from time to time deems appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.
- 9.8 <u>Collateral and Guaranty Matters</u>. (A) The Lenders irrevocably authorize and direct the Administrative Agent, upon the written direction of the Required Lenders and at the expense of the Borrower:
 - (i) to release any Lien on any property granted to or held by the Administrative Agent under any Loan Document (a) upon termination of all DIP Term Loan Commitments and payment in full of all Obligations (other than contingent indemnification obligations), (b) that is sold or otherwise disposed of or to be sold or otherwise disposed of as part of or in connection with any sale or other disposition permitted under the Loan Documents or (c) subject to Section 10.3, if approved, authorized or ratified in writing by the Required Lenders;
 - (ii) to subordinate any Lien on any property granted to or held by the Administrative Agent under any Loan Document to the holder of any Lien on such property that is otherwise permitted by this Agreement; and
 - (iii) to release any Subsidiary Guarantor from its obligations under the Guaranty Agreement if such Person ceases to be a Subsidiary as a result of a transaction permitted under the Loan Documents.

Upon request by the Administrative Agent at any time, the Required Lenders will confirm in writing the Administrative Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Subsidiary Guarantor from its obligations under the Guaranty Agreement pursuant to this <u>Section 9.8</u>.

(B) The Administrative Agent is not responsible for, and has no duty to ascertain or inquire into any representation or warranty regarding, the existence, value or collectability of the Collateral, the existence, priority or perfection of the Administrative Agent's Lien thereon, or any certificate prepared by any Loan Party in connection therewith, nor shall the Administrative Agent be responsible or liable to the Lenders for any failure to monitor or maintain any portion of the Collateral.

9.9 Erroneous Payment.

(A) If the Administrative Agent (x) notifies a Lender or any other Secured Party, or any Person who has received funds on behalf of a Lender or other Secured Party (any such Lender, other Secured Party or other recipient (and each of their respective successors and assigns), a "Payment Recipient") that the Administrative Agent has determined in its reasonable discretion (whether or not after receipt of any notice under immediately succeeding clause (B)) that any funds (as set forth in such notice from the Administrative Agent) received by such Payment Recipient from the Administrative Agent or any of its Affiliates were erroneously or mistakenly transmitted to, or otherwise erroneously or mistakenly received by, such Payment Recipient (whether or not known to such Lender, Secured Party or other Payment Recipient on its behalf) (any such funds, whether transmitted or received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise, individually and collectively, an "Erroneous Payment") and (y) demands in writing the return of such Erroneous Payment (or a portion thereof), such Erroneous Payment shall at all times remain the property of the Administrative Agent pending its return or repayment as contemplated below in this Section 9.9 and held in trust for the benefit of the Administrative Agent, and such Lender, or such other Secured Party shall (or, with respect to any Payment Recipient who received such funds on its behalf, shall cause such Payment Recipient to) promptly, but in no event later than two Business Days thereafter (or such later date as the Administrative Agent may, in its sole discretion, specify in writing), return to the Administrative Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made, in same day funds (in the currency so received), together with interest thereon (except to the extent waived in writing by the Administrative Agent) in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Payment Recipient to the date such amount is repaid to the Administrative Agent in same day funds at the greater of the Federal Funds Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation from time to time in effect provided that any such interest shall only be payable by a Lender to the extent of, and in an amount equal to, interest actually accrued on the Erroneous Payment in favor of that Lender during such period. A

- notice of the Administrative Agent to any Payment Recipient under this clause (a) shall be conclusive, absent manifest error.
- (B) Without limiting immediately preceding Section 9.9(A), each Lender, each other Secured Party or any Person who has received funds on behalf of a Lender or other Secured Party (and each of their respective successors and assigns), agrees that if it receives a payment, prepayment or repayment (whether received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise) from the Administrative Agent (or any of its Affiliates) (x) that is in a different amount than, or on a different date from, that specified in this Agreement or in a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates), or (z) that such Lender or such other Secured Party, or other such recipient, otherwise becomes aware was transmitted, or received, in error or by mistake (in whole or in part), then in each such case:
 - (i) it acknowledges and agrees that (a) in the case of immediately preceding clauses (x) or (y), an error and mistake shall be presumed to have been made (absent written confirmation from the Administrative Agent to the contrary) or (b) an error and mistake has been made (in the case of immediately preceding clause (z)), in each case, with respect to such payment, prepayment or repayment; and
 - (ii) such Lender or such other Secured Party shall use commercially reasonable efforts to (and shall use commercially reasonable efforts to cause any other recipient that receives funds on its respective behalf to) promptly (and, in all events, within one Business Day of its knowledge of the occurrence of any of the circumstances described in immediately preceding clauses (x), (y) and (z)) notify the Administrative Agent of its receipt of such payment, prepayment or repayment, the details thereof (in reasonable detail) and that it is so notifying the Administrative Agent pursuant to this Section 9.9(B).

For the avoidance of doubt, the failure to deliver a notice to the Administrative Agent pursuant to this Section 9.9(B) shall not have any effect on a Payment Recipient's obligations pursuant to Section 9.9(A) or on whether or not an Erroneous Payment has been made.

(C) Each Lender or other Secured Party hereby authorizes the Administrative Agent to set off, net and apply any and all amounts at any time owing to such Lender or such other Secured Party under any Loan Document, or otherwise payable or distributable by the Administrative Agent to such Lender or such other Secured Party under any Loan Document with respect to any payment of principal, interest, fees or other amounts, against any amount that the Administrative Agent has demanded to be returned under the preceding Section 9.9(A).

- (D) (i) In the event that an Erroneous Payment (or portion thereof) is not recovered by the Administrative Agent for any reason, after demand therefor in accordance with preceding clause (A), from any Lender that has received such Erroneous Payment (or portion thereof) (and/or from any Payment Recipient who received such Erroneous Payment (or portion thereof) on its respective behalf) (such unrecovered amount, an "Erroneous Payment Return Deficiency"), upon not less than five (5) Business Days' notice from the Administrative Agent to such Lender at any time, then effective thereupon (with the consideration therefor being acknowledged by the parties hereto), (a) such Lender shall be deemed to have assigned its Loans (but not its Commitments) with respect to which such Erroneous Payment was made (the "Erroneous Payment Impacted Class") in an amount equal to the Erroneous Payment Return Deficiency (or such lesser amount as the Administrative Agent may specify) (such assignment of the Loans (but not Commitments) of the Erroneous Payment Impacted Class, the "Erroneous Payment Deficiency Assignment") (on a cashless basis and such amount calculated at par plus any accrued and unpaid interest (with the assignment fee to be waived by the Administrative Agent in such instance)), and is hereby (together with the Borrower) deemed to execute and deliver an Assignment and Acceptance (or, to the extent applicable, an agreement incorporating an Assignment and Acceptance by reference pursuant to an approved electronic platform as to which the Administrative Agent and such parties are participants) with respect to such Erroneous Payment Deficiency Assignment, and such Lender shall deliver any notes evidencing such Loans to the Borrower or the Administrative Agent (but the failure of such Person to deliver any such notes shall not affect the effectiveness of the foregoing assignment), (b) the Administrative Agent as the assignee Lender shall be deemed to have acquired the Erroneous Payment Deficiency Assignment, (c) upon such deemed acquisition, the Administrative Agent as the assignee Lender shall become a Lender, as applicable, hereunder with respect to such Erroneous Payment Deficiency Assignment and the assigning Lender shall cease to be a Lender, as applicable, hereunder with respect to such Erroneous Payment Deficiency Assignment, excluding, for the avoidance of doubt, its obligations under the indemnification provisions of this Agreement and its applicable Commitments which shall survive as to such assigning Lender, (d) the Administrative Agent and the Borrower shall each be deemed to have waived any consents required under this Agreement to any such Erroneous Payment Deficiency Assignment, and (e) the Administrative Agent will reflect in the Register its ownership interest in the Loans subject to the Erroneous Payment Deficiency Assignment. For the avoidance of doubt, no Erroneous Payment Deficiency Assignment will reduce the Commitments of any Lender and such Commitments shall remain available in accordance with the terms of this Agreement.
 - (ii) Subject to Section 10.04 (but excluding, in all events, any assignment consent or approval requirements (whether from the Borrower or otherwise)), the Administrative Agent may, in its discretion, sell any Loans acquired pursuant to an Erroneous Payment Deficiency Assignment and upon receipt of the proceeds of such sale, the Erroneous Payment Return Deficiency owing by the applicable Lender shall be reduced by the net proceeds of the sale of such Loan (or portion

thereof), and the Administrative Agent shall retain all other rights, remedies and claims against such Lender (and/or against any recipient that receives funds on its respective behalf). In addition, an Erroneous Payment Return Deficiency owing by the applicable Lender (x) shall be reduced by the proceeds of prepayments or repayments of principal and interest, or other distribution in respect of principal and interest, received by the Administrative Agent on or with respect to any such Loans acquired from such Lender pursuant to an Erroneous Payment Deficiency Assignment (to the extent that any such Loans are then owned by the Administrative Agent) and (y) may, in the sole discretion of the Administrative Agent, be reduced by any amount specified by the Administrative Agent in writing to the applicable Lender from time to time.

- The parties hereto agree that irrespective of whether the Administrative Agent may (E) be equitably subrogated, in the event that an Erroneous Payment (or portion thereof) is not recovered from any Payment Recipient that has received such Erroneous Payment (or portion thereof) for any reason, the Administrative Agent shall be subrogated to all the rights and interests of such Payment Recipient (and, in the case of any Payment Recipient who has received funds on behalf of a Lender or other Secured Party, to the rights and interests of such Lender or such other Secured Party, as the case may be) under the Loan Documents with respect to such amount (the "Erroneous Payment Subrogation Rights") (provided that the Obligations under the Loan Documents in respect of the Erroneous Payment Subrogation Rights shall not be duplicative of such Obligations in respect of Loans that have been assigned to the Administrative Agent under an Erroneous Payment Deficiency Assignment); provided that this Section 9.9(E) shall not be interpreted to increase (or accelerate the due date for), or have the effect of increasing (or accelerating the due date for), the Obligations of the Borrower relative to the amount (and/or timing for payment) of the Obligations that would have been payable had such Erroneous Payment not been made by the Administrative Agent; provided, further, that for the avoidance of doubt, the immediately preceding clause shall not apply to the extent any such Erroneous Payment is, and solely with respect to the amount of such Erroneous Payment that is, comprised of funds received by the Administrative Agent from the Borrower or from the proceeds of Collateral.
- (F) To the extent permitted by applicable law, no Payment Recipient shall assert any right or claim to an Erroneous Payment, and hereby waives, and is deemed to waive, any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by the Administrative Agent for the return of any Erroneous Payment received, including, without limitation, any defense based on "discharge for value" or any similar doctrine.
- (G) Each party's obligations, agreements and waivers under this <u>Section 9.9</u> shall survive the resignation or replacement of the Administrative Agent, any transfer of rights or obligations by, or the replacement of, a Lender, the termination of the Commitments and/or the repayment, satisfaction or discharge of all Obligations (or any portion thereof) under any Loan Document.

SECTION 10. MISCELLANEOUS:

- 10.1 <u>Governing Law.</u> This Agreement and the other Loan Documents and any claims, controversy, dispute or cause of action (whether in contract or tort or otherwise) based upon, arising out of or relating to this Agreement or any other Loan Document (except, as to any other Loan Document, as expressly set forth therein) and the transactions contemplated hereby and thereby are governed by, and construed in accordance with, the law of the State of Delaware.
 - 10.2 Expenses; Indemnity; Damage Waiver.
 - (A) Costs and Expenses. The Borrower shall pay (i) all reasonable out of pocket expenses incurred by the Administrative Agent and the Lenders (including the reasonable fees, charges and disbursements of counsel for the Administrative Agent and for the Lenders), and shall pay all fees and time charges and disbursements for attorneys who may be employees of the Administrative Agent, in connection with the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents, or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby are consummated) and (ii) all out of pocket expenses incurred by the Administrative Agent or any Lender (including the fees, charges and disbursements of any counsel for the Administrative Agent or for any Lender), and shall pay all fees and time charges for attorneys who may be employees of the Administrative Agent or any Lender, in connection with the enforcement or protection of its rights (a) in connection with this Agreement and the other Loan Documents, including its rights under this Section 10.2, or (b) in connection with the DIP Term Loans and Roll-Up Loans made hereunder, including all such out of pocket expenses incurred during any workout, restructuring or negotiations in respect of such DIP Term Loans or Roll-Up Loans.
 - (B) Indemnification by the Borrower. The Borrower shall indemnify the Administrative Agent (and any sub-agent thereof), each Lender and each Related Party of any of the foregoing Persons (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all fees, losses, claims, damages, liabilities and related expenses (including the reasonable fees, charges and disbursements of any counsel for any Indemnitee and including out-of-pocket attorneys' fees and expenses and court costs incurred by any Indemnitee in enforcing the indemnity), and shall indemnify and hold harmless each Indemnitee from all reasonable fees and time charges and disbursements for attorneys who may be employees of any Indemnitee, incurred by any Indemnitee or asserted against any Indemnitee by any Person (including the Borrower or any other Loan Party) arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance by the Parties of their respective obligations hereunder or thereunder or the consummation of the transactions contemplated hereby or thereby, (ii) any DIP Term Loan or the use or proposed use of the proceeds therefrom, (iii) any actual or alleged presence or Release of Hazardous Materials on or from any property owned or operated by

the Borrower or any of its Subsidiaries, or any Environmental Liability related in any way to the Borrower or any of its Subsidiaries, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or a Lender or by the Borrower or any other Loan Party, and regardless of whether any Indemnitee is a party thereto; <u>provided</u> that such indemnity is not, as to any Indemnitee, available to the extent that such losses, claims, damages, liabilities or related expenses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee. This <u>Section 10.2(B)</u> does not apply with respect to Taxes other than any Taxes that represent losses, claims, damages, or similar costs arising from any non-Tax claim.

- (C) Reimbursement by Lenders. To the extent that the Borrower for any reason fails to indefeasibly pay any amount required under Sections 3.7(D), 10.2(A) or 10.2(B) to be paid by it to the Administrative Agent (or any sub-agent thereof) or any Related Party of the Administrative Agent, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent) or such Related Party, as the case may be, such Lender's pro rata share (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought based on each Lender's Commitment Percentage at such time) of such unpaid amount (including any such unpaid amount in respect of a claim asserted by such Lender); provided the unreimbursed expense or indemnified loss, fee, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent) or against any Related Party acting for the Administrative Agent (or any such sub-agent) in connection with such capacity.
- Waiver of Consequential Damages. To the fullest extent permitted by Applicable (D) Law, the Borrower shall not assert, and hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any DIP Term Loan or the use of the proceeds thereof. No Indemnitee referred to in Section 10.2(B) will be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby other than for direct and actual damages resulting from the gross negligence or willful misconduct of such Indemnitee as determined by a final and nonappealable judgment of a court of competent jurisdiction.
- (E) <u>Payments</u>. All amounts due under this <u>Section 10.2</u> are payable promptly after demand therefor.

- (F) <u>Survival</u>. Each Party's obligations under this <u>Section 10.2</u> survive the resignation or replacement of the Administrative Agent hereunder and the termination of the Loan Documents and payment of the obligations hereunder.
- 10.3 <u>Amendments, Waivers and Consents</u>. Except as set forth below or as specifically provided in any Loan Document, any term, covenant, agreement or condition of this Agreement or any of the other Loan Documents may be amended or waived by the Lenders, and any consent given by the Lenders, if, but only if, such amendment, waiver or consent is in writing signed by the Required Lenders (or by the Administrative Agent at the written direction of the Required Lenders) and delivered to the Administrative Agent and, in the case of an amendment, signed by the Borrower; provided no amendment, waiver or consent may:
 - (A) increase the DIP Term Loan Commitment of any Lender (or reinstate any DIP Term Loan Commitment terminated pursuant to <u>Section 8.2</u>) or the amount of DIP Term Loans of any Lender, in any case, without the written consent of such Lender;
 - (B) waive, extend or postpone any date fixed by this Agreement or any other Loan Document for any payment (it being understood that a waiver of a mandatory prepayment under Section 2.5 only requires the consent of the Required Lenders) of principal, interest, fees or other amounts due to the Lenders (or any of them) or any mandatory reduction of the DIP Term Loan Commitment hereunder or under any other Loan Document without the written consent of each Lender directly and adversely affected thereby;
 - (C) reduce the principal of, or the rate of interest specified herein on, any DIP Term Loan or (subject to <u>Section 10.3(I)</u>) any fees or other amounts payable hereunder or under any other Loan Document without the written consent of each Lender directly and adversely affected thereby; <u>provided</u> only the consent of the Required Lenders is necessary to waive any obligation of the Borrower to pay interest at the rate set forth in <u>Section 3.2(C)</u> during the continuance of an Event of Default;
 - (D) change Section 3.5 or Section 8.5 in a manner that would alter the pro rata sharing of payments or order of application required thereby without the written consent of each Lender directly and adversely affected thereby;
 - (E) change any provision of this <u>Section 10.3</u> or reduce the percentages specified in the definition of "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender directly affected thereby;
 - (F) consent to the assignment or transfer by any Loan Party of such Loan Party's rights and obligations under any Loan Document to which it is a party (except as permitted pursuant to <u>Section 7.12</u>), in each case, without the written consent of each Lender;
 - (G) release the Guarantor from the Guaranty Agreement (other than as authorized in Section 9.8), without the written consent of each Lender;

- (H) release all or substantially all of the Collateral or release any Security Document (other than as authorized in <u>Section 9.8</u> or as otherwise specifically permitted or contemplated in this Agreement or the applicable Security Document) without the written consent of each Lender; or
- (I) affect the rights or duties of the Administrative Agent under this Agreement or any other Loan Document without the written consent of the Administrative Agent in addition to the Lenders required hereunder;

Notwithstanding anything to the contrary herein, no Defaulting Lender has any right to approve or disapprove any amendment, waiver or consent hereunder, except that the DIP Term Loan Commitment of such Lender may not be increased or extended without the consent of such Lender.

10.4 Successors and Assigns.

- (A) Successors and Assigns Generally. The provisions of this Agreement are binding upon and inure to the benefit of the Parties and their respective successors and assigns permitted hereby, except that neither the Borrower nor any other Loan Party may assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of each Lender, and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of Section 10.4(B), (ii) by way of participation in accordance with the provisions of Section 10.4(D) or (iii) by way of pledge or assignment of a security interest subject to the restrictions of Section 10.4(E) (and any other attempted assignment or transfer by any Party hereto is null and void). Nothing in this Agreement, expressed or implied, may be construed to confer upon any Person (other than the Parties, their respective successors and assigns permitted hereby, Participants to the extent provided in Section 10.4(D) and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (B) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its DIP Term Loan Commitment and the DIP Term Loans and Roll-Up Loans at the time owing to it); <u>provided</u> any such assignment is subject to the following conditions:

(i) Minimum Amounts.

(a) in the case of an assignment of the entire remaining amount of the assigning Lender's DIP Term Loan Commitment or the DIP Term Loans or Roll-Up Loans at the time owing to it or contemporaneous assignments to related Approved Funds (determined after giving effect to such assignments) that equal at least the amount specified in clause (b) below in the aggregate or

- in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and
- (b) in any case not described in clause (a) above, the aggregate amount of the DIP Term Loan Commitment (which for this purpose includes DIP Term Loans outstanding thereunder) or, if the applicable DIP Term Loan Commitment is not then in effect, the principal outstanding balance of the DIP Term Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent) is not less than \$1,000,000 unless each of the Required Lenders and, so long as no Default or Event of Default exists, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed).
- (ii) <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the DIP Term Loan or the DIP Term Loan Commitment assigned.
- (iii) Required Consents. No consent is required for any assignment except to the extent required by Section 10.4(B)(i)(b) and, in addition:
 - (a) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) is required unless (1) a Default or Event of Default exists at the time of such assignment or (2) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund; <u>provided</u> the Borrower is deemed to have consented to any such assignment unless it has objected thereto by written notice to the Administrative Agent within ten Business Days after having received notice thereof; and
 - (b) the consent of the Required Lenders (or the Administrative Agent at the direction of the Required Lenders, with such consent not to be unreasonably withheld or delayed) is required for assignments in respect of any DIP Term Loans or Roll-Up Loans to a Person who is not a Lender, an Affiliate of a Lender or an Approved Fund.
- (iv) <u>Assignment and Assumption</u>. The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee of \$3,500; <u>provided</u> that the Administrative Agent may, in its sole discretion, elect to waive such processing and recording fee in the case of any assignment. The assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire.
- (v) <u>No Assignment to Certain Persons</u>. No such assignment may be made to (1) the Borrower or any of the Borrower's Affiliates or Subsidiaries or (2) any Defaulting Lender or any of its Subsidiaries, or any Person who, upon becoming a Lender hereunder, would constitute a Defaulting Lender or a Subsidiary thereof.

- (vi) <u>No Assignment to Natural Persons</u>. No such assignment may be made to a natural Person (or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person).
- Certain Additional Payments. In connection with any assignment of rights and (vii) obligations of any Defaulting Lender hereunder, no such assignment is effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower, the applicable pro rata share of DIP Term Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (1) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent and each Lender hereunder (and interest accrued thereon) and (2) acquire (and fund as appropriate) its full pro rata share of all DIP Term Loans in accordance with its Commitment Percentage. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder becomes effective under Applicable Law without compliance with the provisions of this Section 10.4(B)(iv), then the assignee of such interest is deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to Section 10.4(C), from and after the effective date specified in each Assignment and Assumption, the assignee thereunder is a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, has the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder is, to the extent of the interest assigned by such Assignment and Assumption, released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender ceases to be a Party) but continues to be entitled to the benefits of Sections 3.6, 3.7 and 10.2 with respect to facts and circumstances occurring prior to the effective date of such assignment; provided except to the extent otherwise expressly agreed by the affected parties, no assignment by a Defaulting Lender constitutes a waiver or release of any claim of any Party arising from that Lender's having been a Defaulting Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this paragraph is treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 10.4(D).

(C) Register. The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the DIP Term Loan Commitments of, and principal amounts (and stated interest) of the DIP Term Loans and Roll-Up Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register are conclusive absent manifest error, and the Borrower, the Administrative Agent

- and the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (D) Participations. Any Lender may at any time, without the consent of, or notice to, the Borrower or the Administrative Agent, sell participations to any Person (other than a natural Person, or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person, or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights or obligations under this Agreement (including all or a portion of its DIP Term Loan Commitment or the DIP Term Loans or Roll-Up Loans owing to it); provided (i) such Lender's obligations under this Agreement remain unchanged, (ii) such Lender remains solely responsible to the other Parties for the performance of such obligations and (iii) the Borrower, the Administrative Agent and Lenders continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. For the avoidance of doubt, each Lender is responsible for the indemnity under Section 10.3(C) with respect to any payments made by such Lender to its Participants.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender retains the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver described in Section 10.3 that affects such Participant. The Borrower agrees that each Participant is entitled to the benefits of Sections 3.6 and 3.7 (subject to the requirements and limitations therein (it being understood that the documentation required under Section 3.7 shall be delivered to the participating Lender)) to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to Section 10.4(B); provided such Participant (A) agrees to be subject to the provisions of Section 3.8 as if it were an assignee under Section 10.4(B) and (B) is not entitled to receive any greater payment under Sections 3.6 or 3.7, with respect to any participation, than its participating Lender would have been entitled to receive, except to the extent such entitlement to receive a greater payment results from a Change in Law that occurs after the Participant acquired the applicable participation. Each Lender that sells a participation agrees, at the Borrower's request and expense, to use reasonable efforts to cooperate with the Borrower to effectuate the provisions of Section 3.8(B) with respect to any Participant. To the extent permitted by law, each Participant also is entitled to the benefits of Section 8.3 as though it were a Lender; provided such Participant agrees to be subject to Section 3.5 as though it were a Lender. Each Lender that sells a participation shall, acting solely for this purpose as a nonfiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under the Loan Documents (the "Participant Register"); provided no Lender has any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury

Regulations. The entries in the Participant Register are conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, the Administrative Agent (in its capacity as Administrative Agent) has no responsibility for maintaining a Participant Register.

- (E) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; <u>provided</u> no such pledge or assignment releases such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a Party.
- 10.5 Treatment of Certain Information; Confidentiality. Each of the Administrative Agent and the Lenders shall maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its Related Parties (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent required or requested by any regulatory authority purporting to have jurisdiction over such Person or its Related Parties (including any self-regulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent required by Applicable Laws or regulations or by any subpoena or similar legal process, (d) to any other Party, (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section 10.5, to any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights and obligations under this Agreement, (g) on a confidential basis to (i) any rating agency in connection with rating the Borrower or their Subsidiaries or the DIP Term Loans or Roll-Up Loans or (ii) the CUSIP Service Bureau or any similar agency in connection with the issuance and monitoring of CUSIP numbers with respect to the DIP Term Loans or Roll-Up Loans, (h) with the consent of the Borrower or (i) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section 10.5 or (ii) becomes available to the Administrative Agent, any Lender or any of their respective Affiliates on a nonconfidential basis from a source other than the Loan Parties. In addition, the Administrative Agent and the Lenders may disclose the existence of this Agreement and information about this Agreement to market data collectors, similar service providers to the lending industry and service providers to the Agents and the Lenders in connection with the administration of this Agreement, the other Loan Documents and the DIP Term Loan Commitments.

For purposes of this Section, "Information" means all information received from the Borrower or any of its Subsidiaries relating to the Borrower or any of its Subsidiaries or any of their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by the Borrower or any of its Subsidiaries; provided, in the case of information received from the Borrower or any of its Subsidiaries after the Closing Date, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section 10.5 is considered to have complied with its obligation to do so if such Person has exercised the same

degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

- 10.6 Submission to Jurisdiction; Venue; Waiver of Jury Trial.
- (A) The Borrower and each other Loan Party irrevocably and Jurisdiction. unconditionally agrees that it shall not commence any action, litigation or proceeding of any kind or description, whether in law or equity, whether in contract or in tort or otherwise, against the Administrative Agent, any Lender or any Related Party of the foregoing in any way relating to this Agreement or any other Loan Document or the transactions relating hereto or thereto, in any forum other than the Bankruptcy Court, or, if the Bankruptcy Court lacks or cannot exercise subject matter jurisdiction, the courts of the State of Delaware sitting in New Castle County, and of the United States District Court of the District of Delaware, and any appellate court from any thereof, and each of the Parties irrevocably and unconditionally submits to the jurisdiction of such courts and agrees that all claims in respect of any such action, litigation or proceeding may be heard and determined in such Delaware court or, to the fullest extent permitted by Applicable Law, in such federal court. Each of the Parties agrees that a final judgment in any such action, litigation or proceeding is conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Nothing in this Agreement or in any other Loan Document affects any right that the Administrative Agent or any Lender may otherwise have to bring any action or proceeding relating to this Agreement or any other Loan Document against the Borrower or any other Loan Party or its properties in the courts of any jurisdiction.
- (B) <u>Waiver of Venue</u>. The Borrower and each other Loan Party irrevocably and unconditionally waives, to the fullest extent permitted by Applicable Law, any objection that it may now or hereafter have to the laying of venue of any action or proceeding arising out of or relating to this Agreement or any other Loan Document in any court referred to in <u>Section 10.6(A)</u>. Each of the Parties irrevocably waives, to the fullest extent permitted by Applicable Law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court.
- (C) <u>Service of Process</u>. Each Party other than the Administrative Agent irrevocably consents to service of process in the manner provided for notices in <u>Section 10.7</u>. Nothing in this Agreement affects the right of any Party to serve process in any other manner permitted by Applicable Law.
- (D) WAIVER OF JURY TRIAL. EACH PARTY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY (i) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF

ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (ii) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10.6(D).

10.7 Notices.

- (A) <u>Notices Generally</u>. Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in <u>Section 10.7(B)</u>) below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by facsimile as follows:
- (i) if to the Borrower or any other Loan Party, to it at the addresses for it as set forth on Annex C;
- (ii) if to the Administrative Agent, to it at the addresses for it as set forth on Annex C;
- (iii) if to a Lender party to this Agreement on the Closing Date, to it at the addresses for it as set forth on Annex C;
- (iv) if to a Lender party to this Agreement after the Closing Date, to it at its address (or facsimile number) set forth in its Administrative Questionnaire;
- (v) if to the United States Trustee appointed in the Bankruptcy Case, to it at its address (or facsimile number) which will be provided at the outset of the Bankruptcy Case; and
- (vi) if to any statutory committee appointed in the Bankruptcy Case, to it at its address (or facsimile number) to be provided upon appointment.

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, are deemed to have been given when received and notices sent by facsimile are deemed to have been given when sent (except that, if not given during normal business hours for the recipient, is deemed to have been given at the opening of business on the next business day for the recipient). Notices delivered through electronic communications, to the extent provided in <u>Section 10.7(B)</u>, are effective as provided in <u>Section 10.7(B)</u>.

(B) <u>Electronic Communications</u>. Notices and other communications to the Lenders hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Administrative Agent; <u>provided</u> the foregoing does not apply to notices to any Lender pursuant to <u>Section 2.2</u> if such Lender has notified the Administrative Agent that it is incapable of receiving notices under such section by electronic communication. The Administrative Agent or the Borrower may, in its discretion,

agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it; <u>provided</u> approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (a) notices and other communications sent to an e-mail address are deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement) and (b) notices or communications posted to an Internet or intranet website are deemed received upon the deemed receipt by the intended recipient, at its e-mail address as described in clause (a) above, of notification that such notice or communication is available and identifying the website address therefor; <u>provided</u>, for both clauses (a) and (b) above, if such notice, email or other communication is not sent during the normal business hours of the recipient, such notice or communication is deemed to have been sent at the opening of business on the next business day for the recipient.

- (C) <u>Change of Address</u>. Any Party may change its address or facsimile number for notices and other communications hereunder by notice to the other Parties.
- (D) Reserved.
- 10.8 Reinstatement of Obligations. To the extent any Loan Party makes a payment or payments to the Administrative Agent for the ratable benefit of the Lenders or the Administrative Agent receives any payment or proceeds of the Collateral which payments or proceeds or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid to a trustee, receiver or any other party under any Debtor Relief Law, other Applicable Law or equitable cause, then, to the extent of such payment or proceeds repaid, the Obligations or part thereof intended to be satisfied are revived and continued in full force and effect as if such payment or proceeds had not been received by the Administrative Agent.
- 10.9 <u>Severability</u>. Any provision of this Agreement or any other Loan Document that is prohibited or unenforceable in any jurisdiction is, as to such jurisdiction, ineffective only to the extent of such prohibition or unenforceability without invalidating the remainder of such provision or the remaining provisions hereof or thereof or affecting the validity or enforceability of such provision in any other jurisdiction.
- 10.10 <u>Integration</u>. This Agreement and the other Loan Documents, and any separate letter agreements with respect to fees payable to the Administrative Agent, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof.
- 10.11 No Implied Waivers. No failure to exercise and no delay in exercising any right or remedy under this Agreement operates as a waiver thereof. No single or partial exercise of any right or remedy under this Agreement, or any abandonment or discontinuance thereof, precludes any other or further exercise thereof or the exercise of any other right or remedy. No waiver or consent under this Agreement is applicable to any events, acts or circumstances except those specifically covered thereby.

- 10.12 <u>USA PATRIOT Act</u>. The Administrative Agent and each Lender hereby notifies the Borrower that pursuant to the requirements of the PATRIOT Act, each of them is required to obtain, verify and record information that identifies each Loan Party, which information includes the name and address of each Loan Party and other information that will allow such Lender to identify each Loan Party in accordance with the PATRIOT Act.
- by different Parties in different counterparts), each of which constitutes an original, but all of which when taken together constitute a single contract. Except as provided in Section 4.1, this Agreement becomes effective when it has been executed by the Administrative Agent and when the Administrative Agent has received counterparts hereof that, when taken together, bear the signatures of each of the other Parties. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or in electronic (i.e., "pdf" or "tif") format is effective as delivery of a manually executed counterpart of this Agreement. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption are deemed to include electronic signatures or the keeping of records in electronic form, each of which is of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any Applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws or regulations based on the Uniform Electronic Transactions Act.

(Signature pages follow)

Case 25-10979-JKS Doc 50-1 Filed 06/02/25 Page 136 of 298

The Parties have executed and delivered this Agreement as of the date first above written

Borrower:
VHS HOLDCO, INC.
By:
Name:
Title:
Guarantor:
SHOOSMITH BROS., INC.
By:
Name:
Title:

Administrative Agent:
VOLUNTEER ENTERPRISES, LLC, as Administrative Agent
By: Name:
Title:

Case 25-10979-JKS Doc 50-1 Filed 06/02/25 Page 138 of 298

<u>Lenders</u> :
VOLUNTEER ENTEPRISES, LLC, as a Lender
By:Name:
Title:

ANNEX A

Rules of Construction

- 1. <u>Definitions</u>. As used in this Agreement, the following terms have the following meanings:
 - "Acceptable Plan" means a bankruptcy plan acceptable to the Required Lenders.
- "Acceptable Stalking Horse Purchase Agreement" means the Bankruptcy Court shall have entered an order (in form and substance reasonably satisfactory to the DIP Lender) authorizing and approving a stalking horse purchase agreement for all or substantially all of the Debtors' assets(s) that (a) is executed, effective, and binding on all parties thereto and (b) either (x) is approved by the DIP Lender in its reasonable discretion or (y) (1) would generate cash proceeds sufficient for the payment in full in cash of the Obligations (pursuant to a signed commitment acceptable to the DIP Lender in its reasonable discretion to lend from a recognized lender or another source of funding acceptable to the DIP Lender in its reasonable discretion) upon the closing date of such sale, (2) closes on or before the eighty-fifth (85th) day after the Petition Date, and (3) requires, pursuant to an order of the Bankruptcy Court, the payment in full in cash of the Obligations as set forth in clause (1).
- "Account Control Agreement" means an agreement which grants the Administrative Agent "control" as defined in the Uniform Commercial Code in effect in the applicable jurisdiction over the applicable Deposit Account.
- "Administrative Agent" has the meaning set forth for such term in the introduction to this Agreement.
- "<u>Administrative Agent's Office</u>" means the office of the Administrative Agent specified in or determined in accordance with the provisions of <u>Section 10.7</u>.
- "<u>Administrative Questionnaire</u>" means an administrative questionnaire in a form supplied by the Administrative Agent.
- "Affiliate" means, with respect to a specified Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.
 - "Agreement" has the meaning set forth for such term in the introduction to this Agreement.
- "Anti-Corruption Laws" means all laws, rules, and regulations of any jurisdiction applicable to the Borrower or its Subsidiaries from time to time concerning or relating to terrorism, trade sanctions programs and embargoes, import/export licensing, money laundering, bribery or corruption, including the United States Foreign Corrupt Practices Act of 1977 and the rules and regulations thereunder.

"Applicable Law" means all applicable provisions of constitutions, laws, statutes, ordinances, rules, treaties, regulations, permits, licenses, approvals, interpretations and orders of Governmental Authorities and all orders and decrees of all courts and arbitrators.

"Approved Counterparty" means (a) a Lender or any Affiliate of a Lender and (b) any other Person whose issuer rating or long term senior unsecured debt rating, at the time the Swap Agreement is entered into, is A-/A3 by S&P or Moody's (or their equivalent) or higher (or whose obligations under the applicable Swap Agreement are guaranteed by an Affiliate or other credit support provider of such Person meeting such minimum rating standards at the time the Swap Agreement is entered into).

"Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"Assignment and Assumption" means an assignment and assumption entered into by a Lender and an Eligible Assignee (with the consent of any party whose consent is required by Section 10.4), and accepted by the Administrative Agent, in substantially the form of Exhibit D or any other form approved by the Administrative Agent.

"Asset Disposition" means the sale, transfer, assignment, conveyance, farm-out, license, lease or other disposition of any Property (including any disposition of Equity Interests) by any Loan Party or any Subsidiary thereof (or the granting of any option or other right to do any of the foregoing). The term "Asset Disposition" does not include (a) the sale of inventory in the ordinary course of business, (b) the transfer of assets to the Borrower or any Subsidiary Guarantor pursuant to any other transaction permitted by this Agreement, (c) the write-off, discount, sale or other disposition of defaulted or past-due receivables and similar obligations in the ordinary course of business and not undertaken as part of an accounts receivable financing transaction, (d) the disposition of any Swap Agreement, and (e) dispositions of Investments in cash and Cash Equivalents, and (f) the transfer by any Loan Party of its assets to any other Loan Party.

"Bankruptcy Code" means Chapter 11, 11 U.S.C. 101 et seq.

"Bankruptcy Court" means the United States Bankruptcy Court for the District of Delaware having jurisdiction over the Chapter 11 Cases.

"Board" means the Board of Governors of the Federal Reserve System of the United States or any successor Governmental Authority.

"Borrower" has the meaning set forth for such term in the introduction to this Agreement.

"Budget" means the latest thirteen-week (13-week) detailed cash projection and statement of sources and uses of the Borrower, including anticipated uses of the DIP Term Loans, to be delivered prior to the funding of the DIP Term Loans and each week thereafter, and which shall be approved by the Lenders of the DIP Term Loans each week, in their sole discretion; provided, that unused variance amounts in any Budget for a prior period shall not be carried forward on a line item basis without the approval of the Lenders of the DIP Term Loans; provided, further, that such amounts shall be counted for purposes of calculating any variances in the Budget hereunder.

"Budget Period" means the period commencing on the date that the Interim DIP Financing Order is entered and ending on the Effective Date.

"Business Day" means any day other than a Saturday, Sunday or legal holiday on which banks in New York City are closed.

"Capital Leases" means, in respect of any Person, all leases which shall have been, or should have been, in accordance with GAAP, recorded as capital leases on the balance sheet of the Person liable (whether contingent or otherwise) for the payment of rent thereunder.

"Carve-Out" has the meaning assigned to such term in the Interim DIP Financing Order or the Final DIP Financing Order, as applicable.

"<u>Cash Equivalents</u>" means Investments of the type described in the definition of Permitted Investments.

"Change in Control" means an event or series of events by which:

- (A) at any time, the Parent fails to own all of the Equity Interests of the Borrower; or
- (B) (i) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act, but excluding any employee benefit plan of such person or its Subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) other than a Permitted Holder becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a "person" or "group" shall be deemed to have "beneficial ownership" of all Equity Interests that such "person" or "group" has the right to acquire, whether such right is exercisable immediately or only after the passage of time (such right, an "option right")), directly or indirectly, of more than 50% of the Equity Interests of the Parent entitled to vote in the election of members of the board of directors (or equivalent governing body) of the Parent or (ii) a majority of the members of the board of directors (or other equivalent governing body) of the Parent do not constitute Continuing Directors; or
- (C) there has occurred under any indenture or other instrument evidencing any Debt or Equity Interests in excess of \$1,000,000 any "change in control" or similar provision (as set forth in the indenture, agreement or other evidence of such Debt) obligating the Borrower or any of its Subsidiaries to repurchase, redeem or repay all or any part of the Debt or Equity Interests provided for therein.

"Change in Law" means the occurrence, after the Closing Date, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty; (b) any change in any law, rule, regulation or treaty or in the administration, interpretation, implementation or application thereof by any Governmental Authority; or (c) the making or issuance of any request, rule, guideline or directive (whether or not having the force of law) by any Governmental Authority; provided notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (ii) all requests, rules, guidelines or directives promulgated by the

Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, are in each case deemed to be a "Change in Law," regardless of the date enacted, adopted or issued.

"Chapter 11 Cases" has the meaning set forth for such term in the introduction to this Agreement.

"Closing Date" means the date on which the conditions specified in Section 4.1 are satisfied (or waived in writing by the Required Lenders in accordance with Section 10.3).

"Code" means the Internal Revenue Code of 1986.

"Collateral" means the collateral security for the Obligations pledged or granted pursuant to the Security Documents.

"Commitment Percentage" means, with respect to any Lender at any time, the percentage of the total Term DIP Term Loan Commitments of all the Lenders represented by such Lender's DIP Term Loan Commitment. If the DIP Term Loan Commitments have terminated or expired, the Commitment Percentage means, with respect to any Lender at any time, the percentage of the total outstanding principal balance of the DIP Term Loans represented by the outstanding principal balance of such Lender's DIP Term Loans. The Commitment Percentage of each Lender in respect of the DIP Term Loans as of the Closing Date is set forth opposite the name of such Lender on Annex B.

"Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"Consolidated" (or "consolidated") or "Consolidating" (or "consolidating") means, when used with reference to any financial term in this Agreement, the aggregate for two or more Persons of the amounts signified by such term for all such Persons determined on a consolidated (or consolidating) basis in accordance with GAAP, applied on a consistent basis.

"Continuing Directors" means the directors (or equivalent governing body) of the Parent on the Closing Date and each other director (or equivalent) of the Parent if, in each case, such other Person's nomination for election to the board of directors (or equivalent governing body) of the Parent is approved by more than 50% of the then Continuing Directors.

"Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. For the purposes of this definition, and without limiting the generality of the foregoing, any Person that owns directly or indirectly 10% or more of the Equity Interests having ordinary voting power for the election of the directors or other governing body of a Person (other than as a limited partner of such other Person) will be deemed to "control" such other Person. "Controlling" and "Controlled" have meanings correlative thereto.

"Covered Entity" means (a) each Loan Party, any other Persons that guaranty the Obligations and/or pledge collateral to secure the Obligations, (b) each Person that, directly or indirectly, is in control of a Person described in clause (a) above, including without limitation, Parent, and (c) all brokers or other agents of any Loan Party acting in any capacity in connection with this Agreement. For purposes of this definition, control of a Person shall mean the direct or indirect (x) ownership of, or power to vote, 25% or more of the issued and outstanding Equity Interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for such Person, or (y) power to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.

"Debt" means, for any Person, the sum of the following (without duplication): (a) all obligations of such Person for borrowed money or evidenced by bonds, bankers' acceptances, debentures, notes or other similar instruments; (b) all obligations of such Person (whether contingent or otherwise) in respect of letters of credit, surety or other bonds and similar instruments; (c) all accounts payable and all accrued expenses, liabilities or other obligations of such Person to pay the deferred purchase price of Property or services; (d) all obligations under Capital Leases; (e) all obligations under Synthetic Leases; (f) all Debt (as defined in the other clauses of this definition) of others secured by (or for which the holder of such Debt has an existing right, contingent or otherwise, to be secured by) a Lien on any Property of such Person, whether or not such Debt is assumed by such Person; (g) all Debt (as defined in the other clauses of this definition) of others guaranteed by such Person or in which such Person otherwise assures a creditor against loss of the Debt (howsoever such assurance shall be made) to the extent of the lesser of the amount of such Debt and the maximum stated amount of such guarantee or assurance against loss; (h) all obligations or undertakings of such Person to maintain or cause to be maintained the financial position or covenants of others or to purchase the Debt or Property of others; (i) obligations to deliver commodities, goods or services, including Hydrocarbons, in consideration of one or more advance payments, other than gas balancing arrangements in the ordinary course of business; (j) obligations to pay for goods or services even if such goods or services are not actually received or utilized by such Person; (k) any Debt of a partnership for which such Person is liable either by agreement, by operation of Applicable Law but only to the extent of such liability; (1) Disqualified Capital Stock; and (m) the undischarged balance of any production payment created by such Person or for the creation of which such Person directly or indirectly received payment. The Debt of any Person includes all obligations of such Person of the character described above to the extent such Person remains legally liable in respect thereof notwithstanding that any such obligation is not included as a liability of such Person under GAAP.

"<u>Debt Issuance</u>" means the issuance of any Debt for borrowed money by any Loan Party or any of its Subsidiaries.

"Debtors" means, collectively, the following Borrower and Guarantor (each, also, a "Debtor").

"<u>Default</u>" means any of the events specified in <u>Section 8.1</u> that with the passage of time, the giving of notice or any other condition, would constitute an Event of Default.

"<u>Default Rate</u>" means an interest rate equal to the Interest Rate plus 2.0%.

"<u>Defaulting Lender</u>" means, subject to <u>Section 3.10(B)</u>, any Lender that (a) has failed to (i) fund all or any portion of its DIP Term Loans within two (2) Business Days of the date such DIP Term Loans were required to be funded hereunder unless such Lender notifies the Administrative Agent and the Borrower in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied or (ii) pay to the Administrative Agent or any Lender (upon notice by such Lender to the Administrative Agent that it has not been paid) any other amount required to be paid by it hereunder within two (2) Business Days of the date when due, (b) has notified the Borrower (and the Borrower has so notified the Administrative Agent) or the Administrative Agent in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a DIP Term Loan hereunder and states that such position is based on such Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable default, is specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three Business Days after written request by the Administrative Agent or the Borrower, to confirm in writing to the Administrative Agent and the Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender will cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by the Administrative Agent and the Borrower) or (d) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; provided that a Lender is not a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any Lender who is determined to be a Defaulting Lender under any one or more of clauses (a) through (d) above is deemed to be a Defaulting Lender (subject to Section 3.10(B)) upon delivery of written notice of such determination to the Borrower and each Lender. Any determination by the Administrative Agent that a Lender is a Defaulting Lender under any one or more of clauses (a) through (d) above is conclusive and binding absent manifest error.

"<u>Deposit Account</u>" means any operating, administrative, cash management, collection activity, demand, time, savings, passbook or other deposit account maintained with a bank or other financial institution.

"DIP Facility" has the meaning set forth for such term in the introduction to this Agreement.

"<u>DIP Financing Orders</u>" means the Interim DIP Financing Order and/or the Final DIP Financing Order, as applicable.

"<u>DIP Term Loan</u>" means the First DIP Term Loan Advances, Second DIP Term Loan Advances and Third DIP Term Loan Advances.

"<u>DIP Term Loan Commitment</u>" means (a) as to any Lender, the obligation of such Lender to make a portion of the DIP Term Loans to the account of the Borrower hereunder on the applicable borrowing date in an aggregate principal amount not to exceed the amount set forth opposite such Lender's name on <u>Annex B</u>, as such amount may be reduced or otherwise modified at any time or from time to time pursuant to the terms hereof and (b) as to all Lenders, the aggregate commitment of all Lenders to make such DIP Term Loans.

"Disqualified Capital Stock" means any Equity Interest that, by its terms (or by the terms of any security into which it is convertible or for which it is exchangeable) or upon the happening of any event, requires the payment of dividends (other than dividends payable solely in Equity Interests which do not otherwise constitute Disqualified Capital Stock) or matures or is mandatorily redeemable for any consideration other than other Equity Interests (which would not constitute Disqualified Capital Stock), pursuant to a sinking fund obligation or otherwise, or is convertible or exchangeable for Debt or redeemable for any consideration other than other Equity Interests (which would not constitute Disqualified Capital Stock) at the option of the holder thereof, in whole or in part, on or prior to the date that is one year after the earlier of (a) the Maturity Date and (b) the date on which there are no DIP Term Loans or other Obligations hereunder outstanding and all of the DIP Term Loan Commitments are terminated.

"<u>Distribution</u>" s has the meaning set forth for such term in <u>Schedule 7.1</u> hereto.

"<u>Dollars</u>" or "<u>\$</u>" means, unless otherwise qualified, dollars in lawful currency of the United States.

"<u>Domestic Subsidiary</u>" means any Subsidiary organized under the laws of any political subdivision of the United States.

"Eligible Assignee" means any Person that meets the requirements to be an assignee under Section 10.4(B)(iii), 10.4(B)(v) and 10.4(B)(v) (subject to such consents, if any, as may be required under Section 10.4(B)(iii)).

"Environmental Laws" means any and all Applicable Laws pertaining in any way to health, safety, the environment, the preservation or reclamation of natural resources, or the management, Release or threatened Release of any Hazardous Materials, in effect in any and all jurisdictions in which the Borrower or any of the Borrower's Subsidiaries is conducting, or at any time has conducted, business, or where any Property of the Borrower or any Subsidiaries of Borrower is located, including the Oil Pollution Act of 1990 ("OPA"), the Clean Air Act, the Comprehensive Environmental, Response, Compensation, and Liability Act of 1980 ("CERCLA"), the Federal Water Pollution Control Act, the Occupational Safety and Health Act of 1970, the Resource Conservation and Recovery Act of 1976 ("RCRA"), the Safe Drinking Water Act, the Toxic Substances Control Act, the Superfund Amendments and Reauthorization Act of 1986, the Hazardous Materials Transportation Law, the Outer Continental Shelf Lands Act of 1953 and other environmental conservation or protection Applicable Laws.

"Equity Interests" means (a) in the case of a corporation, capital stock, (b) in the case of an association or business entity, any and all shares, interests, participations, rights or other equivalents (however designated) of capital stock, (c) in the case of a partnership, partnership

interests (whether general or limited), (d) in the case of a limited liability company, membership interests, (e) any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, the issuing Person and (f) any and all warrants, rights or options to purchase any of the foregoing.

"Equity Issuance" means (a) any issuance by any Loan Party or any Subsidiary thereof of shares of its Equity Interests to any Person that is not a Loan Party (including in connection with the exercise of options or warrants or the conversion of any debt securities to equity) and (b) any capital contribution from any Person that is not a Loan Party into any Loan Party or any Subsidiary thereof. The term "Equity Issuance" does not include (i) any Asset Disposition or (ii) any Debt Issuance.

"ERISA" means the Employee Retirement Income Security Act of 1974 and the rules and regulations thereunder.

"ERISA Affiliate" means any Person who together with any Loan Party or any of its Subsidiaries is treated as a single employer within the meaning of Section 414(b), (c), (m) or (o) of the Code or Section 4001(b) of ERISA.

"Erroneous Payment" has the meaning set forth for such term in Section 9.9(A).

"Erroneous Payment Deficiency Assignment" has the meaning set forth for such term in Section 9.9(D).

"Erroneous Payment Impacted Class" has the meaning set forth for such term in Section 9.9(D).

"Erroneous Payment Return Deficiency" has the meaning set forth for such term in Section 9.9(D).

"Erroneous Payment Subrogation Rights" has the meaning set forth for such term in Section 9.9(E).

"Event of Default" has the meaning set forth for such term in Section 8.1.

"Excepted Liens" means, at any time, Liens in respect of property of the Borrower or any of its Subsidiaries:

- (A) (i) Liens granted pursuant to any Loan Document securing the Obligations pursuant to the Loan Documents and (ii) Liens granted as adequate protection on account of the secured Prepetition Loans pursuant to the DIP Financing Orders;
- (B) Liens for Taxes, assessments or other governmental charges or levies which are not delinquent or which are being contested in good faith by appropriate action and for which adequate reserves have been maintained in accordance with GAAP;
- (C) Liens in connection with workers' compensation, unemployment insurance or other social security, old age pension or public liability obligations which are not

- delinquent or which are being contested in good faith by appropriate action and for which adequate reserves have been maintained in accordance with GAAP;
- (D) statutory landlord's liens, operators', vendors', carriers', warehousemen's, repairmen's, mechanics', suppliers', workers', materialmen's, construction or other like Liens arising by operation of law in the ordinary course of business, each of which is in respect of obligations that are not delinquent or which are being contested in good faith by appropriate action and for which adequate reserves have been maintained in accordance with GAAP;
- (E) contractual Liens which arise in the ordinary course of business under operating agreements, joint venture agreements and other agreements which are usual and customary in the ordinary course of business for claims which are not delinquent or which are being contested in good faith by appropriate action and for which adequate reserves have been maintained in accordance with GAAP; <u>provided</u> that any such Lien referred to in this clause (E) does not materially impair the use of the Property covered by such Lien for the purposes for which such Property is held by the Borrower or any of the Borrower's Subsidiaries or materially impair the value of such Property subject thereto;
- (F) Liens arising solely by virtue of any statutory or common law provision relating to banker's liens, rights of set-off or similar rights and remedies and burdening only deposit accounts or other funds maintained with a creditor depository institution; provided that no such deposit account is a dedicated cash collateral account or is subject to restrictions against access by the depositor in excess of those set forth by regulations promulgated by the Board and no such deposit account is intended by the Borrower or any of the Borrower's Subsidiaries to provide collateral to the depository institution;
- (G) easements, restrictions, servitudes, permits, conditions, covenants, exceptions or reservations in any Property of the Borrower or any of the Borrower's Subsidiaries for the purpose of roads, pipelines, transmission lines, transportation lines, distribution lines for the removal of gas, oil, coal or other minerals or timber, and other like purposes, or for the joint or common use of real estate, rights of way, facilities and equipment, that do not secure any monetary obligations and which in the aggregate do not materially impair the use of such Property for the purposes of which such Property is held by the Borrower or any of the Borrower's Subsidiaries or materially impair the value of such Property subject thereto;
- (H) Liens on cash or securities pledged to secure performance of tenders, surety and appeal bonds, government contracts, performance and return of money bonds, bids, trade contracts, leases, statutory obligations, regulatory obligations and other obligations of a like nature incurred in the ordinary course of business; and
- (I) judgment and attachment Liens not giving rise to an Event of Default; <u>provided</u> that any appropriate legal proceedings which may have been duly initiated for the review of such judgment shall not have been finally terminated or the period within

which such proceeding may be initiated shall not have expired and no action to enforce such Lien has been commenced;

<u>provided</u> Liens described in clauses (A) through (E) above remain "Excepted Liens" only for so long as no action to enforce such Lien has been commenced, and no intention to subordinate the superpriority Lien granted in favor of the Administrative Agent and the Secured Parties is to be implied or expressed by the permitted existence of such Excepted Liens.

"Exchange Act" means the Securities Exchange Act of 1934.

"Excluded Deposit Account" means any Deposit Account, existing as of the Closing Date, which is solely used for purposes of funding payroll, payroll taxes or employee benefit payments or which solely contains cash of any Person, other than any Subsidiary, and which cash is held in such Deposit Account solely on behalf of, and for the benefit of, such third party.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient: (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes and branch profits Taxes, in each case, (i) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes; (b) in the case of a Lender, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a DIP Term Loan, Roll-Up Loan or DIP Term Loan Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in the DIP Term Loan, Roll-Up Loan or DIP Term Loan Commitment (other than pursuant to an assignment request by the Borrower under Section 3.9(B)) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 3.8, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a Party or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Recipient's failure to comply with Section 3.8(G) and (d) any U.S. federal withholding Taxes imposed under FATCA.

"Extraordinary Receipt" means any cash received by or paid to or for the account of any Person not in the ordinary course of business, including, without limitation, tax refunds, pension plan reversions, proceeds of insurance (other than proceeds of business interruption insurance to the extent such proceeds constitute compensation for lost earnings), condemnation awards (and payments in lieu thereof), indemnity payments and any purchase price adjustments and any cash received in connection with the settlement or other resolution (including by judgment) of any litigation, arbitration or other dispute.

"FATCA" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code.

"<u>Final DIP Financing Order</u>" means the order of the Bankruptcy Court approving the DIP Facility on a final basis in form and substance satisfactory to the Administrative Agent and the

Required Lenders in their sole discretion, as the same may be amended, modified or supplemented from time to time with, the consent of the Required Lenders (and with respect to amendments, modifications or supplements that adversely affect the rights or duties of the Administrative Agent in any respect, the Administrative Agent).

"<u>First DIP Term Loan Commitment</u>" means the loans available to be advanced to the Borrower pursuant to <u>Sections 2.3(A)</u> and <u>4.1</u> in the aggregate amount of \$1,800,000 authorized in accordance with the Interim DIP Financing Order.

"Fiscal Year" means the 12-month period ending on December 31.

"<u>Fund</u>" means any Person (other than a natural Person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans, bonds and similar extensions of credit in the ordinary course of its activities.

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"Governmental Authority" means the government of the United States or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"Governmental Obligations" means noncallable direct general obligations of the United States of America or obligations the payment of principal of and interest on which is unconditionally guaranteed by the United States of America.

"Guarantor" has the meaning set forth for such term in the introduction to this Agreement.

"Guaranty Agreement" means the unconditional guaranty agreement of even date herewith executed by the Guarantor in favor of the Administrative Agent, for the benefit of the Secured Parties.

"Hazardous Material" means any substance regulated or as to which liability might arise under any applicable Environmental Law including (a) any chemical, compound, material, product, byproduct, substance or waste defined as or included in the definition or meaning of "hazardous substance," "hazardous material," "hazardous waste," "solid waste," "toxic waste," "extremely hazardous substance," "toxic substance," "contaminant," "pollutant," or words of similar meaning or import found in any applicable Environmental Law, (b) Hydrocarbons, petroleum products, petroleum substances, natural gas, oil, oil and gas waste, crude oil and any components, fractions, or derivatives thereof and (c) radioactive materials, explosives, asbestos or asbestos containing materials, polychlorinated biphenyls, radon, infectious or medical wastes.

"Indemnified Taxes" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of any Loan Party under any Loan Document and (b) to the extent not otherwise described in clause (a) above, Other Taxes.

"Initial Approved Budget" has the meaning assigned to such term in Section 4.1(E).

"Insurance and Condemnation Event" means the receipt by any Loan Party or any of its Subsidiaries of any cash insurance proceeds or condemnation award payable by reason of theft, loss, physical destruction or damage, taking or similar event with respect to any of their respective Property.

"Interim DIP Financing Order" means the order of the Bankruptcy Court approving the DIP Facility on an interim basis, in the form of **Exhibit F** hereto or otherwise satisfactory to the Administrative Agent and the Required Lenders in their sole discretion, as the same may be amended, modified or supplemented from time to time with the consent of the Required Lenders (and with respect to amendments, modifications or supplements that affect the rights or duties of the Administrative Agent, the Administrative Agent).

"IRS" means the United States Internal Revenue Service.

"Interest Payment Date" means the last Business Day of each calendar month and the Maturity Date.

"Interest Rate" means a rate *per annum* equal to twelve percent (12.0%).

"Investment" means, for any Person, (a) the acquisition (whether for cash, Property, services or securities or otherwise) of Equity Interests of any other Person or any agreement to make any such acquisition (including any "short sale" or any sale of any securities at a time when such securities are not owned by the Person entering into such short sale), (b) the making of any deposit with, or advance, loan or capital contribution to, the purchase or other acquisition of any other Debt of or equity participation or interest in, or other extension of credit to, any other Person (including the purchase of Property from another Person subject to an understanding or agreement, contingent or otherwise, to resell such Property to such Person, but excluding any such advance, loan or extension of credit having a term not exceeding 90 days representing the purchase price of inventory or supplies sold by such Person in the ordinary course of business), (c) the purchase or acquisition (in one or a series of transactions) of Property of another Person that constitutes a business unit or (d) the entering into of any guarantee of, or other contingent obligation (including the deposit of any Equity Interests to be sold) with respect to, Debt or other liability of any other Person and (without duplication) any amount committed to be advanced, lent or extended to such Person.

"Involuntary Asset Disposition" means any loss of, damage to or destruction of, or any condemnation or other taking for public use of, any property of the Loan Parties.

"<u>Lenders</u>" means the Persons listed on <u>Annex B</u> and any other Person that becomes party to this Agreement pursuant to an Assignment and Assumption, other than any such Person that ceases to be a party to this Agreement pursuant to an Assignment and Assumption.

"<u>Lien</u>" means, with respect to any asset, any mortgage, leasehold mortgage, lien, pledge, charge, security interest, hypothecation or encumbrance of any kind in respect of such asset. For the purposes of this Agreement, a Person shall be deemed to own subject to a Lien any asset which it has acquired or holds subject to the interest of a vendor or lessor under any conditional sale agreement, Capital Lease Obligation or other title retention agreement relating to such asset.

"Loan" means DIP Term Loans, Roll-Up Loans (including any PIK Amounts), as applicable.

"Loan Documents" means, collectively, this Agreement, Security Documents, the Guaranty Agreement, and each other document, instrument, certificate and agreement executed and delivered by the Loan Parties or any of their respective Subsidiaries in favor of or provided to the Administrative Agent or any Lender in connection with this Agreement or otherwise referred to herein or contemplated hereby.

"Loan Parties" means, collectively, the Borrower and the Guarantor.

"Material Adverse Effect" means, with respect to the Borrower and the other Loan Parties, (a) a material adverse effect on the operations, business, assets, properties, liabilities (actual or contingent) or condition (financial or otherwise) of such Persons, taken as a whole, (b) a material impairment of the ability of any such Person to perform its obligations under the Loan Documents to which it is a party, (c) a material impairment of the rights and remedies of the Administrative Agent or any Lender under any Loan Document or (d) a material adverse effect on the legality, validity, binding effect or enforceability against any Loan Party of any Loan Document to which it is a party.

"Material Agreement" means (a) the agreements listed on any schedule to this Agreement and all replacements thereof and (b) any other agreement, document or instrument to which any Loan Party is a party or by which any Loan Party or its assets are bound and which (i) involves monetary liability of or to any Person in an amount in excess of \$250,000 in any Fiscal Year or (ii) the breach, nonperformance, cancellation or failure to renew by any party thereto would have a Material Adverse Effect.

"Maturity Date" means the date that is the earliest of (i) the date that is one-hundred ten (110) days after the Petition Date, (ii) the effective date or date of the substantial consummation (as defined in Section 1101(2) of the Bankruptcy Code) of a Plan of Reorganization or similar dispositive restructuring plan that has been confirmed by an order of the Bankruptcy Court; (iii) the date the Bankruptcy Court orders the conversion of the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code; (iv) the date the Bankruptcy Court orders the dismissal of the Chapter 11 Cases; (v) the date on which the Borrower consummates a sale of all or substantially all of the assets of the Borrower or its Subsidiaries pursuant to Section 363 of the Bankruptcy Code or otherwise; and (vi) the occurrence of any Event of Default or the date of acceleration of the DIP Term Loans or early termination of the DIP Term Loan Commitments hereunder or the DIP Financing Orders, including as a result of the occurrence of an Event of Default.

"Moody's" means Moody's Investors Service, Inc.

"Mortgages" means each mortgage, deed of trust or other real property security document encumbering any real property and any other Property the Required Lenders require to be mortgaged, whether now or hereafter owned by any Loan Party, executed by such Loan Party in favor of the Administrative Agent, for the benefit of the Secured Parties.

"Net Cash Proceeds" means, as applicable, (a) with respect to any Asset Disposition, Involuntary Asset Disposition, or Insurance and Condemnation Event, the gross proceeds received by any Loan Party or any of its Subsidiaries therefrom (including any cash, Cash Equivalents, deferred payment pursuant to, or by monetization of, a note receivable or otherwise, as and when received) less the sum of (i) in the case of an Asset Disposition or Involuntary Asset Disposition, all income taxes and other taxes assessed by, or reasonably estimated to be payable to, a Governmental Authority as a result of such transaction (provided that if such estimated taxes exceed the amount of actual taxes required to be paid in cash in respect of such Asset Disposition or Involuntary Asset Disposition, the amount of such excess constitutes Net Cash Proceeds), (ii) all reasonable and customary out-of-pocket fees and expenses incurred in connection with such transaction or event and (iii) the principal amount of, premium, if any, and interest on any Debt secured by a Lien on the asset (or a portion thereof) disposed of, which Debt is required to be repaid in connection with such transaction or event and (b) with respect to any Equity Issuance or Debt Issuance, the gross cash proceeds received by any Loan Party or any of its Subsidiaries therefrom less all reasonable and customary out-of-pocket legal, underwriting and other fees and expenses incurred in connection therewith.

"Non-Consenting Lender" means any Lender that does not approve any consent, waiver or amendment that (a) requires the approval of all or all affected Lenders in accordance with the terms of Section 10.3 and (ii) has been approved by the Required Lenders.

"Non-Defaulting Lender" means, at any time, each Lender that is not a Defaulting Lender at such time.

"Notice of Borrowing" means a written notice substantially in the form of **Exhibit A**.

"Notice of Prepayment" means a written notice substantially in the form of **Exhibit B**.

"Obligations" means all amounts owing to any Administrative Agent or any Lender pursuant to the terms of this Agreement or any other Loan Document (including principal, interest, any contingent indemnification and reimbursement obligations, or other payments required under Section 10.2, and all fees and other expenses required to be paid pursuant to the Loan Documents).

"OFAC" means the United States Department of the Treasury's Office of Foreign Assets Control.

"Officer's Compliance Certificate" means a certificate of a Responsible Officer of the Borrower substantially in the form of $\underline{Exhibit\ C}$.

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security

interest under, engaged in any other transaction pursuant to any Loan Document or enforced any Loan Document, or sold or assigned an interest in any Loan or Loan Document).

"Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to Section 3.8(B)).

"Parent" has the meaning set forth for such term in the recitals to this Agreement.

"Parties" has the meaning set forth for such term in the introduction to this Agreement.

"Participant" has the meaning set forth for such term in Section 10.4(D).

"Participant Register" has the meaning set forth for such term in Section 10.4(D).

"PATRIOT Act" means the USA PATRIOT Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)).

"Payment Recipient" has the meaning set forth for such term in Section 9.9(A).

"<u>Permitted Holders</u>" means (i) Volunteer Enterprises, LLC, (ii) Environmental Services Management of Virginia, LLC, (iii) Fred Nichols, or (iv) Larry McGee, together with any Affiliates of any of the foregoing.

"Permitted Investments" means with respect to any Person:

- (a) Governmental Obligations;
- (b) Obligations of a state or commonwealth of the United States or the obligations of the District of Columbia or any possession of the United States, or any political subdivision of any of the foregoing, which are described in Section 103(a) of the Internal Revenue Code and are graded in any of the highest three (3) major grades as determined by at least one Rating Agency, or secured, as to payments of principal and interest, by a letter of credit provided by a financial institution or insurance provided by a bond insurance company which in each case is itself or its debt is rated in one of the highest three (3) major grades as determined by at least one Rating Agency;
- (c) Banker's acceptances, commercial accounts, demand deposit accounts, certificates of deposit, other time deposits or depository receipts issued by or maintained with any Investor or any Affiliate thereof, or any bank, trust company, savings and loan association, savings bank or other financial institution whose deposits are insured by the Federal Deposit Insurance Corporation and whose reported capital and surplus equal at least \$250,000,000, provided that such minimum capital and surplus

- requirement shall not apply to demand deposit accounts maintained by any Credit Party in the ordinary course of business;
- (d) Commercial paper rated at the time of purchase within the two highest classifications established by not less than two Rating Agencies, and which matures within 270 days after the date of issue;
- (e) Secured repurchase agreements against obligations itemized in paragraph (a) above, and executed by a bank or trust company or by members of the association of primary dealers or other recognized dealers in United States government securities, the market value of which must be maintained at levels at least equal to the amounts advanced; and
- (f) Any fund or other pooling arrangement which exclusively purchases and holds the investments itemized in (a) through (e) above.
- "Permitted Variance" has the meaning set forth for such term in Section 6.18(D).
- "Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.
 - "Petition Date" means June 1, 2025.
 - "PIK Amounts" has the meaning set forth for such term in Section 3.2(C).
 - "Plan Effective Date" means the effective date of an Acceptable Plan.
- "<u>Plan of Reorganization</u>" a plan of reorganization with respect to the Debtors pursuant to the Chapter 11 Cases.
- "<u>Prepetition Debt</u>" means all Debt and other obligations of the Debtors arising before the Petition Date.
 - "Prepetition Loan Agreement" has the meaning set forth in the recitals hereto.
 - "Prepetition Loans" means the Bridge Notes as defined in the Prepetition Loan Agreement.
- "Property" means any right or interest in or to property of any kind whatsoever, whether real, personal or mixed, movable or immovable, and whether tangible or intangible, including Equity Interests.
 - "Recipient" means the Administrative Agent or any Lender, as applicable.
- "Redemption" means, with respect to any Debt, the repurchase, redemption, prepayment, repayment, defeasance or any other acquisition or retirement for value (or the segregation of funds with respect to any of the foregoing) of such Debt. "Redeem" has the correlative meaning thereto.

"Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives of such Person and of such Person's Affiliates.

"Release" means any depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, dumping or disposing into the environment.

"Reportable Compliance Event" means that any Covered Entity becomes a Sanctioned Person, or is indicted, arraigned, investigated or custodially detained, or receives an inquiry from regulatory or law enforcement officials, in connection with any Anti-Corruption Law or any predicate crime to any Anti-Corruption Law, or has knowledge of or self-discovers facts or circumstances implicating any aspect of its operations with the actual or possible violation of any Anti-Corruption Law.

"Required Lenders" means, at any date, any combination of Lenders holding more than 66 2/3% of the sum of the aggregate amount of the DIP Term Loan Commitments or, if the DIP Term Loan Commitments have been reduced to zero or terminated, any combination of Lenders holding more than 66 2/3% of the aggregate outstanding principal amount of the DIP Facility; provided the DIP Term Loan Commitment of, and the portion of the DIP Term Loans held or deemed held by, any Defaulting Lender are excluded for purposes of making a determination of Required Lenders.

"Responsible Officer" means as to any Loan Party, the chief executive officer, president, chief financial officer, controller, treasurer, assistant treasurer or any executive vice president of such Person or any other officer of such Person designated in writing by the Borrower and reasonably acceptable to the Required Lenders; provided, to the extent requested thereby, the Administrative Agent has received a certificate of such Person certifying as to the incumbency and genuineness of the signature of each such officer. Any document delivered hereunder or under any other Loan Document that is signed by a Responsible Officer of a Person is conclusively presumed to have been authorized by all necessary corporate, partnership or other action on the part of such Person and such Responsible Officer is conclusively presumed to have acted on behalf of such Person.

"Roll-Up Loans" has the meaning set forth for such term in the introduction to this Agreement.

"S&P" means Standard & Poor's Financial Services LLC, a part of McGraw-Hill Financial and any successor thereto.

"Sale Transaction" means the contemplated sale of all or substantially all of the Debtors' assets and business, pursuant to sections 363 and 365 of the Bankruptcy Code and seeking the entry of orders by the Bankruptcy Court approving such a transaction and authorizing the Debtors to consummate the transactions contemplated therein.

"Sanctions" means economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by the United States government (including those

administered by OFAC), the European Union, Her Majesty's Treasury, or other relevant sanctions authority.

"Sanctioned Country" means at any time, a country or territory which is itself the subject or target of any Sanctions (including Cuba, Iran, North Korea, Sudan and Syria).

"Sanctioned Person" means, at any time, (a) any Person listed in any Sanctions-related list of designated Persons maintained by OFAC, the United States Department of State, the United Nations Security Council, the European Union, Her Majesty's Treasury, or other relevant sanctions authority, (b) any Person operating, organized or resident in a Sanctioned Country or (c) any Person owned or controlled by any such Person or Persons described in clauses (a) and (b) above.

"Second DIP Term Loan Commitment" means the loans available to be advanced to the Borrower pursuant to Sections 2.3(B) and 4.2 in the aggregate amount of \$1,550,000 authorized in accordance with the Final DIP Financing Order.

"Secured Parties" means, collectively, the Administrative Agent and the Lenders.

"Security Documents" means the collective reference to the Mortgages, the Guaranty Agreement, any Account Control Agreements, and each other agreement or writing pursuant to which any Loan Party pledges or grants a security interest in any Property securing the Obligations.

"Shoosmith" has the meaning set forth for such term in the introduction to this Agreement.

"Subordinated Debt" means and Debt of any Loan Party which has been subordinated in right of payment and priority to the Obligations, all on terms and conditions satisfactory to the Administrative Agent.

"Subsidiary" means as to any Person, any corporation, partnership, limited liability company or other entity of which more than 50% of the outstanding Equity Interests having ordinary voting power to elect a majority of the board of directors (or equivalent governing body) or other managers of such corporation, partnership, limited liability company or other entity is at the time owned by (directly or indirectly) or the management is otherwise controlled by (directly or indirectly) such Person (irrespective of whether, at the time, Equity Interests of any other class or classes of such corporation, partnership, limited liability company or other entity have or might have voting power by reason of the happening of any contingency). Unless otherwise qualified, references to "Subsidiary" or "Subsidiaries" refer to a Subsidiary (including the Guarantor) of the Borrower.

"Subsidiary Guarantors" means, collectively, all direct and indirect Subsidiaries of the Borrower in existence on the Closing Date or which become a party to the Guaranty Agreement pursuant to Section 6.13(A).

"Swap Agreement" means any agreement with respect to any swap, forward, future or derivative transaction or option or similar agreement, whether exchange traded, "over-the-counter" or otherwise, involving, or settled by reference to, one or more rates, currencies, commodities, equity or debt instruments or securities, or economic, financial or pricing indices or measures of economic, financial or pricing risk or value or any similar transaction or any combination of these

transactions (including any agreement, contract or transaction that constitutes a "swap" within the meaning of section 1a(47) of the Commodity Exchange Act); <u>provided</u> that no phantom stock or similar plan providing for payments only on account of services provided by current or former directors, officers, employees or consultants of the Borrower or the Subsidiaries is a Swap Agreement.

"Synthetic Leases" means, in respect of any Person, all leases which shall have been, or should have been, in accordance with GAAP, treated as operating leases on the financial statements of the Person liable (whether contingently or otherwise) for the payment of rent thereunder and which were properly treated as indebtedness for borrowed money for purposes of U.S. federal income taxes, if the lessee in respect thereof is obligated to either purchase for an amount in excess of, or pay upon early termination an amount in excess of, 80% of the residual value of the Property subject to such operating lease upon expiration or early termination of such lease.

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"Third DIP Term Loan Commitment" means the loans available to be advanced to the Borrower pursuant to Sections 2.3(A) and 4.3 from and after July 25, 2025 in the aggregate amount of \$1,550,000 authorized in accordance with the Final DIP Financing Order.

"<u>Total DIP Term Loan Commitment</u>" means, as of the Closing Date, the amount of DIP Term Loan Commitments, plus the Roll-Up Loan Amount.

"Transactions" means, with respect to (a) the Borrower, the execution, delivery and performance by the Borrower of this Agreement and each other Loan Document to which it is a party, the borrowing of DIP Term Loans and Roll-Up Loans, the use of the proceeds thereof and the grant of Liens by the Borrower on the Collateral pursuant to the Security Documents and (b) each Guarantor, the execution, delivery and performance by such Guarantor of each Loan Document to which it is a party, the guaranteeing of the Obligations and the other obligations under the Guaranty Agreement by such Guarantor and such Guarantor's grant of Liens by such Guarantor on the Collateral pursuant to the Security Documents.

"Type" shall mean the type of DIP Term Loan to be borrowed, either a (i) First DIP Term Loan Advance, (b) Second DIP Term Loan Advance or (iii) Third DIP Term Loan Advance.

"United States" or "U.S." means the United States of America.

"<u>U.S. Person</u>" means any Person that is a "United States Person" as defined in Section 7701(a)(30) of the Code.

"VDEQ" means the Virginia Department of Environmental Quality or any successor agency.

"Wholly-Owned" (or "wholly-owned") means, with respect to a Subsidiary, that all of the Equity Interests of such Subsidiary are, directly or indirectly, owned or controlled by the Borrower or one or more of its Wholly-Owned Subsidiaries (except for directors' qualifying shares or other

shares required by Applicable Law to be owned by a Person other than the Borrower or one or more of its Wholly-Owned Subsidiaries).

"Withholding Agent" means any Loan Party and the Administrative Agent.

2. <u>Accounting Terms</u>.

- (A) All accounting terms not specifically or completely defined herein are construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with GAAP, applied on a consistent basis, as in effect from time to time and in a manner consistent with that used in preparing the audited financial statements required by Section 6.1(A) except as otherwise specifically prescribed herein.
- (B) If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and either the Borrower or the Required Lenders so request, the Lenders and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Borrower shall provide to the Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.
- 3. <u>Use of Certain Terms</u>. As used in this Agreement, the plural includes the singular and the singular includes the plural. All pronouns and any variations thereof refer to masculine, feminine, neuter, singular or plural as the identity of the Person or Persons may require. As used in this Agreement, "include," "includes" and "including" have the inclusive meaning of "including without limitation."
- References. Section and other headings are for reference only, and do not affect the interpretation or meaning of any provision of this Agreement. Unless otherwise provided, references to articles, sections, clauses, annexes, schedules and exhibits refer to articles, sections, clauses, annexes, schedules and exhibits of this Agreement. The words "hereof," "herein," "hereby," "hereunder" and other similar terms of this Agreement refer to this Agreement as a whole and not exclusively to any particular provision of this Agreement. Unless otherwise expressly indicated in this Agreement, the words "above" and "below," when following a reference to a clause of any Loan Document, refer to a clause within the same section of such Loan Document. References in this Agreement to this Agreement, any other Loan Document or any other agreement are deemed to (a) refer to this Agreement, such other Loan Document or such other agreements, as the case may be, as the same may be amended, restated, supplemented or otherwise modified from time to time under the provisions hereof or thereof, unless expressly stated otherwise or unless such amendment, restatement, supplement or modification is not permitted by the terms of this Agreement and (b) include all schedules, exhibits and appendices thereto. References in this Agreement to any law, rule, statute or regulation are deemed to refer to such law, rule, statute or regulation as it may be amended, supplemented or otherwise modified from time to time, and any

successor law, rule, statute or regulation, in each case as in effect at the time any such reference is operative. Any reference to a Person includes the successors, assigns, participants and transferees of such Person, but such reference will not increase, decrease or otherwise modify in any way the provisions in this Agreement or any other Loan Document governing the assignment of rights and obligations under or the binding effect of any provision of this Agreement or any other Loan Document. The words "asset" and "property" are construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights. The term "documents" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form. In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including;" the words "to" and "until" each mean "to but excluding;" and the word "through" means "to and including"

- 5. <u>Times of Day</u>. Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).
- 6. <u>Rounding</u>. Any financial ratios required to be maintained pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio or percentage is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).
- 7. <u>Inconsistencies with Other Documents</u>. In the event there is a conflict or inconsistency between this Agreement and any other Loan Document, the terms of this Agreement control; <u>provided</u> any provision of the Security Documents which imposes additional burdens on a Loan Party or any of its Subsidiaries or further restricts the rights of a Loan Party or any of its Subsidiaries or gives the Administrative Agent or Lenders additional rights is not deemed to be in conflict or inconsistent with this Agreement and shall be given full force and effect.

ANNEX B

DIP Term Loan Commitments and Commitment Percentages

<u>DIP TERM LOANS MADE PURSUANT TO SECTION 2.1(B) – FIRST DIP TERM LOAN ADVANCE</u>:

<u>Lender</u>	First DIP Term Loan Advance Commitment	Commitment Percentage
VOLUNTEER ENTERPRISES, LLC	\$1,800,000	100%
Total	\$1,800,000	100%

<u>DIP TERM LOANS MADE PURSUANT TO SECTION 2.1(C) – SECOND DIP TERM LOAN ADVANCE</u>:

<u>Lender</u>	Second DIP Term Loan Advance Commitment	Commitment Percentage
VOLUNTEER ENTERPRISES, LLC	\$1,550,000	100%
Total	\$1,550,000	100%

<u>DIP TERM LOANS MADE PURSUANT TO SECTION 2.1(D) – THIRD DIP TERM LOAN ADVANCE</u>:

<u>Lender</u>	Third DIP Term Loan Advance Commitment	<u>Commitment</u> <u>Percentage</u>
VOLUNTEER ENTERPRISES, LLC	\$1,550,000	100%
Total	\$1,550,000	100%

ROLL-UP LOANS PURSUANT TO SECTION 2.2(A):

<u>Lender</u>	Roll-Up Loans for 2022 Term Loans
VOLUNTEER ENTERPRISES, LLC	\$500,000
The state of the s	0,500,000
Total	\$500,000

ANNEX C

Addresses for Notices

If to the Loan Parties:

VWS Holdco, Inc.
[ADDRESS]
[ATTENTION]
[TELEPHONE]
[EMAIL]

With copies to:

[LAW FIRM] [ADDRESS] [ATTENTION] [TELEPHONE] [EMAIL]

If to the Administrative Agent or the Lenders:

Volunteer Enterprises, LLC [ADDRESS] [ATTENTION] [TELEPHONE] [EMAIL]

With a copy to:

Cole Schotz P.C. 1325 Avenue of the Americas, 20th Floor New York, NY 10019

Attention of: James Stefanick, Daniel Geoghan

Telephone No.: 646-563-8949

E-mail: jstefanick@coleschotz.com; dgeoghan@coleschotz.com

ANNEX D

Closing Deliverables

<u>Executed Loan Documents</u>. This Agreement, the Guaranty Agreement and the Security Documents, together with any other applicable Loan Documents.

Officer's Certificate. A certificate from a Responsible Officer of the Borrower to the effect that (a) all representations and warranties of the Loan Parties contained in this Agreement and the other Loan Documents are true, correct and complete, (b) none of the Loan Parties is in violation of any of the covenants contained in this Agreement and the other Loan Documents, (c) after giving effect to the Transactions, no Default or Event of Default has occurred and is continuing, (d) since the Petition Date, no event has occurred or condition arisen, either individually or in the aggregate, that has had or could reasonably be expected to have a Material Adverse Effect and (e) each of the Loan Parties, as applicable, has satisfied each of the conditions set forth in Section 4.1.

Certificate of Secretary of Each Loan Party. A certificate of a Responsible Officer of each Loan Party certifying as to the incumbency and genuineness of the signature of each officer of such Loan Party executing Loan Documents to which it is a party and certifying that attached thereto is a true, correct and complete copy of (a) the articles or certificate of incorporation or formation (or equivalent), as applicable, of such Loan Party and all amendments thereto, certified as of a recent date by the appropriate Governmental Authority in its jurisdiction of incorporation, organization or formation (or equivalent), as applicable, (b) the bylaws or other governing document of such Loan Party as in effect on the Closing Date and (c) resolutions duly adopted by the board of directors (or other governing body) of such Loan Party authorizing and approving the transactions contemplated hereunder and the execution, delivery and performance of this Agreement and the other Loan Documents to which it is a party.

<u>Certificates of Good Standing</u>. Certificates as of a recent date of the good standing of each Loan Party under the laws of its jurisdiction of incorporation, organization or formation (or equivalent), as applicable, and, to the extent requested by the Lenders, each other jurisdiction where such Loan Party is qualified to do business and, to the extent available, a certificate of the relevant taxing authorities of such jurisdictions certifying that such Loan Party has filed required tax returns and owes no delinquent taxes.

Other Documents. Copies of all other documents, certificates and instruments reasonably requested by the Administrative Agent or any Lender.

Exhibit A

Form of Notice of Borrowing

Case 25-10979-JKS Doc 50-1 Filed 06/02/25 Page 262 of 298

Dated as of:,
Volunteer Enterprises, LLC [ADDRESS] [ATTENTION] [TELEPHONE] [EMAIL]
Ladies and Gentlemen:
This irrevocable Notice of Borrowing is delivered to you pursuant to Section 2.3 of the Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of June 2 2025, as amended and as may be further amended from time to time (the "DIP Agreement") among VWS Holdco, Inc. a Delaware corporation (the "Borrower"), Shoosmith Bros., Inc., a Virginia Corporation (the "Guarantor"), the Lenders party thereto from time to time, and Voluntee Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the "Administrative Agent"). Capitalized terms used herein and not otherwise defined have the meanings set forth for such terms in the DIP Agreement. 1. The Borrower hereby requests that the Lenders make the DIP Term Loans to the Borrower in the aggregate principal amount of \$
3. The Person(s) to whom the proceeds of such DIP Terms Loan(s) are to be disbursed and the wire instructions of the account(s) to which funds are to be disbursed are as follows:
VWS Holdco, Inc. Bank: [] ABA: [] Account Name: [] Account Number: []
4. The Borrower has previously delivered the appropriate tax forms with respect to each Person listed in paragraph 3 above as required by the DIP Agreement.

- 5. The Type of DIP Term Loan to be borrowed is: Base Rate Loans
- 6. The aggregate principal amount of all DIP Term Loans outstanding as of the date hereof (including the DIP Term Loans requested herein) does not exceed the maximum amount permitted to be outstanding pursuant to the terms of the DIP Agreement
- 7. All of the conditions applicable to the DIP Term Loan requested herein as set forth in the DIP Agreement have been satisfied as of the date hereof and will remain satisfied to the date of such DIP Term Loan.

Case 25-10979-JKS Doc 50-1 Filed 06/02/25 Page 268 of 298

Executed and delivered as of the date first above	written.
	VWS HOLDCO, INC.
	By: Name: [] Title: []

Exhibit B

Form of Notice of Prepayment

FORM OF PREPAYMENT NOTICE

Date: $[\bullet]$, $20[\bullet]^2$

Volunteer Enterprises, LLC, as Administrative Agent [ADDRESS]
[ATTENTION]
[TELEPHONE]
[EMAIL]

Re: <u>VWS Holdco, Inc. – Prepayment Notice</u>

Ladies and Gentlemen:

Reference is made to the Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of [____], 2025, as amended and as may be further amended from time to time (the "DIP Agreement"), among VWS Holdco, Inc. a Delaware corporation (the "Borrower"), Shoosmith Bros., Inc., a Virginia Corporation (the "Guarantor"), the Lenders party thereto from time to time, and Volunteer Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the "Administrative Agent"). Each capitalized term used but not otherwise defined in this Prepayment Notice shall have the meaning assigned to such term in the DIP Credit Agreement.

This Prepayment Notice is delivered³ to you pursuant to <u>Section 2.5</u> of the DIP Credit Agreement. The Borrower hereby gives notice of a prepayment of Loans as follows:

Borrowing with an Interest Period ending $[\bullet]$, $20[\bullet]$ in the aggregate principal amount of $[\bullet]$.

On $[\bullet]$, 20 $[\bullet]$ (a Business Day).

[3. Attached hereto as Appendix A is a reasonably detailed calculation of the amount of the prepayment contemplated hereby.] 4

[Signature Page Follows]

To be delivered not later than 12:00 noon, New York City time, three (3) Business Days before the scheduled date of such repayment. Each repayment of a Borrowing shall be applied to the DIP Term Loans included in the repaid Borrowing such that each Lender receives its ratable share of such repayment (based upon the respective Percentages at the time of such repayment).

The notice of which may be given by facsimile or email.

This information is only required for a mandatory prepayment.

Case 25-10979-JKS Doc 50-1 Filed 06/02/25 Page 266 of 298

IN WITNESS WHEREOF, the undersigned Responsible Officer has executed and delivered this Prepayment Notice on behalf of the Borrower, in his or her capacity as a duly elected, qualified and acting Responsible Officer of the Borrower (and not in the Responsible Officer's individual capacity), and has made the certifications and statements contained herein in the capacity stated herein, as of the date first above written.

VWS HOLDCO, INC.	
as Borrower	
By:	
Name: []	
Title: []	

Exhibit C

Form of Notice of Responsible Officer's Compliance Certificate

RESPONSIBLE OFFICER CERTIFICATE OF VWS HOLDCO, INC.

June 2, 2025

This Responsible Officer's Certificate is delivered in connection with that certain Senior
Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of June 2, 2025, as
amended and as may be further amended from time to time (the "DIP Agreement"), among VWS
Holdco, Inc. a Delaware corporation (the "Borrower"), Shoosmith Bros., Inc., a Virginia
Corporation (the "Guarantor"), the Lenders party thereto from time to time, and Volunteer
Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the
"Administrative Agent"). The undersigned, [], the Responsible Officer of Borrower,
hereby certifies to Lenders that he is the Responsible Officer of Borrower, and in such capacity, is
duly authorized to execute this Responsible Officer's Certificate and further represents, warrants
and certifies to Lenders as of the date hereof:

- 1. Attached hereto as <u>Exhibit A</u> is a true, correct and complete copy of the Written Consent of the Board of Directors of Borrower approving and authorizing the execution, delivery and performance of the DIP Credit Agreement and other agreements, instruments and documents contemplated thereby to which it is a party, and the consummation of all of the transactions contemplated thereby. Such resolutions have not been amended, rescinded or modified since their adoption and remain in effect as of the date hereof.
- 2. The person whose name appears below is duly elected, qualified and an acting officer of Borrower, occupying the position set forth in the signature block below, and the signature set forth below is a true signature, and such officer is duly authorized to execute and deliver, on behalf of Borrower, the DIP Credit Agreement and other agreements, instruments and documents contemplated thereby to which it is a party.

Capitalized terms used herein and not otherwise defined shall have the meanings set forth in the DIP Credit Agreement.

IN WITNESS WHEREOF, the undersigned Responsible Officer of Borrower has executed this certificate as of the date first written above.

Name: [
Title:	1

Exhibit D

Form of Assignment and Assumption

SENIOR SECURED SUPERPRIORITY DEBTOR-IN-POSSESSION TERM LOAN AGREEMENT (THE "DIP CREDIT AGREEMENT")¹

STANDARD TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION

- 1. Representations and Warranties.
- 1.1 Assignor. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any Lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations of any Person other than the Assignor made in or in connection with the DIP Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral thereunder, (iii) the financial condition of the Borrower, the Guarantor, or any Affiliate or Subsidiary of the foregoing or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, the Guarantor, or any Affiliate or Subsidiary of the foregoing or any other Person of any of such Person's respective obligations under any Loan Document.
- Assignee. The Assignee (a) represents and warrants to the Assignor and the Borrower that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the DIP Credit Agreement, (ii) it has satisfied all required conditions pursuant Section 10.4 of the DIP Credit Agreement, (iii) from and after the Effective Date, it shall be bound by the provisions of the DIP Credit Agreement and the other Loan Documents as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it is sophisticated with respect to decisions to acquire assets of the type represented by the Assigned Interest and either it, or the Person exercising discretion in making its decision to acquire the Assigned Interest, is experienced in acquiring assets of such type, (v) it has received a copy of the DIP Credit Agreement, and has received or has been accorded the opportunity to receive copies of the most recent financial statements delivered pursuant to Section 6.1 or Section 6.18 thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest, (vi) it has, independently and without reliance on the Administrative Agent, the Assignor or any other Lender and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into

¹ Capitalized terms used herein and not otherwise defined have the meanings set forth for such terms in the Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of June 2, 2025, as amended and as may be further amended from time to time (the "<u>DIP Agreement</u>"), among VWS Holdco, Inc. a Delaware corporation (the "<u>Borrower</u>"), Shoosmith Bros., Inc., a Virginia Corporation (the "<u>Guarantor</u>"), the Lenders party thereto from time to time, and Volunteer Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the "<u>Administrative Agent</u>").

this Assignment and Assumption and to purchase the Assigned Interest, (vii) attached to the Assignment and Assumption is any documentation required to be delivered by it pursuant to the terms of the DIP Credit Agreement, duly completed and executed by the Assignee; (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender and (c) authorizes the Administrative Agent to take such actions on behalf of the Assignee and to exercise such powers as are specifically delegated to the Administrative Agent by the terms and provisions of the DIP Credit Agreement and of the other Loan Documents, together with such actions and powers as are reasonably incidental thereto and (viii) it is a "qualified purchaser" ("Qualified Purchaser") as defined under Section 2(a)(51) of the Investment Company Act of 1940 (the "1940 Act") that either: (1)(A) acting for its own account or the account of other Qualified Purchasers in the aggregate owns and invests on a discretionary basis not less than \$25,000,000 in "investments" as defined under Rule 2a51-1 of the 1940 Act, and (B) is not organized for the specific purpose of acquiring the Assigned Interest; does not have shareholders, partners or other holders of equity or beneficial interests that have the right to decide individually whether to participate, or the extent of their participation, in such entity's acquisition of the Assigned Interest; and is not acquiring an Acquired Interest that exceeds 40% of such entity's total assets (on a consolidated basis with its subsidiaries); or (2) is a company in which each beneficial owner of such company's securities is a Qualified Purchaser.

- 2. *Payments*. From and after the Effective Date, the Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the Effective Date and to the Assignee for amounts which have accrued from and after the Effective Date.
- 3. Effect of Assignment. As of the Effective Date, (i) the Assignee shall be a party to the DIP Credit Agreement and, to the extent provided in this Assignment and Assumption, have the rights and obligations of a Lender under the DIP Credit Agreement and under the other Loan Documents and (ii) the Assignor shall, to the extent provided in this Assignment and Assumption, relinquish its rights and be released from its obligations under the DIP Credit Agreement and the other Loan Documents.
- 4. General Provisions. This Assignment and Assumption shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment and Assumption may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment and Assumption by facsimile or other electronic means, which shall include DocuSign and similar electronic signature platforms and digital copies of a signatory's manual signature, will for all purposes be treated as the equivalent of delivery of a manually executed counterpart of this Assignment and Assumption. The words "execution," "signed," "signature," and words of like import in this Assignment and Assumption shall be deemed to include electronic signatures, including by DocuSign and similar electronic signature platforms and digital copies of a signatory's manual signature, or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a

Case 25-10979-JKS Doc 50-1 Filed 06/02/25 Page 272 of 298

paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act. This Assignment and Assumption shall be construed in accordance with and governed by the laws of the State of New York.

Exhibit F

Proposed Interim DIP Financing Order

Schedule 5.4 – Ownership and Control

SCHEDULE 6.1

REPORTING AND INFORMATION REQUIREMENTS

Each Loan Party covenants and agrees, so long as any of the Obligations remains outstanding and unpaid, that it will, and, as applicable, it will cause each of its Subsidiaries to:

- 1. <u>Financial Statements</u>. Furnish to the Administrative Agent, in form and detail satisfactory to the Administrative Agent, with sufficient copies for each Lender, the following documents:
- (a) as soon as available, but in any event within one hundred twenty (120) days after the end of each Fiscal Year, a copy of the audited Consolidated and unaudited Consolidating financial statements of Borrower and its Consolidated Subsidiaries as at the end of such Fiscal Year and the related audited Consolidated and unaudited Consolidating statements of income, stockholders equity, and cash flows of the Borrower and its Consolidated Subsidiaries for such Fiscal Year or partial Fiscal Year and underlying assumptions, setting forth in each case in comparative form the figures for the previous Fiscal Year, certified as being fairly stated in all material respects by an independent, nationally recognized certified public accounting firm reasonably satisfactory to the Administrative Agent;
- (b) as soon as available, but in any event within forty-five (45) days after the end of each fiscal quarter of the Loan Parties (including the last quarter of each Fiscal Year), Borrower prepared unaudited Consolidated and Consolidating balance sheets of Borrower and its Consolidated Subsidiaries as at the end of such quarter and the related unaudited statements of income, stockholders equity and cash flows of Borrower and its Consolidated Subsidiaries for the portion of the Fiscal Year through the end of such quarter, setting forth in each case in comparative form the figures for the corresponding periods in the previous Fiscal Year, and certified by a Responsible Officer of Borrower as being fairly stated in all material respects;
- (c) as soon as available, but in any event within thirty (30) days after the end of each month (unless such month end is also the end of a fiscal quarter or a Fiscal Year), commencing with the first full month after the Closing Date, Borrower prepared unaudited Consolidated and Consolidating balance sheets of Borrower and its Consolidated Subsidiaries as at the end of such month and the related unaudited statements of income, stockholders equity and cash flows of Borrower and its Consolidated Subsidiaries for the portion of the Fiscal Year through the end of such fiscal month, setting forth in each case in comparative form (i) the figures for the corresponding periods in the previous year and (ii) the figures for the relevant period set forth in the projections delivered for such year pursuant to Section 2(d) of this Schedule 6.1, and certified by a Responsible Officer of Borrower as being fairly stated in all material respects; and

all such financial statements to be complete and correct in all material respects and to be prepared in reasonable detail and in accordance with GAAP throughout the periods reflected therein and with prior periods (except as approved by a Responsible Officer and disclosed therein), provided however that the financial statements delivered pursuant to clauses (b) and (c) hereof will

not be required to include footnotes and will be subject to change from audit and year-end adjustments.

- 2. <u>Certificates, Other Information</u>. Furnish to the Administrative Agent, in form and detail acceptable to the Administrative Agent, with sufficient copies for each Lender, the following documents:
 - (a) Promptly upon receipt thereof, copies of all significant reports submitted by the Loan Parties' firm(s) of certified public accountants in connection with each annual, interim or special audit or review of any type of the financial statements or related internal control systems of the Loan Parties made by such accountants, including any comment letter submitted by such accountants to management in connection with their services;
 - (b) Any financial reports, statements, press releases, other material information or written notices delivered to the holders of the Subordinated Debt pursuant to any applicable Subordinated Debt documents (to the extent not otherwise required hereunder), as and when delivered to such Persons;
 - (c) Within thirty (30) days after the end of each Fiscal Year, projections for the Loan Parties for the next succeeding Fiscal Year, on a monthly basis and for the following Fiscal Year on an annual basis, including a balance sheet, as at the end of each relevant period and for the period commencing at the beginning of the Fiscal Year and ending on the last day of such relevant period, such projections certified by a Responsible Officer of Borrower as being based on reasonable estimates and assumptions taking into account all facts and information known (or reasonably available to any Loan Party) by a Responsible Officer of Borrower;
 - (d) Any additional information as required by any Loan Document, and such additional schedules, certificates and reports respecting all or any of the collateral, the items or amounts received by the Loan Parties in full or partial payment thereof, and any goods (the sale or lease of which shall have given rise to any of the collateral) possession of which has been obtained by the Loan Parties, all to such extent as the Administrative Agent may reasonably request from time to time, any such schedule, certificate or report to be certified as true and correct in all material respects by a Responsible Officer of the applicable Loan Party and shall be in such form and detail as the Administrative Agent may reasonably specify; and
 - (e) Such additional financial and/or other information as any Lender may from time to time reasonably request, promptly following such request.
- 3. Notices. Promptly give written notice to the Administrative Agent of:
 - (a) the occurrence of any Default or Event of Default of which any Loan Party has knowledge or the occurrence of any Reportable Compliance Event;

- (b) any (i) litigation or proceeding existing at any time between any Loan Party and any Governmental Authority or other third party, or any investigation of any Loan Party conducted by any Governmental Authority (including, without limitation, any notices, inspections, investigations or proceedings that any Loan Party receives or of which it becomes aware involving the Virginia Department of Environmental Quality, the U.S. Environmental Protection Agency, Chesterfield County, or any other Governmental Authority or Person relating to Hazardous Material Laws), which in any case if adversely determined would have a Material Adverse Effect or (ii) any material adverse change in the financial condition of any Loan Party since the date of the last audited financial statements delivered pursuant to Section 1(a) of this Schedule 6.1;
- (c) the occurrence of any event which any Loan Party believes could reasonably be expected to have a Material Adverse Effect, promptly after concluding that such event could reasonably be expected to have such a Material Adverse Effect;
- (d) promptly after becoming aware thereof, the taking by the Internal Revenue Service or any foreign taxing jurisdiction of a written tax position (or any such tax position taken by any Loan Party in a filing with the Internal Revenue Service or any foreign taxing jurisdiction) which could reasonably be expected to have a Material Adverse Effect, setting forth the details of such position and the financial impact thereof;
- (e) (i) all jurisdictions in which any Loan Party proposes to become qualified after the Closing Date to transact business, (ii) the acquisition or creation of any new Subsidiaries, (iii) any material change after the Closing Date in the authorized and issued Equity Interests of any Loan Party or any other material amendment to any Loan Party's charter, by-laws or other organizational documents, such notice, in each case, to identify the applicable jurisdictions, capital structures or amendments as applicable, provided that such notice shall be given not less than ten (10) Business Days prior to the proposed effectiveness of such changes, acquisition or creation, as the case may be (or such shorter period to which Administrative Agent may consent);
- (f) not less than fifteen (15) Business Days (or such other shorter period to which Administrative Agent may agree) prior to the proposed effective date thereof, any proposed material amendments, restatements or other modifications to any Subordinated Debt documents;
- (g) any default or event of default by any Person under any Subordinated Debt document, concurrently with delivery or promptly after receipt (as the case may be) of any notice of default or event of default under the applicable document, as the case may be; and

(h) promptly after becoming aware thereof, any action taken by the VDEQ with respect to any permit related to the operation of any landfill operated by any Loan Party.

Each notice pursuant to this Section shall be accompanied by a statement of a Responsible Officer of Borrower setting forth details of the occurrence referred to therein and, in the case of notices referred to in clauses (a), (b), (c), (d) and (g) hereof stating what action the applicable Loan Party has taken or proposes to take with respect thereto.

- 4. Management Report. Concurrently with each delivery of financial statements pursuant to Section 1 above, and to the extent such information is not otherwise already included with such financial statements, the Loan Parties will deliver a management report, certified by a Responsible Officer of the Borrower, (a) describing the operations and financial condition of the Loan Parties for the month then ended and the portion of the current Fiscal Year then elapsed (or for the Fiscal Year then ended in the case of year-end financials), (b) setting forth in comparative form the corresponding figures for the corresponding periods of the previous Fiscal Year and the corresponding figures from the most recent projections for the current Fiscal Year, (c) discussing the reasons for any significant variations, and (d) stating that such officer has no knowledge of any Default except as specified in such certificate.
- 5. <u>Tax Returns</u>. As soon as practicable, and in any event within 15 days after the Administrative Agent' request therefor, the Borrower will furnish to the Administrative Agent copies of all Federal, state, local and foreign tax returns (together with all schedules and attachments thereto) filed or required to be filed by any of the Loan Parties.
- 6. Notice of Change of Management or Ownership. Promptly after becoming aware thereof, the Borrower will give the Administrative Agent notice of the departure or anticipated departure of any member of the senior executive management of any Loan Party or any Loan Party's Board of Directors (or similar governing body) or any change in the holders of the Borrower's or any Loan Party's Equity Interests or in the percentages of such securities held by such holders.
- 7. <u>Further Information</u>. The Borrower will promptly furnish to the Administrative Agent such other information regarding the Loan Parties as the Administrative Agent may reasonably request.

SCHEDULE 7.1

NEGATIVE COVENANTS

Each Loan Party covenants and agrees that, so long as any of the Obligations remains outstanding and unpaid, it will not, and, as applicable, it will not permit any of its Subsidiaries to:

- 1. <u>Limitation on Debt.</u> Create, incur, assume or suffer to exist any Debt, except:
 - (a) Obligations of any Loan Party to the Lenders under this Agreement and/or the other Loan Documents;
 - (b) any Debt existing on the Closing Date and set forth in Attachment 1 attached to this Schedule 7.1 and any renewals or refinancing of such Debt; provided that (i) the aggregate principal amount of such renewed or refinanced Debt shall not exceed the aggregate principal amount of the original Debt outstanding on the Closing Date (less any principal payments and the amount of any commitment reductions made thereon on or prior to such renewal or refinancing) plus accrued interest, closing costs, fees and expenses, (ii) the renewal or refinancing of such Debt shall be on substantially the same or better terms as in effect with respect to such Debt on the Closing Date, and shall otherwise be in compliance with this Agreement, and (iii) at the time of such renewal or refinancing no Default or Event of Default has occurred and is continuing or would result from the renewal or refinancing of such Debt:
 - (c) any Debt of Borrower or any Subsidiary incurred to finance the acquisition of fixed or capital assets, whether pursuant to a loan or a Capital Lease provided that both at the time of and immediately after giving effect to the incurrence thereof (i) no Default or Event of Default shall have occurred and be continuing, and (ii) the aggregate amount of all such Debt at any one time outstanding (including, without limitation, any Debt of the type described in this clause (c) which is set forth on Attachment 1 attached to this Schedule 7.1) shall not exceed \$3,300,000, and any renewals or refinancings of such Debt on terms substantially the same or better than those in effect at the time of the original incurrence of such Debt;
 - (d) [Reserved];
 - (e) Debt under any Swap Agreements, provided that such transaction is entered into for risk management purposes and not for speculative purposes;
 - (f) Debt arising from judgments or decrees not deemed to be a Default or Event of Default under subsection (H) of Section 8.1;
 - (g) Debt owing to a Person that is a Loan Party, but only to the extent permitted under Section 7 of this <u>Schedule 7.1</u>;

- (h) Debt of any Loan Party to any other Loan Party;
- (i) Guarantees by any Loan Party of Debt or other obligations arising in the ordinary course of business of any other Loan Party;
- (j) Debt under surety, performance or similar instruments incurred in the ordinary course of business; and
- (k) additional unsecured Debt not otherwise described above, provided that both at the time of and immediately after giving effect to the incurrence thereof (i) no Default or Event of Default shall have occurred and be continuing or result therefrom and (ii) the aggregate amount of all such additional unsecured Debt shall not exceed \$5,500,000 at any one time outstanding.
- 2. <u>Limitation on Liens</u>. Create, incur, assume or suffer to exist any Lien upon any of its property, assets or revenues, whether now owned or hereafter acquired, except for Excepted Liens. Regardless of the provisions of this Section 2 of <u>Schedule 7.1</u>, no Lien over the Equity Interests of Borrower or any Subsidiary of Borrower shall be permitted under the terms of this Agreement.
- 3. <u>Business Activities</u>. Engage in any line of business other than the businesses engaged in as of the Closing Date and businesses reasonably related thereto.
- 4. <u>Limitation on Mergers, Dissolution or Sale of Assets</u>. Enter into any merger or consolidation or convey, sell, lease, assign, transfer or otherwise dispose of any of its property, business or assets (including, without limitation, Equity Interests, receivables and leasehold interests), whether now owned or hereafter acquired or liquidate, wind up or dissolve, in each case unless all Obligations are paid in full in cash prior to or in connection with the consummation of any such transaction, except for:
 - (a) Inventory leased or sold in the ordinary course of business;
 - (b) obsolete, damaged, uneconomic or worn out machinery, parts, property or equipment, or property or equipment no longer used or useful in the conduct of the applicable Loan Party's business;
 - (c) [Reserved];
 - (d) mergers or consolidations of any Subsidiary of Borrower with or into Borrower or any Guarantor so long as the Borrower or such Guarantor shall be the continuing or surviving entity, provided that at the time of each such merger or consolidation, both before and after giving effect thereto, no Default or Event of Default shall have occurred and be continuing or result from such merger or consolidation;

- (e) any Subsidiary of Borrower may liquidate or dissolve into Borrower or a Guarantor if Borrower determines in good faith that such liquidation or dissolution is in the best interests of Borrower, so long as no Default or Event of Default has occurred and is continuing or would result therefrom;
- (f) sales or transfers, including without limitation upon voluntary liquidation from any Loan Party to Borrower or a Guarantor, provided that the applicable Borrower or Guarantor takes such actions as the Administrative Agent may reasonably request to ensure the perfection and priority of the Liens in favor of the Lenders over such transferred assets;
- (g) Asset Dispositions (exclusive of asset sales permitted pursuant to all other subsections of this Section 4 of Schedule 7.1) in which the sales price is at least equal to the fair market value of the assets sold and the consideration received is cash or cash equivalents or Debt of any Loan Party being assumed by the purchaser, provided that the aggregate amount of such Asset Sales does not exceed \$1,100,000 in any Fiscal Year and no Default or Event of Default has occurred and is continuing at the time of each such sale both before and after giving effect to such Asset Disposition;
- (h) the sale or disposition of Permitted Investments and other cash equivalents in the ordinary course of business; and
- (i) dispositions of owned or leased vehicles in the ordinary course of business.

The Lenders hereby consent and agree to the release by the Administrative Agent of any and all Liens on the property sold or otherwise disposed of in compliance with this Section 4 of Schedule 7.1.

- 5. Restricted Payments. Declare or make any distributions, dividend, payment or other distribution of assets, properties, cash, rights, obligations or securities (collectively, "Distributions") on account of any of its Equity Interests, as applicable, or purchase, redeem or otherwise acquire for value any of its Equity Interests, as applicable, or any warrants, rights or options to acquire any of its Equity Interests, now or hereafter outstanding (collectively, "Purchases"), except that:
 - (a) each Subsidiary of the Borrower may pay cash Distributions to the Borrower;
 - (b) each Loan Party may pay reasonable fees and expenses to its directors for attending directors meetings to the extent not exceeding in the aggregate for all Loan Parties \$200,000 per Fiscal Year;
 - (c) each Loan Party may declare and make Distributions payable in kind; and
 - (d) each Loan Party may declare and make Distributions payable in the Equity Interests of such Loan Party, provided that the issuance of such Equity Interests does not otherwise violate the terms of this Agreement and no

Default or Event of Default has occurred and is continuing at the time of making such Distribution or would result from the making of such Distribution.

6. [Reserved].

- 7. <u>Limitation on Investments, Loans and Advances</u>. Make or allow to remain outstanding any Investment (whether such investment shall be of the character of investment in shares of stock, evidences of indebtedness or other securities or otherwise) in, or any loans or advances to, any Person other than:
 - (a) Permitted Investments;
 - (b) Investments existing on the Closing Date and listed on <u>Attachment 3</u> to this Schedule 7.1;
 - (c) sales on open account in the ordinary course of business;
 - (d) intercompany loans or intercompany Investments made by any Loan Party to or in any Guarantor or Borrower, provided that, in the case of any intercompany loans or intercompany Investments made by any Borrower in any Guarantor, the aggregate amount from time to time outstanding in respect thereof shall not exceed \$1,100,000, and provided, further, that in each case, no Default or Event of Default shall have occurred and be continuing at the time of making such intercompany loan or intercompany Investment or result from such intercompany loan or intercompany Investment being made and that any intercompany loans shall be evidenced by and funded under an intercompany note pledged to the Administrative Agent on behalf of the Lenders under the appropriate Security Documents.
 - (e) Investments in respect of Swap Agreements provided that such transaction is entered into for risk management purposes and not for speculative purposes;
 - (f) loans and advances to employees, officers and directors of any Loan Party for moving, entertainment, travel and other similar expenses in the ordinary course of business not to exceed \$100,000 in the aggregate at any time outstanding;
 - (g) [Reserved];
 - (h) Investments constituting deposits made in connection with the purchase of goods or services in the ordinary course of business in an aggregate amount for such deposits not to exceed \$1,100,000 at any one time outstanding; and
 - (i) other Investments not described above provided that both at the time of and immediately after giving effect to any such Investment (i) no Default or Event of Default shall have occurred and be continuing or shall result from the

making of such Investment and (ii) the aggregate amount of all such Investments shall not exceed \$1,100,000 at any time outstanding.

In valuing any Investments for the purpose of applying the limitations set forth in this Section 7 of Schedule 7.1 (except as otherwise expressly provided herein), such Investment shall be taken at the original cost thereof, without allowance for any subsequent write-offs or appreciation or depreciation, but less any amount repaid or recovered on account of capital or principal.

- 8. Transactions with Affiliates. Enter into, amend, restate, or otherwise modify any transaction, agreement, or arrangement, including, without limitation, any purchase, sale, lease or exchange of property or the rendering of any service, with any Affiliates of the Loan Parties, except with respect to any transactions, agreements, or arrangements: (a) with Affiliates that are the Borrower or Guarantor, (b) otherwise permitted under this Agreement, (c) existing as of the Closing Date (on the same terms and conditions as in effect on the Closing Date), and (d) transactions in the ordinary course of a Loan Party's business, consistent with past practice, and upon fair and reasonable terms no less favorable to such Loan Party than it would obtain in a comparable arms'-length transaction from unrelated third parties.
- 9. <u>Limitations on Other Restrictions</u>. Except for this Agreement or any other Loan Document enter into any agreement, document or instrument which would (a) restrict the ability of any Subsidiary of the Borrower to pay or make dividends or distributions in cash or kind to Borrower or Guarantor, to make loans, advances or other payments of whatever nature to any Loan Party, or to make transfers or distributions of all or any part of its assets to any Loan Party, or (b) restrict or prevent any Loan Party from granting Administrative Agent on behalf of the Lenders Liens upon, security interests in and pledges of their respective assets, except to the extent such restrictions exist in documents creating Liens permitted by Section 2 of this <u>Schedule 7.1</u>.
- 10. <u>Prepayment of Debt.</u> Make any prepayment (whether optional or mandatory), repurchase, redemption, defeasance or any other payment in respect of any Subordinated Debt.
- 11. <u>Modification of Certain Agreements</u>. Make, permit or consent to any amendment or other modification to the constitutional documents of any Loan Party or any Material Agreement except to the extent that any such amendment or modification (i) does not violate the terms and conditions of this Agreement or any of the other Loan Documents, (ii) does not materially adversely affect the interest of the Lenders as creditors and/or secured parties under any Loan Document and (iii) would not reasonably be expected to have a Material Adverse Effect.
- 12. [Reserved].

13. <u>Fiscal Year</u>. Permit the Fiscal Year of any Loan Party to end on a day other than December 31.

Schedule 7.5 – Existing Investments

GUARANTY AGREEMENT

This Guaranty Agreement, dated effective as of June 2, 2025 (this "Guaranty Agreement"), is executed by Shoosmith Bros., Inc., a Virginia corporation ("Shoosmith," or, the "Guarantor"), and Volunteer Enterprises, LLC as administrative agent (the "Administrative Agent"), for the benefit of the Lenders party to that certain DIP Credit Agreement (as defined below). The parties agree as follows:

RECITALS

- A. The Guarantor, Administrative Agent, VWS Holdco, Inc., a Delaware corporation (the "Borrower"), and the Lenders (as defined in the DIP Credit Agreement, and together with the Administrative Agent, the "Secured Parties") party thereto from time to time are parties to that certain Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of June 2, 2025 (the "DIP Credit Agreement").
- B. The Lenders require, in order to ensure the prompt and complete payment and performance of the obligations, indebtedness, and liabilities of Borrower to the Secured Parties (the "Obligations," including the meaning assigned to it in the DIP Credit Agreement), that the Guarantor execute and deliver this Guaranty Agreement.
- C. The Guarantor will receive substantial benefit from the Borrower entering into the DIP Credit Agreement.
- D. Capitalized terms used but not defined herein shall have the meanings assigned to them in the DIP Credit Agreement.

AGREEMENT

NOW THEREFORE, in consideration of the Recitals, which are deemed a material part of this Guaranty Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, it is agreed as follows:

1. <u>Incorporation of Recitals</u>. The Recitals set forth above are incorporated into this Guaranty Agreement by this reference thereto and are made a part hereof.

2. Guaranty.

- (a) The Guarantor unconditionally and irrevocably guarantees the due and punctual payment by the Borrower of the Obligations. The Guarantor further agrees that the Obligations may be extended or renewed, in whole or in part, without notice to or further assent from it, and it will remain bound upon this Agreement notwithstanding any extension or renewal of any of the Obligations.
- (b) The Guarantor waives presentation to, demand for payment from and protest to the Borrower or any other Guarantor, and also waives notice of protest for nonpayment. The Obligations of the Guarantor hereunder shall not be affected by (i) the failure of the Administrative Agent or the Lenders to assert any claim or demand or to

enforce any right or remedy against the Borrower or any other Guarantor under the provisions of this Agreement or any other Loan Document or otherwise; (ii) any extension or renewal of any provision hereof or thereof; (iii) any rescission, waiver, compromise, acceleration, amendment or modification of any of the terms or provisions of any of the Loan Documents; (iv) the release, exchange, waiver or foreclosure of any security held by the Administrative Agent for the Obligations or any of them; (v) the failure of the Administrative Agent or the Lenders to exercise any right or remedy against any other Guarantor; or (vi) the release or substitution of any Guarantor or any other Guarantor.

- (c) The Guarantor further agrees that this guaranty constitutes a guaranty of payment when due and not just of collection, and waives any right to require that any resort be had by the Administrative Agent or the Lenders to any security held for payment of the Obligations or to any balance of any deposit, account or credit on the books of the Administrative Agent or the Lenders in favor of the Borrower or any other Guarantor, or to any other Person.
- (d) The Guarantor hereby waives any defense that it might have based on a failure to remain informed of the financial condition of the Borrower and of any other Guarantor and any circumstances affecting the ability of the Borrower to perform under this Agreement.
- (e) The Guarantor's guaranty shall not be affected by the genuineness, validity, regularity or enforceability of the Obligations or any other instrument evidencing any Obligations, or by the existence, validity, enforceability, perfection, or extent of any collateral therefor or by any other circumstance relating to the Obligations which might otherwise constitute a defense to this Guaranty. Neither of the Administrative Agent, nor any of the Lenders makes any representation or warranty in respect to any such circumstances or shall have any duty or responsibility whatsoever to any Guarantor in respect of the management and maintenance of the Obligations.
- (f) Subject to the terms of the DIP Credit Agreement and DIP Financing Orders, upon the Obligations becoming due and payable (by acceleration or otherwise), the Lenders shall be entitled to immediate payment of such Obligations by the Guarantor upon written demand by the Administrative Agent, without further application to or order of the Bankruptcy Court.
- 3. <u>No Impairment of Guaranty.</u> The obligations of the Guarantor hereunder shall not be subject to any reduction, limitation, impairment or termination for any reason, including, without limitation, any claim of waiver, release, surrender, alteration or compromise, and shall not be subject to any defense or set-off, counterclaim, recoupment or termination whatsoever by reason of the invalidity, illegality or unenforceability of the Obligations. Without limiting the generality of the foregoing, the obligations of the Guarantor hereunder shall not be discharged or impaired or otherwise affected by the failure of the Administrative Agent or the Lenders to assert any claim or demand or to enforce any remedy under this Agreement or any other agreement, by any waiver or modification of any provision thereof, by any default, failure or delay, willful or otherwise, in the performance of the Obligations, or by any other act or thing or omission or delay to do any other act or thing which may or might in any manner or to any extent vary the

risk of the Guarantor or would otherwise operate as a discharge of the Guarantor as a matter of law, unless and until the Obligations are paid in full.

- 4. <u>Subrogation</u>. Upon payment by any Guarantor of any sums to the Administrative Agent or the Lenders under this Agreement, all rights of such Guarantor against the Borrower arising as a result thereof by way of right of subrogation or otherwise, shall in all respects be subordinate and junior in right of payment to the prior final and indefeasible payment in full of all the Obligations. If any amount shall be paid to such Guarantor for the account of the Borrower, such amount shall be held in trust for the benefit of the Administrative Agent and the Lenders and shall forthwith be paid to the Administrative Agent and the Lenders to be credited and applied to the Obligations, whether matured or unmatured.
- 5. <u>Reliance</u>. Guarantor acknowledges and agrees that the Lenders will have relied upon and will continue to rely upon this Agreement and the provisions hereof in continuing to deal with Borrower under the DIP Credit Agreement or otherwise.
- 6. <u>Costs of Collection</u>. The liabilities and obligations of Guarantor hereunder will be for the full amount of the Obligations, whatever they may be, and will include all reasonable attorneys' fees, costs of collection and enforcement expenses incurred by Administrative Agent or the Lenders in performing or collecting any of the Obligations at any time. The Guarantor also agrees to pay to the Administrative Agent and the Lenders, on demand, all reasonable costs and expenses incurred by any of them, including the reasonable fees and disbursements of counsel, in connection with the enforcement of and preservation of any of its rights hereunder.
- 7. Certain Actions Not to Affect Obligations. Nothing in this Agreement or otherwise will impair or adversely affect any of the rights or remedies of the Administrative Agent (on behalf of the Lenders) in connection with the DIP Credit Agreement or otherwise, all of which rights and remedies are reserved by the Administrative Agent (on behalf of the Lenders). Nothing in this Agreement will create any liability or obligation of the Administrative Agent or Lenders to the Guarantor or Borrower with respect to how the Administrative Agent or Lenders exercise their rights with respect to the Guarantor or Borrower. Without limiting the Administrative Agent's or Lenders' other rights, regardless of whether or not any existing relationship between the Guarantor and Borrower or any other Party or Affiliate of any of them has been changed or ended, the Guarantor agrees that at any time and from time to time, without notice to or consent of the Guarantor and without incurring responsibility to the Guarantor and without impairing or releasing the liabilities or obligations provided for herein or otherwise impairing the rights of the Administrative Agent hereunder:
 - (a) the time for performance of or compliance with any covenant or agreement in any Loan Document or otherwise may be extended or such performance may be waived by any applicable party under any Loan Document;
 - (b) any covenant or agreement in any Loan Document may from time to time be replaced or modified, for the purpose of adding any requirement thereto or changing in any manner any right, liability or obligation of any party to any Loan Document;

- (c) any manner, place or term for payment of any Guaranteed Obligation may be altered or any term for payment extended, or any Guaranteed Obligation may be renewed in whole or in part; and
- (d) the maturity of any of the Guaranteed Obligations may be accelerated in accordance with the terms of any present or future agreement between the applicable parties to any Loan Document.
- 8. Events of Default. The occurrence of any of the following events shall constitute an Event of Default by the Guarantor under this Agreement (referred to herein as an "Event of Default"): (i) any default under the DIP Credit Agreement and the expiry of any grace periods contained therein and applicable thereto; (ii) breach of any covenant, provision, or term of this Agreement; (iii) the Secured Parties' lien on the Collateral fails to constitute a security interest on such Collateral; or (iv) the Guarantor does, or permits to be done, anything that in any way may materially impair the security and lien of this Agreement or materially impairs the value of the Collateral, or any part thereof, or materially weakens or diminishes the security intended to be given under and by virtue of Security Documents.
- 9. <u>Reinstatement</u>. If any payment applied to any of the Obligations is thereafter set aside, recovered, rescinded or required to be returned for any reason, the Obligations to which such payment was applied will for the purposes of this Agreement be deemed to have continued in existence, notwithstanding such application, and this Agreement will be enforceable as to such Obligations as fully as if such application had never been made.
- 10. <u>Specific Performance</u>. The Guarantor hereby authorizes the Administrative Agent (on behalf of the Lenders) to demand specific performance of this Agreement at any time when the Guarantor shall have failed to comply with any provision hereof, and the Guarantor hereby irrevocably waives any defense based on the adequacy of a remedy at law which might be asserted as a bar to the remedy of specific performance hereof in any action brought therefor.
- 11. <u>Cumulative Remedies</u>. The rights and remedies herein provided are cumulative, may be exercised singly or concurrently and are not exclusive of any rights or remedies provided by applicable law or otherwise.
- 12. <u>No Third Party Beneficiaries</u>. This Agreement is entered into for the sole protection and benefit of the Secured Parties and their successors and assigns, and no other Person will be a direct or indirect beneficiary of, or will have any direct or indirect cause of action or claim in connection with, this Agreement. This Agreement will be binding on and inure to the benefit of the respective permitted successors and assigns of the Secured Parties.
- 13. <u>Headings</u>. Section headings herein are included for convenience of reference only and shall not affect the interpretation of this Agreement.
- 14. Governing Law; Submission to Jurisdiction; Waiver of Venue, Service of Process, Waiver of Right to Jury Trial. The terms of Section 10.1 and Section 10.6 of the DIP Credit Agreement with respect to governing law, submission to jurisdiction, waiver of venue, service of process and waiver of the right to a jury trial are each incorporated herein by reference, mutatis mutandis, and the parties hereto agree to such terms.

- 15. <u>Severability</u>. If any provision of this Agreement is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 16. <u>Entirety</u>. This Agreement, the other Loan Documents and the other documents relating to the Obligations represent the entire agreement of the parties hereto and thereto, and supersede all prior agreements and understandings, oral or written, if any, including any proposal letters or correspondence relating to the Loan Documents, any other documents relating to the Obligations, or the transactions contemplated herein and therein.
- 17. <u>Survival</u>. All representations and warranties of the Guarantor hereunder shall survive the execution and delivery of this Agreement, the other Loan Documents and the other documents relating to the Obligations, and the investments made thereunder or in connection therewith.
- 18. <u>Rights of Required Lenders</u>. All rights of the Administrative Agent hereunder, if not exercised by the Administrative Agent, may be exercised by the Required Lenders.
- 19. <u>Termination</u>. This Agreement will terminate upon the payment and performance in full of all of the Obligations, all without delivery of any instrument or performance of any act by any party. At the request of the Guarantor following any such termination, the Administrative Agent shall deliver to the Guarantor such documents as the Guarantor shall reasonably request to evidence such termination.

[Remainder of page intentionally left blank. Signature page follows.]

Case 25-10979-JKS Doc 50-1 Filed 06/02/25 Page 296 of 298

IN WITNESS WHEREOF, the date first written above.	e parties have executed this Guaranty Agreement as of the
	GUARANTOR:
	SHOOSMITH BROS., INC.
	By: Name: Title:
ACKNOWLEDGED BY:	
	ADMINISTRATIVE AGENT:
	VOLUNTEER ENTERPRISES, LLC
	By:
	Name: Title:
	BORROWER:
	VWS HOLDCO, INC.
	By:
	Name: Fred Nichols

Title: President

EXHIBIT 2

INITIAL DIP BUDGET

VWS Holdco, Inc., et al.

DIP Budget (\$000's USD)

Week#	Week 1		Week 3	Week 4	Week 5	Week 6	Week 7	Week 8	Week 9		Week 11	Week 12	Week 13	Total
Actual / Forecast				Forecast			Forecast		Forecast		Forecast	Forecast	Forecast	0/4/05
Start Date	6/1/25			6/21/25	6/28/25	7/5/25	7/12/25		7/26/25		8/9/25	8/16/25	8/23/25	6/1/25
End Date	6/6/25	6/13/25	6/20/25	6/27/25	7/4/25	7/11/25	7/18/25	7/25/25	8/1/25	8/8/25	8/15/25	8/22/25	8/29/25	8/29/25
Receipts														
Receipts	\$ - \$	- \$	- \$	- \$	- \$	- \$		- \$	- \$	- \$	- \$	- \$	-	\$ -
Gas Royalty Receipts	-	-	50	-	-	-	50	-	-	-	50	-	-	150
Other Receipts	-	-	-	-	-	-	-	-	-	-	-	-	-	 -
Subtotal	\$ - \$	- \$	50 \$	- \$	- \$	- \$	50 \$	5 - \$	- \$	- \$	50 \$	- \$	<u>-</u>	\$ 150
Operating Disbursements														
Salaries & Benefits	\$ (13) \$	(21) \$	(13) \$	(21) \$	(13) \$	(21) \$	(13)	(21) \$	(13) \$	(21) \$	(13) \$	(21) \$	(13)	\$ (216)
Processing & Transportation	(90)	(90)	(90)	(90)	(90)	(90)	(90)	(90)	(90)	(90)	(90)	(90)	(90)	(1,170)
Rent	(11)	-	-	-	(11)	-	-	-	(11)	-	-	-	-	(33)
Fuel	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(12)
Equipment Leasing	-	-	-	-	(6)	-	-	-	(6)	-	-	-	(6)	(18)
Repairs	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(124)
Engineering	(42)	-	-	-	(42)	-	-	-	(42)	-	-	-	-	(126)
Insurance	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(5)
Utilities	(11)	-	-	-	(11)	-	-	-	(11)	-	-	-	-	(33)
Other	 (6)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	 (47)
Subtotal	\$ (184) \$	(126) \$	(117) \$	(126) \$	(187) \$	(126) \$	(117)	(126) \$	(187) \$	(126) \$	(117) \$	(126) \$	(123)	\$ (1,783)
Operating Cash Flow	\$ (184) \$	(126) \$	(67) \$	(126) \$	(187) \$	(126) \$	(67)	(126) \$	(187) \$	(126) \$	(67) \$	(126) \$	(123)	\$ (1,633)
Non-Operating Disbursements														
Bankruptcy-Related Costs	\$ - \$	- \$	(515) \$		(200) \$	- \$			- \$		- \$			\$ (715)
Professional Fees	(196)	(196)	(196)	(196)	(196)	(196)	(196)	(196)	(196)	(196)	(196)	(196)	(188)	 (2,540)
Subtotal	\$ (196) \$	(196) \$	(711) \$	(196) \$	(396) \$	(196) \$	(196)	(196) \$	(196) \$	(196) \$	(196) \$	(196) \$	(188)	\$ (3,255)
Net Cash Flow	\$ (380) \$	(322) \$	(778) \$	(322) \$	(583) \$	(322) \$	(263)	322) \$	(383) \$	(322) \$	(263) \$	(322) \$	(311)	\$ (4,888)
Book Cash														
Beginning Balance	\$ 155 \$	1,575 \$	1,254 \$	476 \$	154 \$	1,122 \$	800	538 \$	1,766 \$	1,383 \$	1,062 \$	799 \$	478	\$ 155
DIP Funding	1,800	-	-	-	1,550	-	-	1,550	-	-	-	-	-	4,900
Repayments	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Net Cash Flow	(380)	(322)	(778)	(322)	(583)	(322)	(263)	(322)	(383)	(322)	(263)	(322)	(311)	 (4,888)
Ending Balance	\$ 1,575 \$	1,254 \$	476 \$	154 \$	1,122 \$	800 \$	538	1,766 \$	1,383 \$	1,062 \$	799 \$	478 \$	167	\$ 167
DIP Balance														
Beginning Balance	\$ - \$	1,800 \$	1,800 \$	1,800 \$	1,800 \$	3,618 \$	3,618	3,618 \$	5,418 \$	5,458 \$	5,458 \$	5,458 \$	5,458	\$ -
Borrowings	1,800	-	-	-	1,550	-	-	1,550	-	-	-	-	-	4,900
Repayments	-	-	-	-	-	-	-	-	-	-	-	-	-	-
Bridge Roll-Up	-	-	-	-	250	-	-	250	-	-	-	-	-	500
PIK Interest	-	-	-	-	18	-	-	-	40	-	-	-	54	 112
Ending Balance	\$ 1,800 \$	1,800 \$	1,800 \$	1,800 \$	3,618 \$	3,618 \$	3,618	5,418 \$	5,458 \$	5,458 \$	5,458 \$	5,458 \$	5,512	\$ 5,512

Exhibit B

Interim Order

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In mor) Chapter 11
In re:) Case No. 25-10979 (JKS)
VWS HOLDCO, INC., et al.,) (Joint Administration Requested
Debtors. ¹)
) Re: D.I. 15

INTERIM ORDER PURSUANT TO

SECTIONS 105, 361, 362, 363, 364, 503, 506, 507 AND 552 OF THE BANKRUPTCY CODE AND RULES 2002, 4001, 6004 AND 9014 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE (I) AUTHORIZING THE DEBTORS TO (A) USE CASH COLLATERAL, (B) OBTAIN SENIOR SECURED SUPERPRIORITY POSTPETITION FINANCING AND GRANTING LIENS AND SUPERPRIORITY ADMINISTRATIVE CLAIMS, AND (C) PROVIDE ADEQUATE PROTECTION, (II) MODIFYING THE AUTOMATIC STAY, (III) SCHEDULING A FINAL HEARING, AND (IV) GRANTING RELATED RELIEF

Upon consideration of the motion (the "Motion") of the above-captioned debtors and debtors in possession (the "Debtors") in the above-captioned chapter 11 cases (the "Chapter 11 Cases"), for the entry of an order pursuant to sections 105, 361, 362, 363, 364, 503, 506, 507 and 552 of title 11 of the United States Code (as amended, the "Bankruptcy Code"), rules 2002, 4001, 6004, and 9014 of the Federal Rules of Bankruptcy Procedure (as amended, the "Bankruptcy Rules") and rules 2002-1, 4001-1, 4001-2, 9006-1, and 9013-1 of the Local Rules of Bankruptcy Practice and Procedure for the District of Delaware (the "Local Rules") (i) authorizing the Debtors to (a) use cash collateral, (b) obtain senior secured superpriority postpetition financing and granting liens and superpriority administrative expenses claims and (c) provide adequate

The Debtors in these Chapter 11 Cases, along with the last four digits of each Debtor's federal tax identification numbers are as follows: VWS Holdco, Inc. (5412) and Shoosmith Bros., Inc. (6914). The Debtors' mailing address is P.O. Box 2770, Chesterfield, VA 23832.

protection; (ii) modifying the automatic stay; (iii) scheduling interim and final hearings; and (iv) granting related relief, the Debtors sought, among other things, the following relief:

(i) authorizing Debtor VWS Holdco, Inc, as borrower, (the "Borrower") and Debtor Shoosmith Bros., Inc. as guarantor, (the "Guarantor") to obtain a senior secured superpriority debtor-in-possession financing facility in an aggregate amount of \$5,400,000.00 (the "DIP Facility"), comprised of \$4,900,000.00 of new money, pursuant to the terms herein, the terms and conditions set forth in the credit agreement attached hereto as Exhibit 1 (as may be amended, restated, supplemented or otherwise modified from time to time, in each case with the consent of the DIP Lenders (as defined below), the "DIP Credit Agreement," collectively with this order, this "Interim Order", the Final Order (as defined below), the DIP Budget (as defined below), and all other agreements, documents, and instruments delivered or executed in connection therewith to the extent that the DIP Agent (as defined below) and the DIP Lenders determine there is a need for such other agreements, documents, and instruments, in each case as amended, restated, supplemented, or otherwise modified from time to time, the "DIP Loan Documents") provided by Volunteer Enterprises, LLC, as administrative agent (together with any successor administrative agents and collateral agents, the "DIP Agent"), and the parties listed on Annex B to the DIP Credit Agreement, as lenders (in such capacity, together with any successors and assigns permitted under the DIP Loan Documents, the "DIP Lenders" and, together with the DIP Agent, the "DIP Secured Parties"), which DIP Facility shall be available, subject to the terms and conditions set forth in this Interim Order and the other DIP Loan Documents, as follows: (i) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.1, the DIP Lenders shall make available during the period (the "Interim Period") from the date hereof through and including the

Capitalized terms used but not defined herein have the same meaning ascribed to such terms in the DIP Credit Agreement or the Motion, as appropriate.

date of entry of the Final Order by this Court, in one or more additional draws the aggregate amount of \$1,800,000.00 (the "First Interim DIP Loan"), to include a roll-up based on the interim funding of \$500,000.00 (the "Interim Roll-Up Amount"); (ii) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.2, the DIP Lenders shall make available in one or more additional draws the aggregate amount of \$1,550,000.00 (the "Second Interim DIP Loan"); and (iii) subject to the terms and conditions set forth in the DIP Loan Documents, including Section 4.3, the DIP Lenders shall make available in one or more additional draws the aggregate amount of \$1,550,000.00 (the "Third Interim DIP Loan"), and with the First Interim DIP Loan and the Second Interim DIP Loan, collectively, the "DIP Loans;

- (ii) the Court's authorization for the Debtors to execute, deliver, and perform, as applicable, under the DIP Credit Agreement and all other DIP Loan Documents, and to perform such other and further acts as may be necessary or appropriate in connection therewith;
- (iii) the Court's authorization for the Debtors to use proceeds of the DIP Loans immediately upon entry of the Interim Order solely in accordance with the budget and cash flow forecast prepared by the Debtors and annexed hereto as **Exhibit 2** (as updated from time to time in accordance with the terms of the DIP Credit Agreement and the other DIP Loan Documents, subject to the prior approval of the DIP Lenders in their sole discretion, the "Initial DIP Budget"), subject to the Permitted Variances;
- (iv) the Court's authorization to grant to the DIP Agent, for the benefit of the DIP Secured Parties, in respect of the DIP Obligations (as defined below), subject to the Carve-Out (as defined below) a superpriority administrative claim pursuant to section 364(c)(1) of the Bankruptcy Code and first priority priming liens on and security interests in substantially all assets and property of the Debtors (now owned or hereafter acquired) pursuant to sections 364(c)(2),

(c)(3), and (d)(1) of the Bankruptcy Code, in each case as and to the extent, set forth more fully below and in the DIP Loan Documents (including the DIP Credit Agreement);

- (v) the Court's authorization for the Debtors to use "cash collateral" as such term is defined in section 363 of the Bankruptcy Code in which the Prepetition Secured Parties (as defined below) or the DIP Secured Parties has an interest (the "Cash Collateral") solely in accordance with the DIP Budget (subject to Permitted Variances) and the DIP Credit Agreement;
- (vi) the Court's authorization to grant, as of the Petition Date, adequate protection for the benefit of the Prepetition Secured Parties, as set forth more fully below, including the Adequate Protection Superpriority Claim, Adequate Protection Liens, and Adequate Protection Payments (each as defined below), to the extent of and as compensation for any Diminution in Value (as defined below), the payment of fees of the Prepetition Secured Parties, if any, and the Debtors' satisfaction of the Milestones (as defined below) in accordance with the DIP Credit Agreement;
- (vii) the modification or waiver by the Court of the automatic stay imposed by section 362 of the Bankruptcy Code and any other applicable stay (including Bankruptcy Rule 6004) to the extent necessary to implement and effectuate the terms and provisions of the DIP Facility and this Interim Order and the other DIP Loan Documents, and to provide for the immediate effectiveness of this Interim Order;
- (viii) the scheduling by the Court of a final hearing (the "<u>Final Hearing</u>") to consider entry of an order (the "<u>Final Order</u>") granting the relief requested in the Motion on a final basis and approving the form of notice with respect to the Final Hearing and the transactions contemplated by the Motion; and
 - (ix) entry of the Final Order.

The Court having considered the Motion, the terms of the DIP Facility and the DIP Loan Documents (including the DIP Credit Agreement), the Declaration of Steven Agran in Support of Debtors' Motion for Entry of Interim and Final Orders (I) Authorizing the Debtors to (A) Use Cash Collateral, (B) Obtain Senior Secured Superpriority Postpetition Financing and Granting Liens and Superpriority Administrative Claims, and (C) Provide Adequate Protection, (II) Scheduling a Final Hearing, and (III) Granting Related Relief, and the Declaration of Steven Agran, as Chief Restructuring Officer, in Support of Chapter 11 Petitions and First Day Pleadings (collectively, the "Declarations") and the evidence submitted at the hearing held before this Court on June 3, 2025, to consider entry of this Interim Order (the "Interim Hearing"); and due and sufficient notice of the Motion and the Interim Hearing having been provided by the Debtors as set forth in paragraph E of this Interim Order; and the Interim Hearing having been held and concluded; and it appearing that approval of the interim relief requested in the Motion is necessary to avoid immediate and irreparable harm to the Debtors pending the Final Hearing and is otherwise fair and reasonable and in the best interests of the Debtors, their creditors, and their estates and stakeholders, and essential for the continued operation of the Debtors' businesses and preservation of the value of the Debtors' assets; and it appearing that the Debtors' entry into the DIP Documents is a sound and prudent exercise of the Debtors' business judgment; and all objections, if any, to the entry of this Interim Order having been withdrawn, resolved, or overruled by the Court; and after due deliberation and consideration, and for good and sufficient cause appearing therefor:

BASED UPON THE RECORD ESTABLISHED AT THE INTERIM HEARING, IT IS FOUND AND DETERMINED THAT:³

- A. <u>Petition Date</u>. On June 1, 2025 (the "<u>Petition Date</u>"), the Debtors filed voluntary petitions under Chapter 11 of the Bankruptcy Code with the United States Bankruptcy Court for the District of Delaware (the "Court").
- B. <u>Debtors in Possession</u>. The Debtors are authorized to continue in the management and operation of their businesses and property as debtors in possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee or examiner has been appointed in the Chapter 11 Cases.
- C. <u>Jurisdiction and Venue</u>. The Court has jurisdiction over this proceeding, pursuant to 28 U.S.C. §§ 157 and 1334 and the *Amended Standing Order of Reference* dated February 29, 2012, from the United States District Court for the District of Delaware. Consideration of the Motion constitutes a core proceeding under 28 U.S.C. § 157(b)(2), and the Court may enter a final order consistent with Article III of the United States Constitution. Venue for the Chapter 11 Cases and the proceeding on the Motion is proper in this district pursuant to 28 U.S.C. §§ 1408 and 1409.
- D. <u>Committee Formation</u>. As of the date hereof, no official committee of unsecured creditors has been appointed in the Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code (a "Committee").
- E. <u>Notice</u>. The Interim Hearing was held pursuant to Bankruptcy Rule 4001(b)(2) and (c)(2). Proper, timely, adequate, and sufficient notice of the Interim Hearing and the relief

6

³ The findings and conclusions set forth herein constitute the Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent that any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent that any of the following conclusions of law constitute findings of fact, they are adopted as such.

requested in the Motion has been provided to (a) the Office of the United States Trustee for the District of Delaware (the "U.S. Trustee"), (b) counsel to the Prepetition Secured Parties and DIP Secured Parties, (c) all other parties asserting a lien on or a security interest in the assets of the Debtors to the extent reasonably known to the Debtors, (d) the Office of the United States Attorney General for the State of Delaware, (e) the Internal Revenue Service, (f) those creditors holding the 20 largest unsecured claims against the Debtors' estates (excluding insiders), (g) the Securities and Exchange Commission, and (h) all parties who have filed a notice of appearance and request for service of papers pursuant to Bankruptcy Rule 2002 (the "Notice Parties"), as evidenced by the certificate of service [D.I_]. Under the circumstances, such notice of the Interim Hearing and the relief requested in the Motion complies with section 102(1) of the Bankruptcy Code, Bankruptcy Rules 2002 and 4001(b), (c), and (d), and the Local Rules, and no other or further notice need be provided for entry of this Interim Order.

- F. <u>Prepetition Indebtedness</u>. The Borrower is a party to that certain Amended and Restated Securities Purchase Agreement (as amended, the "<u>Prepetition Term Loan Agreement</u>"), originally dated as of April 30, 2020, by and among the Borrower, as borrower, Guarantor, Volunteer Enterprises, LLC (as successor in interest to PNC Mezzanine Partners III, L.P., the "<u>Prepetition Agent</u>") and the lender parties identified on Annex B thereto (the "<u>Prepetition Lenders</u>" or the "<u>Prepetition Secured Parties</u>").
- G. <u>Stipulations as to Prepetition Obligations</u>. After consultation with their attorneys and financial advisors, as a condition to provide the DIP Facility as set forth herein and subject to the rights of any party in interest as and to the extent set forth in paragraph 11 hereof, the Debtors, on their behalf, acknowledge, stipulate, and agree with the following: (collectively, the "Stipulations"):

- (i) **Prepetition Obligations**. Pursuant to that certain Prepetition Term Loan Agreement, the Debtors issued the Senior Notes, Junior Notes and Bridge Notes (collectively, the "<u>Prepetition Loans</u>") to the Prepetition Secured Parties. As of the Petition Date, the aggregate principal and accrued interest owed to the Prepetition Secured Parties was not less than \$183,050,584.19, plus other interest, fees and all other obligations under the Prepetition Term Loan Agreement, each of the Prepetition Loans and the other documents related thereto (collectively, the "<u>Prepetition Obligations</u>").
- (ii) Enforceability, etc. of the Prepetition Obligations. The Debtors acknowledge that the Prepetition Term Loan Agreement and the Prepetition Obligations are legal, valid, binding and enforceable obligations against the Debtors. As a condition to providing the DIP Facility, the DIP Lenders have required and the Debtors have agreed not to contest or challenge the Prepetition Obligations under the Bankruptcy Code, under applicable non-bankruptcy law or otherwise.
- that the liens granted by the Debtors under the Prepetition Term Loan Agreement to the Prepetition Agent (the "Prepetition Liens") for the benefit of the Prepetition Secured Parties as security for the Prepetition Obligations encumber all Collateral (as defined in the Prepetition Term Loan Agreement), which includes substantially all of the Debtors' assets and property and all proceeds thereof (collectively, the "Prepetition Collateral") were granted to or for the benefit of the Prepetition Secured Parties for fair consideration and reasonably equivalent value. The Debtors acknowledge that as of the Petition Date, (a) the Debtors granted to the Prepetition Secured Parties the Prepetition Liens, which Prepetition Liens are subject only to any permitted Liens set forth in section 2 of Schedule 3.14 of the Prepetition Term Loan Agreement (the "Prepetition Permitted Liens"); (b) the

Prepetition Obligations constitute legal, valid and binding obligations of the Debtors enforceable in accordance with the terms of the Prepetition Term Loan Agreement; (c) the Debtors have no claims, objections, causes of action, and/or choses in action, including avoidance claims under Chapter 5 of the Bankruptcy Code or applicable state law equivalents or actions for recovery or disgorgement, against any of the Prepetition Secured Parties or any of their respective affiliates, agents, attorneys, advisors, professionals, officers, directors, and employees arising out of, based upon, or related to the Prepetition Term Loan Agreement; (d) the Debtors have waived, discharged, and released any right to challenge any of the Prepetition Obligations, the priority of the Debtors' obligations thereunder, and the validity, extent and priority of the liens securing the Prepetition Obligations; and (e) the Prepetition Obligations constitute allowed, secured claims within the meaning of sections 502 and 506 of the Bankruptcy Code.

(iv) Indemnity. The Prepetition Secured Parties and the DIP Secured Parties and their Related Parties (as defined below) have acted in good faith and without negligence, misconduct, or violation of public policy or law, in respect of all actions taken by them in connection with or related in any way to negotiating, implementing, documenting, or obtaining requisite approvals of the DIP Facility and the use of Cash Collateral, including in respect of the granting of the DIP Liens, the DIP Superpriority Claim, the Adequate Protection Liens, the Adequate Protection Superpriority Claim (each as defined below), and any of the other rights, privileges, remedies, and protections granted hereunder or under the DIP Loan Documents (including the DIP Credit Agreement), any challenges or objections to the DIP Facility or the use of Cash Collateral, and all documents related to and all transactions contemplated by the foregoing. Accordingly, without limitation to any other right to indemnification (including any and all rights of the Prepetition Secured Parties to indemnification under the Prepetition Term Loan Agreement),

the Prepetition Secured Parties and the DIP Secured Parties and each of their respective affiliates, officers, directors, fiduciaries, employees, agents, advisors, attorneys, and representatives (collectively, the "Related Parties") shall be and hereby are indemnified and held harmless by the Debtors in respect of any claim or liability incurred in respect thereof or in any way related thereto; provided, however, that the Debtors shall not indemnify the Prepetition Secured Parties against (i) any successful Challenge under the Challenge Period (each as defined herein) or (ii) any losses, claims, damages, liabilities, or related expenses to the extent determined by a final, unappealable judgment of a court of competent jurisdiction to have been incurred solely by reason of the gross negligence, willful misconduct, or fraud of such Related Parties. No exception or defense in contract, law, or equity exists as to any obligation set forth, as the case may be, in this paragraph G (iv), in the Prepetition Term Loan Agreement, or in the DIP Credit Agreement, to indemnify and/or hold harmless any Prepetition Secured Party, DIP Secured Party, and any Related Party, as the case may be, and any such defenses are hereby waived.

- Parties, in their capacities as DIP Secured Parties or Prepetition Secured Parties, are control persons or insiders of the Debtors or any of their affiliates by virtue of any of the actions taken with respect to, in connection with, related to, or arising from the DIP Facility, the DIP Loan Documents (including the DIP Credit Agreement), and/or the Prepetition Term Loan Agreement. For the avoidance of doubt, nothing in this paragraph G (v) is intended to release or change the status of parties as control persons, if they are, outside of their capacity as DIP Secured Parties or Prepetition Secured Parties.
- (vi) No Claims, Causes of Action. As of the date hereof, other than as previously disclosed to the Lenders, there exist no other claims or causes of action against any of

the Prepetition Secured Parties, the DIP Secured Parties, or their Related Parties with respect to, in connection with, related to, or arising from the Prepetition Term Loan Agreement and/or the DIP Loan Documents (including the DIP Credit Agreement) that may be asserted by the Debtors.

- (vii) **Cash Collateral**. Any and all of the Debtors' cash, whether existing as of the Petition Date or thereafter, wherever located (including, without limitation, all cash or cash equivalents in the control of or on deposit or maintained by the Debtors in any account or accounts, any amounts generated by the sale or other disposition of Prepetition Collateral, and all income, proceeds, products, rents or profits of any Prepetition Collateral), constitutes Cash Collateral of the Prepetition Secured Parties.
- (viii) **Default**. The Debtors are in default under the Prepetition Term Loan Agreement and an event of default has occurred thereunder.
- (ix) Release. Subject to entry of the Final Order, the Debtors hereby forever and irrevocably release, discharge, and acquit all former and current (a) DIP Secured Parties, (b) Prepetition Secured Parties, (c) the Related Parties, (d) Affiliates of the DIP Secured Parties and Prepetition Secured Parties, and (e) officers, employees, directors, agents, representatives, owners, members, partners, financial and other advisors and consultants, legal advisors, shareholders, managers, consultants, accountants, attorneys, investment committee members, sub advisors and predecessors and successors in interest of each of the DIP Secured Parties and Prepetition Secured Parties and each of their respective Affiliates, in each case acting in their respective capacities as such (collectively, the "Releasees") of and from any and all claims, demands, liabilities, responsibilities, disputes, remedies, causes of action, indebtedness and obligations, rights, assertions, allegations, actions, suits, controversies, proceedings, losses, damages, injuries, attorneys' fees, costs, expenses, or judgments of every type, whether known, unknown, asserted,

unasserted, suspected, unsuspected, accrued, unaccrued, fixed, contingent, pending or threatened including, without limitation, all legal and equitable theories of recovery, arising under common law, statute or regulation or by contract, of every nature and description, arising out of, in connection with, or relating to the Prepetition Loans, the DIP Facility, the DIP Loan Documents (including the DIP Credit Agreement), the Prepetition Term Loan Agreement, and/or the transactions contemplated hereunder or thereunder including, without limitation, (x) any so-called "lender liability" or equitable subordination claims or defenses, (y) any and all claims and causes of action arising under the Bankruptcy Code, and (z) any and all claims and causes of action with respect to the validity of the liens or claims of any of the Prepetition Secured Parties or the DIP Secured Parties (the "Releases"); provided that nothing in this paragraph is intended to limit or release (i) any commitments and obligations of any of the DIP Secured Parties under the DIP Credit Agreement or (ii) any claims, causes of action, costs or demands of whatever kind or nature, whether known or unknown, liquidated or unliquidated, fixed or contingent, asserted or unasserted, foreseen or unforeseen, or matured or unmatured, resulting solely from the gross negligence or willful misconduct or fraud of the Releasees, as determined by a court of competent jurisdiction in a final and unappealable judgment or order. The Debtors further waive and release any defense, right of counterclaim, right of setoff or deduction to the payment of the Prepetition Obligations and the DIP Obligations which the Debtors now have or may claim to have against the Releasees arising out of, connected with, or relating to any and all acts, omissions, or events occurring prior to the entry of this Interim Order by the Court.

(x) **Sale and Credit Bidding**. The DIP Secured Parties and the Prepetition Secured Parties shall have the right, as provided for in section 363(k) of the Bankruptcy Code, to credit bid (independently or together) up to the full amount of the applicable outstanding Prepetition

Obligations (including any Adequate Protection Superpriority Claims) and the DIP Obligations, in each case including, without limitation, any accrued interest and fees, in a sale of any DIP Collateral (as defined below) or Prepetition Collateral, as applicable, and whether such sale is effectuated through sections 363 or 1129 of the Bankruptcy Code, or otherwise.

H. <u>Immediate Need for Postpetition Financing and Use of Cash Collateral.</u> The Debtors have requested immediate entry of this Interim Order pursuant to Bankruptcy Rules 4001(b)(2) and (c)(2). Good cause has been shown for entry of this Interim Order as the Debtors do not have sufficient liquidity to operate and maintain their business in the ordinary course of business without the financing requested in the Motion. The Debtors' ability to maintain business relationships with their vendors and suppliers, to pay their employees and to otherwise fund their operations, payroll and other efforts and activities is essential to the Debtors' continued viability as the Debtors seek to maximize and preserve the value of their estates pending the Final Hearing. The ability of the Debtors to finance their operations through the incurrence of the First Interim DIP Loan and use of Cash Collateral is vital to the preservation and maintenance of the going concern value of the Debtors' estates, to permit the orderly continuation of the operation and maintenance of their businesses, minimize the disruption of their operations, and implement and pursue a comprehensive marketing and sale process that maximizes the value of the Debtors' assets for the benefit of their creditors, and to otherwise avoid immediate and irreparable harm to the Debtors, their estates, and their creditors.

I. No Credit Available on More Favorable Terms. The Debtors have been unable to obtain credit on more favorable terms and conditions than those provided in this Interim Order, including for (a) adequate unsecured credit allowable under section 503(b)(1) of the Bankruptcy Code as an administrative expense, (b) credit for money borrowed with priority over any or all

administrative expenses of the kind specified in sections 503(b) or 507(b) of the Bankruptcy Code, (c) credit for money borrowed secured by a lien on property of the estates that is not otherwise subject to a lien, or (d) credit for money borrowed secured by a junior lien on property of the estates which is subject to a lien. The Debtors are unable to obtain credit for borrowed money without granting the DIP Liens and the DIP Superpriority Claim to (or for the benefit of) the DIP Secured Parties and without granting the adequate protection as set forth herein (in each case, subject to the Carve-Out (as defined below)). The Debtors have therefore concluded, in the exercise of their sound business judgment, that the DIP Loans represent the best financing available to the Debtors at this time and are in the best interests of their estates and all stakeholders.

- J. <u>Use of Cash Collateral and Proceeds of the DIP Facility, DIP Collateral, and Prepetition Collateral.</u> Subject to the Challenge Period, the Debtors represent and stipulate that all of the Debtors' cash, cash equivalents, negotiable instruments, investment property, and securities constitute Cash Collateral of the Prepetition Agent on behalf of the Prepetition Secured Parties and the DIP Agent on behalf of the DIP Secured Parties. All Cash Collateral, all proceeds of the Prepetition Collateral and the DIP Collateral, including proceeds realized from any sale or disposition thereof, or from payment thereon, and all proceeds of the DIP Facility (net of any amounts used to pay fees, costs, and expenses payable under the DIP Facility pursuant to this Interim Order or the Final Order) shall be used or applied in accordance with the terms and conditions of this Interim Order, the DIP Budget (subject to the Permitted Variances), and the DIP Loan Documents (including the DIP Credit Agreement) and for no other purpose.
- K. <u>Adequate Protection for the Prepetition Secured Parties</u>. The Prepetition Secured Parties have negotiated in good faith regarding the Debtors' use of the Prepetition Collateral (including the Cash Collateral) to fund the administration of the Debtors' estates and

the continued operation of their businesses, in accordance with the terms hereof and the DIP Budget (subject to the Permitted Variances). The Prepetition Secured Parties have agreed to permit the Debtors to use the Prepetition Collateral, including the Cash Collateral, in accordance with the terms hereof and the DIP Budget (subject to the Permitted Variances) during the Interim Period subject to the terms and conditions set forth herein, including the protections afforded parties acting in "good faith" under section 364(e) of the Bankruptcy Code. The Prepetition Secured Parties are entitled to the adequate protection as and to the extent set forth herein pursuant to sections 361, 362, 363, and 364 of the Bankruptcy Code. Based on the Motion and on the record presented to the Court at the Interim Hearing, the terms of the proposed adequate protection arrangements and of the use of the Prepetition Collateral (including the Cash Collateral) are fair and reasonable, reflect the Debtors' prudent exercise of business judgment and constitute reasonably equivalent value and fair consideration for the Prepetition Agent's consent thereto; provided that nothing in this Interim Order shall (i) be construed as a consent by any Prepetition Secured Party (a) that it would be adequately protected in the event debtor-in-possession financing is provided by a third party (i.e., other than the DIP Lenders) or (b) to the terms of any other such financing, including the consent to any lien encumbering the Prepetition Collateral (whether senior or junior) or to the use of Cash Collateral (except under the terms hereof), or (ii) prejudice, limit, or otherwise impair the rights of the Prepetition Agent (for the benefit of the Prepetition Secured Parties) to seek new, different, or additional adequate protection under any circumstances. The Prepetition Secured Parties' consent to the Debtors' use of Cash Collateral and the granting of DIP Liens on the Prepetition Collateral is expressly conditioned upon entry of this Interim Order and does not and shall not constitute consent other than pursuant to this Interim Order and on the terms set forth herein.

- L. Sections 506(b) and 552; Marshaling. In light of and in exchange for (i) the DIP Secured Parties' willingness to provide the DIP Facility to the extent set forth herein, (ii) the DIP Secured Parties' agreement that their liens and superpriority claims shall be subject to the Carve-Out, as set forth herein, (ii) the Prepetition Secured Parties' agreement that their Prepetition Liens and claims, including any adequate protection liens and claims, shall be subject to the Carve-Out, as set forth herein, (iii) the consensual use of Cash Collateral consistent with the DIP Budget and the terms of this Interim Order, and (iv) the DIP Secured Parties' and the Prepetition Secured Parties' agreement to the payment (in a manner consistent with the DIP Budget (subject to the Permitted Variances) and subject to the terms and conditions of this Interim Order) of certain expenses of administration of these Chapter 11 Cases, the DIP Secured Parties, and subject to entry of the Final Order, the Prepetition Secured Parties are each entitled to a waiver of any "equities of the case" exception under section 552(b) of the Bankruptcy Code and a waiver of the provisions of section 506(c) of the Bankruptcy Code and of the equitable doctrine of marshaling and other similar doctrines upon entry of the Final Order.
- M. Extension of Financing. The DIP Secured Parties have indicated a willingness to provide financing to the Debtors in accordance with the DIP Loan Documents (including the DIP Credit Agreement and the DIP Budget) and subject to (i) the entry of this Interim Order and the Final Order, (ii) approval of the terms and provisions of this Interim Order and the DIP Loan Documents (including the DIP Credit Agreement), including the Milestones (as defined in the DIP Credit Agreement) and the waivers set forth herein, and (iii) findings by this Court that such financing is essential to the Debtors' estates, that the DIP Secured Parties are good faith financiers, and that the reversal or modification on appeal of the authorization hereunder for the Debtors to incur the debt under the DIP Facility, or the grant hereunder of the priority of the DIP Liens and

the Adequate Protection Liens, does not affect the validity of such debt, or any priority of any such lien so granted as and to the extent provided in section 364(e) of the Bankruptcy Code.

N. <u>Limitation of Liability</u>. The Debtors stipulate and, subject to paragraph 11 and solely with respect to the DIP Loans, this Court finds that in making decisions to advance loans to the Debtors, in administering any loans, in permitting the Debtors to use Cash Collateral, in accepting the DIP Budget, or in taking any other actions permitted by this Interim Order or the DIP Loan Documents (including the DIP Credit Agreement), none of the DIP Secured Parties or Prepetition Secured Parties shall be deemed to be in control of the operations of the Debtors or to be acting as a "responsible person" or "owner or operator" with respect to the operation or management of the Debtors.

O. Business Judgment and Good Faith Pursuant to Section 364(e).

- (i) Based on the Declarations and the record at the Interim Hearing, the extensions of credit under the DIP Facility are fair, reasonable, and the best available under the circumstances, reflect the Debtors' exercise of prudent business judgment, and are supported by reasonably equivalent value and fair consideration;
- (ii) all obligations incurred, payments made, and transfers or grants of security set forth in this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement) by the Debtors are granted to or for the benefit of the DIP Secured Parties for fair consideration and reasonably equivalent value and are granted contemporaneously with the making of the loans and commitments and other financial accommodations secured thereby;
- (iii) the DIP Facility was negotiated in good faith and at arm's length among the Debtors and the DIP Secured Parties; and

- (iv) the use of the proceeds to be extended under the DIP Facility will be so extended in good faith and for valid business purposes and uses, as a consequence of which the DIP Secured Parties are entitled to the protection and benefits of section 364(e) of the Bankruptcy Code.
- P. Relief Essential; Best Interest. The relief requested in the Motion (and provided in this Interim Order) is necessary, essential, and appropriate for the continued operation of the Debtors' businesses and the management and preservation of the Debtors' assets and property, and satisfies the requirements of Bankruptcy Rule 6003. It is in the best interest of the Debtors' estates and consistent with the Debtors' exercise of their fiduciary duties that the Debtors are allowed to enter into the DIP Facility, incur the DIP Obligations, grant the liens and claims contemplated herein and under the DIP Loan Documents (including the DIP Credit Agreement) to the DIP Secured Parties and the Prepetition Secured Parties, and use Prepetition Collateral, including Cash Collateral, as it will provide the Debtors with the immediate and necessary liquidity to (i) minimize disruption to the Debtors' businesses and on-going operations, (ii) preserve and maximize the value of the Debtors' estates for the benefit of all the Debtors' creditors and stakeholders, (iii) avoid immediate and irreparable harm to the Debtors' estates and (iv) to otherwise maximize the value of the Debtors' estates.
- Q. <u>Findings Regarding Corporate Authority</u>. The Debtors have all requisite corporate power and authority to enter into, ratify, and perform all of their obligations under the DIP Loan Documents (including the DIP Credit Agreement) to which they are a party.
- R. <u>Immediate Entry</u>. Sufficient cause exists for immediate entry of this Interim Order pursuant to Bankruptcy Rules 4001(b)(2), (c)(2) and (d).

NOW, THEREFORE, on the Motion of the Debtors and the record before this Court with respect to the Motion, including the record made during the Interim Hearing, and with the consent of the Debtors, the Prepetition Secured Parties, and the DIP Secured Parties, and good and sufficient cause appearing therefor,

IT IS HEREBY ORDERED, ADJUDGED AND DECREED that:

1. <u>Motion Granted</u>. The Motion is granted on an interim basis in accordance with the terms and conditions set forth in this Interim Order. Any objections to the Motion with respect to entry of this Interim Order, to the extent not withdrawn, waived or otherwise resolved, and all reservation of rights included therein, are hereby denied and overruled on the merits.

2. **DIP Facility**.

empowered to enter into the DIP Facility and to incur and to perform the DIP Obligations in accordance with and subject to this Interim Order (and, upon its entry, a Final Order) and the other DIP Loan Documents, to execute and deliver all DIP Loan Documents and all other related instruments, certificates, agreements, and documents, and to take all actions which may be reasonably required or otherwise necessary for the performance by the Debtors under the DIP Facility, including the creation and perfection of the DIP Liens described and provided for herein. The Debtors are hereby authorized and directed to pay all fees, expenses and other amounts described herein, in the DIP Credit Agreement, and in the other DIP Loan Documents as such shall accrue and become due hereunder or thereunder, including, without limitation, the reasonable and documented fees and expenses of the attorneys and financial and other advisors and consultants of the DIP Agent and the DIP Lenders, as and to the extent provided for herein, in the DIP Credit Agreement, and in the other DIP Loan Documents (collectively, all loans, advances, extensions of credit, financial

accommodations, fees (which shall be fully earned and non-refundable upon entry of this Interim Order), expenses, and other liabilities and obligations (including indemnities and similar obligations) in respect of DIP Loans, the DIP Facility, the DIP Credit Agreement, and the other DIP Loan Documents, including all "Obligations" under and as defined in the DIP Credit Agreement, the "DIP Obligations"). The DIP Credit Agreement, the other DIP Loan Documents, and all DIP Obligations shall represent, constitute, and evidence, as the case may be, valid and binding obligations of the Debtors, enforceable against the Debtors, their estates, and any successor thereto in accordance with their terms. All obligations incurred, payments made, and transfers or grants of security set forth in this Interim Order and in the other DIP Loan Documents by any DIP Loan Party are granted to or for the benefit of the DIP Secured Parties for fair consideration and reasonably equivalent value and are granted contemporaneously with the making of the loans and commitments and other financial accommodations secured thereby. No obligation, payment, transfer, or grant of security under the DIP Loan Documents (including the DIP Credit Agreement) as approved under this Interim Order shall be voided, voidable, or recoverable under the Bankruptcy Code or under any applicable non-bankruptcy law, or subject to any defense, reduction, setoff, recoupment, or counterclaim; provided, however, that the adequate protection provided in paragraph 4 of this Interim Order shall be subject to paragraph 11 below. The term of the DIP Facility shall commence on the date of entry of this Interim Order and end on the Maturity Date (as defined below), subject to the terms and conditions set forth herein and in the DIP Loan Documents (including the DIP Credit Agreement).

(b) **Authorization to Borrow**. In order to continue to operate its business, subject to the terms and conditions of this Interim Order and the other DIP Loan Documents (including the

DIP Credit Agreement), the Debtors are hereby authorized to borrow under the DIP Facility and incur DIP Obligations during the Interim Period.

- (c) Roll-Up. Upon entry of this Interim Order and the Final Order, and subject to paragraph 11 below, the Prepetition Obligations, on a 1 to 1 cashless basis, in the aggregate amounts equal to the Interim Roll-Up Amount, shall be converted into DIP Obligations, without any further action by the Debtors or any other party. Upon any of the foregoing Interim Roll-Up Amount, and subject to paragraph 11 below, the Interim Roll-Up Amount shall cease to be indebtedness under the Prepetition Term Loan Agreement and shall be deemed DIP Obligations under the DIP Facility in all respects, including for purposes of having the benefit of section 364(e) of the Bankruptcy Code.
- (d) Conditions Precedent. The DIP Lenders shall have no obligation to make any DIP Loans or any other financial accommodation hereunder or under the other DIP Loan Documents (including the DIP Credit Agreement) (and the Debtors shall not make any request therefor) unless all conditions precedent to making DIP Loans under the DIP Credit Agreement have been satisfied or waived in accordance with the terms of the DIP Credit Agreement.
- (e) **DIP Collateral**. As used herein, "<u>DIP Collateral</u>" shall mean all assets, interests, rights, and property of any nature whatsoever of the Debtors, including, without limitation, all property in which the Debtors and their estates have an interest (whether tangible, intangible, real, personal or mixed), whether now owned or hereafter acquired and wherever located, before or after the Petition Date, including, without limitation, all accounts, proceeds of leases, inventory, equipment, equity interests or capital stock in subsidiaries, investment property, instruments, chattel paper, contracts, patents, copyrights, trademarks and other general intangibles, commercial litigation claims, cash, any investment of such cash, inventory, accounts receivable, including

intercompany accounts (and all rights associated therewith), other rights to payment whether arising before or after the Petition Date, any deposit accounts, and "cash collateral accounts", and, in each case all amounts on deposit therein from time to time, the proceeds of all claims or causes of action, and all rents, products, offspring, profits, proceeds, and substitutions thereof (including, without limitation, all Prepetition Collateral and, subject to the entry of the Final Order, all claims or causes of action of the Borrower arising under sections 502(d), 542, 544, 545, 547, 548, 549, 550, and 553 of the Bankruptcy Code and any other avoidance or similar action under the Bankruptcy Code or similar state law avoidance actions under chapter 5 of the Bankruptcy Code (the "Avoidance Actions") provided that the lien on Avoidance Actions shall be limited to the proceeds and property recovered in connection therewith).

- (f) **DIP Liens**. Effective immediately upon the entry of this Interim Order and subject to the Carve-Out, as set forth more fully in this Interim Order, the DIP Agent for the benefit of the DIP Secured Parties is hereby granted the following security interests and liens, which shall immediately be valid, binding, perfected, continuing, enforceable, and non-avoidable without the need for execution by the Borrower or Guarantors or the recordation or other filing by the DIP Secured Parties of security agreements, control agreements, pledge agreements, financing statements, or other similar documents or the possession or control by the DIP Secured Parties of any DIP Collateral (all liens and security interests granted to the DIP Agent for the benefit of the DIP Secured Parties pursuant to this Interim Order, any Final Order, and the DIP Credit Agreement, the "DIP Liens"):
 - (i) pursuant to section 364(c)(2) of the Bankruptcy Code, valid, binding, continuing, enforceable, fully perfected first-priority liens on and security interests in all DIP Collateral that is not subject to any liens or encumbrances immediately prior to the Petition Date, subject only to the Carve-Out; and

- (ii) pursuant to section 363(c)(3) and 364(d)(1) of the Bankruptcy Code, valid, binding, continuing, enforceable, fully perfected first priority, priming liens on and security interests in all other DIP Collateral, which liens and security interests shall be subject only to (a) any valid, enforceable, perfected, and non-avoidable lien or security interest in favor of any person other than the Prepetition Secured Parties that was in existence immediately prior to the Petition Date or that is perfected as permitted by Section 546(b) of the Bankruptcy Code, in each case, with respect to any DIP Collateral comprised of Prepetition Collateral, solely to the extent such lien is senior to the Prepetition Liens (a "Permitted Encumbrance")⁴ and (b) the Carve-Out, and senior to all other liens and encumbrances in respect of the DIP Collateral.
- Other Provisions Relating to the DIP Liens. Effective immediately upon entry of this Interim Order, subject to the Carve-Out, the DIP Liens shall secure all of the DIP Obligations, to the extent and subject to the priorities set forth herein, and such liens (and any superpriority claims and other senior liens, including adequate protection liens, granted hereunder) shall at all times have a higher priority and shall remain senior to the rights of the Debtors, any chapter 7 or chapter 11 trustee, and any secured, administrative priority, unsecured or other claims of any party in these Chapter 11 Cases under the Bankruptcy Code (except as (and solely to the extent) expressly provided herein), and the DIP Liens, the DIP Superpriority Claim, the Adequate Protection Claims, and the Adequate Protection Liens granted herein shall not be made or become subject, junior, or subordinated to any "priming" or other liens, nor made *pari passu* with any other lien, security interest, or claim heretofore or hereafter granted under Bankruptcy Code section 364 or otherwise, in these Cases (except as (and solely to the extent) expressly provided herein) or any

Nothing herein shall constitute a finding or ruling by this Court that any such Permitted Encumbrance is valid, senior, enforceable, prior, perfected, or non-avoidable. Moreover, nothing shall prejudice the rights of any party in interest, including but not limited to the Debtors, the DIP Secured Parties, the Prepetition Secured Parties, or Committee (if any), to challenge the validity, priority, enforceability, seniority, avoidability, perfection, or extent of any alleged Permitted Encumbrance and/or security interest. Notwithstanding anything to the contrary in this Interim Order or the DIP Loan Documents, nothing in this Interim Order or the DIP Loan Documents limits or adversely affects the rights, claims or defenses of any party asserting any Prepetition Permitted Liens or Excepted Liens (as defined in the DIP Credit Agreement) and the rights, claims or defenses of any such party are expressly reserved and preserved including but not limited to any rights of recoupment and/or setoff, subject to the Debtors' right to oppose or contest any of the foregoing.

Successor Case (as defined below), including to (i) any lien or security interest that is avoided and preserved for the benefit of the Debtors and their estates under section 551 of the Bankruptcy Code, (ii) any lien or security interest existing on or arising on or after the Petition Date, including, without limitation, any lien or security interests granted in favor of any federal, state, municipal or other domestic or foreign governmental unit (including any regulatory body), commission, board or court for any liability of the Debtors, (iii) any intercompany or affiliate lien or claim; or (iv) any other lien, claim, or security interest under sections 361, 363, or 364 of the Bankruptcy Code or otherwise, in each case other than as (and solely to the extent) expressly set forth herein. The DIP Liens and the Adequate Protection Liens shall be valid and enforceable against any trustee appointed in the Chapter 11 Cases, upon the conversion of the Chapter 11 Cases to cases under chapter 7 of the Bankruptcy Code or in any other proceedings related to any of the foregoing (each such case or proceeding, a "Successor Case"), and/or upon the dismissal of the Chapter 11 Cases. The DIP Liens and, subject to the entry of the Final Order, the Adequate Protection Liens shall not be subject to section 510 of the Bankruptcy Code, to the "equities of the case" exception of section 552 of the Bankruptcy Code, or to section 506(c) of the Bankruptcy Code, or sections 549, 550, or 551 of the Bankruptcy Code.

(h) **Superpriority Administrative Claim Status**. The DIP Obligations shall, subject to the Carve-Out, pursuant to section 364(c)(1) of the Bankruptcy Code, at all times constitute an allowed superpriority claim (the "<u>DIP Superpriority Claim</u>") of the DIP Agent for the benefit of the DIP Secured Parties against the Debtors, and be payable from all assets and properties of the Debtors, with priority over any and all other administrative expenses, adequate protection claims, diminution in value claims, and all other claims asserted against the Debtors now existing or hereafter arising of any kind whatsoever, including, without limitation, all administrative expenses

of the kind specified in sections 503(b) and 507(b) of the Bankruptcy Code, and over any and all other administrative expenses or other claims arising under any other provisions of the Bankruptcy Code, including sections 105, 326, 327, 328, 330, 331, 363, 364, 365, 503, 506(b), 506(c), 507(a), 507(b), 507(d), 546, 726, 1113, or 1114 of the Bankruptcy Code, whether or not such expenses or claims may become secured by a judgment lien or other nonconsensual lien, levy, or attachment, subject only to the Carve-Out, which DIP Superpriority Claim shall for purposes of section 1129(a)(9)(A) of the Bankruptcy Code be considered an administrative expense allowed under section 503(b) of the Bankruptcy Code and shall be payable from all pre- and postpetition assets and property, whether existing on the Petition Date or thereafter acquired, of the Debtors and all proceeds thereof; provided, however, that the DIP Superpriority Claim shall not be payable from the proceeds of Avoidance Actions or the proceeds thereof before entry of the Final Order. The DIP Superpriority Claim shall be subject only to the Carve-Out. Other than as expressly provided herein, including in paragraph 13 hereof with respect to the Carve-Out, no costs or expenses of administration, including, without limitation, professional fees allowed and payable under sections 328, 330, and 331 of the Bankruptcy Code, or otherwise, that have been or may be incurred in these Cases or in any Successor Case and no priority claims are or will be senior to, prior to, or pari passu with the DIP Liens, the DIP Superpriority Claim, or any of the DIP Obligations or with any other claims of the DIP Secured Parties arising hereunder or under the DIP Loan Documents (including the DIP Credit Agreement), or otherwise in connection with the DIP Facility.

(i) **Proceeds of DIP Loans**. All proceeds of the DIP Loans shall be funded and held in the account specified under the terms of the DIP Loan Documents (including the DIP Credit Agreement) and as otherwise agreed by the DIP Secured Parties, and such account (and any funds therein) shall be subject to a first-priority senior security interest and lien in favor of the DIP Agent.

- (j) **Professional Fees and Expenses**. The Debtors shall pay in cash and in full on the dates provided for in the Budget, all reasonable and documented out of pocket fees, costs and expenses of the DIP Agent and the DIP Lenders (including, without limitation, fees and disbursements of counsel, including Cole Schotz P.C., financial and accounting advisors, and any other advisors) in connection with the Chapter 11 Cases, including, without limitation (i) the negotiation, preparation, execution, and entry, as applicable, of this DIP Credit Agreement, (ii) any litigation, contest, dispute, suit, proceeding or action (whether instituted by DIP Agent, the Borrower, the Guarantors, or any other person) in any way relating to this DIP Credit Agreement, and (iii) the enforcement of any rights and remedies under the DIP Credit Agreement, including, without limitation, any accrued and unpaid fees, costs, and expenses of the DIP Agent and the DIP Lenders (including, without limitation, any fees and disbursements of counsel and any other advisors to the DIP Agent and to the DIP Lenders) prior to the Petition Date. Payment of any amounts set forth in this clause (j) shall not be subject to disgorgement.
- 3. Authorization and Approval to Use Cash Collateral and Proceeds of DIP Facility. Subject to the terms and conditions of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement) and unless otherwise agreed or ordered by the Court, the Debtors are authorized during the Interim Period (and not beyond) to (a) use the Cash Collateral and (b) request and use proceeds of the DIP Loans, in each case as set forth in the DIP Budget (subject to the Permitted Variances) and in accordance with this Interim Order the other DIP Loan Documents (including the DIP Credit Agreement). Notwithstanding anything herein to the contrary, subject only to the Debtors' rights under paragraph 21(b) hereof and the Carve-Out, the Debtors' right (a) to request proceeds of the DIP Loans shall terminate on the earlier of (i) the date the DIP Term Loans are fully funded or reduced to zero and (ii) the Maturity Date and (b) to use

proceeds of the DIP Loans or Cash Collateral shall terminate on the Maturity Date. Nothing in this Interim Order shall authorize the disposition of any assets of the Debtors or their estates or proceeds resulting therefrom outside the ordinary course of business except as expressly permitted herein and in the other DIP Loan Documents (including the DIP Credit Agreement) (subject to any required Court approval).

- 4. Adequate Protection for Prepetition Secured Parties. The Prepetition Secured Parties shall receive adequate protection of their interests in the Prepetition Collateral, including the Cash Collateral, in an amount equal to the aggregate diminution in the value of the Prepetition Secured Parties' interests in the Prepetition Collateral (including Cash Collateral) from and after the Petition Date, if any, for any reason provided for under the Bankruptcy Code, including, without limitation, as a result of the use of Prepetition Collateral, including Cash Collateral, the priming of their interests in Prepetition Collateral by the DIP Facility and the Carve-Out, and the imposition or enforcement of the automatic stay (collectively, "Diminution in Value"). As adequate protection, the Prepetition Secured Parties will be granted, subject to paragraph 11 below, the following:
- (a) Adequate Protection Liens. Pursuant to sections 361, 363(e), and 364(d) of the Bankruptcy Code, effective as of the Petition Date and perfected without the need for execution by the Borrower or the Guarantors or the recordation or other filing by the DIP Lenders of security agreements, control agreements, pledge agreements, financing statements, or other similar documents or the possession or control by the of any DIP Collateral, solely to the extent of any Diminution in Value, shall be granted valid, binding, continuing, enforceable, fully perfected replacement and additional liens on and security interests in (collectively, the "Adequate Protection Liens") all DIP Collateral. The Adequate Protection Liens shall be junior only to the

DIP Liens, the Carve-Out, and any Permitted Encumbrance, and senior to all other security interests in, liens on, or claims against the DIP Collateral. The Adequate Protection Liens shall be non-avoidable and shall not be subject to sections 510, 549, 550, or 551 of the Bankruptcy Code. The Adequate Protection Liens shall be enforceable against and binding upon the Borrower, the Guarantors, their estates and any successors thereto, including, without limitation, any trustee or other estate representative appointed in any Successor Case.

- (b) Adequate Protection Superpriority Claim. To the extent of the aggregate Diminution in Value, the Prepetition Secured Parties shall be granted allowed superpriority administrative expense claims against the Borrower or the Guarantors, and their estates, as provided in section 507(b) of the Bankruptcy Code, immediately junior to the DIP Obligations and subject only to the Carve-Out, and with priority in payment over any and all other claims and administrative expense claims against the Borrower or the Guarantors, now existing or hereafter arising, of any kind or nature whatsoever, including, without limitation, administrative expenses of the kinds specified in or ordered pursuant to sections 105, 326, 327, 328, 330, 331, 361, 362, 363, 364, 365, 503(a), 503(b), 507(a), 507(b), 546(c), 546(d), 552(b), 726, 1113, 1114, and any other provision of the Bankruptcy Code, and shall at all times be senior to the rights of the Borrower and the Guarantors, and any successor trustee or creditor in the Chapter 11 Cases or any Successor Case (the "Adequate Protection Superpriority Claims").
- (c) Adequate Protection Payments, Etc. As further adequate protection, from and after entry of this Interim Order, the Prepetition Agent (on behalf of the Prepetition Secured Parties) shall receive from the Debtors, (1) payment in cash, as and when due under the terms of the Prepetition Term Loan Agreement, of all interest due under the Prepetition Term Loan Agreement, calculated at the default rates under the Prepetition Term Loan Agreement, and

(2) (x) upon entry of this Interim Order, immediate cash payment of all accrued and unpaid fees and disbursements (including all reasonable and documented fees, out-of-pocket costs and expenses of legal, financial, and other advisory professionals of the Prepetition Secured Parties (including, without limitation, Cole Schotz P.C.) owing to the Prepetition Secured Parties under the Prepetition Term Loan Agreement, whether incurred prior to or after the Petition Date and (y) thereafter, when due, all fees and other amounts (including all reasonable and documented legal and advisory fees, out-of-pocket costs and expenses of legal, financial, and other advisory professionals of the Prepetition Secured Parties (including, without limitation, Cole Schotz P.C.) owing to the Prepetition Secured Parties by the Debtors under the Prepetition Term Loan Agreement, provided that payment of any fees and expenses of the Prepetition Secured Parties' professionals incurred after entry of this Interim Order shall be subject to the notice and objection provisions of paragraph 27(b) (the "Adequate Protection Payments"). Payment of any amounts set forth in this clause (c)(2) shall not be subject to disgorgement or recharacterization.

5. **DIP Budget**.

(a) The DIP Budget attached hereto as <u>Exhibit 2</u> constitutes an "Initial DIP Budget" under the DIP Credit Agreement. The Borrower will subsequently provide the DIP Agent with an updated budget before noon every Friday of each week commencing with the second week following the Petition Date, prior to the end of the DIP Budget Period (defined below) for the remaining weeks left under the Initial DIP Budget. The "<u>DIP Budget Period</u>" shall refer to period commencing on the date that this Interim Order is entered and ending on the Outside Effective Date (as defined in the DIP Credit Agreement). Each updated budget shall be subject to the approval of the DIP Lenders in their sole discretion (each such approved budget, a "<u>Supplemental Approved DIP Budget</u>" and together with the Initial DIP Budget, the "<u>DIP Budget</u>"); provided that

until such time as the DIP Agent approves any updated budget, the Borrower shall be subject to and be governed by the terms of the Initial DIP Budget or such Supplemental Approved DIP Budget, as applicable, then in effect in accordance with this Interim Order. The DIP Budget may only be amended, supplemented, modified, restated, replaced, or extended in accordance with this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement). Any such amendment, supplement, modification, restatement, replacement, or extension in accordance with the DIP Credit Agreement may be effected without further order of the Court, so long as approved by the Required Lenders (as such term is defined in the DIP Credit Agreement) of DIP Lenders. The DIP Budget is an integral part of this Interim Order and has been relied upon by the DIP Secured Parties to provide the DIP Facility and consent to this Interim Order.

Agreement), the Borrower shall provide to the DIP Agent a budget variance report (the "Budget Variance Report") before noon on Friday that is the second week following the Petition Date, and measured weekly thereafter, and the DIP Budget Variance Report shall set forth in reasonable detail actual receipts and disbursements of Borrower, together with a statement certifying compliance with the DIP Budget Covenant (as defined below) set forth below. The Borrower shall not permit or suffer to exist, a negative variance of 15% or more of total the total spending set forth in the Budget for such period (as reflected in the DIP Budget) in the aggregate from the amounts set forth in the DIP Budget (each, a "Permitted Variance"), tested on a cumulative basis (the "DIP Budget Covenant"). The Debtors shall operate in accordance with the DIP Budget and all disbursements of the Debtors shall be consistent with the provisions of the DIP Budget, subject in each case to any Permitted Variance provided herein and in the DIP Credit Agreement.

- (c) The DIP Secured Parties shall have no obligation to permit the use of proceeds of DIP Loans or Cash Collateral, and the Borrower shall have no authority to use proceeds of DIP Loans or Cash Collateral, other than in accordance with the DIP Budget, subject to the DIP Budget Covenant and as set forth in the DIP Order.
- 6. Monitoring of Collateral. The Prepetition Secured Parties, the DIP Secured Parties, and their consultants and advisors shall be given reasonable access to the Debtors' books, records, assets, and properties for purposes of monitoring the Debtors' businesses and the value of the DIP Collateral, including view access to the Debtors' accounts, at the Debtors' cost and expense, as provided in the DIP Credit Agreement.
- Financial Reporting, Etc. Without limitation of the requirements of the DIP Credit Agreement, the Debtors shall provide to the DIP Secured Parties and to the Prepetition Secured Parties (and, in each case, their consultants, advisors, and professionals) (a) all financial information required under the DIP Loan Documents (including the DIP Credit Agreement) and (b) access upon reasonable notice and during regular business hours to the Debtors' books and records, assets, and properties for purposes of monitoring its businesses and the value of the DIP Collateral. The Debtors shall also provide such reports and information required to be provided in the DIP Credit Agreement and reasonably cooperate, discuss with, and provide to the DIP Secured Parties and the Prepetition Secured Parties (and, in each case, their professionals) all such information as may be reasonably requested. In addition, the Debtors hereby authorize their accountants, attorneys, financial advisors, bankruptcy professionals, and consultants to cooperate, consult with, and provide to the DIP Secured Parties and the Prepetition Secured Parties (and, in each case, their consultants, advisors and professionals) all such information as may be reasonably requested with respect to the business, results of operations and financial condition of the Debtors

consistent with the requirements set forth in the DIP Loan Documents (including the DIP Credit Agreement).

- 8. <u>Milestones</u>. It is a condition to the DIP Facility and the use of Cash Collateral that the Debtors shall comply with the milestones set forth in the DIP Credit Agreement (the "<u>Milestones</u>"). The failure to comply with any Milestone shall constitute an Event of Default (as defined in the DIP Credit Agreement) in accordance with the DIP Credit Agreement.
- 9. <u>Subordination of Intercompany, Affiliate Liens</u>. All intercompany or affiliate liens of the Debtors, if any (other than any liens granted to the DIP Secured Parties), will be contractually subordinated to the DIP Liens and the Adequate Protection Liens granted pursuant to this Interim Order and under the DIP Credit Agreement, on terms satisfactory to the DIP Lenders.
- DIP Lien and Adequate Protection Replacement Lien Perfection. Automatically upon entry of this Interim Order, the DIP Liens and the Adequate Protection Liens, shall be deemed to be valid, binding, perfected, enforceable, non-avoidable, and effective by operation of law, and not subject to challenge as of the Petition Date, without the need of any further action of any kind; provided, however, that the liens associated with the Interim Roll-Up Amount and the Adequate Protection Liens shall both be subject to the Challenge Period. This Interim Order shall be sufficient and conclusive evidence of the validity, perfection, and priority of the DIP Liens and the Adequate Protection Liens without the necessity of filing or recording any financing statement, deed of trust, mortgage, or other instrument or document which may otherwise be required under the law of any jurisdiction or the taking of any other action to validate or perfect the DIP Liens and the Adequate Protection Liens or to entitle the DIP Liens and the Adequate Protection Liens to the priorities granted herein. Notwithstanding the foregoing, the DIP Agent may, at the direction of the Required Lenders, file such financing statements, deeds of trust, mortgages, security

C6355 & 52-51-0.00297-9-14/18/S DDo 5/04-22 Filled 006/04/125 Page 334 off 6/293

agreements, notices of liens, and other similar documents and is hereby granted relief from the automatic stay of section 362 of the Bankruptcy Code in order to do so, and all such financing statements, deeds of trust, mortgages, security agreements, notices, and other agreements or documents shall be deemed to have been filed or recorded at the time and on the date of the commencement of the Chapter 11 Cases. The Debtors shall execute and deliver to the DIP Agent all such financing statements, mortgages, security agreements, notices and other documents as the DIP Agent may reasonably request to evidence, confirm, validate, or perfect or to insure the contemplated priority of the DIP Liens and the Adequate Protection Liens, and the Debtors shall take all such further actions that may be required under any applicable law or that the DIP Secured Parties or Prepetition Secured Parties, as applicable, may reasonably request in order to grant, preserve, protect or perfect the DIP Liens and Adequate Protection Liens granted pursuant to this Interim Order. The DIP Agent, at the direction of the Required Lenders, may file a photocopy of this Interim Order as a financing statement with any recording officer designated to file financing statements or with any registry of deeds or similar office in any jurisdiction in which the Debtors have real or personal property and, in such event, the subject filing or recording officer shall be authorized to file or record such copy of this Interim Order. To the extent that a Prepetition Secured Party is the secured party under any account control agreements, listed as loss payee under any of the Debtors' insurance policies or is the secured party under any loan document, financing statement, deed of trust, mortgage, or other instrument or document which may otherwise be required under the law of any jurisdiction to create, validate, attach, perfect, or prioritize liens (any such instrument or document, a "Security Document"), then (i) such Prepetition Secured Party shall be deemed to maintain such possession or notation or exercise such control as a gratuitous bailee and/or gratuitous agent for perfection for the benefit of the DIP Agent and the DIP Lenders,

and such Prepetition Secured Party shall comply with the instructions of the DIP Agent with respect to the exercise of such control and (ii) the DIP Agent is also hereby deemed to be the secured party under such account control agreements, loss payee under the Debtors' insurance policies and the secured party under each such Security Document, and shall have all rights and powers attendant to that position (including, without limitation, rights of enforcement) and shall act in that capacity and distribute any proceeds recovered or received in accordance with the terms of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement).

11. Reservation of Certain Third-Party Rights and Bar of Challenges and Claims.

- (a) The Debtors' stipulations, admissions, agreements, Releases, and waivers contained in this Interim Order, including the Stipulations, are and shall be irrevocably binding upon the Debtors and any and all of the Debtors' successors in interest and assigns (subject to the Challenge Period, including, without limitation, any chapter 7 or chapter 11 trustee, responsible person, examiner with expanded powers, or other estate representative) in all circumstances and for all purposes.
- (b) The Stipulations and all other admissions, agreements, releases, and waivers set forth in this Interim Order also are and shall be binding upon all other persons and entities (including any Committee) and each of their respective successors in interest and assigns in all circumstances and for all purposes, unless, and solely to the extent that (i) such parties in interest (including any Committee) in each case with standing and requisite authority to do so (subject in all respects to any agreement or applicable law which may limit or affect such entity's right or ability to do so) have timely filed the proper pleadings, and timely commenced the appropriate proceedings under the Bankruptcy Code and Bankruptcy Rules, including, without limitation, as required pursuant to Part VII of the Bankruptcy Rules (in each case subject to the limitations set

forth below in this paragraph 11), (x) objecting to or challenging any of the Stipulations or (y) otherwise asserting or prosecuting any action against the Prepetition Secured Parties or any of their respective affiliates, agents, attorneys, advisors, professionals, officers, directors, or employees in connection with or related to the matters covered by the Stipulations (each such proceeding or appropriate pleading commencing a proceeding or other contested matter, a "Challenge"), no later than seventy-five (75) calendar days after the date of entry of this Interim Order for any party in interest with requisite standing (such period, the "Challenge Period" and the date that is the next calendar day after the termination of the Challenge Period shall be referred to as the "Challenge Period Termination Date"), as such date may be extended to any such party in interest by the Prepetition Agent and each applicable Prepetition Secured Party that is the subject of a Challenge or by any such later date as has been ordered by the Court for cause upon a motion filed and served within the Challenge Period (before giving effect to such extension) and (ii) the Court enters judgment in favor of the plaintiff or movant in any such timely and properly commenced Challenge proceeding, and any such judgment has become final and is not subject to any further review or appeal.

(c) Any Challenge not asserted by the timely and proper filing of a pleading by a party in interest with the requisite standing and authority as contemplated herein prior to the Challenge Period Termination Date (or any extension thereto) shall be deemed forever waived, released, and barred with respect to such party in interest. To the extent a party in interest with requisite standing and authority timely and properly commences a Challenge prior to the Challenge Period Termination Date, all claims, causes of action and other matters not specifically set forth in such Challenge shall be deemed forever waived, released, and barred with respect to such party in interest.

- (d) To the extent any Stipulations are (i) not subject to a Challenge timely and properly commenced prior to the Challenge Period Termination Date or (ii) subject to a Challenge timely and properly commenced prior to the Challenge Period Termination Date, to the extent any such Challenge does not result in a final and non-appealable judgment or order of the Court that is inconsistent with such Stipulations, then, in each case, without further notice, motion or application to, or order of, or hearing before this Court and without the need or requirement to file any proof of claim: (1) any and all such Challenges by any Committee or any other party in interest shall be deemed to be forever waived, released, and barred (including in any Successor Case), and (2) the Prepetition Obligations shall be deemed to be an allowed secured claim within the meaning of sections 502 and 506 of the Bankruptcy Code for all purposes in connection with these Chapter 11 Cases, and all of the Stipulations and all other waivers, releases, and affirmations set forth in this Interim Order (or any not properly and timely challenged) shall be in full force and effect and shall be binding, conclusive, and final on any person, entity, or party in interest, including any Committee, (in each case, and their successors and assigns) in these Chapter 11 Cases and in any Successor Case for all purposes, without any further order of the Court, and shall not be subject to challenge or objection by the Committee or any other party in interest, including, without limitation, any other statutory committee, any trustee, responsible individual, examiner with expanded powers or other representative of the Debtors' estates.
- (e) If any Challenge is timely and properly filed during the Challenge Period, the Stipulations and all other waivers, releases, and affirmations contained in this Interim Order shall nonetheless remain binding and preclusive, as provided in this paragraph 11, on all other parties in interest (including any Committee) and on any other person or entity, except for the plaintiff or movant timely and successfully asserting such Challenge, as set forth in a final, non-appealable

order of a court of competent jurisdiction. Notwithstanding anything to the contrary herein, the right to commence any Challenge under this Interim Order is preserved only as against any particular Prepetition Obligation or against any Prepetition Secured Party to the extent such Challenge is commenced timely and properly prior to the Challenge Period Termination Date and, in respect of such Prepetition Obligation or Prepetition Secured Party, is otherwise waived as set forth in this paragraph 11.

- (f) All remedies or defenses of any party with respect to any Challenge are hereby preserved. Nothing in this Interim Order vests or confers on any person (as defined in the Bankruptcy Code), including any Committee (if any) appointed in these Chapter 11 Cases, standing or authority to pursue any cause of action belonging to the Debtors or their estates, including, without limitation, any Challenges with respect to the Prepetition Obligations, and an order of the Court (or any other court of competent jurisdiction) conferring such standing on a Committee or other party in interest shall be a prerequisite for the prosecution of a Challenge by the Committee or such other party in interest. If a Committee or other party in interest files a motion for standing to pursue a Challenge that includes a draft of the complaint or motion such party in interest seeks to pursue, then the Challenge Period Termination Date shall be stayed as to such movant(s) until the Court rules on the motion for standing.
- (g) For the avoidance of doubt, if a chapter 7 or 11 trustee is appointed before the Challenge Period expires, then such trustee shall be deemed to be a party in interest other than the Debtors and shall not be bound by the waivers, releases, affirmations, and Stipulations of the Debtors in this Interim Order until the Challenge Period expires. Notwithstanding anything to the contrary herein, if a Committee or other party in interest timely and properly files a Challenge within the Challenge Period that is pending as of the time a chapter 7 or chapter 11 trustee is

appointed or elected in these cases, such trustee shall have the right (but not the obligation) to intervene in such Challenge and, upon such intervention, shall be deemed to have timely and properly filed such Challenge within the Challenge Period to the same extent as the party originally instituting the Challenge.

12. Limitations on Use of DIP Facility and Cash Collateral. Notwithstanding anything herein to the contrary, no portion of the proceeds of the DIP Facility, the DIP Collateral, the Prepetition Collateral, any Cash Collateral, or the Carve-Out and no disbursements set forth in the DIP Budget may be used, in any way, directly or indirectly (a) for any investigation, adversary action, suit, arbitration, proceeding, application, motion, objection, defense, other contested matter, or other litigation of any type, including any Challenge, adverse to the interests of any or all of DIP Lenders, the DIP Agent, the Prepetition Secured Parties, or their respective rights and remedies under the DIP Facility, this Interim Order, the Final Order, the DIP Credit Agreement, or the Prepetition Term Loan Agreement and other documents and instruments in respect of the Prepetition Obligations, including without limitation (i) any action arising under the Bankruptcy Code, (ii) any so-called "lender liability" claims and causes of action, (iii) any action with respect to the validity and extent of the DIP Obligations or Prepetition Obligations or the validity or extent of the DIP Liens or Prepetition Liens, (iv) any action seeking to invalidate, set aside, avoid, reduce, set off, offset, recharacterize, subordinate (whether equitable, contractual, or otherwise), recoup against, disallow, impair, raise any defenses, cross-claims, or counterclaims, or raise any other challenges under the Bankruptcy Code or any other applicable domestic or foreign law or regulation against or with respect to the DIP Liens or the Prepetition Liens, in whole or in part, or (v) appeal or otherwise challenge this Interim Order or the Final Order; provided that no more than \$25,000 in the aggregate of the proceeds of the DIP Facility, the DIP Collateral, the Prepetition

Collateral, and the Cash Collateral (including any proceeds of the DIP Facility, the DIP Collateral, the Prepetition Collateral, or the Cash Collateral used to fund the Carve-Out) may be used by a Committee (if any) appointed in these Chapter 11 Cases to investigate (but not prosecute or Challenge, or commence, or initiate the prosecution of, any Challenge, including the preparation of any complaint or motion on account of, or objection to) the Stipulations before termination of the Challenge Period, (b) for any other action, which with the giving of notice or passing of time, would result in an Event of Default under the DIP Facility, (c) for any purpose that is prohibited under the Bankruptcy Code, DIP Loan Documents (including the DIP Credit Agreement), this Interim Order, or the Final Order or that is not in accordance with the DIP Budget, (d) to make any payment, advance, intercompany advance or transfer, or any other remittance or transfer whatsoever (including any intercompany loans and investments) that is not in accordance with the express terms of the DIP Budget, (e) to make any payment in settlement of any claim, action, or proceeding without Bankruptcy Court approval and the prior written consent of the DIP Lenders, which consent shall not be unreasonably withheld or delayed, (f) to prevent, hinder, impede, or delay any of the DIP Secured Parties or Prepetition Secured Parties' enforcement or realization upon or exercise of rights in respect of any of the DIP Collateral or Prepetition Collateral in accordance with the DIP Credit Agreement, or (g) to seek to amend or modify any of the rights or interests granted to any of the DIP Secured Parties or Prepetition Secured Parties under this Interim Order, the DIP Credit Agreement, or the Prepetition Term Loan Agreement, in a manner adverse to any DIP Secured Party or Prepetition Secured Party, including seeking to use Cash Collateral on a contested basis, without the prior written consent of the DIP Agent.

13. Carve-Out.

As used in this Interim Order, the "Carve-Out" means the sum of (i) all fees required (a) to be paid to the Clerk of the Bankruptcy Court ("Court Fees") and all statutory fees required to be paid to the U.S. Trustee under section 1930(a) of title 28 of the United States Code plus interest at the statutory rate ("US Trustee Fees") (without regard to the notice set forth in (iii) below), (ii) up to the amounts set forth in the respective DIP Budget line items and to the extent allowed by the Court pursuant to a fee application or fee statement on notice or other procedure permitted by any Court order allowing interim compensation or the payment of fees of ordinary course professionals, whether by interim order, final order, procedural order, or otherwise, all reasonable and documented unpaid fees and expenses, all unpaid fees and expenses (the "Allowed Professional Fees") incurred by persons or firms retained by the Debtors pursuant to section 327, 328, or 363 of the Bankruptcy Code (collectively, the "Debtor Professionals") and the Committee, if any, appointed in the Bankruptcy Case (the "Committee Professionals," and together with the Debtor Professionals, the "Professionals") at any time on or before one (1) business day following delivery by the DIP Agent of a Carve-Out Trigger Notice (as defined below), whether allowed by the Bankruptcy Court prior to or after delivery of a Carve-Out Trigger Notice, and (iii) Allowed Professional Fees of Professionals in an aggregate amount not to exceed \$100,000 incurred after delivery by the DIP Agent of the Carve-Out Trigger Notice (the amount set forth in this clause (iii) being the "Post Carve-Out Trigger Notice Cap"); provided that, under no circumstances shall any success, completion, transaction or similar fees be payable from the Carve-Out; provided, further, that the Post Carve-Out Trigger Notice Cap shall be reduced, dollar-for-dollar, by the amount of any fees and expenses incurred and accruing by the Debtors and paid to the applicable Professionals following delivery of a Carve-Out Trigger Notice. For purposes of this Interim Order, "Carve-Out Trigger Notice" shall mean a written notice delivered by email (or other electronic means) by the DIP Agent to counsel to the Debtors, the U.S. Trustee, and counsel to the Committee (if any), which notice may be delivered following the occurrence and during the continuation of an Event of Default, stating that the Post Carve-Out Trigger Notice Cap has been invoked.

- (b) The Debtors shall be authorized and permitted to establish an escrow account, to be held by a qualified institutional escrow agent to be agreed upon by the Debtors and the DIP Secured Parties (the "Escrow Holder"), for the benefit of the Professionals and to fund by wire transfer such escrow on a weekly basis, in each case as provided herein and in the DIP Loan Documents (including the DIP Credit Agreement) (such account, to the extent of such deposited amounts, the "Professional Fee Reserve Account"), in an amount equal to, but not to exceed, the professional fees and costs set forth in the DIP Budget line item for Debtor Professionals for such week. The Professional Fee Reserve Account shall be funded on a weekly basis in accordance with the DIP Budget and ending on the date upon which a Carve-Out Trigger Notice is delivered, plus any amounts provided for in the DIP Budget, if any, since the most recent funding of the Professional Fee Reserve Account through the delivery of the Carve-Out Trigger Notice, in accordance with the DIP Budget. After the delivery of a Carve-Out Trigger Notice, notwithstanding anything herein to the contrary, the Debtors shall be authorized to deposit any cash on hand into the Professional Fee Reserve Account up to the amount of any remaining unpaid professional fees for each of the Professionals set forth in the DIP Budget for the period until delivery of the Carve-Out Trigger Notice plus an amount equal to the Post Carve-Out Trigger Notice Cap.
- (c) From the funds held in the Professional Fee Reserve Account, the Escrow Holder shall release to the Debtor Professionals solely such amounts as are payable from time to time pursuant to the DIP Budget and an applicable order of the Bankruptcy Court, including an order

C6355 & 52-51-0.00297-9-14/18/S DDo 5/04-22 Filled 006/04/125 Page 412 of 612/3

approving interim compensation procedures in the Chapter 11 Cases and any order granting interim or final fee applications for Professionals (each, a "Fee Order"). For the avoidance of doubt, (a) in making payments from the Professional Fee Reserve Account, Escrow Holder shall be entitled to rely upon written certifications of each Debtor Professional as to the amount such Debtor Professional is authorized pursuant to the DIP Budget and an applicable order of the Bankruptcy Court to be paid at such time from the Professional Fee Reserve Account; and (b) in no circumstances shall the Escrow Holder be obligated to pay any Debtor Professional other than from the funds held, from time to time, in the Professional Fee Reserve Account in accordance with the DIP Budget and an applicable order of the Bankruptcy Court. Funds held in the Professional Fee Reserve Account shall be applied to Allowed Professional Fees that have been incurred following the Petition Date in accordance with procedures to be established in the Chapter 11 Cases. Payments and reimbursements made to a Professional prior to delivery of the Carve-Out Trigger Notice shall reduce the amounts available to such Professional under section (ii) of the Carve-Out, and neither the Professional Fee Reserve Account nor payments therefrom shall in any way increase the Carve-Out. All Allowed Professional Fees payable to Debtor Professionals shall be paid first from funds in the Professional Fee Reserve Account, if any, and, upon exhaustion thereof, from the Carve-Out (less any amounts previously funded into the Professional Fee Reserve Account). For the avoidance of doubt the DIP Liens, Prepetition Liens, and Adequate Protection Liens shall attach to, and shall have a residual right to and lien on, any excess funds in the Professional Fee Reserve Account after satisfaction of the Carve-Out in respect of Allowed Professional Fees, which excess funds shall then be used to satisfy amounts due under the DIP Facility, the Prepetition Obligations, and the Adequate Protection Payments, as applicable, in

accordance with their rights and priorities as of the Petition Date and under this Interim Order, in each case, until paid in full in cash.

- (d) None of the DIP Secured Parties or the Prepetition Secured Parties shall be responsible for the payment or reimbursement of any fees or disbursements of any Professional incurred in connection with the Chapter 11 Cases or any Successor Case under any chapter of the Bankruptcy Code. Nothing in this Interim Order or otherwise shall be construed to obligate the DIP Secured Parties or the Prepetition Secured Parties, in any way, to pay compensation to, or to reimburse expenses of, any Professional or to guarantee that the Debtors have sufficient funds to pay such compensation or reimbursement. The DIP Secured Parties and the Prepetition Secured Parties reserve the right to review and object to any fee statement, interim application, or monthly application issued or filed by the Professionals.
- 14. Payment of Compensation. Nothing herein shall be construed as a consent to the allowance of any professional fees or expenses of any of the Debtors or the Committee (if any) or shall limit or otherwise affect the right of the DIP Secured Parties or the Prepetition Secured Parties or any other party in interest to object to the allowance and payment of any such fees and expenses. No professional fees shall be paid absent a Court order allowing such payment, pursuant to a fee application or fee statement on notice, or other procedure permitted by any Court order allowing interim compensation or the payment of fees of ordinary course professionals. So long as no Event of Default exists that has not been waived in writing, the Debtors shall be permitted to pay compensation and reimbursement of expenses allowed by the Court and payable under sections 330 and 331 of the Bankruptcy Code or compensation procedures approved by the Court and in form and substance reasonably acceptable to the Debtors and the DIP Secured Parties, as the same may be due and payable, and the same shall not reduce the Post Carve-Out Trigger Notice Cap.

- 15. Section 506(c) Claims. As a further condition of the DIP Facility, any obligation of the DIP Secured Parties to make DIP Loans, and the Debtors' authorization to use the Cash Collateral and the consent of the DIP Secured Parties and the Prepetition Secured Parties to the payment of the Carve-Out to the extent provided herein, the Debtors (and any successors thereto or any representatives thereof, including any trustees appointed in the Chapter 11 Cases or any Successor Case) shall be deemed to have waived any rights, benefits, or causes of action under section 506(c) of the Bankruptcy Code as they may relate to or be asserted against the DIP Secured Parties, the DIP Liens, and the DIP Collateral and, subject to the entry of the Final Order, the Prepetition Secured Parties, the Adequate Protection Liens, the Prepetition Liens, and the Prepetition Collateral, and except to the extent of the Carve-Out, nothing contained in this Interim Order, the Final Order, or the DIP Credit Agreement shall be deemed a consent by the Prepetition Secured Parties or the DIP Secured Parties to any charge, lien, assessment, or claim against or in respect of the DIP Collateral or the Prepetition Collateral under sections 105 or 506(c) of the Bankruptcy Code or otherwise, and no such consent shall be implied from any other action, inaction, or acquiescence by any by any such parties.
- 16. Collateral Rights; Limitations in Respect of Subsequent Court Orders. Without limiting any other provisions of this Interim Order, unless the DIP Agent and the Prepetition Agent have provided their prior written consent, no credit shall be obtained, or indebtedness incurred that is secured by a security, mortgage, or collateral interest or other lien on all or any portion of the DIP Collateral or entitled to priority administrative status which is superior to or *pari passu* with those granted pursuant to this Interim Order to or for the benefit of the DIP Secured Parties or the Prepetition Secured Parties. Further, the proceeds of the DIP Facility and the Cash Collateral shall

not be used for any purpose other than as permitted under the DIP Credit Agreement and consistent with this Interim Order and the DIP Budget (subject to Permitted Variances).

- Proceeds of Subsequent Financing. Without limiting the provisions and protections of paragraph 16 hereof, if at any time prior to the Maturity Date, the Debtors, the Debtors' estates, any trustee, any examiner with enlarged powers or any responsible officer subsequently appointed shall obtain credit or incur debt in violation of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement), then all of the cash proceeds derived from such credit or debt and all Cash Collateral in the amount necessary to satisfy all DIP Obligations then outstanding shall immediately be turned over to the DIP Agent for the indefeasible repayment and satisfaction in full in cash of all DIP Obligations then outstanding. For the avoidance of doubt, if the Debtors, any trustee, any examiner with expanded powers, or any responsible officer subsequently appointed in these Chapter 11 Cases or any Successor Case shall obtain credit or incur debt pursuant to section 364(d) of the Bankruptcy Code at any time prior to the Maturity Date, the Prepetition Secured Parties' rights to object to the Debtors' use of Cash Collateral and assert a lack of adequate protection shall be fully preserved.
- Cash Management. From and after the date of the entry of this Interim Order, the Debtors shall maintain a cash management system that is the same as or substantially similar to their prepetition cash management system or as otherwise agreed by the DIP Lenders and authorized by DIP Lenders (in their sole discretion) and the Court. The Debtors shall maintain no accounts except those identified in any order of this Court approving the maintenance of the Debtors' cash management system (the "Cash Management Order") or that may be subsequently opened pursuant to the Cash Management Order. It shall constitute an Event of Default (as defined in the DIP Credit Agreement) if the Debtors' cash management system shall not at all times be

maintained in accordance with the terms of the DIP Credit Agreement, the Cash Management Order, and this Interim DIP Order in all material respects, and otherwise in a manner which shall be reasonably satisfactory to the DIP Secured Parties in accordance with the terms of the DIP Loan Documents (including the DIP Credit Agreement). The Debtors and the financial institutions where the Debtors maintain deposit accounts (as identified in the Cash Management Order) are authorized to remit, without offset or deduction, funds in such deposit accounts upon receipt of any direction to that effect from the DIP Agent (acting at the direction of the DIP Lenders). The DIP Agent shall be deemed to have "control" over all cash management accounts for all purposes of perfection under the Uniform Commercial Code pursuant to this Interim Order and, if required or requested by the DIP Agent, pursuant to control agreements reasonably acceptable to the DIP Agent.

- 19. <u>Disposition of DIP Collateral</u>. Unless otherwise ordered by the Court, it shall constitute an Event of Default (as defined in the DIP Credit Agreement) if the Debtors sell (including, without limitation, any sale and leaseback transaction), transfer (including any assignment of rights), lease, encumber or otherwise dispose of any portion of the DIP Collateral, except in the ordinary course of business or as otherwise expressly permitted under the DIP Credit Agreement.
- 20. <u>Survival of Certain Provisions</u>. In the event of the entry of any order converting any of these Chapter 11 Cases into a Successor Case, the DIP Liens, the DIP Superpriority Claim, the Adequate Protection Liens, the Adequate Protection Superpriority Claim, and the Carve-Out shall continue in this proceeding and in any Successor Case, and such DIP Liens, DIP Superpriority Claim, Adequate Protection Liens, Adequate Protection Superpriority Claim, and Carve-Out shall maintain their respective priorities as provided by this Interim Order.

21. Events of Default; Maturity Date; Rights and Remedies Upon Event of Default.

Any automatic stay otherwise applicable to the DIP Secured Parties and the (a) Prepetition Agent, whether arising under sections 105 or 362 of the Bankruptcy Code or otherwise, is hereby modified so that, (i) upon the occurrence and during the continuance of an Event of Default, the DIP Secured Parties, in their sole and absolute discretion, may immediately (w) deliver a notice of an Event of Default; (x) terminate any pending DIP Loans and use of Cash Collateral; (y) sweep all cash in any controlled accounts, and (z) terminate the DIP Facility and all commitments thereunder and (ii) upon and after the occurrence of the Maturity Date, the DIP Agent and the Prepetition Agent shall, without further notice to, hearing of, or order from this Court, to the extent necessary to permit the DIP Secured Parties to take any or all of the following actions, at the same time or different times, unless the Court orders otherwise, and subject only to subparagraph (b) of this paragraph 21, as applicable, be immediately entitled to exercise all of their rights and remedies in respect of the DIP Collateral and the Prepetition Collateral, in accordance with this Interim Order, the DIP Documents, and/or the Prepetition Term Loan Agreement, as applicable. The term "Maturity Date" shall mean the date that is the earliest of (i) the date that is one-hundred ten (110) days after the Petition Date, (ii) the effective date or date of the substantial consummation (as defined in Section 1101(2) of the Bankruptcy Code) of a Plan of Reorganization or similar dispositive restructuring plan that has been confirmed by an order of the Bankruptcy Court; (iii) the date the Bankruptcy Court orders the conversion of the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code; (iv) the date the Bankruptcy Court orders the dismissal of the Chapter 11 Cases; (v) the date on which the Borrower consummates a sale of all or substantially all of the assets of the Borrower or its Subsidiaries pursuant to Section 363 of the Bankruptcy Code or otherwise; (vi) with the consent of the DIP Lenders (solely to the extent the

DIP Obligations have not been satisfied in full), the confirmation of a chapter 11 plan pursuant to section 1129 of the Bankruptcy Code; and (vii) the occurrence of any Event of Default or the date of acceleration of the DIP Term Loans or early termination of the DIP Term Loan Commitments hereunder or the DIP Financing Orders, including as a result of the occurrence of an Event of Default.

(b) Notwithstanding the foregoing subparagraph (a) of this paragraph 21, immediately following the giving of written notice by the DIP Agent to counsel to the Debtors, the U.S. Trustee, and counsel to the Committee, if any, of the occurrence of an Event of Default under the DIP Loan Documents (the "Termination Notice") (i) all Commitments of the DIP Lenders to provide any DIP Loans shall immediately be suspended; (ii) the Debtors shall have no right to request or use any proceeds of any DIP Loans or DIP Collateral, or to use Cash Collateral, other than towards the satisfaction of the DIP Obligations and the Carve-Out, as provided in this Interim Order and the DIP Loan Documents (including the DIP Credit Agreement); provided that, during the Default Notice Period (as defined below), the Debtors shall be permitted solely to continue to use drawn proceeds of the DIP Facility and any Cash Collateral subject to the consent of the DIP Lenders for any critical business-related expenses necessary to operate the Debtors' businesses and preserve the DIP Collateral as determined by the Debtors in their reasonable discretion and in good faith; (iii) the Debtors shall deliver and cause the delivery of the proceeds of the DIP Loans and the DIP Collateral to the DIP Agent as provided herein and in the DIP Loan Documents (including the DIP Credit Agreement) subject to the funding of the Carve-Out, and (iv) the DIP Agent shall be permitted to apply such proceeds in accordance with the terms of this Interim Order and the other DIP Loan Documents (including the DIP Credit Agreement). The Debtors and the Committee (if any) shall be entitled to an emergency hearing before this Court (subject to the Court's availability)

within five (5) calendar days after the Termination Notice is sent by the DIP Agent to the Debtors, the U.S. Trustee, and counsel to the Committee (if any) (such 5-calendar-day period, the "Default Notice Period"). Unless the Court orders otherwise, if the Debtors, the Committee (if any), or any other party in interest does not contest the occurrence of the Event of Default and the Court after notice and a hearing declines to stay the enforcement thereof, the Maturity Date shall be deemed to have occurred for all purposes and the automatic stay shall be modified to permit the DIP Agent and the DIP Lenders to exercise all remedies under this Interim Order, the Final Order, the other DIP Loan Documents (including the DIP Credit Agreement), and applicable law, including, without limitation, to (i) set-off any and all amounts in accounts maintained by the Debtors with the DIP Agent or the DIP Lenders against the DIP Obligations, (ii) to otherwise enforce any and all rights against the DIP Collateral in the possession of any of the applicable DIP Lenders, including, without limitation, disposition of the DIP Collateral solely for application towards the DIP Obligations; and (iii) take any other actions or exercise any other rights or remedies permitted under this Interim Order, the Final Order, the other DIP Loan Documents (including the DIP Credit Agreement), or applicable law to effect the repayment of the DIP Obligations.

- (c) Upon the occurrence of the Maturity Date (but subject, only in the case of the occurrence of the Maturity Date resulting from an Event of Default, to the provisions of paragraph 21(b)), the DIP Secured Parties and the Prepetition Secured Parties are authorized to exercise all remedies and proceed under or pursuant to the applicable DIP Loan Documents (including the DIP Credit Agreement), the Prepetition Term Loan Agreement, this Interim Order, and applicable law.
- (d) The automatic stay imposed under section 362(a) of the Bankruptcy Code is hereby modified pursuant to this Interim Order and the DIP Loan Documents (including the DIP Credit Agreement) as necessary to (i) permit the Debtors to grant the Adequate Protection Liens and the

DIP Liens and to incur all DIP Obligations and all liabilities and obligations to the Prepetition Secured Parties hereunder and under the DIP Credit Agreement, as the case may be, and (ii) authorize the DIP Agent and Prepetition Agent to retain and apply payments and otherwise enforce their respective rights and remedies hereunder, subject to the provisions of paragraph 21(b) hereof.

- (e) The Debtors shall reasonably cooperate with the DIP Secured Parties and/or the Prepetition Secured Parties in their efforts to enforce their liens and security interests in the DIP Collateral or the Prepetition Collateral, as applicable, in accordance with this Interim Order and (other than the right to contest whether a DIP Termination Event has occurred and is continuing) the Debtors shall not take or direct any entity to take any action designed or intended to hinder or restrict in any respect such party from enforcing its rights or remedies in the DIP Collateral or the Prepetition Collateral, as applicable.
- (f) Nothing included herein shall prejudice, impair, or otherwise affect the Prepetition Agent's or the DIP Agent's rights to seek (on behalf of the Prepetition Secured Parties and the DIP Secured Parties, respectively) any other or supplemental relief in respect of the Debtors (including, as the case may be, other or additional adequate protection).

22. <u>Applications of Proceeds of Collateral, Payments and Collections.</u>

(a) As a condition to the DIP Loans and the authorization to use Cash Collateral, the Debtors have agreed that proceeds of any DIP Collateral and Prepetition Collateral, any amounts held on account of the DIP Collateral or Prepetition Collateral, and all payments and collections received by the Debtors with respect to all proceeds of DIP Collateral and Prepetition Collateral shall be used and applied in accordance with the DIP Loan Documents (including the DIP Credit

Agreement) (including repayment and reduction of the DIP Obligations), the DIP Budget (subject to Permitted Variances) and this Interim Order.

- (b) Subject to the Debtors' rights under paragraph 21(b) hereof and the funding of the Carve-Out, if applicable, upon and after the occurrence of the Maturity Date, all proceeds of DIP Collateral and Prepetition Collateral, whenever received, shall be paid and applied as follows: (i) *first*, to permanently and indefeasibly repay and reduce the DIP Obligations then due and owing in accordance with the DIP Loan Documents (including the DIP Credit Agreement), until paid and satisfied in full in cash; (ii) *second*, subject to paragraph 11 hereof, to permanently and indefeasibly repay and reduce the Prepetition Obligations (including any adequate protection obligations) then due and owing in accordance with the Prepetition Term Loan Agreement, until paid and satisfied in full in cash; and (iii) *third*, to the Debtors' estates. For avoidance of doubt, nothing in this Interim Order shall be construed to limit the voluntary and mandatory repayment provisions set forth in the DIP Loan Documents (including the DIP Credit Agreement).
- 23. Proofs of Claim, etc. None of the DIP Secured Parties or the Prepetition Secured Parties shall be required to file proofs of claim in the Chapter 11 Cases or any Successor Case for any claim described herein. Notwithstanding any order entered by the Court in relation to the establishment of a bar date in the Chapter 11 Cases or any Successor Case to the contrary, the DIP Agent, on behalf of itself and the DIP Secured Parties, and the Prepetition Agent, on behalf of itself and the Prepetition Secured Parties, respectively, are hereby authorized and entitled (but not required), in each of their sole and absolute discretion to file (and amend or supplement, as each sees fit) a proof of claim or aggregate proofs of claim in the Chapter 11 Cases or any Successor Case for any claim described herein; any proof of claim filed by the DIP Agent or the Prepetition Agent shall be deemed to be in addition to and not in lieu of any other proof of claim that may be

filed by any of the respective DIP Secured Parties or Prepetition Secured Parties. Any order entered by the Court in relation to the establishment of a bar date for any claim (including without limitation administrative claims) in the Chapter 11 Cases or any Successor Case shall not apply to the DIP Agent, the other DIP Secured Parties, the Prepetition Agent or the other Prepetition Secured Parties.

Payments Free and Clear. Any and all payments or proceeds remitted to the DIP Secured Parties or, except as otherwise provided herein, subject to paragraph 11 hereof, the Prepetition Secured Parties, pursuant to the provisions of this Interim Order, the DIP Loan Documents (including the DIP Credit Agreement) or any subsequent order of the Court shall be irrevocable, received free and clear of any claim, charge, assessment or other liability.

25. Other Rights and Obligations.

(a) Good Faith Under Section 364(e) of the Bankruptcy Code; No Modification or Stay of this Interim Order. The DIP Secured Parties and the Prepetition Secured Parties have acted in good faith in connection with the DIP Facility, the DIP Loans, the DIP Loan Documents (including the DIP Credit Agreement), the Prepetition Term Loan Agreement, and this Interim Order and are entitled to rely upon the protections granted herein and by section 364(e) of the Bankruptcy Code. Based on the findings set forth in this Interim Order and in accordance with section 364(e) of the Bankruptcy Code, which is applicable to the DIP Facility as approved by this Interim Order, and notwithstanding any modification, amendment, reversal, vacatur, or stay of any or all of this Interim Order by a subsequent order of the Court or any other court, the DIP Secured Parties and the Prepetition Secured Parties are entitled to the protections provided in section 364(e) of the Bankruptcy Code, and no such appeal, modification, amendment, reversal, vacatur, or stay shall affect the validity and enforceability of any DIP Obligation, DIP Lien, Adequate Protection

Superpriority Claim, or Adequate Protection Lien, or any other advances made hereunder or the claims, liens, security interest, or priority authorized or created hereby or pursuant to the DIP Loan Documents (including the DIP Credit Agreement). Notwithstanding any such modification, amendment, reversal, vacatur, or stay, any claim granted to the DIP Secured Parties hereunder arising prior to the effective date of such modification, amendment, reversal vacatur, or stay of any DIP Liens or of the DIP Superpriority Claim granted to or for the benefit of the DIP Secured Parties shall be governed in all respects by the original provisions of this Interim Order, and the DIP Secured Parties shall be entitled to all of the rights, remedies, privileges, and benefits, including the DIP Liens and the DIP Superpriority Claim granted herein, with respect to any such claim.

(b) **Expenses.** To the fullest extent provided in the Prepetition Term Loan Agreement and this Interim Order, the Debtors shall pay all reasonable and documented expenses incurred by the DIP Secured Parties and the Prepetition Secured Parties in their capacity as such (including, without limitation, the reasonable and documented fees and disbursements of their counsel, including Cole Schotz P.C., any other advisors and any other local counsel that they shall retain and any internal or third-party appraisers, consultants, financial, restructuring or other advisors and auditors advising any such counsel) without duplication, in connection with (i) the preparation, execution, delivery, funding, and administration of the DIP Credit Agreement and the other DIP Loan Documents, including, without limitation, all due diligence fees and expenses incurred or sustained in connection therewith, (ii) the administration of the Prepetition Term Loan Agreement, or (iii) enforcement of any rights or remedies under the DIP Loan Documents (including the DIP Credit Agreement) or the Prepetition Term Loan Agreement, or applicable law, in each case whether or not the transactions contemplated hereby are fully consummated. Notwithstanding any

other provisions of this Interim Order, the Prepetition Secured Parties and the DIP Secured Parties, and their advisors and professionals shall submit an invoice in summary form, which shall not be required to include time entry detail and may be redacted or modified to the extent necessary to delete any information subject to the attorney-client privilege, any information constituting attorney work product, or any other confidential information, and the provision of such invoices shall not constitute any waiver of the attorney-client privilege or of any benefits of the attorney work-product doctrine for fees and expenses incurred after entry of this Interim Order to the Debtors, the U.S. Trustee and counsel for the Committee (if any). Thereafter, within ten (10) calendar days of presentment of such statements, if no written objections to the reasonableness of the fees and expenses charged in any such invoice (or portion thereof) is made, the Debtors shall pay in cash all such fees and expenses of the Prepetition Secured Parties, the DIP Secured Parties, and their advisors and professionals. Any objection to the payment of such fees or expenses shall be made only on the basis of "reasonableness," and shall specify in writing the amount of the contested fees and expenses and the detailed basis for such objection. To the extent an objection only contests a portion of an invoice, the undisputed portion thereof shall be promptly paid. If any such objection to payment of an invoice (or any portion thereof) is not otherwise resolved between the objecting party and the issuer of the invoice, either party may submit such dispute to the Court for a determination as to the reasonableness of the relevant disputed fees and expenses set forth in the invoice. This Court shall resolve any dispute as to the reasonableness of any fees and expenses. For the avoidance of doubt and without limiting any of the foregoing or any other provision of this Interim Order, the fees specified in the "Costs and Expenses" section of the DIP Credit Agreement are, upon entry of this Interim Order and irrespective of any subsequent order approving or denying the DIP Facility or any other financing pursuant to section 364 of the Bankruptcy Code, entitled

to all of the protections of section 364(e) of the Bankruptcy Code that may apply to such fees, and are deemed fully earned, indefeasibly paid, non-refundable, irrevocable and non-avoidable as of the date of this Interim Order.

(c) Credit Bid. Subject to section 363(k) of the Bankruptcy Code, the DIP Secured Parties and the Prepetition Secured Parties shall each have the right to credit bid (either directly or through one or more acquisition vehicles), up to the full amount of the DIP Obligations, the Prepetition Obligations, and the Adequate Protection Superpriority Claims, in connection with any sale of all or any portion of the Prepetition Collateral or the DIP Collateral, including (without limitation) any sale occurring pursuant to section 363 of the Bankruptcy Code or included as part of any chapter 11 plan subject to confirmation under section 1129(b)(2)(A)(ii)-(iii) of the Bankruptcy Code, by a chapter 7 trustee under section 725 of the Bankruptcy Code, or otherwise, without the need for further court order authorizing the same. The Debtors shall not object to any DIP Secured Party or any Prepetition Secured Party credit bidding up to the full amount of the applicable outstanding DIP Obligations or Prepetition Obligations (including any Adequate Protection Superpriority Claims), in each case including any accrued interest, fees, and expenses, in any sale of any DIP Collateral or Prepetition Collateral, as applicable, whether such sale is effectuated through sections 363 or 1129 of the Bankruptcy Code, by a chapter 7 trustee under section 725 of the Bankruptcy Code, or otherwise. If the DIP Agent, DIP Secured Parties, and the Prepetition Secured Parties (either directly or through one or more acquisition vehicles) make a credit bid in connection with any auction or other sale process relating to the sale or other disposition of any Prepetition Collateral or DIP Collateral, then for purposes of such auction or sale process or any applicable order of this Court, each of the DIP Agent, DIP Secured Parties, and Prepetition Secured Parties shall be deemed to be a "qualified bidder" and its bid shall be a

"qualified bid" regardless of whether any qualified bidder or qualified bid requirements are satisfied.

- (d) Lenders Not Responsible Persons. In (i) making the decision to make the DIP Loans and consent to the use of Cash Collateral, (ii) administering the DIP Facility and the DIP Loans, (iii) extending other financial accommodations to the Debtors under the DIP Documents, and (iv) making the decision to collect the indebtedness and obligations of the Debtors, neither the DIP Agent, the DIP Secured Parties nor the Prepetition Secured Parties shall be considered to (x) owe any fiduciary obligation to the Debtors or any other party with respect to their exercise of any consent or other rights afforded them under the DIP Documents or this Interim Order or (y) be exercising control over any operations of the Debtors or acting in any way as a responsible person, or as an owner or operator under any applicable law, including without limitation, any environmental law (including but not limited to the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C. 9601, et seq., and the Resource Conservation and Recovery Act, 42 U.S.C. 6901, et seq., as either may be amended from time to time, or any similar federal or state statute, but in both cases solely by reason of having made the DIP Loans); provided however, that nothing in this paragraph 25(d) is intended to release or change the fiduciary obligations of the Debtors' officers and directors when acting, in the past, present or future, in their capacity as directors and officers of the Debtors.
- (e) **Binding Effect**. The provisions of this Interim Order shall be binding upon and inure to the benefit of the DIP Secured Parties and the Prepetition Secured Parties, the Debtors, and their successors and assigns (including any trustee or other fiduciary hereinafter appointed as a legal representative of the Debtors or with respect to the property of the estates of the Debtors)

whether in the Chapter 11 Cases, in any Successor Case, or upon dismissal of any such Chapter 11 or Chapter 7 case.

(f) No Waiver.

- The failure of the DIP Secured Parties or the Prepetition Secured Parties to (i) seek relief or otherwise exercise their rights and remedies under this Interim Order, the other DIP Loan Documents (including the DIP Credit Agreement), or the Prepetition Term Loan Agreement or otherwise, as applicable, shall not constitute a waiver of any of the DIP Secured Parties' or Prepetition Secured Parties' rights hereunder, thereunder, or otherwise. Notwithstanding anything herein, the entry of this Interim Order is without prejudice to and does not constitute a waiver of, expressly or implicitly, or otherwise impair any of the rights, claims, privileges, objections, defenses, or remedies of the DIP Secured Parties or the Prepetition Secured Parties under the Bankruptcy Code or under non-bankruptcy law against any other person or entity in any court, including without limitation, the rights of the DIP Agent and the Prepetition Agent (1) to request conversion of the Chapter 11 Cases to a case under Chapter 7, dismissal of the Chapter 11 Cases, or the appointment of a trustee in the Chapter 11 Cases, (2) to propose, subject to the provisions of section 1121 of the Bankruptcy Code, a Plan, or (3) to exercise any of the rights, claims or privileges (whether legal, equitable, or otherwise) on behalf of the DIP Secured Parties or the Prepetition Secured Parties.
- (ii) The failure or delay on the part of any DIP Secured Parties or Prepetition Secured Parties to seek relief or otherwise exercise their rights and remedies under this Interim Order, the other DIP Loan Documents (including the DIP Credit Agreement), the Prepetition Term Loan Agreement, or applicable law, as the case may be, shall not constitute a waiver of any of their respective rights hereunder, thereunder, or otherwise. No delay on the part of any party in

the exercise of any right or remedy under this Interim Order shall preclude any other or further exercise of any such right or remedy or the exercise of any other right or remedy. None of the rights or remedies of any party under this Interim Order shall be deemed to have been amended, modified, suspended, or waived unless such amendment, modification, suspension, or waiver is express, in writing and signed by the party against whom such amendment, modification, suspension, or waiver is sought. No consents required hereunder by any DIP Secured Parties or Prepetition Secured Parties shall be implied by any inaction or acquiescence by any of the DIP Secured Parties or Prepetition Secured Parties, respectively.

- (g) **No Third-Party Rights**. Except as explicitly provided for herein, this Interim Order does not create any rights for the benefit of any third party, creditor, equity holder or any direct, indirect, third party, or incidental beneficiary.
- (h) Limitation on Surcharge. Without limiting the terms of the Carve-Out and subject to the entry of the Final Order, no costs or expenses of administration which have been or may be incurred in the Chapter 11 Cases or any Successor Case at any time shall be surcharged against, and no person may seek to surcharge any costs or expenses of administration against the DIP Secured Parties, the Carve-Out (other than parties entitled to assert a right to be paid under the Carve-Out), the DIP Collateral or the Prepetition Collateral, pursuant to sections 105 or 506(c) of the Bankruptcy Code or otherwise, without the prior written consent of the DIP Agent and the Prepetition Agent (and the beneficiaries of the Carve-Out in the case of a surcharge against the Carve-Out). No action, inaction or acquiescence by the DIP Secured Parties or the Prepetition Secured Parties shall be deemed to be or shall be considered evidence of any alleged consent to a surcharge against the DIP Secured Parties, the DIP Collateral, the Prepetition Secured Parties or the Prepetition Collateral.

- (i) **No Marshaling**. Neither the DIP Secured Parties nor, subject to the entry of the Final Order, the Prepetition Secured Parties shall be subject to the equitable doctrine of "marshaling" or any other similar doctrine with respect to any of the DIP Collateral or the Prepetition Collateral, as applicable.
- (j) Section 552(b). The Prepetition Secured Parties and the DIP Secured Parties shall each be entitled to all of the rights and benefits of section 552(b) of the Bankruptcy Code; and, subject to the entry of the Final Order, to the extent provided therein, the "equities of the case" exception under section 552(b) of the Bankruptcy Code shall not apply to the DIP Secured Parties or the Prepetition Secured Parties with respect to proceeds, product, offspring, or profits of any of the Prepetition Collateral or the DIP Collateral.
- (k) Amendment. The Debtors and the DIP Agent may amend, modify, supplement, or waive any provision of the DIP Credit Agreement (a "DIP Loan Amendment") without further approval of the Court; provided that (i) such DIP Loan Amendment is not material (for purposes hereof, a "Material DIP Loan Amendment" shall (a) mean any modification or amendment that (1) increases the interest rate (other than as a result of the imposition of the default rate) or fees charged in connection with the DIP Facility, (2) increases the commitments of the DIP Lenders to make DIP Loans under the DIP Credit Agreement, or (3) changes the Maturity Date, but (b) exclude the extension of any Milestone) and is undertaken in good faith by the DIP Lenders and DIP Agent and (ii) the Debtors provide written notice of the DIP Loan Amendment on the docket. Any Material DIP Loan Amendment to the DIP Credit Agreement must be set forth in writing, signed by, or on behalf of the Debtors and the DIP Agent (after having obtained the approval of the requisite DIP Secured Parties as provided in the DIP Credit Agreement) and approved by the Court on no less than seven (7) days' notice to parties in interest in order to be

effective, *provided* that the DIP Secured Parties and the Debtors reserve the right to seek approval of any Material DIP Loan Amendment on an expedited basis.

- (l) **Priority of Terms**. To the extent of any conflict between or among (i) the express terms or provisions of any of the DIP Loan Documents (including the DIP Credit Agreement), the Motion, or any other agreements, on the one hand, and (b) the terms and provisions of this Interim Order, on the other hand, unless such term or provision herein is phrased in terms of "defined in" or "as set forth in" the DIP Loan Documents (including the DIP Credit Agreement), the terms and provisions of this Interim Order prevail.
- (m) Survival of Interim Order. The provisions of this Interim Order and any actions taken pursuant hereto shall survive entry of any order which may be entered (i) confirming any plan in the Chapter 11 Cases, (ii) converting of the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code, (iii) to the extent authorized by applicable law, dismissing the Chapter 11 Cases, (iv) withdrawing of the reference of the Chapter 11 Cases from this Court, or (v) providing for abstention from handling or retaining of jurisdiction of the Chapter 11 Cases in this Court. The terms and provisions of this Interim Order and the Prepetition Term Loan Agreement, including the DIP Liens and DIP Superpriority Claim granted pursuant to this Interim Order, the DIP Credit Agreement, and the Prepetition Term Loan Agreement and any priorities and protections granted to or for the benefit of the DIP Secured Parties and Prepetition Secured Parties (including the Adequate Protection Liens and the Adequate Protection Superpriority Claim) hereunder and thereunder shall continue in full force and effect to the fullest extent provided by section 364(e) of the Bankruptcy Code.

- (n) **Enforceability**. This Interim Order shall constitute findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052 and shall take effect and be fully enforceable *nunc pro tunc* to the Petition Date immediately upon execution hereof.
- (o) **No Waivers or Modification of Interim Order**. The Debtors irrevocably waive any right to seek any modification or extension of this Interim Order without the prior written consent of the DIP Agent and the Prepetition Agent, and no such consent shall be implied by any other action, inaction, or acquiescence of the DIP Agent or the Prepetition Agent. The Debtors may not seek to modify or to alter relative lien priority of the DIP Liens, the Prepetition Liens and the Adequate Protection Liens set forth in this Interim Order.
- (p) Waiver of any Applicable Stay. Any applicable stay (including, without limitation, under Bankruptcy Rule 6004(h)) is hereby waived and shall not apply to this Interim Order.
- 26. **Reservations to Final Order**. Notwithstanding language in this Interim Order that provides that certain relief is subject to or conditioned upon entry of a Final Order, such provisions are without prejudice to rights of parties in interest to object to the relief on a final basis and the Court's authority to determine the final relief.

27. **Final Hearing**.

- (a) The Final Hearing to consider entry of the Final Order and final approval of the DIP Facility is scheduled for July 2, 2025, at 11:00 a.m. (prevailing Eastern Time) at the United States Bankruptcy Court for the District of Delaware.
- (b) On or before two (2) business days after entry of this Interim Order, the Debtors shall cause to be served, by United States mail, first-class postage prepaid, notice of the entry of this Interim Order and of the Final Hearing (the "Final Hearing Notice"), together with copies of

Cases 25-5-0.00297-9-1485 DDo 504-2 Filled 006/004/25 **Parone** 662 of 61293

this Interim Order and the Motion, on any known party affected by the terms of this Interim Order,

and to any other party that has filed a request for notices with this Court prior thereto. The Final

Hearing Notice shall state that any party in interest objecting to the entry of the proposed Final

Order shall file written objections with the Clerk of the Court no later than June 20, 2025, at 4:00

p.m. (prevailing Eastern Time), which objections shall be served on (i) proposed counsel for the

Debtors, Pashman Stein Walder Hayden, P.C., 824 North Market Street, Suite 800, Wilmington,

Delaware 19801 (Attn.: John Weiss (jweiss @pashmanstein.com), Leah M. Eisenberg

(leisenberg@pashmanstein.com), Richard W. Riley (rriley@pashmanstein.com) and David E.

Sklar (dsklar@pashmanstein.com); (ii) counsel to the DIP Lenders, Cole Schotz P.C., 500

Delaware Avenue, Suite 1410, Wilmington, Delaware 19801 (Attn: Patrick J. Reilley

(PReilley@coleschotz.com) and Melissa M. Hartlipp (MHartlipp@coleschotz.com)) and 1325

Avenue of the Americas, 19th Floor, New York, NY 10019 (Attn: Daniel F.X. Geoghan

(DGeoghan@coleschotz.com)); (iii) the U.S. Trustee, 844 King Street, Suite 2207, Wilmington,

DE 19801 (Attn: Jane M. Leamy) (jane.m.leamy@usdoj.gov); and (iv) counsel to any official

committee appointed in the Chapter 11 Cases.

28. **Retention of Jurisdiction**. The Court has and will retain jurisdiction to enforce this

Interim Order according to its terms.

Dated: June 4th, 2025

Wilmington, Delaware

ED STATES BANKRUPTCY JUDGE

62

Cassee 225-11099799-JK\$\$ DDooc 502-21 FFileed 00690044225 FPaggee 614 of f 11293

EXHIBIT 1

DIP CREDIT AGREEMENT

SENIOR SECURED SUPERPRIORITY DEBTOR-IN-POSSESSION TERM LOAN AGREEMENT

dated as of June 2, 2025,

among

VWS HOLDCO, INC.,

a Debtor and Debtor-in-Possession under Chapter 11 of the Bankruptcy Code as Borrower,

SHOOSMITH BROS., INC.,

a Debtor and Debtor-in-Possession under Chapter 11 of the Bankruptcy Code as Guarantor

the Lenders party hereto from time to time

and

VOLUNTEER ENTERPRISES, LLC,

as Administrative Agent

Casse 225-10097/9-JK\$\$ Door: 502-21 FFileed 0067004/225 Pragge 636 of f 11293

TABLE OF CONTENTS

		Page
SECTION 1.	INTERPRETATION:	2
SECTION 2.	DIP FACILITY:	2
2.1	DIP Term Loans	2
2.2	Roll-Up Loans	3
2.3	Procedure for Advances	3
2.4	Repayment	5
2.5	Optional Prepayments	5
2.6	Mandatory Prepayments	6
2.7	No Reborrowings	7
SECTION 3.	GENERAL LOAN PROVISIONS:	7
3.1	Repayment of Principal and Outstanding Obligations	7
3.2	Interest	8
3.3	ReservedError! Bookmark not	defined.
3.4	Manner of Payment	8
3.5	Evidence of Indebtedness	9
3.6	Sharing of Payments by Lenders	9
3.7	Increased Costs	10
3.8	Taxes	11
3.9	Mitigation Obligations; Replacement of Lenders	14
3.10	Defaulting Lenders	15
3.11	No Discharge; Survival.	16
SECTION 4.	CONDITIONS OF CLOSING AND BORROWING:	16
4.1	Conditions to Closing and Funding of First DIP Term Loan Advance	16
4.2	Conditions to Funding the Second DIP Term Loan Advance	19
SECTION 5.	REPRESENTATIONS:	20
5.1	Organization; Powers	defined.
5.2	Authority; Enforceability	20
5.3	Approvals; No Conflicts	21
5.4	Financial Condition; No Material Adverse Effect Error! Bookmark not	defined.
5.5	LitigationError! Bookmark not	defined.

Casse 255-10097/99-JK\$\$ Doorc 502-21 FFileed 0066/004/225 PPagge 647 off 11293

TABLE OF CONTENTS

(continued)

Page

5.6	Environmental Matters	Error! Bookmark not defined.
5.7	Compliance with the Laws; No Defaults	Error! Bookmark not defined.
5.8	Investment Company Act	Error! Bookmark not defined
5.9	Taxes	Error! Bookmark not defined
5.10	ERISA	Error! Bookmark not defined
5.11	Disclosure; No Material Misstatements	Error! Bookmark not defined
5.12	Insurance	Error! Bookmark not defined
5.13	Restriction on Liens	Error! Bookmark not defined
5.14	Subsidiaries	Error! Bookmark not defined
5.15	Location of Business and Offices	Error! Bookmark not defined
5.16	Properties; Titles	Error! Bookmark not defined
5.17	Maintenance of Properties	Error! Bookmark not defined
5.18	Gas Imbalances, Prepayments	Error! Bookmark not defined
5.19	Marketing of Production	Error! Bookmark not defined
5.20	Swap Agreements	Error! Bookmark not defined
5.21	Use of Proceeds	23
5.22	Anti-Corruption Laws and Sanctions	Error! Bookmark not defined
5.23	Budget	23
SECTION 6.	AFFIRMATIVE COVENANTS:	23
6.1	Financial Statements; Other Information	23
6.2	Notices of Material Events	25
6.3	Existence; Conduct of Business	Error! Bookmark not defined
6.4	Payment of Obligations	Error! Bookmark not defined
6.5	Performance of Obligations	Error! Bookmark not defined
6.6	Operation and Maintenance of Properties	Error! Bookmark not defined
6.7	Insurance	Error! Bookmark not defined
6.8	Books and Records; Inspection Rights	Error! Bookmark not defined
6.9	Compliance with Laws	Error! Bookmark not defined
6.10	Environmental Matters	Error! Bookmark not defined
6.11	Further Assurances.	25

Casse 225-10097/99-JK\$\$ DDooc 502-21 FFileed 0069/094/225 PPagge 668 of f 11293

TABLE OF CONTENTS

(continued)

Page

6.12	Reserved	Error! Bookmark not defined.
6.13	Title Information	Error! Bookmark not defined.
6.14	Additional Collateral; Additional Guarantors	26
6.15	ERISA Compliance	Error! Bookmark not defined.
6.16	Swap Agreements	Error! Bookmark not defined.
6.17	Marketing Activities	Error! Bookmark not defined.
6.18	Use of Proceeds and Budget.	26
6.19	Collateral Matters	27
6.20	Chapter 11 Milestones.	28
6.21	Compliance with Anti-Corruption Laws and Sa	nctions29
6.22	Deposit Accounts	29
SECTION 7.	NEGATIVE COVENANTS:	29
7.1	[Reserved]	29
7.2	Debt	Error! Bookmark not defined.
7.3	Liens	Error! Bookmark not defined.
7.4	Restricted Payments	Error! Bookmark not defined.
7.5	Investments	Error! Bookmark not defined.
7.6	Nature of Business; International Operations	Error! Bookmark not defined.
7.7	Limitation on Leases	Error! Bookmark not defined.
7.8	Use of Proceeds	Error! Bookmark not defined.
7.9	ERISA Compliance	Error! Bookmark not defined.
7.10	Sale or Discount of Receivables	Error! Bookmark not defined.
7.11	Mergers	Error! Bookmark not defined.
7.12	Asset Dispositions	Error! Bookmark not defined.
7.13	New Deposit Accounts	Error! Bookmark not defined.
7.14	Environmental Matters	Error! Bookmark not defined.
7.15	Transactions with Affiliates	Error! Bookmark not defined.
7.16	Subsidiaries	Error! Bookmark not defined.
7.17	Negative Pledge Agreements; Dividend Restric defined.	tions Error! Bookmark not

Casse 225-11099799-JK\$\$ Doorc 542-21 Ffileed 0066/004/225 Pragge 669 of f 11293

TABLE OF CONTENTS

(continued)

T			
ν	a	Œ	C
	a	~	·

7.	.18 Gas Imbalances, Take-or-Pay or Other Prepayment defined.	s Error! Bookmark not
7.	.19 Swap AgreementsE	rror! Bookmark not defined.
7.	.20 Holding CompanyE	rror! Bookmark not defined.
7.	.21 [Reserved] E	rror! Bookmark not defined.
7.	.22 Prepayment of Prepetition Debt	29
7.	.23 Compliance with the Budget	30
7.	.24 Chapter 11 Modifications.	30
7.	.25 Additional Bankruptcy Matters	30
SECTION	N 8. DEFAULT AND REMEDIES:	32
8.	.1 Events of Default	32
8.	.2 Remedies	36
8.	.3 Rights and Remedies Cumulative; Non-Waiver	36
8.	.4 Crediting of Payments and Proceeds	36
8.	.5 Credit Bidding	37
8.	.6 Injunctive Relief	
SECTION	N 9. AGENCY:	38
9.	.1 Appointment and Authority	38
9.	.2 Rights as a Lender	38
9.	.3 Exculpatory Provisions	38
9.	.4 Reliance by Administrative Agent	41
9.	.5 Delegation of Duties	42
9.	.6 Resignation of Administrative Agent	42
9.	.7 Non-Reliance on Administrative Agent and Other I	Lenders
9.	.8 Collateral and Guaranty Matters	43
9.	.9 Erroneous Payment	44
SECTION	N 10. MISCELLANEOUS:	48
10	0.1 Governing Law	48
10	0.2 Expenses; Indemnity; Damage Waiver	48
10	0.3 Amendments Waivers and Consents	50

Casse 225-10097/99-JK\$\$ D000C5402-21 FFileed 0067/0044/225 FPagge 7/0 off 11293

TABLE OF CONTENTS

(continued)

		Page
10.4	Successors and Assigns	51
10.5	Treatment of Certain Information; Confidentiality	55
10.6	Submission to Jurisdiction; Venue; Waiver of Jury Trial	56
10.7	Notices	57
10.8	Reinstatement of Obligations	58
10.9	Severability	58
10.10	Integration	58
10.11	No Implied Waivers	58
10.12	USA PATRIOT Act	59
10.13	Counterparts; Effectiveness	59

Casse 225-1009799-JKSS Door: 502-21 Fileed 006/004/225 FPagge 781 off 11298

ANNEXES

Annex A - Rules of Construction

Annex B - DIP Term Loan Commitments and Commitment Percentages

Annex C - Addresses for Notice Annex D - Closing Deliverables

EXHIBITS

Exhibit A - Form of Notice of Borrowing
Exhibit B - Form of Notice of Prepayment

Exhibit C - Form of Responsible Officer's Compliance Certificate

Exhibit D - Form of Assignment and Assumption
Exhibit E - Proposed Interim DIP Financing Order

SCHEDULES

Schedule 5.4 - Ownership and Control
Schedule 6.1 - Financial Reporting
Schedule 7.1 - Negative Covenants

SENIOR SECURED SUPERPRIORITY DEBTOR-IN-POSSESSION TERM LOAN AGREEMENT

This Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement, dated as of June 2, 2025 (this "<u>Agreement</u>"), is among VWS Holdco, Inc. a Delaware corporation (the "<u>Borrower</u>"), Shoosmith Bros., Inc., a Virginia Corporation ("<u>Shoosmith</u>" or the "<u>Guarantor</u>"), the Lenders (as defined below) party hereto from time to time, and Volunteer Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the "<u>Administrative Agent</u>" and, together with the Borrowers, the Guarantor, and the Lenders, the "<u>Parties</u>" and each, a "<u>Party</u>").

PRELIMINARY STATEMENT:

The Borrower, the Lenders party hereto, the Administrative Agent and VWS Acquisitions, L.L.C. (the "<u>Parent</u>") are parties to that certain Amended and Restated Securities Purchase Agreement, dated as of April 30, 2020 (as amended from time to time, the "<u>Prepetition Loan Agreement</u>").

The Borrower and the Guarantor (each a "<u>Debtor</u>" and together, the "<u>Debtors</u>") each commenced a voluntary case (the "<u>Chapter 11 Cases</u>" or the "<u>Bankruptcy Case</u>") on June 1, 2025 (the "<u>Petition Date</u>") under Chapter 11 of the Bankruptcy Code (as hereinafter defined) in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>"), and the Debtors continue to operate their businesses and manage them as debtors-in-possession pursuant to Sections 1107 and 1108 of the Bankruptcy Code.

The Borrower has asked the Lenders to provide it with a senior secured superpriority debtor-in-possession term loan credit facility (the "<u>DIP Facility</u>"), consisting of (a) \$4,900,000 of new money delayed draw term loans that will be made available to the Borrower pursuant to the terms and subject to the conditions set forth in this Agreement and the DIP Financing Orders (the "<u>DIP Term Loans</u>"); and (b) Prepetition Loans in the amount of \$500,000 will be deemed "rolled up" as Loans hereunder on a dollar-for-dollar basis pursuant to the terms and subject to the conditions set forth in this Agreement and the DIP Financing Orders (the "<u>Roll-Up Loans</u>").

The Guarantor has agreed to guarantee the obligations of the Borrower hereunder;

To provide for the repayment of the DIP Facility, the Borrower and the Guarantor have agreed to secure, and have secured, their respective Obligations by granting the Administrative Agent, for its benefit and the benefit of the other Secured Parties, subject to the Carve-Out, certain rights and protections pursuant to the terms hereof, security interests and liens pursuant to Sections 364(c)(2), 364(c)(3) and 364(d) of the Bankruptcy Code and superpriority administrative expense claims pursuant to Section 364(c)(1) of the Bankruptcy Code, in each case having the relative priorities as set forth in the DIP Financing Orders, and other rights and protections as more fully described herein and in the DIP Financing Orders.

The Borrower believes that the loans and other financial accommodations provided to the Borrower under this Agreement will preserve the value of the Borrower's business and assets during the Chapter 11 Case.

The Borrower has requested, and subject to the terms and conditions set forth in this Agreement, the Lenders have agreed to extend, the DIP Facility to the Borrower.

AGREEMENT:

In consideration of the foregoing and the mutual agreements contained in this Agreement, the receipt and sufficiency of which are acknowledged, the Parties hereby agree as follows:

SECTION 1. INTERPRETATION:

This Agreement is to be interpreted in accordance with the rules of construction set forth on Annex A. Capitalized terms used in this Agreement and not otherwise defined have the meanings set forth for such terms on Annex A. All annexes, schedules and exhibits to this Agreement are deemed to be a part of this Agreement.

SECTION 2. DIP FACILITY:

2.1 DIP Term Loans.

- (A) Subject to the terms and conditions of this Agreement and in the DIP Financing Orders, and in reliance upon the representations set forth in this Agreement and the DIP Financing Orders, the Lenders shall make the DIP Term Loans available to be drawn by the Borrower from time to time in an aggregate original principal amount equal to the Lenders' total DIP Term Loan Commitment (with each Lender severally making DIP Term Loans in an amount equal to such Lender's DIP Term Loan Commitment). Each Lender's DIP Term Loan Commitment shall be permanently reduced on a dollar for dollar basis by the principal amount of the DIP Term Loan made by such Lender.
- (B) Subject to the terms and conditions of this Agreement and the Interim DIP Financing Order, including the satisfaction of the conditions set forth in Section 4.1, on or after the Closing Date the Lenders shall make DIP Term Loans to the Borrower in one or more additional draws (each, a "First DIP Term Loan Advance") in an aggregate amount up to the First DIP Term Loan Commitment.
- (C) Subject to the terms and conditions of this Agreement and the applicable DIP Financing Order, including this Section 2.1(C) and the satisfaction of the conditions set forth in Section 4.2, the Lenders shall make the DIP Term Loans to the Borrower (after making First DIP Term Loan Advances in the amount of the First DIP Term Loan Commitment), in one or more additional draws (each, a "Second DIP Term Loan Advance") in an aggregate amount up to the Second DIP Term Loan Commitment.
- (D) Subject to the terms and conditions of this Agreement and the applicable DIP Financing Order, including this Section 2.1(D) and the satisfaction of the conditions set forth in Section 4.3, the Lenders shall make the DIP Term Loans to the Borrower (after making First DIP Term Loan Advances in the amount of the First DIP Term Loan Commitment and Second DIP Term Loan Advances in the amount of the

- Second DIP Term Loan Commitment), in one or more additional draws (each, a "<u>Third DIP Term Loan Advance</u>") in an aggregate amount up to the Third DIP Term Loan Commitment.
- (E) The obligations of the Lenders hereunder to make the First DIP Term Loan Advances, Second DIP Term Loan Advances and Third DIP Term Loan Advances are several and not joint. The failure of any Lender to make any DIP Term Loan or, as applicable, to make any such payment on any date required hereunder shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so make its applicable DIP Term Loan (including the First DIP Term Loan Advances, Second DIP Term Loan Advances and Third DIP Term Loan Advances).
- 2.2 Roll-Up Loans.
- (A) Subject to the terms and conditions of the Interim DIP Financing Order, simultaneously with the First DIP Term Loan Commitment being available to Borrower, the Prepetition Loans, including accrued and unpaid interest thereon, shall be deemed Loans advanced under this Agreement, shall constitute Roll-Up Loans and, subject to the terms and conditions set forth herein and without any further action by any party to this Agreement, shall be administered hereunder. Roll-Up Loans under this Section 2.2(A) may be prepaid or repaid but may not be reborrowed.
- (B) [Reserved]
- 2.3 Procedure for Advances.
- (A) First DIP Term Loan Advances.
 - The Borrower shall give the Administrative Agent and the Lenders an (i) irrevocable Notice of Borrowing prior to 11:00 a.m. at least two Business Days prior to any proposed date of borrowing (or such shorter period as may be acceptable to the Lenders in their sole discretion), specifying (i) the amount of funds to be disbursed, (ii) wire instructions of the Borrower's account(s) to which funds are to be disbursed, (iii) the requested date of the borrowing, which shall be a Business Day, and (iv) the Type of DIP Term Loans to be borrowed. A Notice of Borrowing received after 11:00 a.m. is deemed received on the next Business Day. Not later than 1:00 p.m. on the date the Borrower requests a First DIP Term Loan Advance pursuant to a Notice of Borrowing provided by Borrower in accordance with this paragraph (A), and after the acceptance by the Lenders of the latest approved Budget as set forth in Section 6.14(B) and the Borrower having met all conditions as set forth in Section 4.1, each Lender will make available to the Administrative Agent, for the account of the Borrower, at the office of the Administrative Agent in funds immediately available to the Administrative Agent, such Lender's Commitment Percentage of the First

DIP Term Loan Advance requested by the Borrower to be made on such date. The Borrower hereby irrevocably authorizes the Administrative Agent to disburse the proceeds of any First DIP Term Loan Advance received by the Administrative Agent in immediately available funds by wire transfer to such Persons as may be designated by the Borrower in writing. The Administrative Agent shall only be required to fund after the Bankruptcy Court approves the Interim DIP Financing Order in the Chapter 11 Cases.

(B) Second DIP Term Loan Advance.

The Borrower shall give the Administrative Agent and the Lenders an (i) irrevocable Notice of Borrowing prior to 11:00 a.m. at least two Business Days prior to the date of any Second DIP Term Loan Advance, specifying (i) the amount of such advance, (ii) wire instructions of the Borrower's account(s) to which funds are to be disbursed, (ii) the requested date of the borrowing, which shall be a Business Day, (iv) the Type of DIP Term Loans to be borrowed, and (v) the Final DIP Financing Order approving such draw has been (or will be) entered and all of the conditions to such draw set forth in Section 4.2 have been (or will be) satisfied on or prior to the date of such draw. A Notice of Borrowing received after 11:00 a.m. is deemed received on the next Business Day. Not later than 1:00 p.m. on the date the Borrower requests a Second DIP Term Loan Advance pursuant to a Notice of Borrowing provided pursuant to a Notice of Borrowing provided by Borrower in accordance with this paragraph (B), and after the acceptance by the Lenders of the latest approved Budget as set forth in Section 6.18(B) and the Borrower having met all conditions as set forth in Sections 4.1 and 4.2, each Lender will make available to the Administrative Agent, for the account of the Borrower, at the office of the Administrative Agent in funds immediately available to the Administrative Agent, such Lender's Commitment Percentage of the Second DIP Term Loan Advance to be made The Borrower hereby irrevocably authorizes the on such date. Administrative Agent to disburse the proceeds of the Second DIP Term Loan Advance received by the Administrative Agent in immediately available funds by wire transfer to such Persons as may be designated by the Borrower in writing.

(C) Third DIP Term Loan Advance.

(i) The Borrower shall give the Administrative Agent and the Lenders an irrevocable Notice of Borrowing prior to 11:00 a.m. at least two Business Days prior to the date of any Third DIP Term Loan Advance, specifying (i) the amount of such advance, (ii) wire instructions of the Borrower's account(s) to which funds are to be disbursed, (ii) the requested date of the borrowing, which shall be a Business Day, (iv) the Type of DIP Term Loans to be borrowed, and (v) the Final DIP Financing Order approving such draw has been (or will be) entered and all of the conditions to such draw set forth

in Section 4.3 have been (or will be) satisfied on or prior to the date of such draw. A Notice of Borrowing received after 11:00 a.m. is deemed received on the next Business Day. Not later than 1:00 p.m. on the date the Borrower requests a Third DIP Term Loan Advance pursuant to a Notice of Borrowing provided by Borrower in accordance with this paragraph (C), and after the acceptance by the Lenders of the latest approved Budget as set forth in Section 6.18(B) and the Borrower having met all conditions as set forth in Sections 4.1, 4.2 and 4.3 each Lender will make available to the Administrative Agent, for the account of the Borrower, at the office of the Administrative Agent in funds immediately available to the Administrative Agent, such Lender's Commitment Percentage of the Third DIP Term Loan Advance to be made on such date. The Borrower hereby irrevocably authorizes the Administrative Agent to disburse the proceeds of the Third DIP Term Loan Advance received by the Administrative Agent in immediately available funds by wire transfer to such Persons as may be designated by the Borrower in writing.

- (D) The Administrative Agent is not obligated to disburse the portion of the proceeds of any DIP Term Loans requested pursuant to this <u>Section 2</u> to the extent that any Lender has not made available to the Administrative Agent its Commitment Percentage of such DIP Term Loans. If a Lender does not make available to the Administrative Agent its Commitment Percentage of any requested DIP Term Loan, then the Administrative Agent shall return the funds it received from the other Lenders with respect to such requested DIP Term Loan.
- 2.4 <u>Repayment</u>. The Borrower shall repay the aggregate outstanding principal amount of the DIP Term Loans and Roll-Up Loans in full, together with accrued interest thereon, on the Maturity Date.
 - 2.5 Optional Prepayments.
 - (A) <u>Prepayments</u>.
 - The Borrower may from time to time, without premium or penalty, prepay (i) the DIP Facility, in whole or in part, upon delivery to the Administrative Agent and the Lenders of a Notice of Prepayment not later than 11:00 a.m. at least three (3) Business Days before its intention to prepay specifying the date, which must be a Business Day, and amount of prepayment. Each optional prepayment of the DIP Facility hereunder must be in an aggregate principal amount of at least \$500,000 or any whole multiple of \$100,000 in excess thereof. A Notice of Prepayment received after 11:00 a.m. is deemed received on the next Business Day. The Administrative Agent shall promptly notify the Lenders of each Notice of Prepayment; provided, however, that the delivery by the Borrower of such Notice of Prepayment to each Lender directly shall satisfy this requirement. Any prepayment of the DIP Facility shall be accompanied by all accrued interest thereon. Notwithstanding the foregoing, any Notice of Prepayment delivered in

- connection with any refinancing of all of the DIP Facility with the proceeds of such refinancing may be, if expressly so stated to be, contingent upon the consummation of such refinancing or incurrence and may be revoked by the Borrower in the event such refinancing or incurrence is not consummated.
- shall be applied first to all costs, expenses, indemnities and other amounts due and payable hereunder, then proportionately (based on the relation of such amounts to the total amount of the relevant payment under this Section 2.5 to the payment or prepayment (as applicable) of the following amounts of the Obligations: default interest, if any, accrued interest and principal. All payments hereunder will be applied on a pro rata basis to the thenoutstanding DIP Facility.

(B) <u>Commitment Termination</u>.

- (i) The DIP Term Loan Commitments of each Lender to fund the DIP Term Loans pursuant to Section 2.1(A), (B) and (C) shall expire upon the earlier of the date (the "DIP Term Loan Commitment Termination Date") that (a) the DIP Term Loans are fully funded or reduced to zero and (b) the Maturity Date.
- (ii) The Borrower may at any time terminate or reduce (including to zero) any undrawn amount of the DIP Term Loan Commitments upon prior notice to the Administrative Agent and the Lenders.
- (iii) The Borrower shall notify the Administrative Agent and the Lenders of any election to reduce (including to zero) the DIP Term Loan Commitments under paragraph (ii) of this Section at least three (3) Business Days prior to the effective date of such reduction, specifying such election and the effective date thereof.

2.6 <u>Mandatory Prepayments</u>.

- (A) <u>Asset Dispositions and Involuntary Asset Dispositions.</u>
 - (i) Subject to (x) the priority of Liens and application of funds set forth in the DIP Financing Orders, and (y) the use of such Net Cash Proceeds as set forth in the most-recent Budget, the Borrower shall promptly (and, in any event, within three (3) Business Days) prepay the DIP Facility in an aggregate amount equal to 100% of the Net Cash Proceeds of any Asset Disposition or Involuntary Asset Disposition received by the Borrower or any other Debtor. Any prepayment pursuant to this Section 2.6(A)(i) shall be applied as set forth in Section 2.6(D) below.
- (B) Extraordinary Receipts.

(i) Subject to (x) the priority of Liens and application of funds set forth in the DIP Financing Orders, and (y) the use of such Net Cash Proceeds as set forth in the most-recent Budget, the Borrower shall promptly (and, in any event, within three (3) Business Days) upon the receipt by the Borrower or any other Debtor of the Net Cash Proceeds of any Extraordinary Receipt, prepay the DIP Facility in an aggregate amount equal to 100% of such Net Cash Proceeds. Any prepayment pursuant to this this Section 2.6(B)(i) shall be applied as set forth in Section 2.6(D) below.

(C) <u>Debt Issuances</u>.

- (i) Subject to the priority of Liens and application of funds set forth in the DIP Financing Orders, the Borrower shall promptly (and, in any event, within three (3) Business Days) upon the receipt by the Borrower or any other Debtor of the Net Cash Proceeds of any Debt Issuance, prepay the DIP Facility in an aggregate amount equal to 100% of such Net Cash Proceeds. Any prepayment pursuant to this Section 2.6(C)(i) shall be applied as set forth in Section 2.6(D) below.
- (D) <u>Application of Mandatory Prepayments</u>. All payments under this <u>Section 2.6</u> shall be applied first to all costs, expenses, indemnities and other amounts due and payable hereunder, then proportionately (based on the relation of such amounts to the total amount of the relevant payment under this <u>Section 2.6</u> to the payment or prepayment (as applicable) of the following amounts of the Obligations: default interest, if any, accrued interest and principal. All payments hereunder will be applied on a pro rata basis to the then-outstanding DIP Facility.
- (E) <u>Notice of Prepayment</u>. Upon the occurrence of any event triggering the prepayment requirement under this <u>Section 2.6</u>, the Borrower shall promptly deliver a Notice of Prepayment to the Administrative Agent and the Lenders.
- (F) <u>Accrued Interest</u>. Any prepayment of the DIP Facility pursuant to this <u>Section 2.6</u> shall be accompanied by all accrued interest thereon.
- 2.7 <u>No Reborrowings</u>. The Borrower may not reborrow any amounts repaid or prepaid under the DIP Term Loans or Roll-Up Loans.

SECTION 3. GENERAL LOAN PROVISIONS:

- 3.1 Repayment of Principal and Outstanding Obligations.
- (A) The Borrower hereby unconditionally promises to pay the Administrative Agent, for the benefit of the Lenders, all amounts owed hereunder with respect to the DIP Term Loans and the Roll-Up Loans (including, for the avoidance of doubt, any PIK Amounts) on the Maturity Date. Each repayment of a DIP Term Loan and/or Roll-Up Loan shall be applied ratably to the Loans. Repayments of Loans shall be accompanied by accrued interest on the amounts repaid. Any reference in this

Agreement or the Loan Documents to the Loans or the outstanding principal balance of the Loans shall include the PIK Amounts.

3.2 Interest.

- (A) <u>Interest Rate</u>. Each Loan (including the PIK Amounts) shall bear interest at a rate *per annum* equal to the Interest Rate.
- (B) <u>Default Rate</u>. Subject to <u>Section 8.3</u>, immediately upon the occurrence and during the continuance of an Event of Default, the Borrower shall pay interest on the aggregate outstanding principal amount of the DIP Term Loans and Roll-Up Loans at the Default Rate, such interest being due and payable on demand.
- (C) <u>Interest Payment and Computation</u>. The Borrower shall pay interest (at the Interest Rate or Default Rate, as applicable) on the DIP Facility in arrears on each Interest Payment Date. The aggregate outstanding principal amount of the DIP Facility shall be automatically increased on each such Interest Payment Date by the amount of such interest paid in kind (the "<u>PIK Amounts</u>"). For the avoidance of doubt interest shall accrue on the principal amount of the DIP Facility increased as a result of payments of interest in kind. All computations of interest are made on the basis of a 360-day year and actual days elapsed.
- (D) Maximum Rate. In no contingency or event whatsoever will the aggregate of all amounts deemed interest under this Agreement charged or collected pursuant to the terms of this Agreement exceed the highest rate permissible under any Applicable Law that a court of competent jurisdiction, in a final determination, deems applicable hereto. In the event that such a court determines that the Lenders have charged or received interest hereunder in excess of the highest applicable rate, the rate in effect hereunder automatically is reduced to the maximum rate permitted by Applicable Law and the Lenders shall apply such excess to the principal balance of the Obligations. It is the intent hereof that the Borrower not pay or contract to pay, and that neither the Administrative Agent nor any Lender receive or contract to receive, directly or indirectly in any manner whatsoever, interest in excess of that which may be paid by the Borrower under Applicable Law.

3.3 Reserved.

3.4 Manner of Payment. The Borrower shall make each payment on account of the principal of or interest on the DIP Facility or of any fee, commission or other amounts payable to the Lenders under this Agreement not later than 1:00 p.m. on the date specified for payment under this Agreement to the Administrative Agent at the Administrative Agent's Office for the account of the Lenders entitled to such payment in Dollars, in immediately available funds and without any set off, counterclaim or deduction whatsoever. Any payment received after 1:00 p.m. is deemed to have been made on the next succeeding Business Day for all purposes. Upon receipt by the Administrative Agent of each such payment, the Administrative Agent shall distribute to each Lender its Commitment Percentage (or other applicable share as provided herein) of such payment and shall provide notice of the amount of such credit to each Lender. The Borrower shall

pay each payment to the Administrative Agent of Administrative Agent's fees, expenses or other amounts payable to the Administrative Agent for the account of the Administrative Agent and shall pay any amount payable to any Lender under Sections 3.7, 3.8 or 10.2 to the Administrative Agent for the account of the applicable Lender. If any payment under this Agreement is specified to be made upon a day that is not a Business Day, it shall be made on the next succeeding day that is a Business Day and such extension of time is included in computing any interest if payable along with such payment. Notwithstanding the foregoing, each payment by the Borrower to a Defaulting Lender is applied in accordance with Section 3.10(A)(ii).

- 3.5 Evidence of Indebtedness. The DIP Term Loans made by each Lender will be evidenced by one or more accounts or records maintained by such Lender in the ordinary course of business and by the Administrative Agent in accordance with this Agreement. The accounts or records maintained by the Administrative Agent and each Lender are conclusive absent manifest error of the amount of the DIP Term Loans made by the Lenders to the Borrower and the interest and payments thereon. Any failure to so record or any error in doing so do not, however, limit or otherwise affect the obligation of the Borrower hereunder to pay any amount owing with respect to the Obligations. In the event of any conflict between the accounts and records maintained by any Lender and the accounts and records of the Administrative Agent in respect of such matters, the accounts and records of the Administrative Agent control in the absence of manifest error.
- 3.6 Sharing of Payments by Lenders. Subject to the DIP Financing Orders, if any Lender, by exercising any right of setoff or counterclaim or otherwise, obtains payment in respect of any principal of or interest on any of its DIP Term Loans or other obligations hereunder resulting in such Lender receiving payment of a proportion of the aggregate amount of its DIP Term Loans and accrued interest thereon or other such obligations greater than its pro rata share thereof as provided herein, then the Lender receiving such greater proportion shall (a) notify the Administrative Agent of such fact and (b) purchase (for cash at face value) participations in the DIP Term Loans and such other obligations of the other Lenders, or make such other adjustments as are equitable, so that the benefit of all such payments are shared by the Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective DIP Term Loans and other amounts owing them; provided:
 - (A) if any such participations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations are rescinded and the purchase price restored to the extent of such recovery, without interest; and
 - (B) the provisions of this Section 3.6 may not be construed to apply to (i) any payment made by the Borrower pursuant to and in accordance with the express terms of this Agreement (including the application of funds arising from the existence of a Defaulting Lender) or (ii) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its DIP Term Loans to any assignee or participant.

Each Loan Party consents to the foregoing and agrees, to the extent it may effectively do so under Applicable Law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against each Loan Party rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of each Loan Party in the amount of such participation.

3.7 Increased Costs.

(A) In the event that any Lender shall have determined (which determination shall, absent manifest error, be final and conclusive and binding upon all parties hereto) any Change in Law shall subject any Recipient to any Taxes (other than (A) Indemnified Taxes, (B) Taxes described in clauses (b) through (d) of the definition of Excluded Taxes and (C) Connection Income Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities or capital attributable thereto;

then, and in any such event, such Lender shall promptly give written notice to the Borrower and to the Administrative Agent of such determination (which notice the Administrative Agent shall promptly transmit to each of the other Lenders). The Borrower agrees to pay to such Lender, upon such Lender's written request therefor, such additional amounts (in the form of an increased rate of, or a different method of calculating, interest or otherwise as such Lender shall determine after consultation with the Borrower) as shall be required to compensate such Lender for such increased costs or reductions in amounts received or receivable hereunder (a written notice as to the additional amounts owed to such Lender, showing in reasonable detail the basis for the calculation thereof, submitted to the Borrower by such Lender shall, absent manifest error, be final and conclusive and binding on all the parties hereto).

(B) If any Lender determines that after the date of this Agreement the introduction of or any change in any Applicable Law or governmental rule, regulation, order, guideline, directive or request (whether or not having the force of law) concerning capital adequacy, or any change in interpretation or administration thereof by the NAIC or any Governmental Authority, central bank or comparable agency, will have the effect of increasing the amount of capital required or expected to be maintained by such Lender or any corporation controlling such Lender based on the existence of such Lender's DIP Term Loan Commitments hereunder or its obligations hereunder, then the Borrower agrees to pay to such Lender, upon its written demand therefor, such additional amounts as shall be required to compensate such Lender or such other corporation for the increased cost to such Lender or such other corporation or the reduction in the rate of return to such Lender or such other corporation as a result of such increase of capital. In determining such additional amounts, each Lender will act reasonably and in good faith and will use averaging and attribution methods which are reasonable; provided that such Lender's determination of compensation owing under this Section 3.7(B) shall, absent manifest error, be final and conclusive and binding on all the parties hereto. Each Lender, upon determining that any additional amounts, will be payable pursuant to this Section 3.7(B), will give prompt written notice thereof to the Borrower, which notice shall show in reasonable detail the basis for calculation of such additional amounts, although the failure to give any such notice shall not release or diminish a Borrower's obligations to pay additional amounts pursuant to this <u>Section 3.7(B)</u> upon the subsequent receipt of such notice.

- (C) Notwithstanding anything in this Agreement to the contrary, (x) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines, requirements and directives thereunder, issued in connection therewith or in implementation thereof and (y) all requests, rules, guidelines or directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States regulatory authorities, in each case pursuant to Basel III, shall be deemed to be a change after the date of this Agreement in a requirement of law or government rule, regulation or order, regardless of the date enacted, adopted, issued or implemented (including for purposes of this Section 3.6).
- 3.8 <u>Taxes</u>.
- (A) <u>Defined Terms</u>. For purposes of this <u>Section 3.8</u>, the term "applicable law" includes FATCA.
- (B) Payments Free of Taxes. Any and all payments by or on account of any obligation of any Loan Party under any Loan Document shall be made without deduction or withholding for any Taxes, except as required by applicable law. If any applicable law (as determined in the good faith discretion of an applicable Withholding Agent) requires the deduction or withholding of any Tax from any such payment by a Withholding Agent, then the applicable Withholding Agent is entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Authority in accordance with applicable law and, if such Tax is an Indemnified Tax, then the sum payable by the applicable Loan Party is increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section 3.8) the applicable Recipient receives an amount equal to the sum it would have received had no such deduction or withholding been made.
- (C) <u>Payment of Other Taxes by the Borrower</u>. The Loan Parties shall timely pay to the relevant Governmental Authority in accordance with applicable law, or at the option of the Administrative Agent timely reimburse it for the payment of, any Other Taxes.
- (D) Indemnification by the Borrower. The Loan Parties shall jointly and severally indemnify each Recipient, within 10 days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by such Recipient or required to be withheld or deducted from a payment to such Recipient and any reasonable expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority, and shall indemnify the Administrative Agent, as Withholding Agent, in connection with any tax withholdings. A certificate as to the amount of such payment or liability delivered to the Borrower by a Lender (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender, is conclusive absent manifest error.

- (E) Indemnification by the Lenders. Each Lender shall severally indemnify the Administrative Agent, within 10 days after demand therefor, for (i) any Indemnified Taxes attributable to such Lender (but only to the extent that any Loan Party has not already indemnified the Administrative Agent for such Indemnified Taxes and without limiting the obligation of the Loan Parties to do so), (ii) any Taxes attributable to such Lender's failure to comply with the provisions of Section 10.4(D) relating to the maintenance of a Participant Register and (iii) any Excluded Taxes attributable to such Lender, in each case, that are payable or paid by the Administrative Agent in connection with any Loan Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. A certificate as to the amount of such payment or liability delivered to any Lender by the Administrative Agent is conclusive absent manifest error. Each Lender hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender under any Loan Document or otherwise payable by the Administrative Agent to the Lender from any other source against any amount due to the Administrative Agent under this Section 3.8(E).
- (F) Evidence of Payments. As soon as practicable after any payment of Taxes by any Loan Party to a Governmental Authority pursuant to this Section 3.8, such Loan Party shall deliver to the Administrative Agent the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to the Required Lenders.
- (G) Status of Lenders. (i) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to the Borrower and the Administrative Agent, at the time or times reasonably requested by the Borrower or the Administrative Agent, such properly completed and executed documentation reasonably requested by the Borrower or the Administrative Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by the Borrower or the Administrative Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Section 3.8(G)(ii) is not required if in the Lender's reasonable judgment such completion, execution or submission would subject such Lender to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Lender.
- (ii) Any Lender that is a U.S. Person shall deliver to the Borrower and the Administrative Agent on or prior to the date on which such Lender becomes a Lender under this Agreement (and from time to time thereafter upon the reasonable request of the Borrower or the

Administrative Agent), executed copies of IRS Form W-9 certifying that such Lender is exempt from U.S. federal backup withholding tax. Each Lender agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify the Borrower and the Administrative Agent in writing of its legal inability to do so.

Volunteer Enterprises, LLC, both in its individual capacity and in its capacity as the Administrative Agent, has no liability to the Borrower, the Lenders or any other Person in connection with any tax withholding amounts paid or withheld from any payment pursuant to Applicable Law or arising from the Borrower's or a Lender's failure, as applicable, to timely provide an accurate, correct and complete IRS Form W-9, an appropriate IRS Form W-8 or such other documentation contemplated under this Agreement.

- (H) Treatment of Certain Refunds. If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this Section 3.8 (including by the payment of additional amounts pursuant to this Section 3.8), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this Section 3.8 with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnified party and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this Section 3.8(H) (plus any penalties, interest or other charges imposed by the relevant Governmental Authority) in the event that such indemnified party is required to repay such refund to such Governmental Authority. Notwithstanding anything to the contrary in this Section 3.8(H), in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this Section 3.8(H) the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnification payments or additional amounts with respect to such Tax had never been paid. This Section 3.8(H) may not be construed to require any indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party or any other Person.
- (I) <u>Survival</u>. Each Party's obligations under this <u>Section 3.8</u> survives the resignation or replacement of the Administrative Agent or any assignment of rights by, or the replacement of, a Lender, the termination of the DIP Term Loan Commitments and the repayment, satisfaction or discharge of all obligations under any Loan Document.
- (J) <u>Tax Reporting</u>. The Lenders and any transferees or assignees after the Closing Date will be required to provide to Administrative Agent or its agents all information, documentation or certifications reasonably requested by Administrative Agent to

- permit Administrative Agent to comply with its tax reporting obligations under Applicable Laws, including any applicable cost basis reporting obligations.
- (K) Not less than four (4) Business Days (or such shorter period as may be agreed to by Administrative Agent) prior to any payment, distribution or transfer of funds by Administrative Agent to any Person under the Loan Documents, the payee shall provide to Administrative Agent such documentation and information as may be requested by Administrative Agent (unless such Person has previously provided the documentation or information, and so long as such documentation or information remain accurate and true). Administrative Agent shall have no duty, obligation or liability to make any payment to any Person unless it has timely received such documentation and information with respect to such Person, which documentation and information shall be reasonably satisfactory to Administrative Agent.
- 3.9 Mitigation Obligations; Replacement of Lenders.
- (A) <u>Designation of a Different Lending Office</u>. If any Lender requests compensation under <u>Section 3.7</u>, or requires the Borrower to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.8</u>, then such Lender shall (at the request of the Borrower) use reasonable efforts to designate a different lending office for funding or booking its DIP Term Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to <u>Section 3.7</u> or <u>3.8</u>, as the case may be, in the future and (ii) would not subject such Lender to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender. The Borrower shall pay all reasonable costs and expenses incurred by any Lender in connection with any such designation or assignment.
- (B) Replacement of Lenders. If any Lender requests compensation under Section 3.7, or if the Borrower is required to pay any Indemnified Taxes or additional amounts to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 3.8 and, in each case, such Lender has declined or is unable to designate a different lending office in accordance with Section 3.9(A), or if any Lender is a Defaulting Lender or a Non-Consenting Lender, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 10.4), all of its interests, rights (other than its existing rights to payments pursuant to Section 3.7 or 3.8) and obligations under this Agreement and the related Loan Documents to an Eligible Assignee that assumes such obligations (which assignee may be another Lender, if a Lender accepts such assignment); provided:
- (i) [reserved];

- (ii) such Lender has received payment of an amount equal to the outstanding principal of its DIP Term Loans or Roll-Up Loans, accrued interest thereon and all other amounts payable to it hereunder and under the other Loan Documents from the assignee (to the extent of such outstanding principal and accrued interest) or the Borrower (in the case of all other amounts);
- (iii) in the case of any such assignment resulting from a claim for compensation under <u>Section 3.7</u> or payments required to be made pursuant to <u>Section 3.8</u>, such assignment will result in a reduction in such compensation or payments thereafter;
- (iv) such assignment does not conflict with Applicable Law; and
- (v) in the case of any assignment resulting from a Lender becoming a Non-Consenting Lender, the applicable assignee has consented to the applicable amendment, waiver or consent.

A Lender is not required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

- 3.10 <u>Defaulting Lenders</u>.
- (A) <u>Defaulting Lender Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as such Lender is no longer a Defaulting Lender, to the extent permitted by Applicable Law:
- (i) Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement is restricted as set forth in the definition of Required Lenders.
- Any payment of principal, interest, fees or other amounts received by the (ii)Administrative Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Section 8 or otherwise) shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; second, as the Borrower may request (so long as no Default or Event of Default exists), to the funding of any DIP Term Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement; third, if so determined by the Administrative Agent and the Borrower, to be held in a deposit account and released pro rata in order to satisfy such Defaulting Lender's potential future funding obligations with respect to its DIP Term Loans under this Agreement; fourth, to the payment of any amounts owing to the Lenders as a result of any judgment of a court of competent jurisdiction obtained by any Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; fifth, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction

obtained by the Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and sixth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (a) such payment is a payment of the principal amount of any DIP Term Loans in respect of which such Defaulting Lender has not fully funded its appropriate share and (b) such DIP Term Loans were made at a time when the conditions set forth in Section 4.1 or 4.2 were satisfied or waived, such payment is applied solely to pay the DIP Term Loans of all Non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any DIP Term Loans of such Defaulting Lender until such time as all DIP Term Loans are held by the Lenders pro rata in accordance with the DIP Term Loan Commitments. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender pursuant to this Section 3.9(A)(ii) are deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.

(B) <u>Defaulting Lender Cure</u>. If the Borrower and the Required Lenders agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent shall so notify the Parties, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein, that Lender shall, to the extent applicable, purchase at par that portion of outstanding DIP Term Loans of the other Lenders or take such other actions as may be necessary to cause the DIP Term Loans to be held pro rata by the Lenders in accordance with the DIP Term Loan Commitments, whereupon such Lender will cease to be a Defaulting Lender. Except to the extent otherwise expressly agreed by the affected Parties, no change hereunder from Defaulting Lender to Lender constitutes a waiver or release of any claim of any Party arising from that Lender's having been a Defaulting Lender.

3.11 No Discharge; Survival.

Unless in accordance with an Acceptable Plan, the Borrower agrees that prior to payment in full in cash of the Obligations and termination of the DIP Term Loan Commitments in accordance herewith, (a) its obligations under the Loan Documents shall not be discharged by the entry of an order confirming a Plan of Reorganization and (b) the superpriority claims granted to the Administrative Agent and the Lenders pursuant to the DIP Financing Orders and the Liens granted to the Administrative Agent and the Lenders pursuant to the DIP Financing Orders shall not be affected in any manner by the entry of an order confirming a Plan of Reorganization, except to the extent set forth in an Acceptable Plan confirmed pursuant to a final, non-appealable order from the Bankruptcy Court that is consummated.

SECTION 4. CONDITIONS OF CLOSING AND BORROWING:

4.1 <u>Conditions to Closing and Funding of any First DIP Term Loan Advance</u>. The obligation of the Lenders to close this Agreement and to make any First DIP Term Loan Advance is subject to the satisfaction of each of the following conditions:

- (A) Executed Loan Documents. Each agreement, instrument or other document set forth on Annex D has been duly authorized, executed and delivered to the Administrative Agent on behalf of the Lenders and to the Lenders by the parties thereto and is in full force and effect.
- (B) <u>Interim DIP Financing Order.</u> Subject to the availability of the Bankruptcy Court shall have entered the Interim DIP Financing Order, in a form acceptable to the Lenders (in their sole and exclusive discretion) and the Administrative Agent (in its reasonable discretion), no later than five (5) Business Days after the Petition Date, which Interim DIP Financing Order shall be in full force and effect and shall not have been reversed, vacated or stayed, and shall not have been amended, supplemented or otherwise modified without the prior written consent of the Lenders; and the Loan Parties shall be in compliance with the Interim DIP Financing Order.
- (C) <u>Governing Documents</u>. The Lenders shall have received a certificate executed by a Responsible Officer of the Borrower certifying:
 - (i) the names and signatures of the officers of each Loan Party authorized to execute the Loan Documents;
 - (ii) the resolutions duly adopted by the Board of Directors (or equivalent governing body) of each Loan Party approving the Loan Documents;
 - (iii) the correctness and completeness of the copy of the Operating Agreements (or equivalent governing document) of each Loan Party attached thereto; and
 - (iv) the correctness and completeness of the copy of the certificate of incorporation (or equivalent governing document) of each Loan Party attached thereto.
- (D) <u>Lender Approval</u>. The Lenders shall have received approval of the transaction contemplated hereby.
- (E) <u>Initial Approved Budget.</u> No later than two (2) days after the Petition Date, the Lenders shall have received, and approved, a cash forecast for the period from the Petition Date through the following thirteen (13) weeks setting forth projected cash flows and disbursements (the "<u>Initial Approved Budget</u>").
- (F) <u>First Day Motions</u>.
 - (i) The Borrower shall have provided the Lenders with a copy of the "first day" motions, including the cash management motion, and proposed orders to be filed with the Bankruptcy Court in connection with the commencement of the Chapter 11 Cases, in form and substance satisfactory to the Required Lenders.

- (ii) The Bankruptcy Court shall have entered orders approving all "first day" motions, either on an interim or final basis as the case may be, other than the Interim DIP Financing Order or as may be waived by the Lenders.
- (G) Restructuring Support Agreement or Plan of Reorganization. Other than as set forth herein, the Borrower shall not have executed, entered into or otherwise committed to any plan or restructuring support agreement or any other agreement or understanding concerning the terms of a Chapter 11 plan or other exit strategy without the consent of the Lenders.
- (H) <u>No Trustee, Examiner, or Receiver</u>. No order shall have been entered appointing a trustee, examiner or receiver (or local law equivalent) with respect to the Borrower or its Subsidiaries' business, properties or assets, and no motion shall be pending seeking any such relief.
- (I) Administrative Agent and Lender Expenses. The Borrower shall have paid to the Administrative Agent and the Lenders, the fees, costs and expenses then earned, due and payable under the Loan Documents (including, without limitation, the fees, costs and expenses of counsel and the other advisors to the Administrative Agent and the Lenders incurred in connection with the Chapter 11 Cases and the negotiation, preparation, execution and delivery of the Loan Documents).
- (J) Payment at Closing. The Borrower has paid or made arrangements to pay contemporaneously with closing (i) all reasonable fees, charges and disbursements of counsel to the Administrative Agent and the Lenders (directly to such counsel if requested by the Administrative Agent) to the extent accrued and unpaid and submitted to the Borrower prior to or on the Closing Date, plus such additional amounts of such fees, charges and disbursements as constitute its reasonable estimate of such fees, charges and disbursements incurred or to be incurred by it through the closing proceedings (provided that such estimate does not thereafter preclude a final settling of accounts between the Borrower and the Administrative Agent and the Lenders) and (ii) to any other Person such amount as may be due thereto in connection with the transactions contemplated hereby, including all taxes, fees and other charges in connection with the execution, delivery, recording, filing and registration of any of the Loan Documents.
- (K) The Administrative Agent and the Lenders shall have received a Notice of Borrowing from the Borrower in accordance with <u>Section 2.3</u>.
- (L) The representations contained in this Agreement and the other Loan Documents are true and correct in all material respects, on and as of the Closing Date with the same effect as if made on and as of such date (except for any such representation that by its terms is made only as of an earlier date, which representation remains true and correct in all material respects as of such earlier date).
- (M) No Default or Event of Default exists on the Closing Date with respect to such DIP Term Loans or after giving effect to the DIP Term Loans to be made on such date.

- (N) <u>Material Adverse Effect</u>. Since the Petition Date, other than those customarily resulting from the commencement of the Chapter 11 Cases:
 - (i) no event, development or circumstance has occurred or exists that has resulted in, or could reasonably be expected to result in, a Material Adverse Effect; and
 - (ii) there has not been any fact, event, change, effect, development, circumstance or occurrence that, individually or together with any other fact, event, change, effect, development, circumstance or occurrence, has had or would reasonably be expected to have a material and adverse effect on the ability of the Loan Parties taken as a whole to perform their respective obligations under, or to consummate the transactions contemplated by, the Loan Documents.

Notwithstanding anything to the contrary contained herein, funding of any First DIP Term Loan Advance shall be subject to entry of the Interim DIP Financing Order, which shall not have been vacated, reversed or stayed, appealed (and for which the appeal period has expired or has been waived), or modified or amended without the prior written consent of the Lenders in their sole discretion.

- 4.2 <u>Conditions to Funding any Second DIP Term Loan Advance</u>. The obligation of the Lenders to make any Second DIP Term Loan Advance is subject to the satisfaction of each of the following conditions, in addition to all of the conditions set forth in <u>Section 4.1</u>:
 - (A) <u>Final DIP Financing Order</u>. The Final DIP Financing Order shall have been entered by the Bankruptcy Court, in a form acceptable to the Lenders, in their sole and exclusive discretion, shall be in full force and effect, shall not have been vacated or reversed, shall not be subject to a stay and shall not have been modified or amended other than as acceptable to the Required Lenders in their sole and exclusive discretion; and the Loan Parties shall be in compliance with the Final DIP Financing Order.
 - (B) <u>Second Day Orders</u>. All "second day orders" approving on a final basis any first day orders intended to be entered on or prior to the date of entry of the Final DIP Financing Order shall have been entered by the Bankruptcy Court, shall be in full force and effect, shall not have been vacated or reversed, shall not be subject to a stay, and shall be in form and substance satisfactory to the Required Lenders.
 - (C) The Debtors shall have (a) entered into and delivered a fully executed and binding Acceptable Stalking Horse Purchase Agreement and (b) satisfied any other applicable Chapter 11 Milestones;
 - (D) The Bankruptcy Court shall have approved the Bidding Procedures Motion in a form satisfactory to the Lenders and scheduled the auction.
 - (E) <u>Compliance with Orders</u>. The Loan Parties shall be in compliance in all material respects with each First Day Order and Second Day Order then in effect.

- (F) <u>Approved Budget</u>. The Lenders shall have received, and approved, the Budget in form and substance acceptable to the Lenders in their sole discretion, at least two (2) Business Days before funding the Second DIP Term Loan Advance.
- (G) The representations contained in this Agreement and the other Loan Documents are true and correct in all material respects, on and as of the Closing Date with the same effect as if made on and as of such date (except for any such representation that by its terms is made only as of an earlier date, which representation remains true and correct in all material respects as of such earlier date).
- (H) <u>Material Adverse Effect</u>. No event shall have occurred, or circumstance exists, that has or could reasonably be expected to have a Material Adverse Effect.
- (I) <u>Legal Restriction</u>. Such advance or financial accommodation shall not be prohibited by any law or regulation or any order of any court or Governmental Authority or agency.
- (J) <u>No Repudiation</u>. No Loan Party shall have repudiated or made any anticipatory breach or repudiation of any of its obligations under any Loan Document.
- 4.3 <u>Conditions to Funding any Third Term Loan Advance</u>. The obligation of the Lenders to make any Third DIP Term Loan Advance shall become effective from and after July ___, 2025, and shall be subject to the satisfaction of each of the conditions set forth in <u>Sections 4.1</u> and 4.2.

SECTION 5. REPRESENTATIONS:

To induce the Administrative Agent and Lenders to enter into this Agreement and to induce the Lenders to make the DIP Term Loans, the Loan Parties represent to the Administrative Agent and the Lenders both before and after giving effect to the transactions contemplated hereunder, which representations and warranties are deemed made on the Closing Date, that:

- Organization; Powers. Subject to any restrictions arising on account of the Loan Parties' status as "debtors" under the Bankruptcy Code, the Loan Parties (a) are duly organized, incorporated or formed, validly existing and (where applicable) in good standing under the Laws of the jurisdiction of its incorporation or organization, (b) have all requisite power and authority and all requisite governmental licenses, authorizations, consents and approvals to (i) own or lease its assets and carry on their businesses and (ii) execute, deliver and perform their obligations under the Loan Documents to which they are a party, and (c) are duly qualified and are licensed and (where applicable) in good standing under the Laws of each jurisdiction where their ownership, lease or operation of properties or the conduct of their businesses require such qualification or license; except in each case referred to in clause (b)(i) or (c), to the extent that failure to do so could not reasonably be expected to have a Material Adverse Effect.
- 5.2 <u>Authority; Enforceability</u>. Subject to the entry of the DIP Financing Orders, the Transactions are within the Borrower's and Guarantor's corporate powers and have been duly authorized by all necessary corporate and, if required, stockholder action (including any action required to be taken by any class of directors of the Borrower or any other Person, whether

interested or disinterested, in order to ensure the due authorization of the Transactions). Subject to the entry of the DIP Financing Orders, each Loan Document to which the Borrower and each Guarantor is a party has been duly executed and delivered by the Borrower and such Guarantor and constitutes a legal, valid and binding obligation of the Borrower and such Guarantor, as applicable, enforceable in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally and subject to general principles of equity, regardless of whether considered in a proceeding in equity or at law.

- Approvals; No Conflicts. Other than as otherwise previously disclosed to the Lenders, subject to the entry of the DIP Financing Orders Transactions (a) do not require any consent or approval of, registration or filing with, or any other action by, any Governmental Authority or any other third Person (including shareholders or any class of directors, whether interested or disinterested, of the Borrower or any other Person), nor is any such consent, approval, registration, filing or other action necessary for the validity or enforceability of any Loan Document or the consummation of the transactions contemplated thereby, except such as have been obtained or made and are in full force and effect other than the recording and filing of the Security Documents as required by this Agreement and those third party approvals or consents which, if not made or obtained, would not cause an Event of Default hereunder, could not reasonably be expected to have a Material Adverse Effect or do not have an adverse effect on the enforceability of the Loan Documents, (b) will not violate any Applicable Law or regulation or the charter, by-laws or other organizational documents of the Loan Parties or any order of any Governmental Authority, (c) will not violate or result in a default under any indenture, agreement or other instrument binding upon it or any of its Subsidiaries or its Properties, or give rise to a right thereunder to require any payment to be made by it or such Subsidiary, (d) breach or result in a default (or an event which, with the giving of notice or the passage of time, or both, would constitute a default) under, require any consent under, result in the imposition of any Lien under or give to others any rights of termination, acceleration, suspension, revocation, cancellation or amendment of any Material Agreement and (e) will not result in the creation or imposition of any Lien on any Property of it or any of its Subsidiaries (other than the Liens created by the Loan Documents).
- Ownership and Control. Schedule 5.4 sets forth a correct and complete list of (a) 5.4 the authorized capitalization of Parent and each Loan Party, (b) the number of Equity Interests of Parent and each Loan Party issued and outstanding, and (c) the names of the record and beneficial owners of the Equity Interests of Parent and each Loan Party and the number and percentage of Parent's and each Loan Party's Equity Interests owned by such Persons, in each case as of the Closing Date. Each Loan Party's and Parent's issued and outstanding Equity Interests are duly authorized, validly issued, fully paid and non-assessable as of the Closing Date. Such securities have been or, at the Closing Date, will be issued in compliance with all applicable state and federal securities laws. There are no outstanding (w) options, warrants or other rights issued by Parent or any Loan Party for the acquisition of its Equity Interests, (x) securities or other obligations of Parent or any Loan Party which are convertible into its Equity Interests, (y) agreements of Parent or any Loan Party to issue or sell its Equity Interests, or (z) options, sale agreements, equityholder agreements, pledges, proxies, voting trusts, powers of attorney, restrictions on transfer or other agreements or instruments which are binding on Parent's or any Loan Party's equityholders and which relate to the ownership, voting or transfer of Parent's or such Loan Party's Equity Interests.

- 5.5 <u>Subsidiaries</u>. Parent has no Subsidiaries other than the Borrower, and the Borrower has no Subsidiaries other than Shoosmith. As of the Closing Date, the Borrower is the sole direct and wholly-owned Subsidiary of Parent, Shoosmith is the sole direct and wholly-owned Subsidiary of the Borrower and Shoosmith has no Subsidiaries. No Loan Party is party to a joint venture arrangement or owns any Equity Interest in any Person who is not a Loan Party.
 - 5.6 Financial Condition.
 - (a) [reserved]
 - (b) [reserved]
- (c) Except as and to the extent otherwise disclosed in the most recent Financial Statements or the most recent balance sheet (collectively, the "<u>Current Financial Statements</u>"), the Loan Parties have no material liabilities of any kind, whether direct or indirect, fixed or contingent or otherwise, other than (i) executory obligations under Material Agreements which are not required to be set forth in the Current Financial Statements in accordance with GAAP and (ii) liabilities incurred in the ordinary course of business since the date of the Current Financial Statements.
- (d) The Loan Parties have previously delivered to the Lenders projected consolidated and consolidating balance sheets and statements of income, retained earnings and cash flows of Parent and its consolidated Subsidiaries for Fiscal Year 2025. Such projections were prepared taking into account the past operations and results of the Loan Parties and are based upon the estimates and assumptions stated therein, all of which the Loan Parties believe to be reasonable and fair in light of current conditions and current facts actually known to the Loan Parties. Such projections reflect, as of the Closing Date, the Loan Parties' good faith and reasonable estimates of the future financial performance of the Loan Parties and of the other information projected therein for the periods set forth therein.
- 5.7 <u>Litigation</u>. Other than the Chapter 11 Cases or as otherwise previously disclosed to the Lenders, there is neither pending nor, to the actual knowledge of the Loan Parties' knowledge, threatened against any Loan Party or its assets any action, proceeding or investigation by or before any Governmental Authority or arbitrator.
- 5.8 <u>Compliance with Requirements of Law, Constituent Documents and Material Agreements</u>. Other than as otherwise previously disclosed to the Lenders, each Loan Party is in compliance with (a) all Requirements of Law applicable to such Loan Party or to its business or assets, (b) its constituent documents, and (c) all Material Agreements, except in each case for such non-compliance as would not, individually or in the aggregate, have a Material Adverse Effect.
- 5.9 Governmental Permits, Intellectual Property. Other than as otherwise previously disclosed to the Lenders, each Loan Party has obtained and maintained in full force and effect, and is in compliance with, (a) all permits, licenses, franchises, authorizations, consents and approvals from governmental entities necessary for the ownership or operation of its assets or for the conduct of its business as presently conducted and as presently planned to be conducted, and (b) all licenses and other agreements necessary for it to lawfully use any intellectual property used in the conduct of its business as presently conducted and as presently planned to be conducted, except in each

case where the failure to obtain or maintain the same or such non-compliance would not, individually or in the aggregate, have a Material Adverse Effect.

5.10 <u>Title to Assets</u>. Other than as otherwise previously disclosed to the Lenders, each Loan Party has (a) good and marketable fee simple title to all real property purported to be owned by it, (b) good and marketable title to all other assets purported to be owned by it, and (c) good leasehold title to all assets purported to be leased by it, in each case free and clear of all Liens other than Excepted Liens.

5.11 Certain Requirements of Law.

- (a) The Loan Parties are not subject to regulation under the Investment Company Act of 1940, the Federal Power Act, the Interstate Commerce Act, the Commodity Exchange Act or any Requirement of Law limiting its ability to incur indebtedness for borrowed money or consummate the transactions contemplated by the Loan Documents.
- (b) No Loan Party (i) is a Person whose property or interest in property is blocked or subject to blocking pursuant to Section 1 of Executive Order 13224 of September 23, 2001 Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)), (ii) engages in any dealings or transactions prohibited by Section 2 of such executive order, or is otherwise associated with any such Person in any manner violative of Section 2, or (iii) is a Person on the list of Specially Designated Nationals and Blocked Persons or subject to the limitations or prohibitions under any other U.S. Department of Treasury's Office of Foreign Assets Control regulation or executive order.
 - (c) The Loan Parties are in compliance, in all material respects, with the Patriot Act.
- 5.12 <u>Use of Proceeds</u>. The Loan Parties have used the proceeds of the DIP Term Loans solely as permitted in <u>Section 6.14</u>. The Loan Parties are not engaged principally, or as one of its or their important activities, in the business of extending credit for the purpose, whether immediate, incidental or ultimate, of buying or carrying margin stock (within the meaning of Regulation T, U or X of the Board).
- 5.13 <u>Budget</u>. Each Budget has been prepared in good faith on the basis of the assumptions stated therein, which assumptions were believed by the Borrower to be reasonable on the date such Budget was delivered.

SECTION 6. AFFIRMATIVE COVENANTS:

Until all of the Obligations (other than contingent indemnification obligations not then due) have been paid and satisfied in full in cash and the DIP Term Loan Commitments terminated or expired, each Loan Party shall, and shall cause each of its Subsidiaries to:

- 6.1 <u>Financial Statements; Other Information</u>. Comply at all times with the covenants regarding information and reporting requirements set forth on <u>Schedule 6.1</u>.
- 6.2 <u>Visitation and Access to Records</u>. The Loan Parties will permit such Persons as the Administrative Agent may designate to visit and inspect any of their respective properties, to examine

and to make copies or make extracts from their respective books and records and to discuss their respective affairs with their respective officers, employees and independent accountants at such times (during normal business hours and upon reasonable prior notice (which prior notice shall not be required following the occurrence and during the continuance of an Event of Default)) and as often as the Administrative Agent may reasonably request. The Loan Parties will bear the reasonable costs and expenses of such visitations and inspections, provided that so long as no Event of Default or Default exists, the Loan Parties shall not be required to reimburse the Administrative Agent for such visitations or inspections more frequently than twice each Fiscal Year.

- 6.3 <u>Financial Accounting Practices</u>. The Loan Parties will make and keep books, records and accounts which, in reasonable detail, accurately and fairly reflect their respective financial positions and operations and which otherwise comply with good accounting practices and GAAP.
- 6.4 Preservation of Existence and Franchises. The Loan Parties will (a) continue to engage in their respective business and operations substantially as conducted immediately prior to the Closing Date, (b) maintain their respective existences, rights and franchises in full force and effect in their respective jurisdictions of organization, (c) qualify and remain qualified as foreign business entities in good standing in each jurisdiction in which the failure to obtain or maintain such qualification would, individually or in the aggregate, have a Material Adverse Effect, and (d) obtain and maintain in full force and effect, and comply with, all permits, licenses, franchises, authorizations, consents and approvals from governmental entities necessary for the ownership or operation of their assets or the conduct of their businesses, except where the failure to obtain or maintain the same or such non-compliance would not, individually or in the aggregate, have a Material Adverse Effect.
- assets and businesses and against such liabilities, casualties and contingencies and of such types, in such amounts and with such financially sound and reputable third party insurers as is customary in the case of Persons engaged in the same or a similar business or having similar properties similarly situated. The Loan Parties will cause to be delivered to the Administrative Agent, upon their written request, certificates for all of the foregoing policies. The activities and operations of the Loan Parties will be conducted at all times in a manner so as to conform in all material respects to such insurance policies, and no Loan Party will take or fail to take any action which would cause any such insurance policy to lapse.
- 6.6 <u>Maintenance of Assets</u>. The Loan Parties will (a) maintain in good repair (ordinary wear and tear excepted), working order and condition the assets now or in the future owned, leased or otherwise possessed by them, (b) make all material repairs, renewals, replacements, improvements and additions to such assets, and (c) not use or permit the use of any of such assets for any unlawful purpose or permit any nuisance to exist thereon.
- 6.7 <u>Payment of Liabilities</u>. The Loan Parties will pay or discharge the following, except those which are being disputed in good faith and by appropriate proceedings, diligently prosecuted to final resolution and as to which appropriate reserves are being maintained to the extent required by GAAP: (a) on or prior to the date on which penalties attach, all taxes imposed upon them or any of their respective assets, (b) on or prior to the date when due, all lawful claims of materialmen,

mechanics, carriers, warehousemen, landlords and other like Persons which, if unpaid, might result in the creation of a Lien upon any of their respective properties or income (other than an Excepted Lien), and (c) on or prior to the date when due, all other lawful claims the nonpayment of which would, individually or in the aggregate, have a Material Adverse Effect.

- 6.8 <u>Compliance With Requirements of Law, Constituent Documents and Material Agreements</u>. The Loan Parties will comply with all applicable Requirements of Law, ¹ their respective constituent documents and all Material Agreements, except for such non-compliance as would not, individually or in the aggregate, have a Material Adverse Effect.
- 6.9 <u>Federal Reserve Board Regulations</u>. The Loan Parties will neither take nor permit to be taken on their behalf any action which might cause any transaction, obligation or right created by any of the Loan Documents to violate any regulation of the Federal Reserve Board applicable to the Administrative Agent.
- 6.10 <u>Anti-Corruption</u>. The Loan Parties shall (i) not permit any Covered Entity to become a Sanctioned Person, (ii) not permit any Covered Entity, either in its own right or through any third party, to (A) have any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Corruption Law; (B) do business in or with, or derive any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Corruption Law; or (C) engage in any dealings or transactions prohibited by any Anti-Corruption Law, and (iii) cause each Covered Entity to comply with all Anti-Corruption Laws.

6.11 Further Assurances

- (A) It shall, and shall cause each of its Subsidiaries to, at its own expense, promptly execute and deliver to the Administrative Agent all such other documents, agreements and instruments reasonably requested by the Required Lenders to comply with, cure any defects or accomplish the conditions precedent, covenants and agreements of it or any of its Subsidiaries, as the case may be, in the Loan Documents or to further evidence and more fully describe the collateral intended as security for the Obligations, or to correct any omissions in this Agreement or the other Loan Documents, or to state more fully the obligations secured by the Loan Documents, or to perfect, protect or preserve any Liens created pursuant to this Agreement or any of the Loan Documents or the priority thereof, or to make any recordings, file any notices or obtain any consents, all as may be reasonably necessary or appropriate, in the sole discretion of the Required Lenders, in connection therewith.
- (B) Each Loan Party authorizes the Administrative Agent and the Lenders to file one or more financing or continuation statements, and amendments thereto, relative to all or any part of the Collateral without the signature of such Loan Party where permitted by law. A carbon, photographic or other reproduction of the Security

¹ NTD – to discuss.

- Documents or any financing statement covering the Collateral or any part thereof shall be sufficient as a financing statement where permitted by law.
- (C) Notwithstanding any provision in any of the Loan Documents to the contrary, in no event will any Excluded Deposit Account be encumbered by any Security Document; <u>provided</u>, the Borrower shall not, and shall not permit any of its Subsidiaries to, permit to exist any Lien on any Excluded Deposit Account.

6.13 Additional Collateral; Additional Guarantors

(A) Borrower shall promptly cause each of its Subsidiaries to guarantee the Obligations pursuant to the Guaranty Agreement. Borrower shall cause such Subsidiary to promptly (but with respect to any Subsidiary formed or acquired after the date hereof, no later than ten days after the date of such formation or acquisition), (i) execute and deliver the Guaranty Agreement or a supplement to the Guaranty Agreement as required by the Required Lenders, (ii) pledge all of the Equity Interests of such Subsidiary (including delivery of original certificates evidencing the Equity Interests of such Subsidiary, together with an appropriate undated stock or equity powers for each certificate duly executed in blank by the registered owner thereof) and (iii) execute and deliver such other additional closing documents, certificates and legal opinions as are reasonably requested by the Required Lenders in connection with this Section 6.13.

6.14 <u>Use of Proceeds and Budget.</u>

- (A) Unless otherwise approved by the Administrative Agent in writing (at the direction of the Required Lenders), the proceeds of the DIP Term Loans shall be used by the Borrower strictly in accordance with the Budget and subject to the covenant set forth below in Section 6.18(B) (including variances).
- (B) The Borrower shall also deliver to the Administrative Agent and Lenders the Budget, delivered before noon on Friday of each week commencing with the second week following the Petition Date, showing actual receipts and disbursements through and including the immediately preceding week. The Budget attached to the Interim DIP Financing Order constitutes the Initial Approved Budget. Commencing with the second week following the Petition Date, the Borrower will subsequently provide the Lenders with an updated budget before noon every Friday prior to the end of the Budget Period for the remaining weeks left under the Initial Approved Budget.
- (C) Every week, before noon beginning on the Friday that is one week following the Petition Date, the Borrower shall deliver to the Administrative Agent and the Lenders supplements to the Budget showing projected receipts and disbursements for the subsequent 13-week period. The timing and delivery of the Budget may be modified by written agreement of the Borrower and the Administrative Agent (acting at the direction of a majority in interest of Lenders). Each updated budget shall be subject to the approval of the Lenders in their sole discretion (each such

approved budget, a "<u>Supplemental Approved DIP Budget</u>" and together with the Initial Approved Budget, the "<u>Budget</u>"); <u>provided</u> that until such time as the Lenders approve any updated budget, the Borrower shall be subject to and be governed by the terms of the Initial Approved Budget or such Supplemental Approved DIP Budget, as applicable, then in effect.

- (D) Beginning on the Friday that is the first week following the Petition Date, and measured weekly thereafter, the Borrower shall not permit the actual total disbursements (excluding any disbursements that are or would be reimbursable by a liquidator) to exceed, on a cumulative basis, 115% of the total disbursements set forth in the Budget for such period (each, a "Permitted Variance").
- (E) The Borrower shall hold weekly calls with the Lenders following the delivery of the updated Budget on a day and time to be mutually agreed upon by the Borrower and the Lenders.

6.15 Collateral Matters.

Subject to the Carve-Out and the DIP Financing Orders, all Obligations of the Loan Parties shall be:

- (A) Entitled to superpriority claim status under section 364(c)(1) of the Bankruptcy Code with priority over all administrative expense claims and unsecured claims existing as of the Petition Date or arising thereafter under the Bankruptcy Code, including, without limitation, the prepetition claims and adequate protection claims of the Prepetition Lenders (the "DIP Superpriority Claims"). The DIP Superpriority Claims may be repaid from any cash of the Debtors, including without limitation, Cash Collateral (within the meaning of the DIP Financing Orders) and, following entry of the Final DIP Financing Order, the proceeds of Avoidance Actions and Avoidance Action Proceeds;
- (B) secured, pursuant to section 364(c)(2) of the Bankruptcy Code, by valid, enforceable, first priority, security interests in and liens on all of the Debtors' rights in property of the Debtors' estates as of the Petition Date that, as of the Petition Date, were unencumbered (and do not become perfected subsequent to the Petition Date as permitted by section 546(b) of the Bankruptcy Code) (including, following entry of the Final DIP Financing Order, Avoidance Action Proceeds);
- (C) secured, pursuant to section 364(c)(3) of the Bankruptcy Code, by valid, enforceable, security interests in and liens on all of the Debtors' rights in property of the Debtors' estates as of the Petition Date that, as of the Petition Date, were subject to valid liens and unavoidable liens in existence immediately prior to the Petition Date, if any, that are perfected subsequent to the Petition Date as permitted by section 546(b) of the Bankruptcy Code (the "Permitted Prior Liens"), which security interests and liens shall be junior and subordinate only to such Permitted Prior Liens and the Carve-Out; and

(D) secured, pursuant to section 364(d)(1) of the Bankruptcy Code, by valid, enforceable, priming first priority security interests in and liens upon all of the Debtors' rights in property of the Debtors' estates as of the Petition Date, and all of the Debtors' rights in property acquired post-petition (and proceeds thereof), whether now existing or hereafter acquired or arising, that secure the obligations in respect of the Prepetition Term Loan Agreement.

The Collateral shall also include any and all hereafter acquired or arising property, rents, issues, products, offspring, proceeds and profits generated by any item of Collateral and, following entry of the Final DIP Financing Order, liens on proceeds of any Avoidance Actions.

The Liens shall not be subject or subordinate to (i) any lien or security interest that is avoided and preserved for the benefit of any Debtor and their estates under section 551 of the Bankruptcy Code, (ii) any liens arising after the Petition Date including, without limitation, any liens or security interests granted in favor of any federal, state, municipal or other governmental unit, commission, board or court for any liability of any Debtor, or (iii) any intercompany or affiliate liens of any Debtor.

The Collateral will be free and clear of other liens, claims and encumbrances, except valid, enforceable liens, rights of recoupment enforceable in bankruptcy, and rights of setoff permissible under section 553 of the Bankruptcy Code, in each case except as otherwise agreed by the applicable creditor or lienholder, as applicable, in existence as of the Petition Date and permitted pursuant to Prepetition Credit Agreements, if any, and any other Permitted Prior Liens.

The Borrower acknowledges that the Liens shall automatically attach to the Collateral and became valid immediately upon entry of the Interim DIP Financing Order without the requirement of any further action by the Lender; provided that if the Lender determines to file any financing statements, notice of liens or similar instruments, the Debtors will cooperate and assist in any such filings.

6.16 Chapter 11 Milestones.

The Loan Parties shall comply with and meet the following milestones as the same may be extended, waived, or otherwise modified with the prior written consent of the Required Lenders in their sole discretion (which may be by email) without further order by the Bankruptcy Court (collectively, the "Chapter 11 Milestones" and each a "Chapter 11 Milestone"):

- (A) The Bankruptcy Court shall have entered the Interim DIP Financing Order by the date that is no later than five (5) Business Days after the Petition Date;
- (B) The Debtors shall have filed the motion seeking entry of the Bidding Procedures Order and Sale Order (the "Bidding Procedures Motion") by no later than ten (10) days after the Petition Date;
- (C) The Bankruptcy Court shall have entered the Final DIP Financing Order by the date that is no later than thirty-three (33) days after the Petition;

- (D) The Bankruptcy Court shall have entered the order approving the Bidding Procedures Motion by no later than thirty-three (33) days after the filing of the Bidding Procedures Motion;
- (E) The Debtors shall have (a) entered into and delivered a fully executed and binding Acceptable Stalking Horse Purchase Agreement and (b) satisfied any other applicable Chapter 11 Milestones by no later than the date of entry of the Final DIP Financing Order;
- (F) The Bankruptcy Court shall have entered the order approving the Sale Transaction by no later than eighty (80) days after the Petition Date;
- (G) The Debtors shall have closed the Sale Transaction by no later than eighty-five (85) days after the Petition Date; and
- (H) The Plan Effective Date shall have occurred by no later than one hundred ten (110) days after the Petition Date.

The Loan Parties shall also provide the Administrative Agent and the Lenders with drafts of all first day pleadings, second day pleadings, sale pleadings and plan pleadings in the Bankruptcy Case two (2) days prior to the anticipated filing date, subject to waiver by the Lenders in their sole discretion.

- 6.17 <u>Compliance with Anti-Corruption Laws and Sanctions</u>. The Borrower shall maintain in effect and enforce policies and procedures designed to ensure compliance by the Borrower and its Subsidiaries and their respective directors, officers, employees and agents with Anti-Corruption Laws and applicable Sanctions.
- 6.18 <u>Deposit Accounts</u>. Each Loan Party shall provide the Administrative Agent with an Account Control Agreement in form and substance reasonably satisfactory to the Required Lenders duly executed by such Loan Party and the financial institute maintaining each Deposit Account of such Loan Party other than (a) Excluded Deposit Accounts and (b) Deposit Accounts used for the payment of corporate credit cards so long as the aggregate amount on deposit in such Deposit Accounts does not exceed \$100,000 at any time.

SECTION 7. NEGATIVE COVENANTS:

Until all of the Obligations (other than contingent, indemnification obligations not then due) have been paid and satisfied in full in cash and the DIP Term Loan Commitments terminated or expired, the Loan Parties shall not, and shall not permit any of their respective Subsidiaries to:

- 7.1 <u>Negative Covenants</u>. Fail to be in compliance at any time with the negative covenants set forth in <u>Schedule 7.1</u>.
 - 7.2 <u>Prepayment of Prepetition Debt.</u>
 - (A) It shall not make (or give any notice with respect thereto) any voluntary or optional payment or prepayment or redemption or acquisition for value of (including without

limitation, by way of depositing money or securities with the Administrative Agent with respect thereto before due for the purpose of paying when due), refund, refinance or exchange of any Debt of the Loan Parties (other than (i) the Obligations arising under the Loan Documents and (ii) payments in respect of Debt expressly contemplated under the Budget (subject to Permitted Variances), in each case, to the extent permitted by the DIP Financing Orders, the First and Second Day Orders and adequate protection payments as set forth in the DIP Financing Orders).

7.3 Compliance with the Budget.

- (A) It shall not, except as approved in advance in writing by the Required Lenders (or the Administrative Agent at the direction of the Required Lenders), directly or indirectly:
 - (i) use any cash or the proceeds of any DIP Term Loans in a manner or for a purpose other than those consistent with this Agreement, the DIP Financing Orders and the approved Budget (other than a Permitted Variance);
 - (ii) permit a disbursement causing any variance from the approved Budget other than a Permitted Variance; or
 - (iii) transfer cash or Cash Equivalents to a Subsidiary or Affiliate other than payments set forth in and consistent with the approved Budget (subject to any Permitted Variance).

It is understood and agreed, for the avoidance of doubt, that Permitted Variances under this Agreement shall be limited to variances in excess of aggregate disbursements of items contemplated by the approved Budget and not for the purposes of additional categories or types of expenditures or disbursements not expressly contemplated in the approved Budget.

7.4 Chapter 11 Modifications.

(A) Borrower shall not, without the prior written consent of the Required Lenders (or the Administrative Agent at the direction of the Required Lenders): (a) make or permit to be made any change, amendment or modification, to the DIP Financing Orders or (b) incur, create, assume or suffer to exist or permit any claim or Lien against the Borrower ranking *pari passu* with or senior to the claims and Liens of the Administrative Agent and the other Secured Parties hereunder, except with respect to the Carve-Out or as expressly permitted under the Interim DIP Financing Order approved by the Bankruptcy Court as of the Effective Date or under the Final DIP Financing Order, as applicable.

7.5 Additional Bankruptcy Matters.

(A) Borrower shall not do any of the following other than as expressly permitted by the DIP Financing Orders or as may be agreed to in writing by the Required Lenders (or the Administrative Agent at the direction of the Required Lenders):

- (i) enter into any agreement to return any of its inventory to any of its creditors for application against any Prepetition Debt, prepetition trade payables or other prepetition claims under Section 546(c) of the Bankruptcy Code or agree that any creditor may take any set-off or recoupment against any of its Prepetition Debt, prepetition trade payables or other prepetition claims based upon any such return pursuant to Section 553(b)(1) of the Bankruptcy Code or otherwise if, after giving effect to any such agreement, set-off or recoupment, the aggregate amount applied to Prepetition Debt, prepetition trade payables and other prepetition claims subject to all such agreements, set-offs and recoupments since the Petition Date would exceed \$100,000;
- (ii) seek, consent to, or permit to exist, without timely filing an objection, without the prior written consent of the Required Lenders (or the Administrative Agent at the direction of the Required Lenders), any order granting authority to take any action that is prohibited by the terms of this Agreement, the DIP Financing Orders or the other Loan Documents or refrain from taking any action that is expressly required to be taken by the terms of this Agreement, the DIP Financing Orders or any of the other Loan Documents;
- (iii) assert, join, investigate, support or prosecute any claim or cause of action against (i) any of the Secured Parties, unless such claim or cause of action is solely to enforce the Loan Documents against any of the Secured Parties or (ii) any Prepetition Secured Party;
- (iv) subject to the terms of the DIP Financing Orders and subject to Section 8, object to, contest, prevent or interfere with in any manner the exercise of rights and remedies by the Administrative Agent, the Lenders or other Secured Parties with respect to the Collateral following the occurrence of an Event of Default, including, without limitation, a motion or petition by any Secured Party to lift an applicable stay of proceedings to do the foregoing (provided, that, the Loan Parties may contest or dispute whether an Event of Default has occurred in accordance with the terms of the DIP Financing Orders);
- (v) hold any proceeds of the DIP Term Loans in any account other than a Controlled Account, pending withdrawal and application thereof in accordance with this Agreement;
- (vi) other than in connection with an Asset Disposition permitted by <u>Section 4(g) of Schedule 7.1</u> or an Acceptable Plan, without the prior written consent of the Required Lenders, move to assume or reject any material lease, license, or other Material Contract of the Borrower pursuant to Section 365 of the Bankruptcy Code; or
- (vii) except as expressly provided or permitted hereunder (including, without limitation, to the extent pursuant to any First and Second Day Orders) and

the then approved Budget (including Permitted Variances thereto), make any payment or distribution to any Subsidiary, Affiliate or insider of the Borrower.

SECTION 8. DEFAULT AND REMEDIES:

- 8.1 <u>Events of Default</u>. Each of the following constitutes an Event of Default:
- (A) the Borrower fails to pay (i) when and as required to be paid herein, any amount of principal of any Loan, or (ii) within three (3) Business Days after the same becomes due, any interest on any Loan, or any repayment premium or fee due hereunder, or any other amount payable hereunder or under any other Loan Document;
- (B) any representation or warranty made or deemed made by or on behalf of the Parent, the Borrower or any Subsidiary in or in connection with any Loan Document or any amendment or modification of any Loan Document or waiver under such Loan Document, or in any report, certificate, financial statement or other document furnished pursuant to or in connection with any Loan Document or any amendment or modification thereof or waiver thereunder, proves to have been materially incorrect when made or deemed made;
- (C) the Borrower or any Subsidiary fails to observe or perform any covenant, condition or agreement contained in <u>Section 6</u> or in <u>Section 7</u>, including, but not limited to those requirements concerning the budget contained in <u>Section 6.14</u>;
- (D) the Borrower or any Subsidiary fails to observe or perform any covenant, condition or agreement contained in this Agreement (other than those specified in Section 8.1(A), 8.1(B), or 8.1(C)) or any other Loan Document, and such failure continues unremedied for a period of 30 days after the earlier to occur of (i) notice thereof from the Administrative Agent (at the written direction of any Lender) or any Lender to the Borrower and (ii) a Responsible Officer of the Borrower or such Subsidiary otherwise becoming aware of such default;
- (E) any event or condition occurs that results in any Debt having a principal amount greater than One Hundred Thousand Dollars (\$100,000) ("Material Debt") becoming due prior to its scheduled maturity or that enables or permits (with or without the giving of notice, the lapse of time or both) the holder or holders of any Material Debt or any trustee or agent on its or their behalf to cause any Material Debt to become due, or to require the Redemption thereof or any offer to Redeem to be made in respect thereof, prior to its scheduled maturity or require the Borrower or any Subsidiary to make an offer in respect thereof;
- (F) the appointment of or application for a receiver, trustee, custodian, examiner, sequestrator, conservator or similar official for the Borrower or any Subsidiary or for a substantial part of its or their assets, and, in any such case, such proceeding or petition continues undismissed for 30 days or an order or decree approving or ordering any of the foregoing is entered.

- (G) the Borrower or any Subsidiary who is not party to the Bankruptcy Case (i) voluntarily commences any proceeding or files any petition seeking liquidation, reorganization or other relief under any Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, aside from the contemplated Bankruptcy Case, (ii) consents to the institution of, or fails to contest in a timely and appropriate manner, any proceedings described in Section 8.1(F), (iii) applies for or consents to the appointment of a receiver, trustee, custodian, examiner, sequestrator, conservator, or similar official for the Borrower, or any Subsidiary or for a substantial part of its or their assets, (iv) files an answer admitting the material allegations of a petition filed against it or them in any such proceeding, (v) makes a general assignment for the benefit of creditors, or (vi) takes any action for the purpose of effecting any of the forgoing;
- (H) (i) one or more judgments for the payment of money in an aggregate amount in excess of \$2,000,000 (to the extent not covered by independent third party insurance provided by insurers of the highest claims paying rating or financial strength as to which the insurer does not dispute coverage and is not subject to an insolvency proceeding) or (ii) any one or more non-monetary judgments that have, or could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, are rendered against the Borrower, any Subsidiary or any combination thereof and the same remain undischarged for a period of 30 consecutive days during which execution is not effectively stayed, or any action is legally taken by a judgment creditor to attach or levy upon any assets of the Borrower or any Subsidiary to enforce any such judgment;
- (I) any Loan Document, for any reason, except to the extent permitted by the terms thereof, ceases to be in full force and effect and valid, binding and enforceable in accordance with their terms against the Borrower or a Guarantor party thereto or is repudiated by any of them, or cease to create a Lien of the priority required thereby on any of the collateral purported to be covered thereby, except to the extent permitted by the terms of this Agreement, or the Borrower or any Subsidiary or any of their Affiliates so state in writing;
- (J) there occurs any circumstance or circumstances that could reasonably be expected, either individually or in the aggregate, to have a Material Adverse Effect;
- (K) any subordination provision in any document or instrument governing Debt that is purported to be subordinated to the Obligations or any subordination provision in any subordination agreement that relates to any Debt that is to be subordinated to the Obligations, or any subordination provision in any guaranty by the Borrower or any of its Subsidiaries of any such Debt, shall cease to be in full force and effect, or any Person (including the holder of any such Debt) shall contest in any manner the validity, binding nature or enforceability of any such provision;
- (L) any court order enjoins, restrains, or prevents the Borrower or any of its Subsidiaries from conducting any material part of its business;

- (M) the Borrower shall file a motion in the Chapter 11 Case without the prior express written consent of the Required Lenders (or the Administrative Agent at the direction of the Required Lenders), to obtain additional financing from a party other than the Lenders under Section 364(d) of the Bankruptcy Code or to use Cash Collateral of a Lender under Section 363(c) of the Bankruptcy Code;
- (N) the Bankruptcy Court shall enter an order (i) approving payment of any prepetition claim other than (x) as provided for in the First and Second Day Orders, (y) as otherwise contemplated hereunder and by the approved Budget or (z) as otherwise consented to by the Required Lenders (or the Administrative Agent at the direction of the Required Lenders) in writing in their reasonable discretion, (ii) granting relief from the automatic stay applicable under Section 362 of the Bankruptcy Code to any holder of any security interest to permit foreclosure on any assets having a book value in excess of \$100,000 in the aggregate, or (iii) except, as expressly provided in the DIP Financing Orders, approving any settlement or other stipulation not approved by the Required Lenders (or the Administrative Agent at the direction of the Required Lenders) with any prepetition secured creditor of the Borrower or its Subsidiary providing for payments as adequate protection or otherwise to such secured creditor;
- (O) an order with respect to the Chapter 11 Cases shall be entered by the Bankruptcy Court (i) appointing a trustee under Section 1104, (ii) appointing an examiner with enlarged powers relating to the operation of the business (powers beyond those set forth in Section 1106(a)(3) and (4) of the Bankruptcy Code) under Section 1106(b) of the Bankruptcy Code, or (iii) converting the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code;
- (P) The entry of a final non-appealable order in the Chapter 11 Cases charging any of the DIP Collateral under section 506(c) of the Bankruptcy Code against the Lenders or Administrative Agent;
- (Q) an order shall be entered by the Bankruptcy Court dismissing the Chapter 11 Cases which does not contain a provision for termination of all DIP Term Loan Commitments, and indefeasible payment in full in cash of all Obligations upon entry thereof;
- (R) an order with respect to the Chapter 11 Cases shall be entered by the Bankruptcy Court without the express prior written consent of the Required Lenders (and with respect to any provisions that affect the rights or duties of the Administrative Agent, the Administrative Agent, not to be unreasonably withheld, conditioned or delayed), (i) to revoke, vacate, reverse, stay, modify, supplement or amend the DIP Financing Orders (other than the modifications to the Interim DIP Financing Order by the Final DIP Financing Order) or (ii) to permit, unless otherwise contemplated by the DIP Financing Orders, any administrative expense or any claim (now existing or hereafter arising, of any kind or nature whatsoever) to have administrative priority equal or superior to the administrative priority of the Obligations (other than the Carve Out);

- (S) No Contestation; Surcharge.
 - (i) an application for any of the orders described in Section 8.1(O) above shall be made by the Borrower or any such application shall be made by a Person other than the Borrower and such application is not contested by the Borrower in good faith or the relief requested is not withdrawn, dismissed or denied within ten (10) days after the filing or (ii) the Borrower shall file a motion seeking (or the Borrower shall not object to any other Person seeking), or any other Person shall obtain an order authorizing, any of the Collateral or collateral securing the Prepetition Debt to be surcharged pursuant to Section 506(c) of the Bankruptcy Code;
- (T) the entry of an order by the Bankruptcy Court terminating or modifying the exclusive right of the Borrower to file a Plan of Reorganization pursuant to Section 1121 of the Bankruptcy Code, without the prior written consent of the Required Lenders;
- (U) the Borrower or any of its Subsidiaries shall seek to, or shall support (in any such case by way of any motion or other pleading filed with the Bankruptcy Court or any other writing to another party-in-interest executed by or on behalf of the Borrower or any of its Subsidiaries) any other Person's motion to, avoid or disallow in whole or in part the Lenders' claims or Liens in respect of the Obligations or contest any provision of any Loan Document or any provision of any Loan Document shall cease to be effective provided, however, that it shall not constitute an Event of Default if the Borrower or any of its Subsidiaries provide information with respect to the Loan Documents or documents relating to the Lenders' Prepetition Debt to a party in interest, or are compelled to provide information by an order of the Bankruptcy Court so long as the Lenders have been served with the request for such an order or, if no such service has been made prior to the time the Lenders are required to provide information, so long as the Borrower or any of its Subsidiaries provide prior written notice to the Lenders of any intention or requirement to do so;
- (V) the Borrower, Guarantor or any of their Subsidiaries or Affiliates shall fail to comply with the Interim DIP Financing Order or the Final DIP Financing Order;
- (W) any order by the Bankruptcy Court is entered granting any administrative or superpriority claim that is *pari passu* with or senior to those of the Secured Parties or any Lien that is senior to the Liens securing the Obligations, other than in accordance with the DIP Financing Order (except with respect to the Carve-Out);
- (X) without the Required Lenders' consent, the filing by the Borrower of any motion or other request with the Bankruptcy Court seeking authority to consummate a sale of material assets or equity or all or substantially all of the Borrower's or any of its Subsidiaries' assets or equity, or the consummation of any such sale;

- (Y) the Borrower or any of its Subsidiaries, or any person claiming by or through the Borrower or any of its Subsidiaries, obtain court authorization to commence, or commence, join in, assist or otherwise participate as an adverse party in any suit or other proceeding against any of the Administrative Agent, the Lenders, or the Prepetition Parties, in each case relating to this Agreement or Prepetition Debt, as applicable;
- (Z) the filing by the Borrower of a Plan of Reorganization other than an Acceptable Plan; or
- (AA) a Change in Control occurs without the Required Lenders consent.
- 8.2 <u>Remedies</u>. Notwithstanding anything in Section 362 of the Bankruptcy Code, but subject to the applicable DIP Financing Order, if any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Lenders, take any or all of the following actions:
 - (A) declare the DIP Term Loan Commitment of each Lender to be terminated, whereupon such DIP Term Loan Commitments shall be terminated;
 - (B) declare the unpaid principal amount of all outstanding DIP Term Loans and, if after the Interim DIP Financing Order is entered, the Roll-Up Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower; and
 - (C) exercise on behalf of any Lender all of its other rights and remedies under this Agreement, the other Loan Documents and Applicable Law, in order to satisfy all of the Obligations.
- Rights and Remedies Cumulative; Non-Waiver. The enumeration of the rights and remedies of the Administrative Agent and the Lenders set forth in this Agreement is not intended to be exhaustive and the exercise by the Administrative Agent and the Lenders of any right or remedy does not preclude the exercise of any other rights or remedies, all of which are cumulative, and are in addition to any other right or remedy given hereunder or under the other Loan Documents or that may now or hereafter exist at law or in equity or by suit or otherwise. No delay or failure to take action on the part of the Administrative Agent or any Lender in exercising any right, power or privilege operates as a waiver thereof, nor does any single or partial exercise of any such right, power or privilege preclude any other or further exercise thereof or the exercise of any other right, power or privilege or be construed to be a waiver of any Event of Default. No course of dealing between the Loan Parties, the Administrative Agent and the Lenders or their respective agents or employees is effective to change, modify or discharge any provision of this Agreement or any of the other Loan Documents or to constitute a waiver of any Event of Default.
- 8.4 <u>Crediting of Payments and Proceeds</u>. In the event that the Obligations have been accelerated pursuant to <u>Section 8.2</u> or the Administrative Agent or any Lender has exercised any remedy set forth in this Agreement or any other Loan Document, all payments received on account

Casse 225-1009799-JK\$ Door 502-21 FFiled 0066 0044 225 Pragge 1438 off 11293

of the Obligations and all net proceeds from the enforcement of the Obligations shall be applied by the Administrative Agent as follows:

- (A) <u>first</u>, to fund the Carve-Out (to the extent not fully funded by Borrower at such time);
- (B) <u>second</u>, to payment of that portion of the Obligations constituting fees, losses, indemnities, expenses and other amounts, including attorney fees, payable to the Administrative Agent under the Loan Documents in its capacity as such;
- (C) third, to payment of that portion of the Obligations constituting fees, indemnities and other amounts (other than principal and interest) payable to the Lenders under the Loan Documents, including attorney fees, ratably among the Lenders in proportion to the respective amounts described in this clause second payable to them;
- (D) <u>fourth</u>, to payment of that portion of the Obligations constituting accrued and unpaid interest on the DIP Facility ratably among the Lenders in proportion to the respective amounts described in this clause <u>third</u> payable to them;
- (E) <u>fifth</u>, to payment of that portion of the Obligations constituting unpaid principal of the DIP Facility, ratably among the Lenders in proportion to the respective amounts described in this clause <u>fourth</u> payable to them; and
- (F) <u>last</u>, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to the Borrower or as otherwise required by Applicable Law.

8.5 <u>Credit Bidding.</u>

(A) The Administrative Agent (at the written direction of the Required Lenders), on behalf of itself and the Lenders, has the right to credit bid and purchase for the benefit of the Administrative Agent and the Lenders all or any portion of Collateral at any sale thereof conducted by the Administrative Agent under the provisions of the Uniform Commercial Code, including pursuant to Sections 9-610 or 9-620 of the Uniform Commercial Code, at any sale thereof conducted under the provisions of the United States Bankruptcy Code, including Section 363 thereof, or a sale under a plan of reorganization, or at any other sale or foreclosure conducted by the Administrative Agent (whether by judicial action or otherwise) in accordance with Applicable Law. Such credit bid or purchase may be completed through one or more acquisition vehicles formed by the Administrative Agent to make such credit bid or purchase and, in connection therewith, the Administrative Agent is authorized, on behalf of itself and the Lenders, to adopt documents providing for the governance of the acquisition vehicle or vehicles, and assign the applicable Obligations to any such acquisition vehicle in exchange for Equity Interests or debt issued by the applicable acquisition vehicle (which shall be deemed to be held for the ratable account of the applicable Lenders on the basis of the Obligations so assigned by each Lender).

- (B) Except as otherwise provided in any Loan Document or with the written consent of the Required Lenders, no Lender shall take any enforcement action, accelerate obligations under any of the Loan Documents, or exercise any right that it might otherwise have under Applicable Law to credit bid at foreclosure sales, UCC sales or other similar dispositions of Collateral.
- 8.6 <u>Injunctive Relief.</u> The Borrower recognizes that, in the event the Borrower fails to perform, observe or discharge any of its obligations or liabilities under this Agreement, any remedy of law may prove to be inadequate relief to the Lenders. Therefore, the Borrower agrees that the Lenders, at the Lenders' option, are entitled to temporary and permanent injunctive relief in any such case without the necessity of proving actual damages.

SECTION 9. AGENCY:

- 9.1 Appointment and Authority. Each of the Lenders irrevocably appoints Volunteer Enterprises, LLC to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents to which the Administrative Agent is a party and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof. The provisions of this Section 9 are solely for the benefit of the Administrative Agent and the Lenders and neither the Borrower nor any other Loan Party have rights as a third-party beneficiary of any of such provisions. It is understood and agreed that the use of the term "agent" herein or in any other Loan Documents (or any other similar term) with reference to the Administrative Agent is not intended to connote any fiduciary or other implied (or express) obligations arising under agency doctrine of any Applicable Law. Instead such term is used as a matter of market custom, and is intended to create or reflect only an administrative relationship between contracting parties.
- 9.2 Rights as a Lender. The Person serving as the Administrative Agent hereunder has the same rights and powers in its capacity as a Lender as any other Lender, if applicable, and may exercise the same as though it were not the Administrative Agent, and the term "Lender" or "Lenders" includes, unless otherwise expressly indicated or unless the context otherwise requires, the Person serving as the Administrative Agent hereunder in its individual capacity, if such Person is a Lender. Such Person and its Affiliates may accept deposits from, lend money to, own securities of, act as the financial advisor or in any other advisory capacity for, and generally engage in any kind of business with, the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.
- 9.3 <u>Exculpatory Provisions</u>. (A) The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents to which it is a party, and its duties hereunder are administrative in nature. Without limiting the generality of the foregoing, the Administrative Agent:
 - (i) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;

- (ii) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as is expressly provided for herein or in the other Loan Documents); provided that the Administrative Agent is not required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or Applicable Law, including for the avoidance of doubt any action that may be in violation of the automatic stay under any Debtor Relief Law or that may affect a forfeiture, modification or termination of property of a Defaulting Lender in violation of any Debtor Relief Law; and
- (iii) shall not, except as expressly set forth herein and in the other Loan Documents to which it is a party, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.
- (B) The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request or direction of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent believes in good faith shall be necessary, under the circumstances) or (ii) in the absence of its own gross negligence or willful misconduct as determined by a court of competent jurisdiction by final and nonappealable judgment. The Administrative Agent is not deemed to have knowledge of any Default or Event of Default unless and until notice describing such Default or Event of Default is given to the Administrative Agent in writing by the Borrower or a Lender.
- (C) The Administrative Agent is not responsible for and has no duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith or therewith and the Administrative Agent is not required to calculate, recalculate, verify, review or certify any calculation or any other mathematical or numerical information, (iii) the performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default or Event of Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document, or the perfection or priority of any Lien created or purported to be created by the Security Documents or (v) the satisfaction of any condition set forth in Section 4 or elsewhere herein other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.
- (D) The Lenders direct the Administrative Agent to execute and deliver the Loan Documents to which it is a party on the Closing Date. Whether or not so expressly stated therein, in entering into, or taking (or forbearing from) any action under or pursuant to, the Loan Documents, the Administrative Agent has all of the rights, immunities, indemnities and other

protections granted to it under this Agreement (in addition to those that may be granted to it under the terms of such other agreement or agreements).

- (E) The Person serving as the Administrative Agent may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Loan Parties or any Affiliate thereof as if such Person was not the Administrative Agent and without any duty to account therefor to the Lenders.
- (F) The powers conferred on the Administrative Agent under the Security Documents are solely to protect the Administrative Agent's interest in the Collateral, do not impose any duty upon it to exercise any such powers and are subject to the provisions of this Agreement. The Administrative Agent is accountable only for amounts that it actually receives as a result of the exercise of such powers and the Administrative Agent and any of its officers, directors, employees or agents are not responsible for any act or failure to act, except for gross negligence or willful misconduct (as determined by a final non-appealable judgment of a court of competent jurisdiction). Unless otherwise directed by the Required Lenders, the Administrative Agent has no responsibility for taking any necessary steps to protect, preserve or exercise rights against any Person with respect to any of the Collateral and is relieved of all responsibility for the Collateral upon surrendering it to the Borrower or any other designated Person. In no event does the Administrative Agent have any duty, responsibility, obligation or liability with respect to monitoring the Collateral or the condition thereof, and the Administrative Agent is not required to take any action to protect against any diminution in value of the Collateral.
- (G) Notwithstanding any provision of this Agreement or the other Loan Documents to the contrary, the Administrative Agent is not required to (i) make or give any determination (including whether a matter is satisfactory to the Administrative Agent or whether to deem a matter necessary, desirable, proper or advisable), agreement, consent, approval, request, notice, consultation, designation, appointment, election, judgment or direction, (ii) file, record or prepare any Uniform Commercial Code financing or continuation statements or similar documents or instruments in any jurisdiction for purposes of creating, perfecting or maintaining any Lien or security interest or otherwise perfect any security interest or maintain any perfection or priority, (iii) make any inspection or (iv) release or sell Collateral or otherwise exercise any rights or remedies of a secured party (including voting rights), in each case, without the written direction of the Required Lenders.
- (H) Notwithstanding any provision of this Agreement or the other Loan Documents to the contrary, before taking or omitting any action to be taken or omitted by the Administrative Agent under the terms of this Agreement and the other Loan Documents, the Administrative Agent may seek the written direction of the Required Lenders (which written direction may be in the form of an e-mail), and the Administrative Agent is entitled to rely (and is fully protected in so relying) upon such direction. The Administrative Agent is not liable with respect to any action taken or omitted to be taken by it in accordance with such direction. If the Administrative Agent requests such direction with respect to any action, the Administrative Agent is entitled to refrain from such action unless and until the Administrative Agent has received such direction, and the Administrative Agent does not incur liability to any Person by reason of so refraining. If the Administrative Agent so requests, it must first be indemnified to its satisfaction by the directing Lenders against any and all liability and expense which may be incurred by the Administrative

Agent by reason of taking or continuing to take, or omitting, any action directed by any of the Lenders. Any provision of this Agreement or the other Loan Documents authorizing the Administrative Agent to take any action does not obligate the Administrative Agent to take such action.

- (I) The Administrative Agent is never required to use, risk or advance its own funds or otherwise incur financial liability in the performance of any of its duties or the exercise of any of its rights and powers hereunder or under the other Loan Documents (including no obligation to grant any credit extension or to make any advance hereunder).
- (J) Delivery of reports, documents and other information to the Administrative Agent is for informational purposes only and the Administrative Agent's receipt of the foregoing does not constitute constructive knowledge of any event or circumstance or any information contained therein or determinable from information contained therein.
- (K) The parties hereto acknowledge that in accordance with the Customer Identification Program (CIP) requirements under the USA PATRIOT Act and its implementing regulations, the Administrative Agent, in order to help fight the funding of terrorism and money laundering, is required to obtain, verify, and record information that identifies each person or legal entity that establishes a relationship or opens an account with the Administrative Agent. The parties hereto shall provide the Administrative Agent with such information as it may request including each party's name, physical address, tax identification number and other information that will help the Administrative Agent identify and verify each party's identity such as organizational documents, certificate of good standing, license to do business, or other pertinent identifying information.
- (L) In no event is the Administrative Agent liable on any theory of liability for any special, indirect, consequential or punitive damages (including lost profits) even if the Administrative Agent has been advised of the possibility of such damages and regardless of the form of action. The Administrative Agent is not responsible for delays or failures in performance resulting from acts beyond its control. Such acts include, but are not limited to, acts of God, strikes, lockouts, riots, acts of war, epidemics, governmental regulations superimposed after the fact, fire, communication line failures, computer viruses, power failures, earthquakes, terrorist attacks or other disasters.
- (M) The Administrative Agent shall have no responsibility for interest or income on any funds held by it under the Loan Documents and any funds so held shall be held un-invested pending distribution thereof.
- (N) In no event shall the Administrative Agent be liable on account of any loss, including without limitation, any loss of principal or interest, or for any breakage fees or penalties in connection with the purchase or liquidation of any investment made in accordance with a written instruction of the Borrower or the Required Lenders.
- 9.4 <u>Reliance by Administrative Agent</u>. The Administrative Agent is entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed,

sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a DIP Term Loan that by its terms must be fulfilled to the satisfaction of a Lender, the Administrative Agent may presume that such condition is satisfactory to such Lender unless the Administrative Agent has received written notice to the contrary from such Lender prior to the making of such DIP Term Loans. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and is not liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

- 9.5 <u>Delegation of Duties</u>. The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub-agents appointed by the Administrative Agent, which sub-agents may be Affiliates of the Administrative Agent. The Administrative Agent and any such sub-agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this <u>Section 9</u> apply to any such sub-agent and to the Related Parties of the Administrative Agent and any such sub-agent. The Administrative Agent is not responsible for the negligence or misconduct of any sub-agents except to the extent that a court of competent jurisdiction determines in a final and nonappealable judgment that the Administrative Agent acted with gross negligence or willful misconduct in the selection of such sub-agents.
- Resignation of Administrative Agent. (A) The Administrative Agent may at any time give notice of its resignation to the Lenders and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders have the right, in consultation with the Borrower, to appoint a successor, which must be a bank or trust company, but in no event may be a Defaulting Lender. If no such successor shall have been so appointed by the Required Lenders and has accepted such appointment within 30 days after the retiring Administrative Agent gives notice of its resignation (or such earlier day as is agreed by the Required Lenders) (the "Resignation Effective Date"), then in any event such resignation becomes effective in accordance with such notice on the Resignation Effective Date and (i) the retiring Administrative Agent is discharged from its duties and obligations hereunder and under the other Loan Documents, except that until a successor Administrative Agent is appointed, any Collateral held by the Administrative Agent on behalf of the Secured Parties under any of the Loan Documents shall continue to be held by the resigning Administrative Agent as bailee until such time as a successor Administrative Agent is appointed, and (ii) except for any fees, expenses or indemnity payments owed to the retiring Administrative Agent, all payments, communications and determinations provided to be made by, to or through such Administrative Agent shall instead be made by or to each Lender directly, until such time as the Required Lenders appoint a successor Administrative Agent as provided in this Section 9.6(A). The Administrative Agent has the right, at the cost and expense of the Borrower, to petition a court of competent jurisdiction regarding the delivery of any Collateral it holds as bailee.
- (B) Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor succeeds to and becomes vested with all of the rights, powers, privileges and duties of the retiring Administrative Agent (other than any rights to fees, expenses and

indemnity payments owed to the retiring Administrative Agent) and the retiring Administrative Agent is discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not discharged earlier as provided in Section 9.6(A)). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Section 9 and Section 10.2 continue in effect for the benefit of such retiring Administrative Agent, its sub-agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

- (C) Any Person into which the Administrative Agent may be merged or converted or with which it may be consolidated, or any Person resulting from any merger, conversion or consolidation to which the Administrative Agent is a party, or any Person succeeding to the corporate trust services business of the Administrative Agent is the successor of the Administrative Agent hereunder without the execution or filing of any paper with any Party or any further act on the part of any of the Parties.
- 9.7 Non-Reliance on Administrative Agent and Other Lenders. Each Lender acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it from time to time deems appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.
- 9.8 <u>Collateral and Guaranty Matters</u>. (A) The Lenders irrevocably authorize and direct the Administrative Agent, upon the written direction of the Required Lenders and at the expense of the Borrower:
 - (i) to release any Lien on any property granted to or held by the Administrative Agent under any Loan Document (a) upon termination of all DIP Term Loan Commitments and payment in full of all Obligations (other than contingent indemnification obligations), (b) that is sold or otherwise disposed of or to be sold or otherwise disposed of as part of or in connection with any sale or other disposition permitted under the Loan Documents or (c) subject to Section 10.3, if approved, authorized or ratified in writing by the Required Lenders;
 - (ii) to subordinate any Lien on any property granted to or held by the Administrative Agent under any Loan Document to the holder of any Lien on such property that is otherwise permitted by this Agreement; and
 - (iii) to release any Subsidiary Guarantor from its obligations under the Guaranty Agreement if such Person ceases to be a Subsidiary as a result of a transaction permitted under the Loan Documents.

Upon request by the Administrative Agent at any time, the Required Lenders will confirm in writing the Administrative Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Subsidiary Guarantor from its obligations under the Guaranty Agreement pursuant to this <u>Section 9.8</u>.

(B) The Administrative Agent is not responsible for, and has no duty to ascertain or inquire into any representation or warranty regarding, the existence, value or collectability of the Collateral, the existence, priority or perfection of the Administrative Agent's Lien thereon, or any certificate prepared by any Loan Party in connection therewith, nor shall the Administrative Agent be responsible or liable to the Lenders for any failure to monitor or maintain any portion of the Collateral.

9.9 Erroneous Payment.

(A) If the Administrative Agent (x) notifies a Lender or any other Secured Party, or any Person who has received funds on behalf of a Lender or other Secured Party (any such Lender, other Secured Party or other recipient (and each of their respective successors and assigns), a "Payment Recipient") that the Administrative Agent has determined in its reasonable discretion (whether or not after receipt of any notice under immediately succeeding clause (B)) that any funds (as set forth in such notice from the Administrative Agent) received by such Payment Recipient from the Administrative Agent or any of its Affiliates were erroneously or mistakenly transmitted to, or otherwise erroneously or mistakenly received by, such Payment Recipient (whether or not known to such Lender, Secured Party or other Payment Recipient on its behalf) (any such funds, whether transmitted or received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise, individually and collectively, an "Erroneous Payment") and (y) demands in writing the return of such Erroneous Payment (or a portion thereof), such Erroneous Payment shall at all times remain the property of the Administrative Agent pending its return or repayment as contemplated below in this Section 9.9 and held in trust for the benefit of the Administrative Agent, and such Lender, or such other Secured Party shall (or, with respect to any Payment Recipient who received such funds on its behalf, shall cause such Payment Recipient to) promptly, but in no event later than two Business Days thereafter (or such later date as the Administrative Agent may, in its sole discretion, specify in writing), return to the Administrative Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made, in same day funds (in the currency so received), together with interest thereon (except to the extent waived in writing by the Administrative Agent) in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Payment Recipient to the date such amount is repaid to the Administrative Agent in same day funds at the greater of the Federal Funds Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation from time to time in effect provided that any such interest shall only be payable by a Lender to the extent of, and in an amount equal to, interest actually accrued on the Erroneous Payment in favor of that Lender during such period. A

- notice of the Administrative Agent to any Payment Recipient under this clause (a) shall be conclusive, absent manifest error.
- (B) Without limiting immediately preceding Section 9.9(A), each Lender, each other Secured Party or any Person who has received funds on behalf of a Lender or other Secured Party (and each of their respective successors and assigns), agrees that if it receives a payment, prepayment or repayment (whether received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise) from the Administrative Agent (or any of its Affiliates) (x) that is in a different amount than, or on a different date from, that specified in this Agreement or in a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment or repayment sent by the Administrative Agent (or any of its Affiliates), or (z) that such Lender or such other Secured Party, or other such recipient, otherwise becomes aware was transmitted, or received, in error or by mistake (in whole or in part), then in each such case:
 - (i) it acknowledges and agrees that (a) in the case of immediately preceding clauses (x) or (y), an error and mistake shall be presumed to have been made (absent written confirmation from the Administrative Agent to the contrary) or (b) an error and mistake has been made (in the case of immediately preceding clause (z)), in each case, with respect to such payment, prepayment or repayment; and
 - (ii) such Lender or such other Secured Party shall use commercially reasonable efforts to (and shall use commercially reasonable efforts to cause any other recipient that receives funds on its respective behalf to) promptly (and, in all events, within one Business Day of its knowledge of the occurrence of any of the circumstances described in immediately preceding clauses (x), (y) and (z)) notify the Administrative Agent of its receipt of such payment, prepayment or repayment, the details thereof (in reasonable detail) and that it is so notifying the Administrative Agent pursuant to this Section 9.9(B).

For the avoidance of doubt, the failure to deliver a notice to the Administrative Agent pursuant to this Section 9.9(B) shall not have any effect on a Payment Recipient's obligations pursuant to Section 9.9(A) or on whether or not an Erroneous Payment has been made.

(C) Each Lender or other Secured Party hereby authorizes the Administrative Agent to set off, net and apply any and all amounts at any time owing to such Lender or such other Secured Party under any Loan Document, or otherwise payable or distributable by the Administrative Agent to such Lender or such other Secured Party under any Loan Document with respect to any payment of principal, interest, fees or other amounts, against any amount that the Administrative Agent has demanded to be returned under the preceding Section 9.9(A).

- (D) (i) In the event that an Erroneous Payment (or portion thereof) is not recovered by the Administrative Agent for any reason, after demand therefor in accordance with preceding clause (A), from any Lender that has received such Erroneous Payment (or portion thereof) (and/or from any Payment Recipient who received such Erroneous Payment (or portion thereof) on its respective behalf) (such unrecovered amount, an "Erroneous Payment Return Deficiency"), upon not less than five (5) Business Days' notice from the Administrative Agent to such Lender at any time, then effective thereupon (with the consideration therefor being acknowledged by the parties hereto), (a) such Lender shall be deemed to have assigned its Loans (but not its Commitments) with respect to which such Erroneous Payment was made (the "Erroneous Payment Impacted Class") in an amount equal to the Erroneous Payment Return Deficiency (or such lesser amount as the Administrative Agent may specify) (such assignment of the Loans (but not Commitments) of the Erroneous Payment Impacted Class, the "Erroneous Payment Deficiency Assignment") (on a cashless basis and such amount calculated at par plus any accrued and unpaid interest (with the assignment fee to be waived by the Administrative Agent in such instance)), and is hereby (together with the Borrower) deemed to execute and deliver an Assignment and Acceptance (or, to the extent applicable, an agreement incorporating an Assignment and Acceptance by reference pursuant to an approved electronic platform as to which the Administrative Agent and such parties are participants) with respect to such Erroneous Payment Deficiency Assignment, and such Lender shall deliver any notes evidencing such Loans to the Borrower or the Administrative Agent (but the failure of such Person to deliver any such notes shall not affect the effectiveness of the foregoing assignment), (b) the Administrative Agent as the assignee Lender shall be deemed to have acquired the Erroneous Payment Deficiency Assignment, (c) upon such deemed acquisition, the Administrative Agent as the assignee Lender shall become a Lender, as applicable, hereunder with respect to such Erroneous Payment Deficiency Assignment and the assigning Lender shall cease to be a Lender, as applicable, hereunder with respect to such Erroneous Payment Deficiency Assignment, excluding, for the avoidance of doubt, its obligations under the indemnification provisions of this Agreement and its applicable Commitments which shall survive as to such assigning Lender, (d) the Administrative Agent and the Borrower shall each be deemed to have waived any consents required under this Agreement to any such Erroneous Payment Deficiency Assignment, and (e) the Administrative Agent will reflect in the Register its ownership interest in the Loans subject to the Erroneous Payment Deficiency Assignment. For the avoidance of doubt, no Erroneous Payment Deficiency Assignment will reduce the Commitments of any Lender and such Commitments shall remain available in accordance with the terms of this Agreement.
 - (ii) Subject to Section 10.04 (but excluding, in all events, any assignment consent or approval requirements (whether from the Borrower or otherwise)), the Administrative Agent may, in its discretion, sell any Loans acquired pursuant to an Erroneous Payment Deficiency Assignment and upon receipt of the proceeds of such sale, the Erroneous Payment Return Deficiency owing by the applicable Lender shall be reduced by the net proceeds of the sale of such Loan (or portion

thereof), and the Administrative Agent shall retain all other rights, remedies and claims against such Lender (and/or against any recipient that receives funds on its respective behalf). In addition, an Erroneous Payment Return Deficiency owing by the applicable Lender (x) shall be reduced by the proceeds of prepayments or repayments of principal and interest, or other distribution in respect of principal and interest, received by the Administrative Agent on or with respect to any such Loans acquired from such Lender pursuant to an Erroneous Payment Deficiency Assignment (to the extent that any such Loans are then owned by the Administrative Agent) and (y) may, in the sole discretion of the Administrative Agent, be reduced by any amount specified by the Administrative Agent in writing to the applicable Lender from time to time.

- The parties hereto agree that irrespective of whether the Administrative Agent may (E) be equitably subrogated, in the event that an Erroneous Payment (or portion thereof) is not recovered from any Payment Recipient that has received such Erroneous Payment (or portion thereof) for any reason, the Administrative Agent shall be subrogated to all the rights and interests of such Payment Recipient (and, in the case of any Payment Recipient who has received funds on behalf of a Lender or other Secured Party, to the rights and interests of such Lender or such other Secured Party, as the case may be) under the Loan Documents with respect to such amount (the "Erroneous Payment Subrogation Rights") (provided that the Obligations under the Loan Documents in respect of the Erroneous Payment Subrogation Rights shall not be duplicative of such Obligations in respect of Loans that have been assigned to the Administrative Agent under an Erroneous Payment Deficiency Assignment); provided that this Section 9.9(E) shall not be interpreted to increase (or accelerate the due date for), or have the effect of increasing (or accelerating the due date for), the Obligations of the Borrower relative to the amount (and/or timing for payment) of the Obligations that would have been payable had such Erroneous Payment not been made by the Administrative Agent; provided, further, that for the avoidance of doubt, the immediately preceding clause shall not apply to the extent any such Erroneous Payment is, and solely with respect to the amount of such Erroneous Payment that is, comprised of funds received by the Administrative Agent from the Borrower or from the proceeds of Collateral.
- (F) To the extent permitted by applicable law, no Payment Recipient shall assert any right or claim to an Erroneous Payment, and hereby waives, and is deemed to waive, any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by the Administrative Agent for the return of any Erroneous Payment received, including, without limitation, any defense based on "discharge for value" or any similar doctrine.
- (G) Each party's obligations, agreements and waivers under this <u>Section 9.9</u> shall survive the resignation or replacement of the Administrative Agent, any transfer of rights or obligations by, or the replacement of, a Lender, the termination of the Commitments and/or the repayment, satisfaction or discharge of all Obligations (or any portion thereof) under any Loan Document.

SECTION 10. MISCELLANEOUS:

- 10.1 <u>Governing Law.</u> This Agreement and the other Loan Documents and any claims, controversy, dispute or cause of action (whether in contract or tort or otherwise) based upon, arising out of or relating to this Agreement or any other Loan Document (except, as to any other Loan Document, as expressly set forth therein) and the transactions contemplated hereby and thereby are governed by, and construed in accordance with, the law of the State of Delaware.
 - 10.2 Expenses; Indemnity; Damage Waiver.
 - (A) Costs and Expenses. The Borrower shall pay (i) all reasonable out of pocket expenses incurred by the Administrative Agent and the Lenders (including the reasonable fees, charges and disbursements of counsel for the Administrative Agent and for the Lenders), and shall pay all fees and time charges and disbursements for attorneys who may be employees of the Administrative Agent, in connection with the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents, or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby are consummated) and (ii) all out of pocket expenses incurred by the Administrative Agent or any Lender (including the fees, charges and disbursements of any counsel for the Administrative Agent or for any Lender), and shall pay all fees and time charges for attorneys who may be employees of the Administrative Agent or any Lender, in connection with the enforcement or protection of its rights (a) in connection with this Agreement and the other Loan Documents, including its rights under this Section 10.2, or (b) in connection with the DIP Term Loans and Roll-Up Loans made hereunder, including all such out of pocket expenses incurred during any workout, restructuring or negotiations in respect of such DIP Term Loans or Roll-Up Loans.
 - (B) Indemnification by the Borrower. The Borrower shall indemnify the Administrative Agent (and any sub-agent thereof), each Lender and each Related Party of any of the foregoing Persons (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all fees, losses, claims, damages, liabilities and related expenses (including the reasonable fees, charges and disbursements of any counsel for any Indemnitee and including out-of-pocket attorneys' fees and expenses and court costs incurred by any Indemnitee in enforcing the indemnity), and shall indemnify and hold harmless each Indemnitee from all reasonable fees and time charges and disbursements for attorneys who may be employees of any Indemnitee, incurred by any Indemnitee or asserted against any Indemnitee by any Person (including the Borrower or any other Loan Party) arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance by the Parties of their respective obligations hereunder or thereunder or the consummation of the transactions contemplated hereby or thereby, (ii) any DIP Term Loan or the use or proposed use of the proceeds therefrom, (iii) any actual or alleged presence or Release of Hazardous Materials on or from any property owned or operated by

the Borrower or any of its Subsidiaries, or any Environmental Liability related in any way to the Borrower or any of its Subsidiaries, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or a Lender or by the Borrower or any other Loan Party, and regardless of whether any Indemnitee is a party thereto; <u>provided</u> that such indemnity is not, as to any Indemnitee, available to the extent that such losses, claims, damages, liabilities or related expenses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee. This <u>Section 10.2(B)</u> does not apply with respect to Taxes other than any Taxes that represent losses, claims, damages, or similar costs arising from any non-Tax claim.

- (C) Reimbursement by Lenders. To the extent that the Borrower for any reason fails to indefeasibly pay any amount required under Sections 3.7(D), 10.2(A) or 10.2(B) to be paid by it to the Administrative Agent (or any sub-agent thereof) or any Related Party of the Administrative Agent, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent) or such Related Party, as the case may be, such Lender's pro rata share (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought based on each Lender's Commitment Percentage at such time) of such unpaid amount (including any such unpaid amount in respect of a claim asserted by such Lender); provided the unreimbursed expense or indemnified loss, fee, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent) or against any Related Party acting for the Administrative Agent (or any such sub-agent) in connection with such capacity.
- Waiver of Consequential Damages. To the fullest extent permitted by Applicable (D) Law, the Borrower shall not assert, and hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any DIP Term Loan or the use of the proceeds thereof. No Indemnitee referred to in Section 10.2(B) will be liable for any damages arising from the use by unintended recipients of any information or other materials distributed by it through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby other than for direct and actual damages resulting from the gross negligence or willful misconduct of such Indemnitee as determined by a final and nonappealable judgment of a court of competent jurisdiction.
- (E) <u>Payments</u>. All amounts due under this <u>Section 10.2</u> are payable promptly after demand therefor.

- (F) <u>Survival</u>. Each Party's obligations under this <u>Section 10.2</u> survive the resignation or replacement of the Administrative Agent hereunder and the termination of the Loan Documents and payment of the obligations hereunder.
- 10.3 <u>Amendments, Waivers and Consents</u>. Except as set forth below or as specifically provided in any Loan Document, any term, covenant, agreement or condition of this Agreement or any of the other Loan Documents may be amended or waived by the Lenders, and any consent given by the Lenders, if, but only if, such amendment, waiver or consent is in writing signed by the Required Lenders (or by the Administrative Agent at the written direction of the Required Lenders) and delivered to the Administrative Agent and, in the case of an amendment, signed by the Borrower; <u>provided</u> no amendment, waiver or consent may:
 - (A) increase the DIP Term Loan Commitment of any Lender (or reinstate any DIP Term Loan Commitment terminated pursuant to <u>Section 8.2</u>) or the amount of DIP Term Loans of any Lender, in any case, without the written consent of such Lender;
 - (B) waive, extend or postpone any date fixed by this Agreement or any other Loan Document for any payment (it being understood that a waiver of a mandatory prepayment under Section 2.5 only requires the consent of the Required Lenders) of principal, interest, fees or other amounts due to the Lenders (or any of them) or any mandatory reduction of the DIP Term Loan Commitment hereunder or under any other Loan Document without the written consent of each Lender directly and adversely affected thereby;
 - (C) reduce the principal of, or the rate of interest specified herein on, any DIP Term Loan or (subject to <u>Section 10.3(I)</u>) any fees or other amounts payable hereunder or under any other Loan Document without the written consent of each Lender directly and adversely affected thereby; <u>provided</u> only the consent of the Required Lenders is necessary to waive any obligation of the Borrower to pay interest at the rate set forth in <u>Section 3.2(C)</u> during the continuance of an Event of Default;
 - (D) change <u>Section 3.5</u> or <u>Section 8.5</u> in a manner that would alter the pro rata sharing of payments or order of application required thereby without the written consent of each Lender directly and adversely affected thereby;
 - (E) change any provision of this <u>Section 10.3</u> or reduce the percentages specified in the definition of "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder, without the written consent of each Lender directly affected thereby;
 - (F) consent to the assignment or transfer by any Loan Party of such Loan Party's rights and obligations under any Loan Document to which it is a party (except as permitted pursuant to <u>Section 7.12</u>), in each case, without the written consent of each Lender;
 - (G) release the Guarantor from the Guaranty Agreement (other than as authorized in Section 9.8), without the written consent of each Lender;

- (H) release all or substantially all of the Collateral or release any Security Document (other than as authorized in <u>Section 9.8</u> or as otherwise specifically permitted or contemplated in this Agreement or the applicable Security Document) without the written consent of each Lender; or
- (I) affect the rights or duties of the Administrative Agent under this Agreement or any other Loan Document without the written consent of the Administrative Agent in addition to the Lenders required hereunder;

Notwithstanding anything to the contrary herein, no Defaulting Lender has any right to approve or disapprove any amendment, waiver or consent hereunder, except that the DIP Term Loan Commitment of such Lender may not be increased or extended without the consent of such Lender.

10.4 Successors and Assigns.

- (A) Successors and Assigns Generally. The provisions of this Agreement are binding upon and inure to the benefit of the Parties and their respective successors and assigns permitted hereby, except that neither the Borrower nor any other Loan Party may assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of each Lender, and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of Section 10.4(B), (ii) by way of participation in accordance with the provisions of Section 10.4(D) or (iii) by way of pledge or assignment of a security interest subject to the restrictions of Section 10.4(E) (and any other attempted assignment or transfer by any Party hereto is null and void). Nothing in this Agreement, expressed or implied, may be construed to confer upon any Person (other than the Parties, their respective successors and assigns permitted hereby, Participants to the extent provided in Section 10.4(D) and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.
- (B) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its DIP Term Loan Commitment and the DIP Term Loans and Roll-Up Loans at the time owing to it); <u>provided</u> any such assignment is subject to the following conditions:

(i) Minimum Amounts.

(a) in the case of an assignment of the entire remaining amount of the assigning Lender's DIP Term Loan Commitment or the DIP Term Loans or Roll-Up Loans at the time owing to it or contemporaneous assignments to related Approved Funds (determined after giving effect to such assignments) that equal at least the amount specified in clause (b) below in the aggregate or

- in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and
- (b) in any case not described in clause (a) above, the aggregate amount of the DIP Term Loan Commitment (which for this purpose includes DIP Term Loans outstanding thereunder) or, if the applicable DIP Term Loan Commitment is not then in effect, the principal outstanding balance of the DIP Term Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent) is not less than \$1,000,000 unless each of the Required Lenders and, so long as no Default or Event of Default exists, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed).
- (ii) <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the DIP Term Loan or the DIP Term Loan Commitment assigned.
- (iii) Required Consents. No consent is required for any assignment except to the extent required by Section 10.4(B)(i)(b) and, in addition:
 - (a) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) is required unless (1) a Default or Event of Default exists at the time of such assignment or (2) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund; provided the Borrower is deemed to have consented to any such assignment unless it has objected thereto by written notice to the Administrative Agent within ten Business Days after having received notice thereof; and
 - (b) the consent of the Required Lenders (or the Administrative Agent at the direction of the Required Lenders, with such consent not to be unreasonably withheld or delayed) is required for assignments in respect of any DIP Term Loans or Roll-Up Loans to a Person who is not a Lender, an Affiliate of a Lender or an Approved Fund.
- (iv) <u>Assignment and Assumption</u>. The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee of \$3,500; <u>provided</u> that the Administrative Agent may, in its sole discretion, elect to waive such processing and recording fee in the case of any assignment. The assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire.
- (v) <u>No Assignment to Certain Persons</u>. No such assignment may be made to (1) the Borrower or any of the Borrower's Affiliates or Subsidiaries or (2) any Defaulting Lender or any of its Subsidiaries, or any Person who, upon becoming a Lender hereunder, would constitute a Defaulting Lender or a Subsidiary thereof.

- (vi) No Assignment to Natural Persons. No such assignment may be made to a natural Person (or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person).
- (vii) Certain Additional Payments. In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment is effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower, the applicable pro rata share of DIP Term Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (1) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent and each Lender hereunder (and interest accrued thereon) and (2) acquire (and fund as appropriate) its full pro rata share of all DIP Term Loans in accordance with its Commitment Percentage. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder becomes effective under Applicable Law without compliance with the provisions of this Section 10.4(B)(iv), then the assignee of such interest is deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to Section 10.4(C), from and after the effective date specified in each Assignment and Assumption, the assignee thereunder is a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, has the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder is, to the extent of the interest assigned by such Assignment and Assumption, released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender ceases to be a Party) but continues to be entitled to the benefits of Sections 3.6, 3.7 and 10.2 with respect to facts and circumstances occurring prior to the effective date of such assignment; provided except to the extent otherwise expressly agreed by the affected parties, no assignment by a Defaulting Lender constitutes a waiver or release of any claim of any Party arising from that Lender's having been a Defaulting Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this paragraph is treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 10.4(D).

(C) Register. The Administrative Agent, acting solely for this purpose as an agent of the Borrower, shall maintain a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the DIP Term Loan Commitments of, and principal amounts (and stated interest) of the DIP Term Loans and Roll-Up Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register are conclusive absent manifest error, and the Borrower, the Administrative Agent

- and the Lenders shall treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- (D) Participations. Any Lender may at any time, without the consent of, or notice to, the Borrower or the Administrative Agent, sell participations to any Person (other than a natural Person, or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person, or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "Participant") in all or a portion of such Lender's rights or obligations under this Agreement (including all or a portion of its DIP Term Loan Commitment or the DIP Term Loans or Roll-Up Loans owing to it); provided (i) such Lender's obligations under this Agreement remain unchanged, (ii) such Lender remains solely responsible to the other Parties for the performance of such obligations and (iii) the Borrower, the Administrative Agent and Lenders continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement. For the avoidance of doubt, each Lender is responsible for the indemnity under Section 10.3(C) with respect to any payments made by such Lender to its Participants.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender retains the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; provided such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, modification or waiver described in Section 10.3 that affects such Participant. The Borrower agrees that each Participant is entitled to the benefits of Sections 3.6 and 3.7 (subject to the requirements and limitations therein (it being understood that the documentation required under Section 3.7 shall be delivered to the participating Lender)) to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to Section 10.4(B); provided such Participant (A) agrees to be subject to the provisions of Section 3.8 as if it were an assignee under Section 10.4(B) and (B) is not entitled to receive any greater payment under Sections 3.6 or 3.7, with respect to any participation, than its participating Lender would have been entitled to receive, except to the extent such entitlement to receive a greater payment results from a Change in Law that occurs after the Participant acquired the applicable participation. Each Lender that sells a participation agrees, at the Borrower's request and expense, to use reasonable efforts to cooperate with the Borrower to effectuate the provisions of Section 3.8(B) with respect to any Participant. To the extent permitted by law, each Participant also is entitled to the benefits of Section 8.3 as though it were a Lender; provided such Participant agrees to be subject to Section 3.5 as though it were a Lender. Each Lender that sells a participation shall, acting solely for this purpose as a nonfiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under the Loan Documents (the "Participant Register"); provided no Lender has any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans or its other obligations under any Loan Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury

Regulations. The entries in the Participant Register are conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, the Administrative Agent (in its capacity as Administrative Agent) has no responsibility for maintaining a Participant Register.

- (E) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; <u>provided</u> no such pledge or assignment releases such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a Party.
- 10.5 Treatment of Certain Information; Confidentiality. Each of the Administrative Agent and the Lenders shall maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its Related Parties (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent required or requested by any regulatory authority purporting to have jurisdiction over such Person or its Related Parties (including any self-regulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent required by Applicable Laws or regulations or by any subpoena or similar legal process, (d) to any other Party, (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section 10.5, to any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights and obligations under this Agreement, (g) on a confidential basis to (i) any rating agency in connection with rating the Borrower or their Subsidiaries or the DIP Term Loans or Roll-Up Loans or (ii) the CUSIP Service Bureau or any similar agency in connection with the issuance and monitoring of CUSIP numbers with respect to the DIP Term Loans or Roll-Up Loans, (h) with the consent of the Borrower or (i) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section 10.5 or (ii) becomes available to the Administrative Agent, any Lender or any of their respective Affiliates on a nonconfidential basis from a source other than the Loan Parties. In addition, the Administrative Agent and the Lenders may disclose the existence of this Agreement and information about this Agreement to market data collectors, similar service providers to the lending industry and service providers to the Agents and the Lenders in connection with the administration of this Agreement, the other Loan Documents and the DIP Term Loan Commitments.

For purposes of this Section, "<u>Information</u>" means all information received from the Borrower or any of its Subsidiaries relating to the Borrower or any of its Subsidiaries or any of their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by the Borrower or any of its Subsidiaries; <u>provided</u>, in the case of information received from the Borrower or any of its Subsidiaries after the Closing Date, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this <u>Section 10.5</u> is considered to have complied with its obligation to do so if such Person has exercised the same

degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

- 10.6 Submission to Jurisdiction; Venue; Waiver of Jury Trial.
- (A) The Borrower and each other Loan Party irrevocably and unconditionally agrees that it shall not commence any action, litigation or proceeding of any kind or description, whether in law or equity, whether in contract or in tort or otherwise, against the Administrative Agent, any Lender or any Related Party of the foregoing in any way relating to this Agreement or any other Loan Document or the transactions relating hereto or thereto, in any forum other than the Bankruptcy Court, or, if the Bankruptcy Court lacks or cannot exercise subject matter jurisdiction, the courts of the State of Delaware sitting in New Castle County, and of the United States District Court of the District of Delaware, and any appellate court from any thereof, and each of the Parties irrevocably and unconditionally submits to the jurisdiction of such courts and agrees that all claims in respect of any such action, litigation or proceeding may be heard and determined in such Delaware court or, to the fullest extent permitted by Applicable Law, in such federal court. Each of the Parties agrees that a final judgment in any such action, litigation or proceeding is conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Nothing in this Agreement or in any other Loan Document affects any right that the Administrative Agent or any Lender may otherwise have to bring any action or proceeding relating to this Agreement or any other Loan Document against the Borrower or any other Loan Party or its properties in the courts of any jurisdiction.
- (B) <u>Waiver of Venue</u>. The Borrower and each other Loan Party irrevocably and unconditionally waives, to the fullest extent permitted by Applicable Law, any objection that it may now or hereafter have to the laying of venue of any action or proceeding arising out of or relating to this Agreement or any other Loan Document in any court referred to in <u>Section 10.6(A)</u>. Each of the Parties irrevocably waives, to the fullest extent permitted by Applicable Law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court.
- (C) <u>Service of Process</u>. Each Party other than the Administrative Agent irrevocably consents to service of process in the manner provided for notices in <u>Section 10.7</u>. Nothing in this Agreement affects the right of any Party to serve process in any other manner permitted by Applicable Law.
- (D) WAIVER OF JURY TRIAL. EACH PARTY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY (i) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF

ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (ii) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10.6(D).

10.7 Notices.

- (A) <u>Notices Generally</u>. Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in <u>Section 10.7(B)</u>) below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by facsimile as follows:
- (i) if to the Borrower or any other Loan Party, to it at the addresses for it as set forth on Annex C;
- (ii) if to the Administrative Agent, to it at the addresses for it as set forth on Annex C;
- (iii) if to a Lender party to this Agreement on the Closing Date, to it at the addresses for it as set forth on Annex C;
- (iv) if to a Lender party to this Agreement after the Closing Date, to it at its address (or facsimile number) set forth in its Administrative Questionnaire;
- (v) if to the United States Trustee appointed in the Bankruptcy Case, to it at its address (or facsimile number) which will be provided at the outset of the Bankruptcy Case; and
- (vi) if to any statutory committee appointed in the Bankruptcy Case, to it at its address (or facsimile number) to be provided upon appointment.

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, are deemed to have been given when received and notices sent by facsimile are deemed to have been given when sent (except that, if not given during normal business hours for the recipient, is deemed to have been given at the opening of business on the next business day for the recipient). Notices delivered through electronic communications, to the extent provided in <u>Section 10.7(B)</u>, are effective as provided in <u>Section 10.7(B)</u>.

(B) <u>Electronic Communications</u>. Notices and other communications to the Lenders hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Administrative Agent; <u>provided</u> the foregoing does not apply to notices to any Lender pursuant to <u>Section 2.2</u> if such Lender has notified the Administrative Agent that it is incapable of receiving notices under such section by electronic communication. The Administrative Agent or the Borrower may, in its discretion,

agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it; <u>provided</u> approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (a) notices and other communications sent to an e-mail address are deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement) and (b) notices or communications posted to an Internet or intranet website are deemed received upon the deemed receipt by the intended recipient, at its e-mail address as described in clause (a) above, of notification that such notice or communication is available and identifying the website address therefor; <u>provided</u>, for both clauses (a) and (b) above, if such notice, email or other communication is not sent during the normal business hours of the recipient, such notice or communication is deemed to have been sent at the opening of business on the next business day for the recipient.

- (C) <u>Change of Address</u>. Any Party may change its address or facsimile number for notices and other communications hereunder by notice to the other Parties.
- (D) <u>Reserved</u>.
- 10.8 Reinstatement of Obligations. To the extent any Loan Party makes a payment or payments to the Administrative Agent for the ratable benefit of the Lenders or the Administrative Agent receives any payment or proceeds of the Collateral which payments or proceeds or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid to a trustee, receiver or any other party under any Debtor Relief Law, other Applicable Law or equitable cause, then, to the extent of such payment or proceeds repaid, the Obligations or part thereof intended to be satisfied are revived and continued in full force and effect as if such payment or proceeds had not been received by the Administrative Agent.
- 10.9 <u>Severability</u>. Any provision of this Agreement or any other Loan Document that is prohibited or unenforceable in any jurisdiction is, as to such jurisdiction, ineffective only to the extent of such prohibition or unenforceability without invalidating the remainder of such provision or the remaining provisions hereof or thereof or affecting the validity or enforceability of such provision in any other jurisdiction.
- 10.10 <u>Integration</u>. This Agreement and the other Loan Documents, and any separate letter agreements with respect to fees payable to the Administrative Agent, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof.
- 10.11 No Implied Waivers. No failure to exercise and no delay in exercising any right or remedy under this Agreement operates as a waiver thereof. No single or partial exercise of any right or remedy under this Agreement, or any abandonment or discontinuance thereof, precludes any other or further exercise thereof or the exercise of any other right or remedy. No waiver or consent under this Agreement is applicable to any events, acts or circumstances except those specifically covered thereby.

- 10.12 <u>USA PATRIOT Act</u>. The Administrative Agent and each Lender hereby notifies the Borrower that pursuant to the requirements of the PATRIOT Act, each of them is required to obtain, verify and record information that identifies each Loan Party, which information includes the name and address of each Loan Party and other information that will allow such Lender to identify each Loan Party in accordance with the PATRIOT Act.
- by different Parties in different counterparts), each of which constitutes an original, but all of which when taken together constitute a single contract. Except as provided in Section 4.1, this Agreement becomes effective when it has been executed by the Administrative Agent and when the Administrative Agent has received counterparts hereof that, when taken together, bear the signatures of each of the other Parties. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or in electronic (i.e., "pdf" or "tif") format is effective as delivery of a manually executed counterpart of this Agreement. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption are deemed to include electronic signatures or the keeping of records in electronic form, each of which is of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any Applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws or regulations based on the Uniform Electronic Transactions Act.

(Signature pages follow)

Casse 255-110097/99-JK\$\$ Doorc 5402-21 FFileed 0066/004/225 PPagge 1681 of f 11293

Title:

The Parties have executed and delivered this Agreement as of the date first above written.

Borrower:
VHS HOLDCO, INC.
By:
Name:
Title:
Guarantor:
SHOOSMITH BROS., INC.
By:
Name:

Casse 225-11099799-JK\$\$ Doorc 502-21 FFileed 00670044225 Pragge 16392 of f 11293

Administrative Agent:	
VOLUNTEER ENTERPRISES, LLC, as Administrative Agent	
By:Name:	
Title:	

Casse 225-11099799-JK\$\$ Doorc 5402-21 FFileed 0069 0044 2255 PPagge 17308 of f 112938

<u>Lenders</u> :
VOLUNTEER ENTEPRISES, LLC, as a Lende
By: Name:
Title:

ANNEX A

Rules of Construction

- 1. <u>Definitions</u>. As used in this Agreement, the following terms have the following meanings:
 - "Acceptable Plan" means a bankruptcy plan acceptable to the Required Lenders.
- "Acceptable Stalking Horse Purchase Agreement" means the Bankruptcy Court shall have entered an order (in form and substance reasonably satisfactory to the DIP Lender) authorizing and approving a stalking horse purchase agreement for all or substantially all of the Debtors' assets(s) that (a) is executed, effective, and binding on all parties thereto and (b) either (x) is approved by the DIP Lender in its reasonable discretion or (y) (1) would generate cash proceeds sufficient for the payment in full in cash of the Obligations (pursuant to a signed commitment acceptable to the DIP Lender in its reasonable discretion to lend from a recognized lender or another source of funding acceptable to the DIP Lender in its reasonable discretion) upon the closing date of such sale, (2) closes on or before the eighty-fifth (85th) day after the Petition Date, and (3) requires, pursuant to an order of the Bankruptcy Court, the payment in full in cash of the Obligations as set forth in clause (1).
- "Account Control Agreement" means an agreement which grants the Administrative Agent "control" as defined in the Uniform Commercial Code in effect in the applicable jurisdiction over the applicable Deposit Account.
- "Administrative Agent" has the meaning set forth for such term in the introduction to this Agreement.
- "<u>Administrative Agent's Office</u>" means the office of the Administrative Agent specified in or determined in accordance with the provisions of <u>Section 10.7</u>.
- "<u>Administrative Questionnaire</u>" means an administrative questionnaire in a form supplied by the Administrative Agent.
- "Affiliate" means, with respect to a specified Person, another Person that directly, or indirectly through one or more intermediaries, Controls or is Controlled by or is under common Control with the Person specified.
 - "Agreement" has the meaning set forth for such term in the introduction to this Agreement.
- "Anti-Corruption Laws" means all laws, rules, and regulations of any jurisdiction applicable to the Borrower or its Subsidiaries from time to time concerning or relating to terrorism, trade sanctions programs and embargoes, import/export licensing, money laundering, bribery or corruption, including the United States Foreign Corrupt Practices Act of 1977 and the rules and regulations thereunder.

"Applicable Law" means all applicable provisions of constitutions, laws, statutes, ordinances, rules, treaties, regulations, permits, licenses, approvals, interpretations and orders of Governmental Authorities and all orders and decrees of all courts and arbitrators.

"Approved Counterparty" means (a) a Lender or any Affiliate of a Lender and (b) any other Person whose issuer rating or long term senior unsecured debt rating, at the time the Swap Agreement is entered into, is A-/A3 by S&P or Moody's (or their equivalent) or higher (or whose obligations under the applicable Swap Agreement are guaranteed by an Affiliate or other credit support provider of such Person meeting such minimum rating standards at the time the Swap Agreement is entered into).

"Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.

"Assignment and Assumption" means an assignment and assumption entered into by a Lender and an Eligible Assignee (with the consent of any party whose consent is required by Section 10.4), and accepted by the Administrative Agent, in substantially the form of Exhibit D or any other form approved by the Administrative Agent.

"Asset Disposition" means the sale, transfer, assignment, conveyance, farm-out, license, lease or other disposition of any Property (including any disposition of Equity Interests) by any Loan Party or any Subsidiary thereof (or the granting of any option or other right to do any of the foregoing). The term "Asset Disposition" does not include (a) the sale of inventory in the ordinary course of business, (b) the transfer of assets to the Borrower or any Subsidiary Guarantor pursuant to any other transaction permitted by this Agreement, (c) the write-off, discount, sale or other disposition of defaulted or past-due receivables and similar obligations in the ordinary course of business and not undertaken as part of an accounts receivable financing transaction, (d) the disposition of any Swap Agreement, and (e) dispositions of Investments in cash and Cash Equivalents, and (f) the transfer by any Loan Party of its assets to any other Loan Party.

"Bankruptcy Code" means Chapter 11, 11 U.S.C. 101 et seq.

"Bankruptcy Court" means the United States Bankruptcy Court for the District of Delaware having jurisdiction over the Chapter 11 Cases.

"Board" means the Board of Governors of the Federal Reserve System of the United States or any successor Governmental Authority.

"Borrower" has the meaning set forth for such term in the introduction to this Agreement.

"Budget" means the latest thirteen-week (13-week) detailed cash projection and statement of sources and uses of the Borrower, including anticipated uses of the DIP Term Loans, to be delivered prior to the funding of the DIP Term Loans and each week thereafter, and which shall be approved by the Lenders of the DIP Term Loans each week, in their sole discretion; provided, that unused variance amounts in any Budget for a prior period shall not be carried forward on a line item basis without the approval of the Lenders of the DIP Term Loans; provided, further, that such amounts shall be counted for purposes of calculating any variances in the Budget hereunder.

"Budget Period" means the period commencing on the date that the Interim DIP Financing Order is entered and ending on the Effective Date.

"Business Day" means any day other than a Saturday, Sunday or legal holiday on which banks in New York City are closed.

"<u>Capital Leases</u>" means, in respect of any Person, all leases which shall have been, or should have been, in accordance with GAAP, recorded as capital leases on the balance sheet of the Person liable (whether contingent or otherwise) for the payment of rent thereunder.

"Carve-Out" has the meaning assigned to such term in the Interim DIP Financing Order or the Final DIP Financing Order, as applicable.

"<u>Cash Equivalents</u>" means Investments of the type described in the definition of Permitted Investments.

"Change in Control" means an event or series of events by which:

- (A) at any time, the Parent fails to own all of the Equity Interests of the Borrower; or
- (B) (i) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act, but excluding any employee benefit plan of such person or its Subsidiaries, and any person or entity acting in its capacity as trustee, agent or other fiduciary or administrator of any such plan) other than a Permitted Holder becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act, except that a "person" or "group" shall be deemed to have "beneficial ownership" of all Equity Interests that such "person" or "group" has the right to acquire, whether such right is exercisable immediately or only after the passage of time (such right, an "option right")), directly or indirectly, of more than 50% of the Equity Interests of the Parent entitled to vote in the election of members of the board of directors (or equivalent governing body) of the Parent or (ii) a majority of the members of the board of directors (or other equivalent governing body) of the Parent do not constitute Continuing Directors; or
- (C) there has occurred under any indenture or other instrument evidencing any Debt or Equity Interests in excess of \$1,000,000 any "change in control" or similar provision (as set forth in the indenture, agreement or other evidence of such Debt) obligating the Borrower or any of its Subsidiaries to repurchase, redeem or repay all or any part of the Debt or Equity Interests provided for therein.

"Change in Law" means the occurrence, after the Closing Date, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty; (b) any change in any law, rule, regulation or treaty or in the administration, interpretation, implementation or application thereof by any Governmental Authority; or (c) the making or issuance of any request, rule, guideline or directive (whether or not having the force of law) by any Governmental Authority; provided notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith and (ii) all requests, rules, guidelines or directives promulgated by the

Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, are in each case deemed to be a "Change in Law," regardless of the date enacted, adopted or issued.

"Chapter 11 Cases" has the meaning set forth for such term in the introduction to this Agreement.

"Closing Date" means the date on which the conditions specified in Section 4.1 are satisfied (or waived in writing by the Required Lenders in accordance with Section 10.3).

"Code" means the Internal Revenue Code of 1986.

"Collateral" means the collateral security for the Obligations pledged or granted pursuant to the Security Documents.

"Commitment Percentage" means, with respect to any Lender at any time, the percentage of the total Term DIP Term Loan Commitments of all the Lenders represented by such Lender's DIP Term Loan Commitment. If the DIP Term Loan Commitments have terminated or expired, the Commitment Percentage means, with respect to any Lender at any time, the percentage of the total outstanding principal balance of the DIP Term Loans represented by the outstanding principal balance of such Lender's DIP Term Loans. The Commitment Percentage of each Lender in respect of the DIP Term Loans as of the Closing Date is set forth opposite the name of such Lender on Annex B.

"Connection Income Taxes" means Other Connection Taxes that are imposed on or measured by net income (however denominated) or that are franchise Taxes or branch profits Taxes.

"Consolidated" (or "consolidated") or "Consolidating" (or "consolidating") means, when used with reference to any financial term in this Agreement, the aggregate for two or more Persons of the amounts signified by such term for all such Persons determined on a consolidated (or consolidating) basis in accordance with GAAP, applied on a consistent basis.

"Continuing Directors" means the directors (or equivalent governing body) of the Parent on the Closing Date and each other director (or equivalent) of the Parent if, in each case, such other Person's nomination for election to the board of directors (or equivalent governing body) of the Parent is approved by more than 50% of the then Continuing Directors.

"Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. For the purposes of this definition, and without limiting the generality of the foregoing, any Person that owns directly or indirectly 10% or more of the Equity Interests having ordinary voting power for the election of the directors or other governing body of a Person (other than as a limited partner of such other Person) will be deemed to "control" such other Person. "Controlling" and "Controlled" have meanings correlative thereto.

"Covered Entity" means (a) each Loan Party, any other Persons that guaranty the Obligations and/or pledge collateral to secure the Obligations, (b) each Person that, directly or indirectly, is in control of a Person described in clause (a) above, including without limitation, Parent, and (c) all brokers or other agents of any Loan Party acting in any capacity in connection with this Agreement. For purposes of this definition, control of a Person shall mean the direct or indirect (x) ownership of, or power to vote, 25% or more of the issued and outstanding Equity Interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for such Person, or (y) power to direct or cause the direction of the management and policies of such Person whether by ownership of Equity Interests, contract or otherwise.

"Debt" means, for any Person, the sum of the following (without duplication): (a) all obligations of such Person for borrowed money or evidenced by bonds, bankers' acceptances, debentures, notes or other similar instruments; (b) all obligations of such Person (whether contingent or otherwise) in respect of letters of credit, surety or other bonds and similar instruments; (c) all accounts payable and all accrued expenses, liabilities or other obligations of such Person to pay the deferred purchase price of Property or services; (d) all obligations under Capital Leases; (e) all obligations under Synthetic Leases; (f) all Debt (as defined in the other clauses of this definition) of others secured by (or for which the holder of such Debt has an existing right, contingent or otherwise, to be secured by) a Lien on any Property of such Person, whether or not such Debt is assumed by such Person; (g) all Debt (as defined in the other clauses of this definition) of others guaranteed by such Person or in which such Person otherwise assures a creditor against loss of the Debt (howsoever such assurance shall be made) to the extent of the lesser of the amount of such Debt and the maximum stated amount of such guarantee or assurance against loss; (h) all obligations or undertakings of such Person to maintain or cause to be maintained the financial position or covenants of others or to purchase the Debt or Property of others; (i) obligations to deliver commodities, goods or services, including Hydrocarbons, in consideration of one or more advance payments, other than gas balancing arrangements in the ordinary course of business; (j) obligations to pay for goods or services even if such goods or services are not actually received or utilized by such Person; (k) any Debt of a partnership for which such Person is liable either by agreement, by operation of Applicable Law but only to the extent of such liability; (1) Disqualified Capital Stock; and (m) the undischarged balance of any production payment created by such Person or for the creation of which such Person directly or indirectly received payment. The Debt of any Person includes all obligations of such Person of the character described above to the extent such Person remains legally liable in respect thereof notwithstanding that any such obligation is not included as a liability of such Person under GAAP.

"<u>Debt Issuance</u>" means the issuance of any Debt for borrowed money by any Loan Party or any of its Subsidiaries.

"Debtors" means, collectively, the following Borrower and Guarantor (each, also, a "Debtor").

"<u>Default</u>" means any of the events specified in <u>Section 8.1</u> that with the passage of time, the giving of notice or any other condition, would constitute an Event of Default.

"<u>Default Rate</u>" means an interest rate equal to the Interest Rate plus 2.0%.

"<u>Defaulting Lender</u>" means, subject to <u>Section 3.10(B)</u>, any Lender that (a) has failed to (i) fund all or any portion of its DIP Term Loans within two (2) Business Days of the date such DIP Term Loans were required to be funded hereunder unless such Lender notifies the Administrative Agent and the Borrower in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied or (ii) pay to the Administrative Agent or any Lender (upon notice by such Lender to the Administrative Agent that it has not been paid) any other amount required to be paid by it hereunder within two (2) Business Days of the date when due, (b) has notified the Borrower (and the Borrower has so notified the Administrative Agent) or the Administrative Agent in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a DIP Term Loan hereunder and states that such position is based on such Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable default, is specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three Business Days after written request by the Administrative Agent or the Borrower, to confirm in writing to the Administrative Agent and the Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender will cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by the Administrative Agent and the Borrower) or (d) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; provided that a Lender is not a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender. Any Lender who is determined to be a Defaulting Lender under any one or more of clauses (a) through (d) above is deemed to be a Defaulting Lender (subject to Section 3.10(B)) upon delivery of written notice of such determination to the Borrower and each Lender. Any determination by the Administrative Agent that a Lender is a Defaulting Lender under any one or more of clauses (a) through (d) above is conclusive and binding absent manifest error.

"<u>Deposit Account</u>" means any operating, administrative, cash management, collection activity, demand, time, savings, passbook or other deposit account maintained with a bank or other financial institution.

"DIP Facility" has the meaning set forth for such term in the introduction to this Agreement.

"<u>DIP Financing Orders</u>" means the Interim DIP Financing Order and/or the Final DIP Financing Order, as applicable.

"<u>DIP Term Loan</u>" means the First DIP Term Loan Advances, Second DIP Term Loan Advances and Third DIP Term Loan Advances.

"<u>DIP Term Loan Commitment</u>" means (a) as to any Lender, the obligation of such Lender to make a portion of the DIP Term Loans to the account of the Borrower hereunder on the applicable borrowing date in an aggregate principal amount not to exceed the amount set forth opposite such Lender's name on <u>Annex B</u>, as such amount may be reduced or otherwise modified at any time or from time to time pursuant to the terms hereof and (b) as to all Lenders, the aggregate commitment of all Lenders to make such DIP Term Loans.

"Disqualified Capital Stock" means any Equity Interest that, by its terms (or by the terms of any security into which it is convertible or for which it is exchangeable) or upon the happening of any event, requires the payment of dividends (other than dividends payable solely in Equity Interests which do not otherwise constitute Disqualified Capital Stock) or matures or is mandatorily redeemable for any consideration other than other Equity Interests (which would not constitute Disqualified Capital Stock), pursuant to a sinking fund obligation or otherwise, or is convertible or exchangeable for Debt or redeemable for any consideration other than other Equity Interests (which would not constitute Disqualified Capital Stock) at the option of the holder thereof, in whole or in part, on or prior to the date that is one year after the earlier of (a) the Maturity Date and (b) the date on which there are no DIP Term Loans or other Obligations hereunder outstanding and all of the DIP Term Loan Commitments are terminated.

"Distribution" s has the meaning set forth for such term in Schedule 7.1 hereto.

"<u>Dollars</u>" or "<u>\$</u>" means, unless otherwise qualified, dollars in lawful currency of the United States.

"<u>Domestic Subsidiary</u>" means any Subsidiary organized under the laws of any political subdivision of the United States.

"Eligible Assignee" means any Person that meets the requirements to be an assignee under Section 10.4(B)(iii), 10.4(B)(v) and 10.4(B)(vi) (subject to such consents, if any, as may be required under Section 10.4(B)(iii)).

"Environmental Laws" means any and all Applicable Laws pertaining in any way to health, safety, the environment, the preservation or reclamation of natural resources, or the management, Release or threatened Release of any Hazardous Materials, in effect in any and all jurisdictions in which the Borrower or any of the Borrower's Subsidiaries is conducting, or at any time has conducted, business, or where any Property of the Borrower or any Subsidiaries of Borrower is located, including the Oil Pollution Act of 1990 ("OPA"), the Clean Air Act, the Comprehensive Environmental, Response, Compensation, and Liability Act of 1980 ("CERCLA"), the Federal Water Pollution Control Act, the Occupational Safety and Health Act of 1970, the Resource Conservation and Recovery Act of 1976 ("RCRA"), the Safe Drinking Water Act, the Toxic Substances Control Act, the Superfund Amendments and Reauthorization Act of 1986, the Hazardous Materials Transportation Law, the Outer Continental Shelf Lands Act of 1953 and other environmental conservation or protection Applicable Laws.

"Equity Interests" means (a) in the case of a corporation, capital stock, (b) in the case of an association or business entity, any and all shares, interests, participations, rights or other equivalents (however designated) of capital stock, (c) in the case of a partnership, partnership

Casse 255-11009779-JKSS Doorc 542-21 FFileed 0006/004/225 PPagge 1749 off 11293

interests (whether general or limited), (d) in the case of a limited liability company, membership interests, (e) any other interest or participation that confers on a Person the right to receive a share of the profits and losses of, or distributions of assets of, the issuing Person and (f) any and all warrants, rights or options to purchase any of the foregoing.

"Equity Issuance" means (a) any issuance by any Loan Party or any Subsidiary thereof of shares of its Equity Interests to any Person that is not a Loan Party (including in connection with the exercise of options or warrants or the conversion of any debt securities to equity) and (b) any capital contribution from any Person that is not a Loan Party into any Loan Party or any Subsidiary thereof. The term "Equity Issuance" does not include (i) any Asset Disposition or (ii) any Debt Issuance.

"ERISA" means the Employee Retirement Income Security Act of 1974 and the rules and regulations thereunder.

"ERISA Affiliate" means any Person who together with any Loan Party or any of its Subsidiaries is treated as a single employer within the meaning of Section 414(b), (c), (m) or (o) of the Code or Section 4001(b) of ERISA.

"Erroneous Payment" has the meaning set forth for such term in Section 9.9(A).

"Erroneous Payment Deficiency Assignment" has the meaning set forth for such term in Section 9.9(D).

"Erroneous Payment Impacted Class" has the meaning set forth for such term in Section 9.9(D).

"Erroneous Payment Return Deficiency" has the meaning set forth for such term in Section 9.9(D).

"Erroneous Payment Subrogation Rights" has the meaning set forth for such term in Section 9.9(E).

"Event of Default" has the meaning set forth for such term in Section 8.1.

"Excepted Liens" means, at any time, Liens in respect of property of the Borrower or any of its Subsidiaries:

- (A) (i) Liens granted pursuant to any Loan Document securing the Obligations pursuant to the Loan Documents and (ii) Liens granted as adequate protection on account of the secured Prepetition Loans pursuant to the DIP Financing Orders;
- (B) Liens for Taxes, assessments or other governmental charges or levies which are not delinquent or which are being contested in good faith by appropriate action and for which adequate reserves have been maintained in accordance with GAAP;
- (C) Liens in connection with workers' compensation, unemployment insurance or other social security, old age pension or public liability obligations which are not

- delinquent or which are being contested in good faith by appropriate action and for which adequate reserves have been maintained in accordance with GAAP;
- (D) statutory landlord's liens, operators', vendors', carriers', warehousemen's, repairmen's, mechanics', suppliers', workers', materialmen's, construction or other like Liens arising by operation of law in the ordinary course of business, each of which is in respect of obligations that are not delinquent or which are being contested in good faith by appropriate action and for which adequate reserves have been maintained in accordance with GAAP;
- (E) contractual Liens which arise in the ordinary course of business under operating agreements, joint venture agreements and other agreements which are usual and customary in the ordinary course of business for claims which are not delinquent or which are being contested in good faith by appropriate action and for which adequate reserves have been maintained in accordance with GAAP; provided that any such Lien referred to in this clause (E) does not materially impair the use of the Property covered by such Lien for the purposes for which such Property is held by the Borrower or any of the Borrower's Subsidiaries or materially impair the value of such Property subject thereto;
- (F) Liens arising solely by virtue of any statutory or common law provision relating to banker's liens, rights of set-off or similar rights and remedies and burdening only deposit accounts or other funds maintained with a creditor depository institution; provided that no such deposit account is a dedicated cash collateral account or is subject to restrictions against access by the depositor in excess of those set forth by regulations promulgated by the Board and no such deposit account is intended by the Borrower or any of the Borrower's Subsidiaries to provide collateral to the depository institution;
- (G) easements, restrictions, servitudes, permits, conditions, covenants, exceptions or reservations in any Property of the Borrower or any of the Borrower's Subsidiaries for the purpose of roads, pipelines, transmission lines, transportation lines, distribution lines for the removal of gas, oil, coal or other minerals or timber, and other like purposes, or for the joint or common use of real estate, rights of way, facilities and equipment, that do not secure any monetary obligations and which in the aggregate do not materially impair the use of such Property for the purposes of which such Property is held by the Borrower or any of the Borrower's Subsidiaries or materially impair the value of such Property subject thereto;
- (H) Liens on cash or securities pledged to secure performance of tenders, surety and appeal bonds, government contracts, performance and return of money bonds, bids, trade contracts, leases, statutory obligations, regulatory obligations and other obligations of a like nature incurred in the ordinary course of business; and
- (I) judgment and attachment Liens not giving rise to an Event of Default; <u>provided</u> that any appropriate legal proceedings which may have been duly initiated for the review of such judgment shall not have been finally terminated or the period within

which such proceeding may be initiated shall not have expired and no action to enforce such Lien has been commenced;

<u>provided</u> Liens described in clauses (A) through (E) above remain "Excepted Liens" only for so long as no action to enforce such Lien has been commenced, and no intention to subordinate the superpriority Lien granted in favor of the Administrative Agent and the Secured Parties is to be implied or expressed by the permitted existence of such Excepted Liens.

"Exchange Act" means the Securities Exchange Act of 1934.

"Excluded Deposit Account" means any Deposit Account, existing as of the Closing Date, which is solely used for purposes of funding payroll, payroll taxes or employee benefit payments or which solely contains cash of any Person, other than any Subsidiary, and which cash is held in such Deposit Account solely on behalf of, and for the benefit of, such third party.

"Excluded Taxes" means any of the following Taxes imposed on or with respect to a Recipient or required to be withheld or deducted from a payment to a Recipient: (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes and branch profits Taxes, in each case, (i) imposed as a result of such Recipient being organized under the laws of, or having its principal office or, in the case of any Lender, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof) or (ii) that are Other Connection Taxes; (b) in the case of a Lender, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a DIP Term Loan, Roll-Up Loan or DIP Term Loan Commitment pursuant to a law in effect on the date on which (i) such Lender acquires such interest in the DIP Term Loan, Roll-Up Loan or DIP Term Loan Commitment (other than pursuant to an assignment request by the Borrower under Section 3.9(B)) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 3.8, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a Party or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Recipient's failure to comply with Section 3.8(G) and (d) any U.S. federal withholding Taxes imposed under FATCA.

"Extraordinary Receipt" means any cash received by or paid to or for the account of any Person not in the ordinary course of business, including, without limitation, tax refunds, pension plan reversions, proceeds of insurance (other than proceeds of business interruption insurance to the extent such proceeds constitute compensation for lost earnings), condemnation awards (and payments in lieu thereof), indemnity payments and any purchase price adjustments and any cash received in connection with the settlement or other resolution (including by judgment) of any litigation, arbitration or other dispute.

"FATCA" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code.

"<u>Final DIP Financing Order</u>" means the order of the Bankruptcy Court approving the DIP Facility on a final basis in form and substance satisfactory to the Administrative Agent and the

Required Lenders in their sole discretion, as the same may be amended, modified or supplemented from time to time with, the consent of the Required Lenders (and with respect to amendments, modifications or supplements that adversely affect the rights or duties of the Administrative Agent in any respect, the Administrative Agent).

"<u>First DIP Term Loan Commitment</u>" means the loans available to be advanced to the Borrower pursuant to <u>Sections 2.3(A)</u> and <u>4.1</u> in the aggregate amount of \$1,800,000 authorized in accordance with the Interim DIP Financing Order.

"Fiscal Year" means the 12-month period ending on December 31.

"<u>Fund</u>" means any Person (other than a natural Person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans, bonds and similar extensions of credit in the ordinary course of its activities.

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"Governmental Authority" means the government of the United States or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"Governmental Obligations" means noncallable direct general obligations of the United States of America or obligations the payment of principal of and interest on which is unconditionally guaranteed by the United States of America.

"Guarantor" has the meaning set forth for such term in the introduction to this Agreement.

"Guaranty Agreement" means the unconditional guaranty agreement of even date herewith executed by the Guarantor in favor of the Administrative Agent, for the benefit of the Secured Parties.

"Hazardous Material" means any substance regulated or as to which liability might arise under any applicable Environmental Law including (a) any chemical, compound, material, product, byproduct, substance or waste defined as or included in the definition or meaning of "hazardous substance," "hazardous material," "hazardous waste," "solid waste," "toxic waste," "extremely hazardous substance," "toxic substance," "contaminant," "pollutant," or words of similar meaning or import found in any applicable Environmental Law, (b) Hydrocarbons, petroleum products, petroleum substances, natural gas, oil, oil and gas waste, crude oil and any components, fractions, or derivatives thereof and (c) radioactive materials, explosives, asbestos or asbestos containing materials, polychlorinated biphenyls, radon, infectious or medical wastes.

"Indemnified Taxes" means (a) Taxes, other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligation of any Loan Party under any Loan Document and (b) to the extent not otherwise described in clause (a) above, Other Taxes.

"Initial Approved Budget" has the meaning assigned to such term in Section 4.1(E).

"Insurance and Condemnation Event" means the receipt by any Loan Party or any of its Subsidiaries of any cash insurance proceeds or condemnation award payable by reason of theft, loss, physical destruction or damage, taking or similar event with respect to any of their respective Property.

"Interim DIP Financing Order" means the order of the Bankruptcy Court approving the DIP Facility on an interim basis, in the form of **Exhibit F** hereto or otherwise satisfactory to the Administrative Agent and the Required Lenders in their sole discretion, as the same may be amended, modified or supplemented from time to time with the consent of the Required Lenders (and with respect to amendments, modifications or supplements that affect the rights or duties of the Administrative Agent, the Administrative Agent).

"IRS" means the United States Internal Revenue Service.

"Interest Payment Date" means the last Business Day of each calendar month and the Maturity Date.

"Interest Rate" means a rate *per annum* equal to twelve percent (12.0%).

"Investment" means, for any Person, (a) the acquisition (whether for cash, Property, services or securities or otherwise) of Equity Interests of any other Person or any agreement to make any such acquisition (including any "short sale" or any sale of any securities at a time when such securities are not owned by the Person entering into such short sale), (b) the making of any deposit with, or advance, loan or capital contribution to, the purchase or other acquisition of any other Debt of or equity participation or interest in, or other extension of credit to, any other Person (including the purchase of Property from another Person subject to an understanding or agreement, contingent or otherwise, to resell such Property to such Person, but excluding any such advance, loan or extension of credit having a term not exceeding 90 days representing the purchase price of inventory or supplies sold by such Person in the ordinary course of business), (c) the purchase or acquisition (in one or a series of transactions) of Property of another Person that constitutes a business unit or (d) the entering into of any guarantee of, or other contingent obligation (including the deposit of any Equity Interests to be sold) with respect to, Debt or other liability of any other Person and (without duplication) any amount committed to be advanced, lent or extended to such Person.

"Involuntary Asset Disposition" means any loss of, damage to or destruction of, or any condemnation or other taking for public use of, any property of the Loan Parties.

"<u>Lenders</u>" means the Persons listed on <u>Annex B</u> and any other Person that becomes party to this Agreement pursuant to an Assignment and Assumption, other than any such Person that ceases to be a party to this Agreement pursuant to an Assignment and Assumption.

"Lien" means, with respect to any asset, any mortgage, leasehold mortgage, lien, pledge, charge, security interest, hypothecation or encumbrance of any kind in respect of such asset. For the purposes of this Agreement, a Person shall be deemed to own subject to a Lien any asset which it has acquired or holds subject to the interest of a vendor or lessor under any conditional sale agreement, Capital Lease Obligation or other title retention agreement relating to such asset.

"Loan" means DIP Term Loans, Roll-Up Loans (including any PIK Amounts), as applicable.

"Loan Documents" means, collectively, this Agreement, Security Documents, the Guaranty Agreement, and each other document, instrument, certificate and agreement executed and delivered by the Loan Parties or any of their respective Subsidiaries in favor of or provided to the Administrative Agent or any Lender in connection with this Agreement or otherwise referred to herein or contemplated hereby.

"Loan Parties" means, collectively, the Borrower and the Guarantor.

"Material Adverse Effect" means, with respect to the Borrower and the other Loan Parties, (a) a material adverse effect on the operations, business, assets, properties, liabilities (actual or contingent) or condition (financial or otherwise) of such Persons, taken as a whole, (b) a material impairment of the ability of any such Person to perform its obligations under the Loan Documents to which it is a party, (c) a material impairment of the rights and remedies of the Administrative Agent or any Lender under any Loan Document or (d) a material adverse effect on the legality, validity, binding effect or enforceability against any Loan Party of any Loan Document to which it is a party.

"Material Agreement" means (a) the agreements listed on any schedule to this Agreement and all replacements thereof and (b) any other agreement, document or instrument to which any Loan Party is a party or by which any Loan Party or its assets are bound and which (i) involves monetary liability of or to any Person in an amount in excess of \$250,000 in any Fiscal Year or (ii) the breach, nonperformance, cancellation or failure to renew by any party thereto would have a Material Adverse Effect.

"Maturity Date" means the date that is the earliest of (i) the date that is one-hundred ten (110) days after the Petition Date, (ii) the effective date or date of the substantial consummation (as defined in Section 1101(2) of the Bankruptcy Code) of a Plan of Reorganization or similar dispositive restructuring plan that has been confirmed by an order of the Bankruptcy Court; (iii) the date the Bankruptcy Court orders the conversion of the Chapter 11 Cases to cases under Chapter 7 of the Bankruptcy Code; (iv) the date the Bankruptcy Court orders the dismissal of the Chapter 11 Cases; (v) the date on which the Borrower consummates a sale of all or substantially all of the assets of the Borrower or its Subsidiaries pursuant to Section 363 of the Bankruptcy Code or otherwise; and (vi) the occurrence of any Event of Default or the date of acceleration of the DIP Term Loans or early termination of the DIP Term Loan Commitments hereunder or the DIP Financing Orders, including as a result of the occurrence of an Event of Default.

"Moody's" means Moody's Investors Service, Inc.

"Mortgages" means each mortgage, deed of trust or other real property security document encumbering any real property and any other Property the Required Lenders require to be mortgaged, whether now or hereafter owned by any Loan Party, executed by such Loan Party in favor of the Administrative Agent, for the benefit of the Secured Parties.

"Net Cash Proceeds" means, as applicable, (a) with respect to any Asset Disposition, Involuntary Asset Disposition, or Insurance and Condemnation Event, the gross proceeds received by any Loan Party or any of its Subsidiaries therefrom (including any cash, Cash Equivalents, deferred payment pursuant to, or by monetization of, a note receivable or otherwise, as and when received) less the sum of (i) in the case of an Asset Disposition or Involuntary Asset Disposition, all income taxes and other taxes assessed by, or reasonably estimated to be payable to, a Governmental Authority as a result of such transaction (provided that if such estimated taxes exceed the amount of actual taxes required to be paid in cash in respect of such Asset Disposition or Involuntary Asset Disposition, the amount of such excess constitutes Net Cash Proceeds), (ii) all reasonable and customary out-of-pocket fees and expenses incurred in connection with such transaction or event and (iii) the principal amount of, premium, if any, and interest on any Debt secured by a Lien on the asset (or a portion thereof) disposed of, which Debt is required to be repaid in connection with such transaction or event and (b) with respect to any Equity Issuance or Debt Issuance, the gross cash proceeds received by any Loan Party or any of its Subsidiaries therefrom less all reasonable and customary out-of-pocket legal, underwriting and other fees and expenses incurred in connection therewith.

"Non-Consenting Lender" means any Lender that does not approve any consent, waiver or amendment that (a) requires the approval of all or all affected Lenders in accordance with the terms of Section 10.3 and (ii) has been approved by the Required Lenders.

"Non-Defaulting Lender" means, at any time, each Lender that is not a Defaulting Lender at such time.

"Notice of Borrowing" means a written notice substantially in the form of **Exhibit A**.

"Notice of Prepayment" means a written notice substantially in the form of **Exhibit B**.

"Obligations" means all amounts owing to any Administrative Agent or any Lender pursuant to the terms of this Agreement or any other Loan Document (including principal, interest, any contingent indemnification and reimbursement obligations, or other payments required under Section 10.2, and all fees and other expenses required to be paid pursuant to the Loan Documents).

"OFAC" means the United States Department of the Treasury's Office of Foreign Assets Control.

"Officer's Compliance Certificate" means a certificate of a Responsible Officer of the Borrower substantially in the form of **Exhibit C**.

"Other Connection Taxes" means, with respect to any Recipient, Taxes imposed as a result of a present or former connection between such Recipient and the jurisdiction imposing such Tax (other than connections arising from such Recipient having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security

interest under, engaged in any other transaction pursuant to any Loan Document or enforced any Loan Document, or sold or assigned an interest in any Loan or Loan Document).

"Other Taxes" means all present or future stamp, court or documentary, intangible, recording, filing or similar Taxes that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, any Loan Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to Section 3.8(B)).

"Parent" has the meaning set forth for such term in the recitals to this Agreement.

"Parties" has the meaning set forth for such term in the introduction to this Agreement.

"Participant" has the meaning set forth for such term in Section 10.4(D).

"Participant Register" has the meaning set forth for such term in Section 10.4(D).

"PATRIOT Act" means the USA PATRIOT Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)).

"Payment Recipient" has the meaning set forth for such term in Section 9.9(A).

"<u>Permitted Holders</u>" means (i) Volunteer Enterprises, LLC, (ii) Environmental Services Management of Virginia, LLC, (iii) Fred Nichols, or (iv) Larry McGee, together with any Affiliates of any of the foregoing.

"Permitted Investments" means with respect to any Person:

- (a) Governmental Obligations;
- (b) Obligations of a state or commonwealth of the United States or the obligations of the District of Columbia or any possession of the United States, or any political subdivision of any of the foregoing, which are described in Section 103(a) of the Internal Revenue Code and are graded in any of the highest three (3) major grades as determined by at least one Rating Agency, or secured, as to payments of principal and interest, by a letter of credit provided by a financial institution or insurance provided by a bond insurance company which in each case is itself or its debt is rated in one of the highest three (3) major grades as determined by at least one Rating Agency;
- (c) Banker's acceptances, commercial accounts, demand deposit accounts, certificates of deposit, other time deposits or depository receipts issued by or maintained with any Investor or any Affiliate thereof, or any bank, trust company, savings and loan association, savings bank or other financial institution whose deposits are insured by the Federal Deposit Insurance Corporation and whose reported capital and surplus equal at least \$250,000,000, provided that such minimum capital and surplus

- requirement shall not apply to demand deposit accounts maintained by any Credit Party in the ordinary course of business;
- (d) Commercial paper rated at the time of purchase within the two highest classifications established by not less than two Rating Agencies, and which matures within 270 days after the date of issue;
- (e) Secured repurchase agreements against obligations itemized in paragraph (a) above, and executed by a bank or trust company or by members of the association of primary dealers or other recognized dealers in United States government securities, the market value of which must be maintained at levels at least equal to the amounts advanced; and
- (f) Any fund or other pooling arrangement which exclusively purchases and holds the investments itemized in (a) through (e) above.
- "Permitted Variance" has the meaning set forth for such term in Section 6.18(D).
- "Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.
 - "Petition Date" means June 1, 2025.
 - "PIK Amounts" has the meaning set forth for such term in Section 3.2(C).
 - "Plan Effective Date" means the effective date of an Acceptable Plan.
- "<u>Plan of Reorganization</u>" a plan of reorganization with respect to the Debtors pursuant to the Chapter 11 Cases.
- "<u>Prepetition Debt</u>" means all Debt and other obligations of the Debtors arising before the Petition Date.
 - "Prepetition Loan Agreement" has the meaning set forth in the recitals hereto.
 - "Prepetition Loans" means the Bridge Notes as defined in the Prepetition Loan Agreement.
- "Property" means any right or interest in or to property of any kind whatsoever, whether real, personal or mixed, movable or immovable, and whether tangible or intangible, including Equity Interests.
 - "Recipient" means the Administrative Agent or any Lender, as applicable.
- "Redemption" means, with respect to any Debt, the repurchase, redemption, prepayment, repayment, defeasance or any other acquisition or retirement for value (or the segregation of funds with respect to any of the foregoing) of such Debt. "Redeem" has the correlative meaning thereto.

Casse 225-1009799-JK\$ Door 502-21 FFiled 0066 0044 225 Pragge 1650 off 11293

"Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives of such Person and of such Person's Affiliates.

"Release" means any depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, dumping or disposing into the environment.

"Reportable Compliance Event" means that any Covered Entity becomes a Sanctioned Person, or is indicted, arraigned, investigated or custodially detained, or receives an inquiry from regulatory or law enforcement officials, in connection with any Anti-Corruption Law or any predicate crime to any Anti-Corruption Law, or has knowledge of or self-discovers facts or circumstances implicating any aspect of its operations with the actual or possible violation of any Anti-Corruption Law.

"Required Lenders" means, at any date, any combination of Lenders holding more than 66 2/3% of the sum of the aggregate amount of the DIP Term Loan Commitments or, if the DIP Term Loan Commitments have been reduced to zero or terminated, any combination of Lenders holding more than 66 2/3% of the aggregate outstanding principal amount of the DIP Facility; provided the DIP Term Loan Commitment of, and the portion of the DIP Term Loans held or deemed held by, any Defaulting Lender are excluded for purposes of making a determination of Required Lenders.

"Responsible Officer" means as to any Loan Party, the chief executive officer, president, chief financial officer, controller, treasurer, assistant treasurer or any executive vice president of such Person or any other officer of such Person designated in writing by the Borrower and reasonably acceptable to the Required Lenders; provided, to the extent requested thereby, the Administrative Agent has received a certificate of such Person certifying as to the incumbency and genuineness of the signature of each such officer. Any document delivered hereunder or under any other Loan Document that is signed by a Responsible Officer of a Person is conclusively presumed to have been authorized by all necessary corporate, partnership or other action on the part of such Person and such Responsible Officer is conclusively presumed to have acted on behalf of such Person.

"Roll-Up Loans" has the meaning set forth for such term in the introduction to this Agreement.

"S&P" means Standard & Poor's Financial Services LLC, a part of McGraw-Hill Financial and any successor thereto.

"Sale Transaction" means the contemplated sale of all or substantially all of the Debtors' assets and business, pursuant to sections 363 and 365 of the Bankruptcy Code and seeking the entry of orders by the Bankruptcy Court approving such a transaction and authorizing the Debtors to consummate the transactions contemplated therein.

"Sanctions" means economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by the United States government (including those

administered by OFAC), the European Union, Her Majesty's Treasury, or other relevant sanctions authority.

"Sanctioned Country" means at any time, a country or territory which is itself the subject or target of any Sanctions (including Cuba, Iran, North Korea, Sudan and Syria).

"Sanctioned Person" means, at any time, (a) any Person listed in any Sanctions-related list of designated Persons maintained by OFAC, the United States Department of State, the United Nations Security Council, the European Union, Her Majesty's Treasury, or other relevant sanctions authority, (b) any Person operating, organized or resident in a Sanctioned Country or (c) any Person owned or controlled by any such Person or Persons described in clauses (a) and (b) above.

"Second DIP Term Loan Commitment" means the loans available to be advanced to the Borrower pursuant to Sections 2.3(B) and 4.2 in the aggregate amount of \$1,550,000 authorized in accordance with the Final DIP Financing Order.

"Secured Parties" means, collectively, the Administrative Agent and the Lenders.

"Security Documents" means the collective reference to the Mortgages, the Guaranty Agreement, any Account Control Agreements, and each other agreement or writing pursuant to which any Loan Party pledges or grants a security interest in any Property securing the Obligations.

"Shoosmith" has the meaning set forth for such term in the introduction to this Agreement.

"Subordinated Debt" means and Debt of any Loan Party which has been subordinated in right of payment and priority to the Obligations, all on terms and conditions satisfactory to the Administrative Agent.

"Subsidiary" means as to any Person, any corporation, partnership, limited liability company or other entity of which more than 50% of the outstanding Equity Interests having ordinary voting power to elect a majority of the board of directors (or equivalent governing body) or other managers of such corporation, partnership, limited liability company or other entity is at the time owned by (directly or indirectly) or the management is otherwise controlled by (directly or indirectly) such Person (irrespective of whether, at the time, Equity Interests of any other class or classes of such corporation, partnership, limited liability company or other entity have or might have voting power by reason of the happening of any contingency). Unless otherwise qualified, references to "Subsidiary" or "Subsidiaries" refer to a Subsidiary (including the Guarantor) of the Borrower.

"Subsidiary Guarantors" means, collectively, all direct and indirect Subsidiaries of the Borrower in existence on the Closing Date or which become a party to the Guaranty Agreement pursuant to Section 6.13(A).

"Swap Agreement" means any agreement with respect to any swap, forward, future or derivative transaction or option or similar agreement, whether exchange traded, "over-the-counter" or otherwise, involving, or settled by reference to, one or more rates, currencies, commodities, equity or debt instruments or securities, or economic, financial or pricing indices or measures of economic, financial or pricing risk or value or any similar transaction or any combination of these

transactions (including any agreement, contract or transaction that constitutes a "swap" within the meaning of section 1a(47) of the Commodity Exchange Act); <u>provided</u> that no phantom stock or similar plan providing for payments only on account of services provided by current or former directors, officers, employees or consultants of the Borrower or the Subsidiaries is a Swap Agreement.

"Synthetic Leases" means, in respect of any Person, all leases which shall have been, or should have been, in accordance with GAAP, treated as operating leases on the financial statements of the Person liable (whether contingently or otherwise) for the payment of rent thereunder and which were properly treated as indebtedness for borrowed money for purposes of U.S. federal income taxes, if the lessee in respect thereof is obligated to either purchase for an amount in excess of, or pay upon early termination an amount in excess of, 80% of the residual value of the Property subject to such operating lease upon expiration or early termination of such lease.

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"Third DIP Term Loan Commitment" means the loans available to be advanced to the Borrower pursuant to Sections 2.3(A) and 4.3 from and after July 25, 2025 in the aggregate amount of \$1,550,000 authorized in accordance with the Final DIP Financing Order.

"<u>Total DIP Term Loan Commitment</u>" means, as of the Closing Date, the amount of DIP Term Loan Commitments, plus the Roll-Up Loan Amount.

"Transactions" means, with respect to (a) the Borrower, the execution, delivery and performance by the Borrower of this Agreement and each other Loan Document to which it is a party, the borrowing of DIP Term Loans and Roll-Up Loans, the use of the proceeds thereof and the grant of Liens by the Borrower on the Collateral pursuant to the Security Documents and (b) each Guarantor, the execution, delivery and performance by such Guarantor of each Loan Document to which it is a party, the guaranteeing of the Obligations and the other obligations under the Guaranty Agreement by such Guarantor and such Guarantor's grant of Liens by such Guarantor on the Collateral pursuant to the Security Documents.

"Type" shall mean the type of DIP Term Loan to be borrowed, either a (i) First DIP Term Loan Advance, (b) Second DIP Term Loan Advance or (iii) Third DIP Term Loan Advance.

"United States" or "U.S." means the United States of America.

"<u>U.S. Person</u>" means any Person that is a "United States Person" as defined in Section 7701(a)(30) of the Code.

"VDEQ" means the Virginia Department of Environmental Quality or any successor agency.

"Wholly-Owned" (or "wholly-owned") means, with respect to a Subsidiary, that all of the Equity Interests of such Subsidiary are, directly or indirectly, owned or controlled by the Borrower or one or more of its Wholly-Owned Subsidiaries (except for directors' qualifying shares or other

Casse 225-1009799-JK\$ Door 502-21 FFiled 0066 0044 225 Pragge 1933 off 11293

shares required by Applicable Law to be owned by a Person other than the Borrower or one or more of its Wholly-Owned Subsidiaries).

"Withholding Agent" means any Loan Party and the Administrative Agent.

2. <u>Accounting Terms</u>.

- (A) All accounting terms not specifically or completely defined herein are construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with GAAP, applied on a consistent basis, as in effect from time to time and in a manner consistent with that used in preparing the audited financial statements required by Section 6.1(A) except as otherwise specifically prescribed herein.
- (B) If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and either the Borrower or the Required Lenders so request, the Lenders and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); <u>provided</u> that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Borrower shall provide to the Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.
- 3. <u>Use of Certain Terms</u>. As used in this Agreement, the plural includes the singular and the singular includes the plural. All pronouns and any variations thereof refer to masculine, feminine, neuter, singular or plural as the identity of the Person or Persons may require. As used in this Agreement, "include," "includes" and "including" have the inclusive meaning of "including without limitation."
- References. Section and other headings are for reference only, and do not affect the interpretation or meaning of any provision of this Agreement. Unless otherwise provided, references to articles, sections, clauses, annexes, schedules and exhibits refer to articles, sections, clauses, annexes, schedules and exhibits of this Agreement. The words "hereof," "herein," "hereby," "hereunder" and other similar terms of this Agreement refer to this Agreement as a whole and not exclusively to any particular provision of this Agreement. Unless otherwise expressly indicated in this Agreement, the words "above" and "below," when following a reference to a clause of any Loan Document, refer to a clause within the same section of such Loan Document. References in this Agreement to this Agreement, any other Loan Document or any other agreement are deemed to (a) refer to this Agreement, such other Loan Document or such other agreements, as the case may be, as the same may be amended, restated, supplemented or otherwise modified from time to time under the provisions hereof or thereof, unless expressly stated otherwise or unless such amendment, restatement, supplement or modification is not permitted by the terms of this Agreement and (b) include all schedules, exhibits and appendices thereto. References in this Agreement to any law, rule, statute or regulation are deemed to refer to such law, rule, statute or regulation as it may be amended, supplemented or otherwise modified from time to time, and any

Casse 225-1009799-JK\$ Door: 502-21 FFiled 0066 0044 225 Pragge 19514 off 11293

successor law, rule, statute or regulation, in each case as in effect at the time any such reference is operative. Any reference to a Person includes the successors, assigns, participants and transferees of such Person, but such reference will not increase, decrease or otherwise modify in any way the provisions in this Agreement or any other Loan Document governing the assignment of rights and obligations under or the binding effect of any provision of this Agreement or any other Loan Document. The words "asset" and "property" are construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract rights. The term "documents" includes any and all instruments, documents, agreements, certificates, notices, reports, financial statements and other writings, however evidenced, whether in physical or electronic form. In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including;" the words "to" and "until" each mean "to but excluding;" and the word "through" means "to and including"

- 5. <u>Times of Day</u>. Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).
- 6. <u>Rounding</u>. Any financial ratios required to be maintained pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio or percentage is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).
- 7. <u>Inconsistencies with Other Documents</u>. In the event there is a conflict or inconsistency between this Agreement and any other Loan Document, the terms of this Agreement control; <u>provided</u> any provision of the Security Documents which imposes additional burdens on a Loan Party or any of its Subsidiaries or further restricts the rights of a Loan Party or any of its Subsidiaries or gives the Administrative Agent or Lenders additional rights is not deemed to be in conflict or inconsistent with this Agreement and shall be given full force and effect.

Casse 225-11099799-JK\$\$ Doorc 5402-21 HFileed 0066/004/225 Prayge 1955 off 11293

ANNEX B

DIP Term Loan Commitments and Commitment Percentages

<u>DIP TERM LOANS MADE PURSUANT TO SECTION 2.1(B) – FIRST DIP TERM LOAN ADVANCE</u>:

<u>Lender</u>	First DIP Term Loan Advance Commitment	<u>Commitment</u> <u>Percentage</u>
VOLUNTEER ENTERPRISES, LLC	\$1,800,000	100%
	21 222 222	1000/
Total	\$1,800,000	100%

Casse 225-11099799-JKSS Doorc 502-21 FFileed 0066/004/225 Pragge 1956 off 11293

<u>DIP TERM LOANS MADE PURSUANT TO SECTION 2.1(C) – SECOND DIP TERM LOAN ADVANCE:</u>

<u>Lender</u>	Second DIP Term Loan Advance Commitment	Commitment Percentage
VOLUNTEER ENTERPRISES, LLC	\$1,550,000	100%
Total	\$1,550,000	100%

Casse 225-11099799-JKSS Doorc 502-21 FFileed 0066/004/225 Pragge 1957 off 11293

<u>DIP TERM LOANS MADE PURSUANT TO SECTION 2.1(D) – THIRD DIP TERM LOAN ADVANCE:</u>

<u>Lender</u>	Third DIP Term Loan Advance Commitment	Commitment Percentage
VOLUNTEER ENTERPRISES, LLC	\$1,550,000	100%
Total	\$1,550,000	100%

ROLL-UP LOANS PURSUANT TO SECTION 2.2(A):

<u>Lender</u>	Roll-Up Loans for 2022 Term Loans
VOLUNTEER ENTERPRISES, LLC	\$500,000
The state of the s	0700 000
Total	\$500,000

ANNEX C

Addresses for Notices

If to the Loan Parties:

VWS Holdco, Inc.
[ADDRESS]
[ATTENTION]
[TELEPHONE]
[EMAIL]

With copies to:

[LAW FIRM]
[ADDRESS]
[ATTENTION]
[TELEPHONE]
[EMAIL]

If to the Administrative Agent or the Lenders:

Volunteer Enterprises, LLC [ADDRESS] [ATTENTION] [TELEPHONE] [EMAIL]

With a copy to:

Cole Schotz P.C. 1325 Avenue of the Americas, 20th Floor New York, NY 10019 Attention of: James Stefanick, Daniel Geoghan

Telephone No.: 646-563-8949

E-mail: jstefanick@coleschotz.com; dgeoghan@coleschotz.com

ANNEX D

Closing Deliverables

<u>Executed Loan Documents</u>. This Agreement, the Guaranty Agreement and the Security Documents, together with any other applicable Loan Documents.

Officer's Certificate. A certificate from a Responsible Officer of the Borrower to the effect that (a) all representations and warranties of the Loan Parties contained in this Agreement and the other Loan Documents are true, correct and complete, (b) none of the Loan Parties is in violation of any of the covenants contained in this Agreement and the other Loan Documents, (c) after giving effect to the Transactions, no Default or Event of Default has occurred and is continuing, (d) since the Petition Date, no event has occurred or condition arisen, either individually or in the aggregate, that has had or could reasonably be expected to have a Material Adverse Effect and (e) each of the Loan Parties, as applicable, has satisfied each of the conditions set forth in Section 4.1.

Certificate of Secretary of Each Loan Party. A certificate of a Responsible Officer of each Loan Party certifying as to the incumbency and genuineness of the signature of each officer of such Loan Party executing Loan Documents to which it is a party and certifying that attached thereto is a true, correct and complete copy of (a) the articles or certificate of incorporation or formation (or equivalent), as applicable, of such Loan Party and all amendments thereto, certified as of a recent date by the appropriate Governmental Authority in its jurisdiction of incorporation, organization or formation (or equivalent), as applicable, (b) the bylaws or other governing document of such Loan Party as in effect on the Closing Date and (c) resolutions duly adopted by the board of directors (or other governing body) of such Loan Party authorizing and approving the transactions contemplated hereunder and the execution, delivery and performance of this Agreement and the other Loan Documents to which it is a party.

<u>Certificates of Good Standing</u>. Certificates as of a recent date of the good standing of each Loan Party under the laws of its jurisdiction of incorporation, organization or formation (or equivalent), as applicable, and, to the extent requested by the Lenders, each other jurisdiction where such Loan Party is qualified to do business and, to the extent available, a certificate of the relevant taxing authorities of such jurisdictions certifying that such Loan Party has filed required tax returns and owes no delinquent taxes.

Other Documents. Copies of all other documents, certificates and instruments reasonably requested by the Administrative Agent or any Lender.

Cassae 225-110097/99-JK\$\$ Doorc 5402-21 Hilleadd 0006/004/225 PPagge 1981 off 11293

Exhibit A

Form of Notice of Borrowing

Casse 225-1009799-JKSS Doorc 502-21 FFileed 0066 0044 225 Pragge 19592 off 11293

Dated as of:,
Volunteer Enterprises, LLC [ADDRESS] [ATTENTION] [TELEPHONE] [EMAIL]
Ladies and Gentlemen:
This irrevocable Notice of Borrowing is delivered to you pursuant to <u>Section 2.3</u> of the Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of June 2 2025, as amended and as may be further amended from time to time (the " <u>DIP Agreement</u> ") among VWS Holdco, Inc. a Delaware corporation (the " <u>Borrower</u> "), Shoosmith Bros., Inc., a Virginia Corporation (the " <u>Guarantor</u> "), the Lenders party thereto from time to time, and Volunteer Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the " <u>Administrative Agent</u> "). Capitalized terms used herein and not otherwise defined have the meanings set forth for such terms in the DIP Agreement.
 The Borrower hereby requests that the Lenders make the DIP Term Loans to the Borrower in the aggregate principal amount of \$
2. The Borrower hereby requests that such DIP Term Loans be made on the following Business Day:
3. The Person(s) to whom the proceeds of such DIP Terms Loan(s) are to be disbursed and the wire instructions of the account(s) to which funds are to be disbursed are as follows:
VWS Holdco, Inc. Bank: [] ABA: [] Account Name: [] Account Number: []
4. The Borrower has previously delivered the appropriate tax forms with respect to each Person listed in paragraph 3 above as required by the DIP Agreement.
5 The Type of DID Term I can to be howevered by Page Pate I cans

- 5. The Type of DIP Term Loan to be borrowed is: Base Rate Loans
- 6. The aggregate principal amount of all DIP Term Loans outstanding as of the date hereof (including the DIP Term Loans requested herein) does not exceed the maximum amount permitted to be outstanding pursuant to the terms of the DIP Agreement
- 7. All of the conditions applicable to the DIP Term Loan requested herein as set forth in the DIP Agreement have been satisfied as of the date hereof and will remain satisfied to the date of such DIP Term Loan.

Case 25-10979-JKS Doc 50-2 Filed 06/04/25 Page 160 of 128

Executed and delivered as of the date first above	e written.
	VWS HOLDCO, INC.
	By: Name: [] Title: []

Exhibit B

Form of Notice of Prepayment

FORM OF PREPAYMENT NOTICE

Date: $[\bullet]$, $20[\bullet]^2$

Volunteer Enterprises, LLC, as Administrative Agent [ADDRESS]
[ATTENTION]
[TELEPHONE]
[EMAIL]

Re: <u>VWS Holdco</u>, Inc. – Prepayment Notice

Ladies and Gentlemen:

Reference is made to the Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of [_____], 2025, as amended and as may be further amended from time to time (the "DIP Agreement"), among VWS Holdco, Inc. a Delaware corporation (the "Borrower"), Shoosmith Bros., Inc., a Virginia Corporation (the "Guarantor"), the Lenders party thereto from time to time, and Volunteer Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the "Administrative Agent"). Each capitalized term used but not otherwise defined in this Prepayment Notice shall have the meaning assigned to such term in the DIP Credit Agreement.

This Prepayment Notice is delivered³ to you pursuant to <u>Section 2.5</u> of the DIP Credit Agreement. The Borrower hereby gives notice of a prepayment of Loans as follows:

Borrowing with an Interest Period ending $[\bullet]$, $20[\bullet]$ in the aggregate principal amount of $[\bullet]$.

On $[\bullet]$, 20 $[\bullet]$ (a Business Day).

[3. Attached hereto as Appendix A is a reasonably detailed calculation of the amount of the prepayment contemplated hereby.] 4

[Signature Page Follows]

To be delivered not later than 12:00 noon, New York City time, three (3) Business Days before the scheduled date of such repayment. Each repayment of a Borrowing shall be applied to the DIP Term Loans included in the repaid Borrowing such that each Lender receives its ratable share of such repayment (based upon the respective Percentages at the time of such repayment).

The notice of which may be given by facsimile or email.

This information is only required for a mandatory prepayment.

Case 25-10979-JKS Doc 50-2 Filed 06/04/25 Page 108 of 128

IN WITNESS WHEREOF, the undersigned Responsible Officer has executed and delivered this Prepayment Notice on behalf of the Borrower, in his or her capacity as a duly elected, qualified and acting Responsible Officer of the Borrower (and not in the Responsible Officer's individual capacity), and has made the certifications and statements contained herein in the capacity stated herein, as of the date first above written.

VWS HOLDCO, INC.
as Borrower
By:
Name: []
Title: [

Exhibit C

Form of Notice of Responsible Officer's Compliance Certificate

RESPONSIBLE OFFICER CERTIFICATE OF VWS HOLDCO, INC.

June 2, 2025

This Responsible Officer's Certificate is delivered in connection with that certain Senior
Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of June 2, 2025, as
amended and as may be further amended from time to time (the "DIP Agreement"), among VWS
Holdco, Inc. a Delaware corporation (the "Borrower"), Shoosmith Bros., Inc., a Virginia
Corporation (the "Guarantor"), the Lenders party thereto from time to time, and Volunteer
Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the
"Administrative Agent"). The undersigned, [], the Responsible Officer of Borrower,
hereby certifies to Lenders that he is the Responsible Officer of Borrower, and in such capacity, is
duly authorized to execute this Responsible Officer's Certificate and further represents, warrants
and certifies to Lenders as of the date hereof:

- 1. Attached hereto as <u>Exhibit A</u> is a true, correct and complete copy of the Written Consent of the Board of Directors of Borrower approving and authorizing the execution, delivery and performance of the DIP Credit Agreement and other agreements, instruments and documents contemplated thereby to which it is a party, and the consummation of all of the transactions contemplated thereby. Such resolutions have not been amended, rescinded or modified since their adoption and remain in effect as of the date hereof.
- 2. The person whose name appears below is duly elected, qualified and an acting officer of Borrower, occupying the position set forth in the signature block below, and the signature set forth below is a true signature, and such officer is duly authorized to execute and deliver, on behalf of Borrower, the DIP Credit Agreement and other agreements, instruments and documents contemplated thereby to which it is a party.

Capitalized terms used herein and not otherwise defined shall have the meanings set forth in the DIP Credit Agreement.

IN WITNESS WHEREOF, the undersigned Responsible Officer of Borrower has executed this certificate as of the date first written above.

Name: [
Title:	1

Exhibit D

Form of Assignment and Assumption

SENIOR SECURED SUPERPRIORITY DEBTOR-IN-POSSESSION TERM LOAN AGREEMENT (THE "DIP CREDIT AGREEMENT")¹

STANDARD TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION

- 1. Representations and Warranties.
- 1.1 Assignor. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any Lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations of any Person other than the Assignor made in or in connection with the DIP Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral thereunder, (iii) the financial condition of the Borrower, the Guarantor, or any Affiliate or Subsidiary of the foregoing or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, the Guarantor, or any Affiliate or Subsidiary of the foregoing or any other Person of any of such Person's respective obligations under any Loan Document.
- Assignee. The Assignee (a) represents and warrants to the Assignor and the Borrower that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the DIP Credit Agreement, (ii) it has satisfied all required conditions pursuant Section 10.4 of the DIP Credit Agreement, (iii) from and after the Effective Date, it shall be bound by the provisions of the DIP Credit Agreement and the other Loan Documents as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it is sophisticated with respect to decisions to acquire assets of the type represented by the Assigned Interest and either it, or the Person exercising discretion in making its decision to acquire the Assigned Interest, is experienced in acquiring assets of such type, (v) it has received a copy of the DIP Credit Agreement, and has received or has been accorded the opportunity to receive copies of the most recent financial statements delivered pursuant to Section 6.1 or Section 6.18 thereof, as applicable, and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest, (vi) it has, independently and without reliance on the Administrative Agent, the Assignor or any other Lender and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into

¹ Capitalized terms used herein and not otherwise defined have the meanings set forth for such terms in the Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of June 2, 2025, as amended and as may be further amended from time to time (the "<u>DIP Agreement</u>"), among VWS Holdco, Inc. a Delaware corporation (the "<u>Borrower</u>"), Shoosmith Bros., Inc., a Virginia Corporation (the "<u>Guarantor</u>"), the Lenders party thereto from time to time, and Volunteer Enterprises, LLC, a Texas limited liability company, as administrative agent (in such capacity, the "Administrative Agent").

this Assignment and Assumption and to purchase the Assigned Interest, (vii) attached to the Assignment and Assumption is any documentation required to be delivered by it pursuant to the terms of the DIP Credit Agreement, duly completed and executed by the Assignee; (b) agrees that (i) it will, independently and without reliance on the Administrative Agent, the Assignor or any other Lender, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Documents, and (ii) it will perform in accordance with their terms all of the obligations which by the terms of the Loan Documents are required to be performed by it as a Lender and (c) authorizes the Administrative Agent to take such actions on behalf of the Assignee and to exercise such powers as are specifically delegated to the Administrative Agent by the terms and provisions of the DIP Credit Agreement and of the other Loan Documents, together with such actions and powers as are reasonably incidental thereto and (viii) it is a "qualified purchaser" ("Qualified Purchaser") as defined under Section 2(a)(51) of the Investment Company Act of 1940 (the "1940 Act") that either: (1)(A) acting for its own account or the account of other Qualified Purchasers in the aggregate owns and invests on a discretionary basis not less than \$25,000,000 in "investments" as defined under Rule 2a51-1 of the 1940 Act, and (B) is not organized for the specific purpose of acquiring the Assigned Interest; does not have shareholders, partners or other holders of equity or beneficial interests that have the right to decide individually whether to participate, or the extent of their participation, in such entity's acquisition of the Assigned Interest; and is not acquiring an Acquired Interest that exceeds 40% of such entity's total assets (on a consolidated basis with its subsidiaries); or (2) is a company in which each beneficial owner of such company's securities is a Qualified Purchaser.

- 2. *Payments*. From and after the Effective Date, the Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assignor for amounts which have accrued to but excluding the Effective Date and to the Assignee for amounts which have accrued from and after the Effective Date.
- 3. Effect of Assignment. As of the Effective Date, (i) the Assignee shall be a party to the DIP Credit Agreement and, to the extent provided in this Assignment and Assumption, have the rights and obligations of a Lender under the DIP Credit Agreement and under the other Loan Documents and (ii) the Assignor shall, to the extent provided in this Assignment and Assumption, relinquish its rights and be released from its obligations under the DIP Credit Agreement and the other Loan Documents.
- 4. General Provisions. This Assignment and Assumption shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment and Assumption may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment and Assumption by facsimile or other electronic means, which shall include DocuSign and similar electronic signature platforms and digital copies of a signatory's manual signature, will for all purposes be treated as the equivalent of delivery of a manually executed counterpart of this Assignment and Assumption. The words "execution," "signed," "signature," and words of like import in this Assignment and Assumption shall be deemed to include electronic signatures, including by DocuSign and similar electronic signature platforms and digital copies of a signatory's manual signature, or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a

Case 25-10979-JKS Doc 50-2 Filed 06/04/25 Page 109 of 198

paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act. This Assignment and Assumption shall be construed in accordance with and governed by the laws of the State of New York.

Exhibit F

Proposed Interim DIP Financing Order

Schedule 5.4 – Ownership and Control

SCHEDULE 6.1

REPORTING AND INFORMATION REQUIREMENTS

Each Loan Party covenants and agrees, so long as any of the Obligations remains outstanding and unpaid, that it will, and, as applicable, it will cause each of its Subsidiaries to:

- 1. <u>Financial Statements</u>. Furnish to the Administrative Agent, in form and detail satisfactory to the Administrative Agent, with sufficient copies for each Lender, the following documents:
- (a) as soon as available, but in any event within one hundred twenty (120) days after the end of each Fiscal Year, a copy of the audited Consolidated and unaudited Consolidating financial statements of Borrower and its Consolidated Subsidiaries as at the end of such Fiscal Year and the related audited Consolidated and unaudited Consolidating statements of income, stockholders equity, and cash flows of the Borrower and its Consolidated Subsidiaries for such Fiscal Year or partial Fiscal Year and underlying assumptions, setting forth in each case in comparative form the figures for the previous Fiscal Year, certified as being fairly stated in all material respects by an independent, nationally recognized certified public accounting firm reasonably satisfactory to the Administrative Agent;
- (b) as soon as available, but in any event within forty-five (45) days after the end of each fiscal quarter of the Loan Parties (including the last quarter of each Fiscal Year), Borrower prepared unaudited Consolidated and Consolidating balance sheets of Borrower and its Consolidated Subsidiaries as at the end of such quarter and the related unaudited statements of income, stockholders equity and cash flows of Borrower and its Consolidated Subsidiaries for the portion of the Fiscal Year through the end of such quarter, setting forth in each case in comparative form the figures for the corresponding periods in the previous Fiscal Year, and certified by a Responsible Officer of Borrower as being fairly stated in all material respects;
- (c) as soon as available, but in any event within thirty (30) days after the end of each month (unless such month end is also the end of a fiscal quarter or a Fiscal Year), commencing with the first full month after the Closing Date, Borrower prepared unaudited Consolidated and Consolidating balance sheets of Borrower and its Consolidated Subsidiaries as at the end of such month and the related unaudited statements of income, stockholders equity and cash flows of Borrower and its Consolidated Subsidiaries for the portion of the Fiscal Year through the end of such fiscal month, setting forth in each case in comparative form (i) the figures for the corresponding periods in the previous year and (ii) the figures for the relevant period set forth in the projections delivered for such year pursuant to Section 2(d) of this Schedule 6.1, and certified by a Responsible Officer of Borrower as being fairly stated in all material respects; and

all such financial statements to be complete and correct in all material respects and to be prepared in reasonable detail and in accordance with GAAP throughout the periods reflected therein and with prior periods (except as approved by a Responsible Officer and disclosed therein), provided however that the financial statements delivered pursuant to clauses (b) and (c) hereof will

not be required to include footnotes and will be subject to change from audit and year-end adjustments.

- 2. <u>Certificates, Other Information</u>. Furnish to the Administrative Agent, in form and detail acceptable to the Administrative Agent, with sufficient copies for each Lender, the following documents:
 - (a) Promptly upon receipt thereof, copies of all significant reports submitted by the Loan Parties' firm(s) of certified public accountants in connection with each annual, interim or special audit or review of any type of the financial statements or related internal control systems of the Loan Parties made by such accountants, including any comment letter submitted by such accountants to management in connection with their services;
 - (b) Any financial reports, statements, press releases, other material information or written notices delivered to the holders of the Subordinated Debt pursuant to any applicable Subordinated Debt documents (to the extent not otherwise required hereunder), as and when delivered to such Persons;
 - (c) Within thirty (30) days after the end of each Fiscal Year, projections for the Loan Parties for the next succeeding Fiscal Year, on a monthly basis and for the following Fiscal Year on an annual basis, including a balance sheet, as at the end of each relevant period and for the period commencing at the beginning of the Fiscal Year and ending on the last day of such relevant period, such projections certified by a Responsible Officer of Borrower as being based on reasonable estimates and assumptions taking into account all facts and information known (or reasonably available to any Loan Party) by a Responsible Officer of Borrower;
 - (d) Any additional information as required by any Loan Document, and such additional schedules, certificates and reports respecting all or any of the collateral, the items or amounts received by the Loan Parties in full or partial payment thereof, and any goods (the sale or lease of which shall have given rise to any of the collateral) possession of which has been obtained by the Loan Parties, all to such extent as the Administrative Agent may reasonably request from time to time, any such schedule, certificate or report to be certified as true and correct in all material respects by a Responsible Officer of the applicable Loan Party and shall be in such form and detail as the Administrative Agent may reasonably specify; and
 - (e) Such additional financial and/or other information as any Lender may from time to time reasonably request, promptly following such request.
- 3. Notices. Promptly give written notice to the Administrative Agent of:
 - (a) the occurrence of any Default or Event of Default of which any Loan Party has knowledge or the occurrence of any Reportable Compliance Event;

- (b) any (i) litigation or proceeding existing at any time between any Loan Party and any Governmental Authority or other third party, or any investigation of any Loan Party conducted by any Governmental Authority (including, without limitation, any notices, inspections, investigations or proceedings that any Loan Party receives or of which it becomes aware involving the Virginia Department of Environmental Quality, the U.S. Environmental Protection Agency, Chesterfield County, or any other Governmental Authority or Person relating to Hazardous Material Laws), which in any case if adversely determined would have a Material Adverse Effect or (ii) any material adverse change in the financial condition of any Loan Party since the date of the last audited financial statements delivered pursuant to Section 1(a) of this Schedule 6.1;
- (c) the occurrence of any event which any Loan Party believes could reasonably be expected to have a Material Adverse Effect, promptly after concluding that such event could reasonably be expected to have such a Material Adverse Effect;
- (d) promptly after becoming aware thereof, the taking by the Internal Revenue Service or any foreign taxing jurisdiction of a written tax position (or any such tax position taken by any Loan Party in a filing with the Internal Revenue Service or any foreign taxing jurisdiction) which could reasonably be expected to have a Material Adverse Effect, setting forth the details of such position and the financial impact thereof;
- (e) (i) all jurisdictions in which any Loan Party proposes to become qualified after the Closing Date to transact business, (ii) the acquisition or creation of any new Subsidiaries, (iii) any material change after the Closing Date in the authorized and issued Equity Interests of any Loan Party or any other material amendment to any Loan Party's charter, by-laws or other organizational documents, such notice, in each case, to identify the applicable jurisdictions, capital structures or amendments as applicable, provided that such notice shall be given not less than ten (10) Business Days prior to the proposed effectiveness of such changes, acquisition or creation, as the case may be (or such shorter period to which Administrative Agent may consent);
- (f) not less than fifteen (15) Business Days (or such other shorter period to which Administrative Agent may agree) prior to the proposed effective date thereof, any proposed material amendments, restatements or other modifications to any Subordinated Debt documents;
- (g) any default or event of default by any Person under any Subordinated Debt document, concurrently with delivery or promptly after receipt (as the case may be) of any notice of default or event of default under the applicable document, as the case may be; and

(h) promptly after becoming aware thereof, any action taken by the VDEQ with respect to any permit related to the operation of any landfill operated by any Loan Party.

Each notice pursuant to this Section shall be accompanied by a statement of a Responsible Officer of Borrower setting forth details of the occurrence referred to therein and, in the case of notices referred to in clauses (a), (b), (c), (d) and (g) hereof stating what action the applicable Loan Party has taken or proposes to take with respect thereto.

- 4. Management Report. Concurrently with each delivery of financial statements pursuant to Section 1 above, and to the extent such information is not otherwise already included with such financial statements, the Loan Parties will deliver a management report, certified by a Responsible Officer of the Borrower, (a) describing the operations and financial condition of the Loan Parties for the month then ended and the portion of the current Fiscal Year then elapsed (or for the Fiscal Year then ended in the case of year-end financials), (b) setting forth in comparative form the corresponding figures for the corresponding periods of the previous Fiscal Year and the corresponding figures from the most recent projections for the current Fiscal Year, (c) discussing the reasons for any significant variations, and (d) stating that such officer has no knowledge of any Default except as specified in such certificate.
- 5. <u>Tax Returns</u>. As soon as practicable, and in any event within 15 days after the Administrative Agent' request therefor, the Borrower will furnish to the Administrative Agent copies of all Federal, state, local and foreign tax returns (together with all schedules and attachments thereto) filed or required to be filed by any of the Loan Parties.
- 6. Notice of Change of Management or Ownership. Promptly after becoming aware thereof, the Borrower will give the Administrative Agent notice of the departure or anticipated departure of any member of the senior executive management of any Loan Party or any Loan Party's Board of Directors (or similar governing body) or any change in the holders of the Borrower's or any Loan Party's Equity Interests or in the percentages of such securities held by such holders.
- 7. <u>Further Information</u>. The Borrower will promptly furnish to the Administrative Agent such other information regarding the Loan Parties as the Administrative Agent may reasonably request.

SCHEDULE 7.1

NEGATIVE COVENANTS

Each Loan Party covenants and agrees that, so long as any of the Obligations remains outstanding and unpaid, it will not, and, as applicable, it will not permit any of its Subsidiaries to:

- 1. <u>Limitation on Debt.</u> Create, incur, assume or suffer to exist any Debt, except:
 - (a) Obligations of any Loan Party to the Lenders under this Agreement and/or the other Loan Documents;
 - (b) any Debt existing on the Closing Date and set forth in Attachment 1 attached to this Schedule 7.1 and any renewals or refinancing of such Debt; provided that (i) the aggregate principal amount of such renewed or refinanced Debt shall not exceed the aggregate principal amount of the original Debt outstanding on the Closing Date (less any principal payments and the amount of any commitment reductions made thereon on or prior to such renewal or refinancing) plus accrued interest, closing costs, fees and expenses, (ii) the renewal or refinancing of such Debt shall be on substantially the same or better terms as in effect with respect to such Debt on the Closing Date, and shall otherwise be in compliance with this Agreement, and (iii) at the time of such renewal or refinancing no Default or Event of Default has occurred and is continuing or would result from the renewal or refinancing of such Debt:
 - (c) any Debt of Borrower or any Subsidiary incurred to finance the acquisition of fixed or capital assets, whether pursuant to a loan or a Capital Lease provided that both at the time of and immediately after giving effect to the incurrence thereof (i) no Default or Event of Default shall have occurred and be continuing, and (ii) the aggregate amount of all such Debt at any one time outstanding (including, without limitation, any Debt of the type described in this clause (c) which is set forth on Attachment 1 attached to this Schedule 7.1) shall not exceed \$3,300,000, and any renewals or refinancings of such Debt on terms substantially the same or better than those in effect at the time of the original incurrence of such Debt;
 - (d) [Reserved];
 - (e) Debt under any Swap Agreements, provided that such transaction is entered into for risk management purposes and not for speculative purposes;
 - (f) Debt arising from judgments or decrees not deemed to be a Default or Event of Default under subsection (H) of Section 8.1;
 - (g) Debt owing to a Person that is a Loan Party, but only to the extent permitted under Section 7 of this <u>Schedule 7.1</u>;

- (h) Debt of any Loan Party to any other Loan Party;
- (i) Guarantees by any Loan Party of Debt or other obligations arising in the ordinary course of business of any other Loan Party;
- (j) Debt under surety, performance or similar instruments incurred in the ordinary course of business; and
- (k) additional unsecured Debt not otherwise described above, provided that both at the time of and immediately after giving effect to the incurrence thereof (i) no Default or Event of Default shall have occurred and be continuing or result therefrom and (ii) the aggregate amount of all such additional unsecured Debt shall not exceed \$5,500,000 at any one time outstanding.
- 2. <u>Limitation on Liens</u>. Create, incur, assume or suffer to exist any Lien upon any of its property, assets or revenues, whether now owned or hereafter acquired, except for Excepted Liens. Regardless of the provisions of this Section 2 of <u>Schedule 7.1</u>, no Lien over the Equity Interests of Borrower or any Subsidiary of Borrower shall be permitted under the terms of this Agreement.
- 3. <u>Business Activities</u>. Engage in any line of business other than the businesses engaged in as of the Closing Date and businesses reasonably related thereto.
- 4. <u>Limitation on Mergers, Dissolution or Sale of Assets.</u> Enter into any merger or consolidation or convey, sell, lease, assign, transfer or otherwise dispose of any of its property, business or assets (including, without limitation, Equity Interests, receivables and leasehold interests), whether now owned or hereafter acquired or liquidate, wind up or dissolve, in each case unless all Obligations are paid in full in cash prior to or in connection with the consummation of any such transaction, except for:
 - (a) Inventory leased or sold in the ordinary course of business;
 - (b) obsolete, damaged, uneconomic or worn out machinery, parts, property or equipment, or property or equipment no longer used or useful in the conduct of the applicable Loan Party's business;
 - (c) [Reserved];
 - (d) mergers or consolidations of any Subsidiary of Borrower with or into Borrower or any Guarantor so long as the Borrower or such Guarantor shall be the continuing or surviving entity, provided that at the time of each such merger or consolidation, both before and after giving effect thereto, no Default or Event of Default shall have occurred and be continuing or result from such merger or consolidation;

- (e) any Subsidiary of Borrower may liquidate or dissolve into Borrower or a Guarantor if Borrower determines in good faith that such liquidation or dissolution is in the best interests of Borrower, so long as no Default or Event of Default has occurred and is continuing or would result therefrom;
- (f) sales or transfers, including without limitation upon voluntary liquidation from any Loan Party to Borrower or a Guarantor, provided that the applicable Borrower or Guarantor takes such actions as the Administrative Agent may reasonably request to ensure the perfection and priority of the Liens in favor of the Lenders over such transferred assets;
- (g) Asset Dispositions (exclusive of asset sales permitted pursuant to all other subsections of this Section 4 of <u>Schedule 7.1</u>) in which the sales price is at least equal to the fair market value of the assets sold and the consideration received is cash or cash equivalents or Debt of any Loan Party being assumed by the purchaser, provided that the aggregate amount of such Asset Sales does not exceed \$1,100,000 in any Fiscal Year and no Default or Event of Default has occurred and is continuing at the time of each such sale both before and after giving effect to such Asset Disposition;
- (h) the sale or disposition of Permitted Investments and other cash equivalents in the ordinary course of business; and
- (i) dispositions of owned or leased vehicles in the ordinary course of business.

The Lenders hereby consent and agree to the release by the Administrative Agent of any and all Liens on the property sold or otherwise disposed of in compliance with this Section 4 of Schedule 7.1.

- 5. Restricted Payments. Declare or make any distributions, dividend, payment or other distribution of assets, properties, cash, rights, obligations or securities (collectively, "Distributions") on account of any of its Equity Interests, as applicable, or purchase, redeem or otherwise acquire for value any of its Equity Interests, as applicable, or any warrants, rights or options to acquire any of its Equity Interests, now or hereafter outstanding (collectively, "Purchases"), except that:
 - (a) each Subsidiary of the Borrower may pay cash Distributions to the Borrower;
 - (b) each Loan Party may pay reasonable fees and expenses to its directors for attending directors meetings to the extent not exceeding in the aggregate for all Loan Parties \$200,000 per Fiscal Year;
 - (c) each Loan Party may declare and make Distributions payable in kind; and
 - (d) each Loan Party may declare and make Distributions payable in the Equity Interests of such Loan Party, provided that the issuance of such Equity Interests does not otherwise violate the terms of this Agreement and no

Default or Event of Default has occurred and is continuing at the time of making such Distribution or would result from the making of such Distribution.

6. [Reserved].

- 7. <u>Limitation on Investments, Loans and Advances</u>. Make or allow to remain outstanding any Investment (whether such investment shall be of the character of investment in shares of stock, evidences of indebtedness or other securities or otherwise) in, or any loans or advances to, any Person other than:
 - (a) Permitted Investments;
 - (b) Investments existing on the Closing Date and listed on <u>Attachment 3</u> to this Schedule 7.1;
 - (c) sales on open account in the ordinary course of business;
 - (d) intercompany loans or intercompany Investments made by any Loan Party to or in any Guarantor or Borrower, provided that, in the case of any intercompany loans or intercompany Investments made by any Borrower in any Guarantor, the aggregate amount from time to time outstanding in respect thereof shall not exceed \$1,100,000, and provided, further, that in each case, no Default or Event of Default shall have occurred and be continuing at the time of making such intercompany loan or intercompany Investment or result from such intercompany loan or intercompany Investment being made and that any intercompany loans shall be evidenced by and funded under an intercompany note pledged to the Administrative Agent on behalf of the Lenders under the appropriate Security Documents.
 - (e) Investments in respect of Swap Agreements provided that such transaction is entered into for risk management purposes and not for speculative purposes;
 - (f) loans and advances to employees, officers and directors of any Loan Party for moving, entertainment, travel and other similar expenses in the ordinary course of business not to exceed \$100,000 in the aggregate at any time outstanding;
 - (g) [Reserved];
 - (h) Investments constituting deposits made in connection with the purchase of goods or services in the ordinary course of business in an aggregate amount for such deposits not to exceed \$1,100,000 at any one time outstanding; and
 - (i) other Investments not described above provided that both at the time of and immediately after giving effect to any such Investment (i) no Default or Event of Default shall have occurred and be continuing or shall result from the

making of such Investment and (ii) the aggregate amount of all such Investments shall not exceed \$1,100,000 at any time outstanding.

In valuing any Investments for the purpose of applying the limitations set forth in this Section 7 of Schedule 7.1 (except as otherwise expressly provided herein), such Investment shall be taken at the original cost thereof, without allowance for any subsequent write-offs or appreciation or depreciation, but less any amount repaid or recovered on account of capital or principal.

- 8. Transactions with Affiliates. Enter into, amend, restate, or otherwise modify any transaction, agreement, or arrangement, including, without limitation, any purchase, sale, lease or exchange of property or the rendering of any service, with any Affiliates of the Loan Parties, except with respect to any transactions, agreements, or arrangements: (a) with Affiliates that are the Borrower or Guarantor, (b) otherwise permitted under this Agreement, (c) existing as of the Closing Date (on the same terms and conditions as in effect on the Closing Date), and (d) transactions in the ordinary course of a Loan Party's business, consistent with past practice, and upon fair and reasonable terms no less favorable to such Loan Party than it would obtain in a comparable arms'-length transaction from unrelated third parties.
- 9. <u>Limitations on Other Restrictions</u>. Except for this Agreement or any other Loan Document enter into any agreement, document or instrument which would (a) restrict the ability of any Subsidiary of the Borrower to pay or make dividends or distributions in cash or kind to Borrower or Guarantor, to make loans, advances or other payments of whatever nature to any Loan Party, or to make transfers or distributions of all or any part of its assets to any Loan Party, or (b) restrict or prevent any Loan Party from granting Administrative Agent on behalf of the Lenders Liens upon, security interests in and pledges of their respective assets, except to the extent such restrictions exist in documents creating Liens permitted by Section 2 of this <u>Schedule 7.1</u>.
- 10. <u>Prepayment of Debt</u>. Make any prepayment (whether optional or mandatory), repurchase, redemption, defeasance or any other payment in respect of any Subordinated Debt.
- 11. <u>Modification of Certain Agreements</u>. Make, permit or consent to any amendment or other modification to the constitutional documents of any Loan Party or any Material Agreement except to the extent that any such amendment or modification (i) does not violate the terms and conditions of this Agreement or any of the other Loan Documents, (ii) does not materially adversely affect the interest of the Lenders as creditors and/or secured parties under any Loan Document and (iii) would not reasonably be expected to have a Material Adverse Effect.
- 12. [Reserved].

13. <u>Fiscal Year</u>. Permit the Fiscal Year of any Loan Party to end on a day other than December 31.

Schedule 7.5 – Existing Investments

GUARANTY AGREEMENT

This Guaranty Agreement, dated effective as of June 2, 2025 (this "Guaranty Agreement"), is executed by Shoosmith Bros., Inc., a Virginia corporation ("Shoosmith," or, the "Guarantor"), and Volunteer Enterprises, LLC as administrative agent (the "Administrative Agent"), for the benefit of the Lenders party to that certain DIP Credit Agreement (as defined below). The parties agree as follows:

RECITALS

- A. The Guarantor, Administrative Agent, VWS Holdco, Inc., a Delaware corporation (the "Borrower"), and the Lenders (as defined in the DIP Credit Agreement, and together with the Administrative Agent, the "Secured Parties") party thereto from time to time are parties to that certain Senior Secured Superpriority Debtor-in-Possession Term Loan Agreement dated as of June 2, 2025 (the "DIP Credit Agreement").
- B. The Lenders require, in order to ensure the prompt and complete payment and performance of the obligations, indebtedness, and liabilities of Borrower to the Secured Parties (the "Obligations," including the meaning assigned to it in the DIP Credit Agreement), that the Guarantor execute and deliver this Guaranty Agreement.
- C. The Guarantor will receive substantial benefit from the Borrower entering into the DIP Credit Agreement.
- D. Capitalized terms used but not defined herein shall have the meanings assigned to them in the DIP Credit Agreement.

AGREEMENT

NOW THEREFORE, in consideration of the Recitals, which are deemed a material part of this Guaranty Agreement, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, it is agreed as follows:

1. <u>Incorporation of Recitals</u>. The Recitals set forth above are incorporated into this Guaranty Agreement by this reference thereto and are made a part hereof.

2. Guaranty.

- (a) The Guarantor unconditionally and irrevocably guarantees the due and punctual payment by the Borrower of the Obligations. The Guarantor further agrees that the Obligations may be extended or renewed, in whole or in part, without notice to or further assent from it, and it will remain bound upon this Agreement notwithstanding any extension or renewal of any of the Obligations.
- (b) The Guarantor waives presentation to, demand for payment from and protest to the Borrower or any other Guarantor, and also waives notice of protest for nonpayment. The Obligations of the Guarantor hereunder shall not be affected by (i) the failure of the Administrative Agent or the Lenders to assert any claim or demand or to

enforce any right or remedy against the Borrower or any other Guarantor under the provisions of this Agreement or any other Loan Document or otherwise; (ii) any extension or renewal of any provision hereof or thereof; (iii) any rescission, waiver, compromise, acceleration, amendment or modification of any of the terms or provisions of any of the Loan Documents; (iv) the release, exchange, waiver or foreclosure of any security held by the Administrative Agent for the Obligations or any of them; (v) the failure of the Administrative Agent or the Lenders to exercise any right or remedy against any other Guarantor; or (vi) the release or substitution of any Guarantor or any other Guarantor.

- (c) The Guarantor further agrees that this guaranty constitutes a guaranty of payment when due and not just of collection, and waives any right to require that any resort be had by the Administrative Agent or the Lenders to any security held for payment of the Obligations or to any balance of any deposit, account or credit on the books of the Administrative Agent or the Lenders in favor of the Borrower or any other Guarantor, or to any other Person.
- (d) The Guarantor hereby waives any defense that it might have based on a failure to remain informed of the financial condition of the Borrower and of any other Guarantor and any circumstances affecting the ability of the Borrower to perform under this Agreement.
- (e) The Guarantor's guaranty shall not be affected by the genuineness, validity, regularity or enforceability of the Obligations or any other instrument evidencing any Obligations, or by the existence, validity, enforceability, perfection, or extent of any collateral therefor or by any other circumstance relating to the Obligations which might otherwise constitute a defense to this Guaranty. Neither of the Administrative Agent, nor any of the Lenders makes any representation or warranty in respect to any such circumstances or shall have any duty or responsibility whatsoever to any Guarantor in respect of the management and maintenance of the Obligations.
- (f) Subject to the terms of the DIP Credit Agreement and DIP Financing Orders, upon the Obligations becoming due and payable (by acceleration or otherwise), the Lenders shall be entitled to immediate payment of such Obligations by the Guarantor upon written demand by the Administrative Agent, without further application to or order of the Bankruptcy Court.
- 3. No Impairment of Guaranty. The obligations of the Guarantor hereunder shall not be subject to any reduction, limitation, impairment or termination for any reason, including, without limitation, any claim of waiver, release, surrender, alteration or compromise, and shall not be subject to any defense or set-off, counterclaim, recoupment or termination whatsoever by reason of the invalidity, illegality or unenforceability of the Obligations. Without limiting the generality of the foregoing, the obligations of the Guarantor hereunder shall not be discharged or impaired or otherwise affected by the failure of the Administrative Agent or the Lenders to assert any claim or demand or to enforce any remedy under this Agreement or any other agreement, by any waiver or modification of any provision thereof, by any default, failure or delay, willful or otherwise, in the performance of the Obligations, or by any other act or thing or omission or delay to do any other act or thing which may or might in any manner or to any extent vary the

risk of the Guarantor or would otherwise operate as a discharge of the Guarantor as a matter of law, unless and until the Obligations are paid in full.

- 4. <u>Subrogation</u>. Upon payment by any Guarantor of any sums to the Administrative Agent or the Lenders under this Agreement, all rights of such Guarantor against the Borrower arising as a result thereof by way of right of subrogation or otherwise, shall in all respects be subordinate and junior in right of payment to the prior final and indefeasible payment in full of all the Obligations. If any amount shall be paid to such Guarantor for the account of the Borrower, such amount shall be held in trust for the benefit of the Administrative Agent and the Lenders and shall forthwith be paid to the Administrative Agent and the Lenders to be credited and applied to the Obligations, whether matured or unmatured.
- 5. <u>Reliance</u>. Guarantor acknowledges and agrees that the Lenders will have relied upon and will continue to rely upon this Agreement and the provisions hereof in continuing to deal with Borrower under the DIP Credit Agreement or otherwise.
- 6. <u>Costs of Collection</u>. The liabilities and obligations of Guarantor hereunder will be for the full amount of the Obligations, whatever they may be, and will include all reasonable attorneys' fees, costs of collection and enforcement expenses incurred by Administrative Agent or the Lenders in performing or collecting any of the Obligations at any time. The Guarantor also agrees to pay to the Administrative Agent and the Lenders, on demand, all reasonable costs and expenses incurred by any of them, including the reasonable fees and disbursements of counsel, in connection with the enforcement of and preservation of any of its rights hereunder.
- otherwise will impair or adversely affect any of the rights or remedies of the Administrative Agent (on behalf of the Lenders) in connection with the DIP Credit Agreement or otherwise, all of which rights and remedies are reserved by the Administrative Agent (on behalf of the Lenders). Nothing in this Agreement will create any liability or obligation of the Administrative Agent or Lenders to the Guarantor or Borrower with respect to how the Administrative Agent or Lenders exercise their rights with respect to the Guarantor or Borrower. Without limiting the Administrative Agent's or Lenders' other rights, regardless of whether or not any existing relationship between the Guarantor and Borrower or any other Party or Affiliate of any of them has been changed or ended, the Guarantor agrees that at any time and from time to time, without notice to or consent of the Guarantor and without incurring responsibility to the Guarantor and without impairing or releasing the liabilities or obligations provided for herein or otherwise impairing the rights of the Administrative Agent hereunder:
 - (a) the time for performance of or compliance with any covenant or agreement in any Loan Document or otherwise may be extended or such performance may be waived by any applicable party under any Loan Document;
 - (b) any covenant or agreement in any Loan Document may from time to time be replaced or modified, for the purpose of adding any requirement thereto or changing in any manner any right, liability or obligation of any party to any Loan Document;

- (c) any manner, place or term for payment of any Guaranteed Obligation may be altered or any term for payment extended, or any Guaranteed Obligation may be renewed in whole or in part; and
- (d) the maturity of any of the Guaranteed Obligations may be accelerated in accordance with the terms of any present or future agreement between the applicable parties to any Loan Document.
- 8. Events of Default. The occurrence of any of the following events shall constitute an Event of Default by the Guarantor under this Agreement (referred to herein as an "Event of Default"): (i) any default under the DIP Credit Agreement and the expiry of any grace periods contained therein and applicable thereto; (ii) breach of any covenant, provision, or term of this Agreement; (iii) the Secured Parties' lien on the Collateral fails to constitute a security interest on such Collateral; or (iv) the Guarantor does, or permits to be done, anything that in any way may materially impair the security and lien of this Agreement or materially impairs the value of the Collateral, or any part thereof, or materially weakens or diminishes the security intended to be given under and by virtue of Security Documents.
- 9. <u>Reinstatement</u>. If any payment applied to any of the Obligations is thereafter set aside, recovered, rescinded or required to be returned for any reason, the Obligations to which such payment was applied will for the purposes of this Agreement be deemed to have continued in existence, notwithstanding such application, and this Agreement will be enforceable as to such Obligations as fully as if such application had never been made.
- 10. <u>Specific Performance</u>. The Guarantor hereby authorizes the Administrative Agent (on behalf of the Lenders) to demand specific performance of this Agreement at any time when the Guarantor shall have failed to comply with any provision hereof, and the Guarantor hereby irrevocably waives any defense based on the adequacy of a remedy at law which might be asserted as a bar to the remedy of specific performance hereof in any action brought therefor.
- 11. <u>Cumulative Remedies</u>. The rights and remedies herein provided are cumulative, may be exercised singly or concurrently and are not exclusive of any rights or remedies provided by applicable law or otherwise.
- 12. <u>No Third Party Beneficiaries</u>. This Agreement is entered into for the sole protection and benefit of the Secured Parties and their successors and assigns, and no other Person will be a direct or indirect beneficiary of, or will have any direct or indirect cause of action or claim in connection with, this Agreement. This Agreement will be binding on and inure to the benefit of the respective permitted successors and assigns of the Secured Parties.
- 13. <u>Headings</u>. Section headings herein are included for convenience of reference only and shall not affect the interpretation of this Agreement.
- 14. Governing Law; Submission to Jurisdiction; Waiver of Venue, Service of Process, Waiver of Right to Jury Trial. The terms of Section 10.1 and Section 10.6 of the DIP Credit Agreement with respect to governing law, submission to jurisdiction, waiver of venue, service of process and waiver of the right to a jury trial are each incorporated herein by reference, mutatis mutandis, and the parties hereto agree to such terms.

- 15. <u>Severability</u>. If any provision of this Agreement is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.
- 16. <u>Entirety</u>. This Agreement, the other Loan Documents and the other documents relating to the Obligations represent the entire agreement of the parties hereto and thereto, and supersede all prior agreements and understandings, oral or written, if any, including any proposal letters or correspondence relating to the Loan Documents, any other documents relating to the Obligations, or the transactions contemplated herein and therein.
- 17. <u>Survival</u>. All representations and warranties of the Guarantor hereunder shall survive the execution and delivery of this Agreement, the other Loan Documents and the other documents relating to the Obligations, and the investments made thereunder or in connection therewith.
- 18. <u>Rights of Required Lenders</u>. All rights of the Administrative Agent hereunder, if not exercised by the Administrative Agent, may be exercised by the Required Lenders.
- 19. <u>Termination</u>. This Agreement will terminate upon the payment and performance in full of all of the Obligations, all without delivery of any instrument or performance of any act by any party. At the request of the Guarantor following any such termination, the Administrative Agent shall deliver to the Guarantor such documents as the Guarantor shall reasonably request to evidence such termination.

[Remainder of page intentionally left blank. Signature page follows.]

Case 25-10979-JKS Doc 50-2 Filed 06/04/25 Page 128 of 128

IN WITNESS WHEREOF, the date first written above.	parties have executed this Guaranty Agreement as of the
	GUARANTOR:
	SHOOSMITH BROS., INC.
	By: Name: Title:
ACKNOWLEDGED BY:	
	ADMINISTRATIVE AGENT:
	VOLUNTEER ENTERPRISES, LLC
	By:
	Name: Title:
	BORROWER:
	VWS HOLDCO, INC.
	By:
	Name: Fred Nichols

Title: President

EXHIBIT 2

INITIAL DIP BUDGET

10,1000	DIP Budget	103779,000\$7

Mach #	Wook 1	CAOOM	Wook 2	Mook	Mooks	Mooke	Viooly 7	Wook 9	Wooko	Wook 10	Mook 11	Wook19	Wook 13		Total
Actual / Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast	Forecast		TOTAL
Start Date End Date	6/1/25	6/7/25	6/14/25	6/21/25	6/28/25	7/5/25	7/12/25	7/19/25	7/26/25	8/2/25	8/9/25	8/16/25	8/23/25		6/1/25
Receipts															
Receipts \$,	· •	· •	•		- 9	· · ·	-	٠			- %	· •	€9	
Gas Royalty Receipts			20				20				20				150
Other Receipts			. 03				. 20			. .	, 02				150
							3		,		3	•		•	201
Operating Disbursements															
Salaries & Benefits \$	(13)	\$ (21)	(13)	\$ (21)	\$ (13)	\$ (21) \$	\$ (13) \$			(21)	\$ (13)	\$ (21)	\$ (13)	€	(216)
Processing & Transportation	(06)	(06)	(06)	(06)	(06)	(06)	(06)	(06)	(06)	(06)	(06)	(06)	(06)		(1,170)
Rent	(11)	,			(11)	,			(11)			,	•		(33)
Fuel	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)	(1)		(12)
Equipment Leasing			•		(9)			,	(9)				(9)		(18)
Repairs	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)	(10)		(124)
Engineering	(42)				(42)				(42)						(126)
Insurance	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)	(0)		(2)
Utilities	(11)	•	•	,	(11)	,		,	(11)	,	,	,			(33)
Other	(9)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)	(3)		(47)
Subtotal	(184)	\$ (126)	\$ (1		\$ (187)	\$ (126) \$	\$ (117) \$	(1)	(187)	(126)	\$ (117)	\$ (126)	\$ (123)	49	(1,783)
					(107)		(10)		(107)	(406)	(10)	(406)		6	(4,699)
Operating Cash From	(184)	¢ (971) ¢	(/o)	* (17p) *	(187)	¢ (971) ¢	e (/a)	¢ (971)	¢ (181)	(170)	(/9)	(170)	(123)		(1,633)
Non-Operating Disbursements															
Bankruptcy-Related Costs \$,	· &9	\$ (515)	· •	\$ (200)	- +	+	٠	٠		-	· •	· •	\$	(715)
Professional Fees	(196)	(196)		(196)	(196)	(196)	(196)	(196)	(196)	(196)	(196)	(196)	(188)	ļ	(2,540)
Subtotal	(196)	(196)	(711)	(196)	\$ (968) \$	\$ (196) \$	\$ (1961) \$	\$ (1961)	\$ (196)	(196)	(196)	(196)	\$ (188)	\$	(3,255)
Net Cash Flow \$	(380)	\$ (322)	(2778)	\$ (322)	\$ (283)	\$ (322) \$	\$ (263) \$	(322) \$	\$ (282)	(322)	\$ (263)	\$ (322)	\$ (311)	s)	(4,888)
Book Cash															
Beginning Balance \$	155	\$ 1.575	\$ 1.254	\$ 476		\$ 1,122	\$ 800	538 \$	1.766 \$	1.383	\$ 1.062	\$ 799	\$ 478	49	155
	_				1.550	. '		1.550	. '	. '	. '				4.900
Repayments		,	٠	,		,			,	,	,	,	٠		
Net Cash Flow	(380)	(322)	(778)	(322)	(283)	(322)	(263)	(322)	(383)	(322)	(263)	(322)	(311)		(4,888)
Ending Balance \$	1	\$	49				\$ 238 \$		1	1,062		\$ 478	\$ 167	es.	167
DIP Batance															
Beginning Balance \$,	\$ 1,800	\$ 1,800	\$ 1,800		\$ 3,618 \$	\$ 3,618 \$		5,418 \$	5,458	\$ 5,458	\$ 5,458	\$ 5,458	↔	,
Borrowings	1,800		٠		1,550			1,550					•		4,900
Repayments			•										•		
Bridge Roll-Up	٠	,	٠		250	,	,	250	,	,		,	•		200
PIK Interest			•		18		•		40		•		54		112
Ending Balance \$	1,800	\$ 1,800	\$ 1,800	\$ 1,800	\$ 3,618	\$ 3,618 \$	\$ 3,618 \$	5,418 \$	5,458 \$	5,458	\$ 5,458	\$ 5,458	\$ 5,512	s	5,512